Energy system analysis of energy autonomous municipalities

Zur Erlangung des akademischen Grades eines

Doktors der Ingenieurwissenschaften

(Dr.-Ing.)

von der KIT-Fakultät für Wirtschaftswissenschaften des Karlsruher Instituts für Technologie (KIT) genehmigte

Dissertation

von

Jann Michael Weinand

Tag der mündlichen Prüfung: Referent: Korreferentin: 15. September 2020Prof. Dr. rer. pol. Wolf FichtnerProf. Dr. rer. pol. Ingela Tietze

Vorwort

Diese Dissertation ist zwischen September 2016 und Juli 2020 während meiner Arbeit am Lehrstuhl für Energiewirtschaft (Institut für Industriebetriebslehre und Industrielle Produktion (IIP) des Karlsruher Instituts für Technologie) entstanden Während dieser Zeit habe ich Unterstützung von vielen Seiten erhalten, für die ich mich an dieser Stelle herzlich bedanken möchte.

Als erstes möchte ich meinem Doktorvater Wolf Fichtner danken, von dem ich in vielen konstruktiven Diskussionen gelernt habe, worauf es bei einer Promotion ankommt und wie ich mein Promotionsthema weiter erforschen könnte. Herrn Fichtner möchte ich außerdem stellvertretend für die schöne Atmosphäre am IIP danken, die geprägt ist durch einen konstruktiven, offenen und freundlichen Austausch. Weiterhin möchte ich gerne meiner Korreferentin Ingela Tietze danken, die meinen Werdegang seit dem Anfang des Promotionskollegs ENRES in 2016 bis zur Abgabe der Dissertation in 2020 begleitet hat. Darüber hinaus gebührt mein Dank auch meinem Prüfer Stefan Nickel und dem Vorsitzenden der Prüfung Marc Wouters.

Besonders bedanken möchte ich mich auch bei meinem langjährigen Gruppenleiter Russell McKenna, der einen großen Anteil zum Gelingen meiner Dissertation beigetragen hat. Die Einbindung in viele interessante wissenschaftliche Projekte hat mir frühzeitig wichtige Erfahrungen in Verbindung mit wissenschaftlichem Arbeiten und Publizieren verschafft. Du bist für mich ein großes Vorbild, was die menschliche und wissenschaftliche Führung einer Forschungsgruppe betrifft.

Hinsichtlich meines wissenschaftlichen Werdegangs muss ich mich auch bei Max bedanken: mit dir hatte ich beinahe täglich Diskussionen bezüglich unserer Forschungsthemen und viele wertvolle Ideen zu meiner Forschung stammen von dir. Ich kann mir wenige Menschen vorstellen, mit denen ich ähnlich gut zusammenarbeite.

Noch wichtiger als die fachliche Unterstützung ist für mich aber die menschliche durch meine Freunde und Familie, für die ich mich auch sehr bedanken möchte. Neben Max, den ich schon seit dem Studium in Aachen kenne, sind am IIP viele neue Freundschaften entstanden, von denen ich folgende besonders hervorheben möchte: Christoph, Daniel, Elias, Emil, Rafael, Richard, Thomas sowie meine ehemaligen Gruppenmitglieder Fritz, Kai, Phuong und Russell. Danke für all den Spaß und die Freude während dieser Zeit!

Die größte Stütze zu jeder Zeit seid aber ihr, meine Familie. Ihr wisst bestimmt, dass ich euch dafür immer dankbar bin, aber ich möchte euch auch hier nochmal meinen tiefempfundenen Dank aussprechen. Zunächst meinen tollen Eltern Jutta und Manfred, dem besten großen Bruder und langjährigen Vorbild Simon und meinen Großeltern Maria-Theresia, Kurt und Hannelore. Weiter auch den verrückten und liebenswürdigen Pia, Werner, Elisabeth, Harald, Miriam, Mathilda, Johanna, Oliver, Franziskus und Jill. Und abschließend natürlich Tim, Laura, Johannes und Max. Ihr alle macht mein Leben um einiges glücklicher, und ich bin sehr froh, es mit euch zu verbringen.

Mein größter Dank geht aber an Magdalena, meine Frau und gleichzeitig beste Freundin, die mich in allen Lebenslagen immer und ohne Einschränkung unterstützt. Ohne dich wäre sowieso gar nichts möglich. Ich liebe dich über alles! Unsere kleine Tochter Theresia hat sich auch noch auf diese Danksagung gemogelt, denn sie wollte es sich nicht nehmen lassen, in der Nacht vor meiner mündlichen Prüfung auf die Welt zu kommen, im Übrigen zwei Wochen vor dem errechneten Termin. Trotz fehlendem Schlaf hätte ich für diese Prüfung niemals besser vorbereitet werden können als durch deine Geburt, die mir Freude, Zuversicht und Gelassenheit geschenkt hat.

Abstract

Attention on decentralised autonomous energy systems has increased exponentially in the past three decades, as demonstrated by the absolute number of real-world projects and the share of publications in the corpus of scientific literature. This is due to the energy transition and the related environmental awareness as well as the desire of citizens to play an active role in energy supply and to be less dependent on central markets and structures. However, local decision-makers, who often lack the necessary expertise, need decision support in energy system planning. To this end, this thesis follows the objective to develop novel methods for the technical, economic and environmental assessment of a large number of completely energy autonomous municipalities are disconnected from the gas and electricity grid and supply themselves with energy from plants owned and operated by the municipality.

Novel methods of energy system analysis were developed in this thesis as part of seven original research articles. Germany is used as a case study, but the general approach, methods and results are transferable to other contexts. First of all, the 11,131 German municipalities were clustered with regard to their suitability for decentralised energy systems. Based on this municipality typology, representative municipalities were selected to be investigated in an already existing holistic municipal energy system optimisation model (RE³ASON). This model was extended by novel and transferable approaches to design deep geothermal plants and district heating networks. These base-load capable technologies were selected to reduce the storage costs in energy autonomous municipalities. The technical feasibility and economic expenditures of energy autonomy could finally be determined in all 11,131 German municipalities by combining the extended energy system optimisation model with a stepwise linear regression.

The energy system optimisations showed that in the case of complete energy autonomy, deep geothermal plants in combination with district heating networks could reduce the total costs by up to 50%. On average, the energy system costs until 2030 in German municipalities increase by about $0.41 \notin kWh$ in the energy autonomous case compared to the optimised reference case with grid connection. While a technical potential to achieve energy autonomy is present in 56% of the German municipalities, there seem to be no economic advantages through energy autonomy compared to the optimised reference energy system. The novel methodological approach of this thesis enabled to obtain optimisation results for a high number of energy systems (6,314 municipalities) with practicable computational expenses. In addition to the original data and planning tools published alongside the articles, the findings of this thesis can also support local decision makers in determining suitable municipal energy systems.

In order to increase the realizability of the case study results, some methodological extensions should be investigated in future studies such as other perspectives than that of a central planner, higher temporal model resolutions or social aspects like consumer acceptance of specific technologies or a security of supply below 100%.

List of appended papers

- Weinand, J. M.; Scheller, F.; McKenna, R. (2020b): Reviewing energy system modelling of decentralized energy autonomy. In *Energy* 203, 117817. DOI: 10.1016/j.energy.2020.117817. © 2020 Elsevier. Reprinted with permission.
- Weinand, J. M.; McKenna, R.; Fichtner, W. (2019a): Developing a municipality typology for modelling decentralised energy systems. In *Utilities Policy* 57, pp. 75–96. DOI: 10.1016/j.jup.2019.02.003. © 2019 Elsevier. Reprinted with permission.
- Weinand, J. M.; McKenna, R.; Mainzer, K. (2019b): Spatial high-resolution socio-energetic data for municipal energy system analyses. In *Scientific data* 6 (1), 243. DOI: 10.1038/s41597-019-0233-0.
- Weinand, J. M.; McKenna, R.; Karner, K.; Braun, L.; Herbes, C. (2019c): Assessing the potential contribution of excess heat from biogas plants towards decarbonising residential heating. In *Journal of Cleaner Production* 238, 117756. DOI: 10.1016/j.jclepro.2019.117756. © 2019 Elsevier. Reprinted with permission.
- 5) Weinand, J. M.; Kleinebrahm, M.; McKenna, R.; Mainzer, K.; Fichtner, W. (2019d): Developing a combinatorial optimisation approach to design district heating networks based on deep geothermal energy. In *Applied Energy* 251, 113367. DOI: 10.1016/j.apenergy.2019.113367. © 2019 Elsevier. Reprinted with permission.
- Weinand, J. M.; McKenna, R.; Kleinebrahm, M.; Mainzer, K. (2019e): Assessing the contribution of simultaneous heat and power generation from geothermal plants in off-grid municipalities. In *Applied Energy* 255, 113824. DOI: 10.1016/j.apenergy.2019.113824. © 2019 Elsevier. Reprinted with permission.
- 7) Weinand, J. M.; Ried, S.; Kleinebrahm, M.; McKenna, R.; Fichtner, W. (2020a): Identification of Potential Off-Grid Municipalities with 100% Renewable Energy Supply. In *Working Paper Series in Production and Energy*. DOI: 10.5445/IR/1000118013.

Contents

Abstra	act i
List of	f appended papersii
Conte	ntsiii
Figure	esv
Tables	SV
Part A	A: Overview1
1.	Introduction
2.	Background on energy autonomy
3.	Literature review on municipal energy autonomy
4.	Methodology
	4.1. Municipality typology
	4.2. Modelling of geothermal plants and district heating networks
	4.3. Energy system optimisations
	4.3.1. Extension of energy system optimisation model RE ³ ASON
	4.3.2. Transfer of results
5.	Results
	5.1. Municipality typology
	5.2. Contribution of excess heat from biogas plants to energy autonomy
	5.3. Design of municipal district heating networks
	5.4. Impacts of geothermal plants
	5.5. Complete energy autonomy in German municipalities
	5.6. Plausibility check of results
6.	Critical appraisal
7.	Conclusions and outlook
Public	cation bibliography
Part B	8: Articles
Article	e 1: Reviewing energy system modelling of decentralized energy autonomy
Article	e 2: Developing a municipality typology for modelling decentralised energy systems
Article	e 3: Spatial high-resolution socio-energetic data for municipal energy system analyses

Article 4: Assessing the potential contribution of excess heat from biogas plants towards decarbonising residential heating	98
Article 5: Developing a combinatorial optimisation approach to design district heating networks bas on deep geothermal energy	ed 122
Article 6: Assessing the contribution of simultaneous heat and power generation from geothermal plants in off-grid municipalities	139
Article 7: Identification of Potential Off-Grid Municipalities with 100% Renewable Energy Supply	165

Figures

Schematic representation of a decentralised energy system
Development of the number of 100%-Renewable-Energy-Communities in Germany since
1995
Development of the number of journal publications for two search queries in Scopus
Overview of the novel methodological approach of this thesis, divided into three parts 14
Schematic illustration of the geothermal plant considered in this study 16
Exemplary illustration of district heating networks for the use of excess heat from biogas
plants in several municipalities in Baden-Württemberg17
Possible locations for building the geothermal / district heating plant in the municipality
Groß Kreutz
Overview of RE ³ ASON and the extensions developed in three of the seven studies of this
doctoral thesis
Illustration of all German municipalities with their allocation in the 10 cluster solution 24
Share of heat demand that can be covered by district heating from biogas in German
municipalities
Illustration of 6,314 (56%) German municipalities that can become completely autonomous
and the associated $\Delta LCOE$
LCOEs of case studies with energy autonomous municipalities in the literature

Tables

Table 1: Overview of the most relevant studies for this thesis from Weinand et al. (2020b)	12
Table 2: Overview of the main novelties of each study of the present thesis	15
Table 3: Summary of the distinguishing characteristics of the ten municipality clusters	23

Part A: Overview

1. Introduction

In 2015, 197 countries agreed on the Paris Climate Convention, which includes limiting global warming to below 2 °C and reducing greenhouse gas emissions (BMWi 2019a). Germany, which is examined as case study in this thesis, aims to reduce its greenhouse gas emissions until 2050 by at least 80% below the 1990 level (BMWi 2019b). To achieve this objective, a decarbonisation of the energy sector is necessary, since this sector is responsible for about 80% of the greenhouse gas emissions (eurostat 2020). Increasing the share of low-emission renewable energy in the energy mix and increasing energy efficiency would contribute to this decarbonisation. In fact, the share of renewable energies in Germany has increased steadily in the past years: in 2019, renewable energies already accounted for 17.1% of gross energy consumption in Germany (UBA 2020). This expansion of renewable energy sources is mainly decentralised due to their characteristics. Thus, about 95% of the installed capacity of renewable energy plants in Germany has changed (trend:research 2017): the majority of renewable energy plants are actually owned and operated by private individuals, farmers and communities.

In this context of local renewable energy expansion, the concept of energy autonomy (cf. section 2) has become established especially for municipalities (McKenna et al. 2015; Rae and Bradley 2012). Many municipalities claim energy autonomy as their central motivation for investing in local renewable energies (McKenna 2018). Section 2 divides energy autonomy into balanced energy autonomy (i.e. annual energy supply by local renewable energy plants is at least as high as local demand) and complete energy autonomy (i.e. disconnection from the gas and electricity grid). Due to the possible impacts of balanced autonomous municipalities on the energy sector mentioned in section 2, the present thesis focuses on complete municipal energy autonomy, as this state could be more advantageous for the national energy system. Furthermore, in contrast to balanced energy autonomy, this state is less dependent on political framework conditions and thus more generally applicable conclusions can be drawn. In addition, novel methodological developments on this topic are also relevant for many other applications, for example in case studies of islands or communities that are not connected to the national transmission grid.

The associated questions to be answered in this thesis would include for which German municipalities complete energy autonomous systems would make economic sense and what the consequences would be for the overarching, mainly centralised energy system. For example, the necessary expansion of the transmission grid could be reduced by many completely energy autonomous municipalities. To analyse this in detail, it would first be necessary to examine the energy systems of a large number (> 1,000) of German municipalities, in order to identify municipalities in which complete energy autonomy based on renewable energies is feasible from a techno-economic perspective. To meet this central objective, it is necessary to develop an approach that allows to analyse a multitude of energy systems with practicable computational expenses. First of all, therefore, a typology of the energy systems to be investigated is

needed. Subsequently, selected energy systems that represent this typology have to be optimised from a techno-economic perspective. In the final step, the results of the optimisations must then be transferred to all energy systems. This approach is generally transferable to other case studies, in the case of this thesis it is applied to energy system analyses of completely autonomous municipalities. Thereby, this thesis also addresses the following research questions:

- 1) What must be improved in energy system analyses of autonomous systems (Weinand et al. 2020b)?
- 2) How can municipalities be grouped using socio-energetic indicators for the purposes of energy system analysis (Weinand et al. 2019a; 2019b)?
- 3) How do additional measures such as exploitation of excess heat potentials of the German biogas stock or technologies like district heating and base-load capable deep geothermal energy affect the technical feasibility and economic expenditure for municipal energy autonomy (Weinand et al. 2019c; 2019d; 2019e)?
- 4) In which municipalities is complete energy autonomy of all consumption sectors (industry, commercial, residential and private transport) technically feasible, and what are the associated costs of each of these municipal energy systems (Weinand et al. 2020a)?

In this thesis, novel methods for techno-economic energy system analysis are developed. The generated planning tools and planning knowledge on municipal energy system analysis can support local planners and decision makers in the design of municipal energy systems. Especially the typically smaller municipalities striving for energy autonomy require support in energy system analyses due to a lack of expertise and resources. In a first step, data on the 11,131 German municipalities are generated and compiled in order to cluster the municipalities with regard to their suitability for decentralised energy systems (Weinand et al. 2019a; 2019b). In addition, a generic model is developed to determine the potential of deep geothermal energy in arbitrary municipalities and to represent the economic and technical restrictions of this technology in an optimisation model (Weinand et al. 2019e). In connection with this, an approach based on combinatorial optimisation is developed for the design of district heating networks in municipalities in order to use the heat from deep geothermal plants in heating networks (Weinand et al. 2019c; 2019d). The combination of the latter two approaches opens the possibility to investigate the heat and electricity side of a deep geothermal plant in an optimisation model for the first time (Weinand et al. 2019e). Furthermore, an existing holistic municipal energy system optimisation model is extended by the above mentioned technologies as well as the consumption sectors industry, commerce, and private transport in the form of electric vehicles to investigate the technical feasibility and economic expenditures of a multitude of energy autonomous German municipalities (Weinand et al. 2020a). The additional consideration of heat grid infrastructure and alternative technologies such as deep geothermal energy provides local decision makers a comprehensive picture of the available resources at their disposal and their economic and technical impacts. Thereby it is important, that the approaches of energy system optimisation in this thesis are all automatically applicable to arbitrary German municipalities in order to be able to analyse a large number of municipalities. The results of the thesis are then to be applied, among other things, to examine scenarios with many energy autonomous municipalities in future national energy system designs and transmission network analyses. Although Germany is used as a case study in the present thesis, the approach and the newly developed methods are also suitable to be transferred to other contexts and countries with similar data availability.

This framework chapter is structured as follows. First, the concept, motives and consequences of energy autonomy are presented in the following section 2. The literature review in section 3 then shows the evolution of the research area of energy autonomy as well as research gaps filled by this thesis. Subsequently, in section 4 the methods of seven studies conducted for this thesis (Weinand et al. 2019a; 2019b; 2019c; 2019d; 2019e; 2020a; 2020b) are summarised before the results of these studies are presented in section 5. Section 6 then shows potential for improvement and possible future research projects. Finally, the thesis is summarised in section 7 and the most important findings are demonstrated.

2. Background on energy autonomy

In the growing body of literature on energy autonomy (cf. section 3), different terms are used to describe this concept: energy autarky (Müller et al. 2011), self-sufficiency (Engelken et al. 2016) or off-grid energy systems (Chmiel and Bhattacharyya 2015) are just a few of those terms. This number of terms alone illustrates the diversity within the literature, which also extends to its definition. Three rough distinctions can be made between complete energy autonomy (i.e. off-grid), balanced energy autonomy and a tendency towards higher energy autonomy through decentralised renewables (McKenna et al. 2015). Whereas in the case of complete energy autonomy no energy imports are permitted, in the case of balanced energy autonomy only the annual energy supply by local energy plants must exceed the annual demand. For the latter, energy imports are therefore permitted to compensate for supply gaps (McKenna et al. 2015).

In particular, autonomous energy systems need to exhibit the basic criteria defined by Rae and Bradley (2012). First, the local system is able to generate at least enough energy to meet the demand. Second, the local system allows energy shifting possibilities for times in which there is a temporal mismatch between demand and supply (i.e. through storage or in the case of balanced autonomy through energy infrastructure). Third, the system is capable of operating independently on a stand-alone or off-grid basis. Thereby, local autonomy efforts are related to active participation of the community or rather the system components are owned by the community members. In general, autonomy efforts are directly related to the notion of self-governance and community ownership (Rae and Bradley 2012). These previously mentioned definitions also apply to this thesis. Figure 1 shows a schematic representation of decentralised (autonomous) energy systems.



Figure 1: Schematic representation of a decentralised energy system (Weinand et al. 2020b). In contrast to the completely energy autonomous case (off-grid), imports from the national level are permitted in the case of balanced energy autonomy.

Whereas these autonomous energy systems exist in developing countries mainly due to cost considerations, there are also efforts by municipalities and regions to become energy autonomous in industrialized countries with complete electrification like Germany (Weinand et al. 2020b). This is due to the energy transition and the related environmental awareness (Engelken et al. 2016) as well as the desire of citizens to play an active role in energy supply and to be less dependent on central markets and structures (Boon and Dieperink 2014). The majority of municipalities with energy autonomy aspirations in Germany strive for balanced energy autonomy and the focus is usually on electrical energy (Engelken et al. 2016). An analysis of the German energy project "100%-Renewable-Energy-Communities", in which the participating municipalities strive for energy autonomy, confirms these findings of Engelken et al. (2016). According to Figure 2, the number of municipalities in this project has increased exponentially since 1995. In 2016, the 1,300 municipalities corresponded to 12% of all municipalities in Germany and account for 15% of the population (Weinand et al. 2019e).



Figure 2: Development of the number of 100%-Renewable-Energy-Communities in Germany since 1995. The grey areas in the maps show the municipalities.

Thus, as described above, a large number of German municipalities are striving for an autonomous energy system. Municipalities that become balanced energy autonomous would utilize the electricity network structure more than traditional consumers. Due to the volatile generation characteristics of renewable energy plants (such as wind and solar), large parts of the electricity generated would have to be fed into the national grid. On the one hand, the municipalities would benefit from the grid feed-in of surplus electricity. On the other hand, the municipalities would make a much lower contribution to the costs of this infrastructure, currently charged per unit of electricity used, due to lower export amounts (McKenna 2018). As a result, economic inefficiencies compared to the established system of centralised generation, transmission and distribution could emerge (Jägemann et al. 2013). Furthermore, the large expansion of renewable energies in the municipalities could make the expansion of the German electricity grid structures even more essential, as it would require the physical integration of the large amount of volatile renewable electricity (i.e. for power transports, Schaber et al. 2012). Alternatively, a nationally or even internationally coordinated expansion of renewable energies could limit the expansion of the electricity grid and the associated costs to a necessary extent. Based on these considerations, the question arises whether the development towards balanced autonomous municipalities will continue at

all. Otherwise, a change in political framework conditions (e.g. adjusted subsidy mechanisms / distribution of network fees) could bring this development to a halt.

3. Literature review on municipal energy autonomy¹

In the context of the decentralised expansion of renewable energies, the number of scientific publications on the topic of municipal energy system analysis is also increasing. The share of these municipal energy system analyses in all publications on energy system analyses has increased from 8% to 20% between 1990 and 2019. Similarly, the share of articles on energy autonomy (cf. dark curve in Figure 3) in publications on municipal energy system analyses (cf. bright curve in Figure 3) has increased from 0% to 15% between 1990 and 2019. Thereby the number of publications has increased exponentially in recent years (cf. Figure 3). Municipal energy system analyses in general and with a focus on energy autonomy are therefore becoming increasingly relevant, as well as the implementation of projects (see municipal energy autonomy projects in section 2). Therefore, this section is intended to provide an overview of the studies on municipal energy autonomy and to highlight research gaps which are addressed by the present thesis. Thereby, this section serves to show the novelty of this thesis as a whole. Further information can be found in the literature reviews of the individual articles (Weinand et al. 2019a; 2019b; 2019c; 2019d; 2019e; 2020a; 2020b).





Figure 3: Development of the number of journal publications for two search queries in Scopus (Weinand et al. 2020b). The bright curve contains the publications on energy system analyses at the local level. For the dark curve, the search query has been adjusted so that the studies also deal with autonomous energy systems. The search in Scopus was performed on November 20, 2019.

This literature review is largely based on the review study Weinand et al. (2020b). However, not all 123 discussed articles can be addressed at this point. Therefore, the most relevant results of the review study

¹ This literature review is largely based on the review study Weinand et al. (2020b).

as well as selected articles (cf. Table 1) are shown in this section. Peter (2013), Scheffer (2008) and Woyke and Forero (2014), which are not English language journal articles, have been added because they deal with energy autonomy in German municipalities and are therefore particularly interesting for this thesis. The criteria in Table 1 will be explained column by column in the following paragraphs.

The methods for investigating municipal energy autonomy range from simple energy balancing calculations (e.g. Chmiel and Bhattacharyya (2015) or Jenssen et al. (2014)) to linear optimisation (e.g. the present thesis, Alhamwi et al. (2019) or Šare et al. (2015)), multi-objective optimisation (e.g. Sameti and Haghighat (2018)) or multi-criteria decision analysis (MCDA, e.g. McKenna et al. (2018)). As is generally the case for energy system analyses, a compromise between scope and level of detail is necessary in order to be able to perform the optimisations at all or at least within a reasonable time. For example, Schmidt et al. (2012) and Sameti and Haghighat (2018), which both apply complex methods, use only two or eighteen hourly time slices for their calculations to obtain results for an entire year. Schmidt et al. (2012) compare energy autonomy with conventional energy supply in Sauwald, a rural region with 21,000 inhabitants in Austria. In order to include land use competition in energy system optimisation, this study combines the energy system model BeWhere with the land use optimisation model PASMA. In Sameti and Haghighat (2018), a net-zero energy district is investigated in three scenarios with grid connection and one as a stand-alone variant without grid connection. The district is examined with a bottom-up method, i.e. individual buildings and streets of the district are part of the analysis.

In most studies, only scheduling of the energy system technologies is considered. In such cases, longterm energy system planning is not really beneficial, since no investment decisions can be made. Instead, studies with investment decisions should always have a long-term planning horizon in order to consider possible future investments. These investments may be influenced by price developments or existing technologies such as already installed photovoltaic modules which have to be decommissioned in the future. Municipal energy autonomy studies that model these long-term investment decisions are the present thesis and McKenna et al. (2018). The study of McKenna et al. (2018) includes a participatory approach to developing feasible energy concepts for small municipalities. Thereby, for determining the optimal energy system in the case study municipality of Ebhausen, a linear optimisation is combined with a MCDA.

Furthermore, grid infrastructures are only marginally considered in municipal energy system analyses or local energy autonomy studies. In the FlexiGIS model in Alhamwi et al. (2019), OpenStreetMap is used to obtain data on lines and substations of the distribution network in order to determine the optimal placement of a battery storage in an urban area. Unfortunately, power grid data is not yet completely included in OpenStreetMap and therefore this method is only usable for very few case studies. In McKenna et al. (2018) the grid infrastructures are at least indirectly considered through network flows between various districts of a municipality. In Sameti and Haghighat (2018), designing of district heating

networks in a city district is included. An approach for the design of district heating networks was also developed in the present thesis (cf. section 4.2). It differs from the bottom-up approach of Sameti and Haghighat (2018) by designing the district heating network for an entire municipality (top-down) and at the same time optimising the location of the district heating plant. Furthermore, this approach is automatically transferable and applicable for each (German) municipality. The reason for the consideration of district heating in this thesis is shown in the next paragraph. However, as an emerging technology, district heating is also the most emerging technology in terms of research on energy system planning at the municipal level (Weinand 2020).

In order to cover the demand of a municipality without imports, either storage systems to flexibly use the fluctuating renewable energies, or base-load capable technologies are needed. Many studies come to the conclusion that energy autonomy is associated with very high costs. For example in Scheffer (2008), high investments are necessary for balanced energy autonomy in a rural model region with 10,000 inhabitants. Storable biomass is highlighted as the most important energy source for achieving autonomy. However, the agricultural land is not sufficient for an additional use of biomass as transport fuel. Jenssen et al. (2014) also come to this latter conclusion in their case study of an "average" German municipality. In addition, they show, that complete energy autonomy is technically achievable through the "Bioenergy Village" approach, albeit at high cost. Peter (2013) demonstrates that fluctuating renewable energies (PV, wind and solar thermal energy) could cover the electricity demand of an "example village" with 3,850 inhabitants but with tremendous storage costs. The author therefore concludes that base-load-capable technologies such as biomass, hydropower or deep geothermal energy should be included in energy system planning of autonomous municipalities. According to Schmidt et al. (2012), energy autonomy implies a decline in local production of food and feed as well as high costs for consumers. Complete energy autonomy in Pellworm, an island municipality in Germany with 1,100 inhabitants, which has been regarded as a model location for the construction of renewable energies, is assessed in Woyke & Forero (2014). Due to grid restrictions, complete energy autonomy is not feasible with the current energy system in Pellworm. Furthermore, the case study in Alhamwi et al. (2019) on Oldenburg, a German city with 165,000 inhabitants, shows that complete energy autonomy in a city of this size is technically and economically not practicable. These examples have led to the consideration of further technologies in this thesis, which could possibly lead to the technical feasibility and cost reduction of autonomous municipal energy systems. In this context, deep geothermal energy was identified as a potentially advantageous technology, as it is base load capable and has a high and widespread potential in Germany (Eyerer et al. 2020). In order to utilise the geothermal heat not only for electricity generation but also as a heating technology, an approach for the design of district heating networks in municipalities had to be developed in this thesis, as already indicated above.

More and more studies emphasize the importance of planning the siting of renewable energy plants together with the expansion of the transmission network (e.g. Neumann and Brown (2019)). For this

purpose, future scenarios are often designed on the development of the share of renewable energies in different regions. However, the simultaneous impact of potential completely energy autonomous municipalities on transmission grid planning has not yet been addressed. In general, studies on municipal energy autonomy have not yet examined any influences on the surrounding (national) energy system. The present thesis provides a starting point for this, by determining the technical feasibility and economic expenditures for complete energy autonomy in all 11,131 German municipalities (cf. section 5.5). As discussed in the outlook in section 7, (extreme) scenarios for future energy system analyses or transmission grid planning at the national level with many energy autonomous municipalities can be derived from this.

In addition to the above-mentioned novel methods and applications, such as the additional consideration of heat grid infrastructure and alternative technologies, this thesis presents a novel and transferable methodology to investigate a large number of municipalities with practical computational expenses (cf. section 4). This is a major improvement to the literature on decentralized energy autonomous energy systems, in which only individual energy systems have been investigated as case studies so far (Weinand et al. 2020b).

Table 1: Overview of the most relevant studies for this thesis from Weinand et al. (2020b) $(B = energy \ balancing, \ LP = linear \ programming, \ MILP = mixed-integer \ linear \ programming, \ MCDA = multi-criteria$ $decision analysis, \ MH = metaheuristics, \ MO = multi-objective, \ S = scheduling, \ I = investment \ planning, \ EN = electricity$ $network, \ HN = heat / gas \ network, \ F = fluctuating, \ BLC = base \ load \ capable)$

Study	Method	Planning	Transition	Network	Techno-	System	Transfer-
		type	pathway	infra- structures	logies	impacts	ability
Alhamwi et al. (2019)	LP	S, I	overnight	EN	F, BLC	X	(√)
Azaza and Wallin (2017)	MO, MH	S, I	overnight	×	F, BLC	×	×
Bonati et al. (2019)	MILP	S	overnight	×	F, BLC	×	×
Chmiel and Bhattacharyya (2015)	В	S	overnight	×	F, BLC	×	×
Dorotić et al. (2019)	MILP	S	long-term	×	F	×	×
Drysdale et al. (2019)	MILP	S	long-term	×	F, BLC	×	×
Jenssen et al. (2014)	В	-	overnight	×	BLC	×	×
Kötter et al. (2016)	MILP	S, I	overnight	×	F, BLC	×	×
Krajačić et al. (2009)	В	-	long-term	×	F	Х	×
McKenna et al. (2018)	MILP & MCDA	S, I	long-term	(EN), (HN)	F, BLC	×	\checkmark
Oldenbroek et al. (2017)	В	-	overnight	×	F	×	×
Østergaard and Lund (2011)	MILP	S	overnight	×	F, BLC	×	×
Peter (2013)	В	-	overnight	×	F	×	×
Peura et al. (2018)	В	-	overnight	×	F, BLC	×	×
Sameti and Haghighat (2018)	MO, MILP	S, I	overnight	HN	F, BLC	×	×
<i>Šare et al. (2015)</i>	MILP	S	overnight	×	F	×	×
Scheffer (2008)	В	-	overnight	×	F, BLC	×	×
Schmidt et al. (2012)	MILP	S, I	overnight	×	F, BLC	Х	×
Woyke and Forero (2014)	В	-	overnight	×	F, BLC	Х	×
This thesis	MILP	S, I	long-term	(EN), HN	F, BLC	(√)	\checkmark

4. Methodology

The frame of the thesis is the comprehensive literature review study Weinand et al. (2020b), which was already presented in part in section 3 and also includes large parts of the critical appraisal in section 6. The other six studies (Weinand et al. 2019a; 2019b; 2019c; 2019d; 2019e; 2020a) comprise the methodology of the present thesis, which can be divided into three parts (cf. Figure 4) and is explained in the following subsections. First of all, in order to investigate a large number of energy systems, a typology of these energy systems was needed. For this purpose, a municipality typology of the 11,131 German municipalities was developed to show the general suitability of municipalities for decentralised energy systems (cf. section 4.1). Subsequently, selected energy systems that represent this typology had to be optimised. Therefore, novel approaches were developed for estimating the potential and costs for deep geothermal plants and district heating networks (cf. section 4.2) as well as for implementing these technologies in a holistic optimisation model of a municipal energy system. The energy system optimisations as well as the necessary extensions of the optimisation model are explained in section 4.3. In the third step, the results of the optimisations were transferred to all German municipalities (cf. section 4.3.2). This overall approach (cf. Figure 4) is generally transferable to other case studies, in this thesis it is applied to energy system analyses of completely autonomous municipalities. The novelties of the individual articles and the methods developed in connection therewith are summarised in Table 2.

4.1. Municipality typology

A central challenge in energy modelling is to make compromises between model resolution, scope and computational feasibility. As already described in sections 1 and 3, the potential for energy autonomy should be determined in all German municipalities in this thesis. A detailed examination of all 11,131 German municipalities is not feasible in a reasonable amount of time. Weinand et al. (2019a) make a significant contribution to complexity reduction by clustering all 11,131 German municipalities using 34 pre-identified socio-energetic indicators. These indicators are mainly based on open data relating to the consumption sectors of Private Households and Transport, as well as indicators relating to the potentials for renewable energies. The method involves two main steps, namely a factor analysis and a cluster analysis. For the former, different methods are compared with each other, and the most effective methods for allocating the indicators to factors are chosen. Selected cluster validation methods are then used to determine an appropriate number of clusters to which the 11,131 municipalities are distributed.

The study not only applies cluster analysis to all 11,131 German municipalities for the first time, but also uses novel indicators to classify the municipalities. Therefore, many of the indicators, such as hydrothermal potential at 5000 m depth or wind energy potential, were determined at the municipal level. The potentials of renewable energies were assigned to the individual municipalities using the *Geographical Information System* (GIS) *QGIS*. Since this and the allocation of census data to the municipalities involved a great effort, the whole data set was published open access in Weinand et al. (2019b).



Figure 4: Overview of the novel methodological approach of this thesis, divided into three parts. Parts of the figure are from Weinand et al (2019a) and Weinand et al (2019e).

Study	Main novelties				
	Methodological	Application-related			
Weinand et al. (2019a)	 Determination of wind potential and deep geothermal energy potential at the municipal level 	 Classification of all 11,131 German municipalities Clustering with socio-energetic indicators 			
Weinand et al. (2019b)		 Open access publication of the calculated and consolidated data from Weinand et al. (2019a), Mainzer et al. (2014) and McKenna et al. (2014) 			
Weinand et al. (2019c)	 Combination of a survey on biogas plants excess heat fraction with a heuristic to connect heat sources (biogas plants) with heat sinks (settlements) Determination of the heat demand of all 38,414 settlements in Germany Heuristic to design up to 1.5 million district heating pipelines simultaneously 	 Case study in Germany to determine the potential of utilising excess heat of the 10,446 biogas plants in residential buildings Determining contribution of biogas excess heat to energy autonomy in the 11,131 German municipalities 			
Weinand et al. (2019d)	 Simultaneous optimisation of district heating network and location of district heating plant in municipalities Combinatorial optimisation problem based on the theory of minimum spanning trees with several adjustments Three-stage heuristic to make the problem feasible for large municipalities Heuristic automatically applicable to arbitrary municipalities 	 District heating network based on deep geothermal energy Application of the problem to several German municipalities 			
Weinand et al. (2019e)	 Method to assess the hydrothermal potential in every German municipality Generic approach for modelling deep geothermal plants Integration of the geothermal plant model and the district heating design from Weinand et al. (2019d) in a holistic energy system optimisation model in order to simultaneously optimise heat and electricity generation 	 Application of the energy system model to several German municipalities Determination of costs for achieving complete energy autonomy in the residential sector 			
Weinand et al. (2020a)	 Modelling approach to consider the demand of commercial, industrial and personal transport sector in arbitrary municipalities Combination of energy system optimisations and regression analyses to transfer results to other municipalities 	 Determination of the technical feasibility and economic expenditure for complete energy autonomy in all 11,131 German municipalities 			
Weinand et al. (2020b)		 Comprehensive overview and quantitative analysis of applied methods in studies on decentralised energy autonomy Compilation of costs for 			
		decentralised energy autonomy			

Table 2: Overview of the main novelties of each study of the present thesis.

4.2. Modelling of geothermal plants and district heating networks

Section 3 already demonstrated that many case studies on municipalities show a limited base load potential (e.g. land-use restrictions for biomass), which could lead to high costs for energy autonomy. Additional base load capable technologies like deep geothermal plants could therefore be an option to reduce these costs. Therefore, a novel generic optimisation model of a geothermal plant (cf. Figure 5) is developed in Weinand et al. (2019e), which for the first time simultaneously optimises the electricity (P_{el}) and heat (\dot{Q}_{th}) side of the the plant (cf. Eqs. 1-2).

$$\dot{V}_B \cdot \rho_W \cdot c_{p,W} \cdot \left(\boldsymbol{T}_{PW}(t) - \boldsymbol{T}_{ORC,out}(t) \right) = \boldsymbol{P}_{el}(t) / \eta_{el} \qquad \forall t \qquad 1$$

$$\dot{V}_B \cdot \rho_W \cdot c_{p,W} \cdot \left(\boldsymbol{T_{ORC,out}}(t) - \boldsymbol{T_{DHP,return}}(t) \right) = \dot{\boldsymbol{Q}}_{th}(t) / \eta_{th} \qquad \forall t \qquad 2$$

with

₿ V _B	_	volumetric flow rate of the geothermal water
ρ_{w}	_	density of the geothermal water
c _{p,w}	-	heat capacity of the geothermal water
T _{PW} (t)	_	temperature of the geothermal water before utilisation in the electricity plant in time slice t
T _{ORC,out} (t)	_	temperature of the geothermal water after utilisation in the electricity plant in time slice t
T _{DHP,return} (t)	_	temperature of the geothermal water after utilisation in the heating network in time slice t
η _{el}	_	efficiency of the electricity generation
η_{th}	_	efficiency of the heat utilisation

The optimisation variables are shown in bold, the others are parameters. Variable drilling depths and thus geothermal water temperatures represent some of the novel modelling approaches. As input for the optimisation, a linear regression estimates the achievable geothermal water temperatures and the required drilling depths in the municipalities. Some cost estimations for the geothermal plant, such as drilling costs, had to be linearised for this purpose.



Figure 5: Schematic illustration of the geothermal plant considered in this study. The heat flow in the condenser of the Organic Rankine Cycle process could be water or air. Adapted figure from Weinand et al. (2019e).

Besides the Organic Rankine Cycle plant for electricity generation, district heating networks are necessary to utilise the heat from the deep geothermal plant within buildings. In addition, as already mentioned in section 3, district heating is one of the key research subjects in the field of municipal energy system planning. Therefore, approaches for district heating network modelling in municipalities were developed.

In a first study on district heating, the technical potential for utilising excess heat from biogas plants is analysed (Weinand et al. 2019c), in order to supply local settlements through district heating. Based on a survey of around 600 German biogas plant operators, the fractions of excess heat from the cogeneration unit in these plants have been analysed. The analysis was carried out for the surveyed population as well as scaled up to the whole German biogas plant stock. A heuristic was developed to connect biogas plants (heat sources) with local settlements (sinks) in order to determine a least-cost district heating supply for residential buildings. A heuristic is necessary since more complex methods like linear optimisations are hardly applicable for these problem sizes. In total, 10,446 biogas plants and 38,414 settlements were included and up to 1.5 million district heating pipelines could be designed simultaneously in the analysis for Germany. Thereby two criteria were employed, namely the CO_2 abatement costs and the payback period, which represent the macro- and microeconomic perspectives respectively. By taking these criteria into account and balancing heat supply and demand, the heuristic can cover different district heating network configurations (cf. Figure 6).



Figure 6: Exemplary illustration of district heating networks for the use of excess heat from biogas plants (green circles) in several municipalities in Baden-Württemberg (Weinand et al. 2019c). The red district heating pipelines lead from a biogas plant to a settlement area (brown shapes) and the blue ones lead from one settlement area to another. The black lines represent municipal borders².

² The background map is from OpenStreetMap Contributors (<u>https://www.openstreetmap.org/</u>).

Either settlements are supplied with heat by one biogas plant (biogas plants 3, 5, 8 in Figure 6) or several biogas plants (1, 2, 5, 6 and 9), or one biogas plant provides heat to several settlements (4, 5, 6), or the biogas plant does not supply heat to any settlement (7). This approach allows to determine the possible contribution of excess heat from biogas plants to the energy autonomy of German municipalities.

The algorithm from Weinand et al. (2019c) can only be used for existing district heating plants. However, for the use in a municipal energy system analysis, in which the location of a potential district heating plant has yet to be determined, the algorithm needs to be extended. This and further extensions of the algorithm are the subjects of Weinand et al. (2019d). Against the background of a trend towards decentralised and community-owned energy systems, the study developed a method to set up a minimum-cost geothermal-based municipal district heating system. To this end, two approaches based on combinatorial optimisation were presented, in order to support local planners in the design of geothermal district heating systems. The first approach involved a combinatorial optimisation of the district heating network layout, including geothermal plant location and network topology, which is only applicable to municipalities with less than eight discrete settlement areas due to its computational complexity. The second approach is a three-stage heuristic, which serves the same purpose but can be applied to a much larger number of municipalities with many more settlement areas. One of the innovations of the developed optimisation model and the three-stage heuristic compared to previous work is the fact that not only the district heating network but also the location of the district heating plant, is optimised. Thereby, the optimal location is selected from a discrete number of possible locations in the municipality (cf. Figure 7). Furthermore, the nodes / settlements to be connected are not fixed in advance and do not have to be supplied completely with heat.



Figure 7: Possible locations (purple circles) for building the geothermal / district heating plant in the municipality Groß Kreutz (Weinand et al. 2019d). The pink areas are the settlements of the municipality. Inadmissible locations in forests, settlements and water areas are automatically excluded by the algorithm².

This approach for modelling deep geothermal plants and district heating networks presented in Weinand et al. (2019d) and Weinand et al. (2019e) can be applied to every municipality in Germany and the

methodology could be extended to an arbitrary country with equivalent data. The 38 other European countries are particularly suitable for this extension due to the availability of *Corine Land Cover* data which is used to represent the settlement areas. Finally, the geothermal plant model as well as the district heating heuristic are implemented in a holistic energy system optimisation model (RE³ASON, cf. section 4.3.1)

4.3. Energy system optimisations

For the determination of the optimal municipal energy system designs in this doctoral thesis, the *Renewable Energies and Energy Efficiency Analysis and System OptimisatioN* (RE³ASON, Mainzer (2019)) model was used and further extended (cf. section 4.3.1). Since this model is applied in Weinand et al. (2019e) and Weinand et al. (2020a), section 4.3.1 first describes it alongside the extensions developed in this thesis. Subsequently, the method for transferring the model results to all German municipalities is explained in section 4.3.2.

4.3.1. Extension of energy system optimisation model RE³ASON

An overview of the two parts "input data determination" and "energy system optimisation" of the RE³ASON model are shown in Figure 8. In the first step of the model ("Input data determination") the required input data are calculated with the use of a Java model (Eclipse). The input data are applied in the second step, the actual optimisation model, which is implemented by using the General Algebraic Modeling System (GAMS). The RE³ASON model consists of several parts, which provide transferable methods for determining the existing technologies, infrastructure, the heat and electricity demand of residential buildings as well as the potential and associated costs for energy supply from photovoltaic (PV), wind and biomass in an arbitrary location. Due to this high transferability, the model is applied in the present doctoral thesis, as many municipalities in different locations are investigated. RE³ASON further provides a deterministic model of optimal investment and dispatch for new energy conversion technologies at the municipality level. In the mixed integer linear program (MILP), the optimal technology investment and unit commitment of all technologies as well as energy flows between districts are identified. The model serves to cope with the complexity resulting from the number and combinations of the individual measures and their dependencies that would otherwise not be feasible. Included in the model are the above-mentioned energy supply technologies as well as measures such as insulation, heating technologies or appliances. A municipality under consideration is divided into districts, in which buildings are grouped into building types according to the TABULA building typology (IWU 2015). The spatial resolution consists of these districts as nodes to which the input, like heat and power demand, is assigned. In this thesis, the model is used to perform a long-term energy system optimisation (from 2015 to 2030), whereby each 5th year is modelled explicitly and divided into 108 time slices (4 seasons, 3 day types, 9 time slices within each day). The model can be used to minimise total discounted system costs, CO₂ emissions or energy imports of the municipal energy system. The RE³ASON model is only employed for minimising the total discounted system costs in this thesis, since complete renewable energy autonomy as a prerequisite simultaneously leads to a minimisation of the other two target criteria.

The RE³ASON model uses the macroeconomic perspective of central planners who develop concepts for the design of energy systems in municipalities. It takes into account all decision-relevant expenditures within the energy system, regardless of which actor is responsible for these expenditures. For example, the optimal investments of private households in new heating systems are determined from a system perspective without considering the economic viability of these investments from the perspective of the individual household. Within this macroeconomic perspective, taxes, subsidies and levies are considered as a redistribution of costs and are therefore not considered in the cost calculation. This means, for example, that the Renewable Energy Sources Act levy is not included in the electricity price for consumers, but at the same time the owners of renewable energy plants do not receive any feed-in remuneration for electricity generation. The reason for this approach is that the legislative situation, which showed frequent changes in the past, cannot be assumed to be constant over the comparatively long-term time horizon of the model. Furthermore, this allows a neutral comparison between individual technologies and measures. Regarding the prices of the energy sources, the model only takes into account the costs of procurement and distribution as well as grid fees (Mainzer 2019). Taxes and levies, which in 2019 accounted for about 52% of the household electricity price as well as about 26% of the natural gas price (Bundesnetzagentur 2019d), are also regarded as redistribution of costs and are therefore not considered in the economic evaluation (Mainzer 2019). For further information about the model including the calculation of renewable energy potentials and the mathematical model formulation the reader is referred to McKenna et al. (2018) and Mainzer (2019).

The RE³ASON model extensions shown in Figure 8 are explained in the specific studies of this thesis and the previous subsections. The extensions are implemented within the program environment of the existing RE³ASON model and comprise

- 1) the design of district heating networks (Weinand et al. (2019d), cf. section 4.2),
- 2) the implementation of potential determination and model equations for deep geothermal energy systems as well as the integration of extreme days (Weinand et al. (2019e), cf. section 4.2),
- 3) the modelling of the industrial and commercial sectors as well as the integration of electric vehicles (Weinand et al. 2020a, cf. section 4.3.2).

In addition, some yet unpublished extensions have been made:

- 1) Some errors in the old model version have been fixed (e.g. for PV potential determination).
- 2) Development of the possibility to determine the input data (e.g. renewable potentials) as well as to optimise the energy system for several municipalities simultaneously.

3) Integration of Value of Lost Load:

The *Value of Lost Load* (VoLL) addresses the economic consequences of power blackouts and the monetary evaluation of uninterruptedness of electricity supply. Various factors influence the level of VoLL: in industrial or commercial enterprises, for example, production holdups or restart times have an influence, whilst in the residential sector lost leisure time is a significant issue (Schröder and Kuckshinrichs 2015). Particularly in the case of energy autonomy, when the LCOEs increase sharply, the idea of dispensing with energy demand in order to save costs is imaginable. Therefore, the VoLL was implemented in the RE³ASON optimisation model version from Weinand et al. (2020a). The amount of energy demand that is dispensed with (*Lost of Load* (*LoL*)) in the respective sector *s* is limited by the energy demand *ED* in this sector in each time step *t* (cf. Eq. 3).

$$LoL_{s,t} \le ED_{s,t}$$
 $\forall t = 1, ..., T; \forall s = [residential, commercial, industrial]$ 3

The VoLL are then used to assign costs to the LoL in the objective function. In line with CEPA (2018), the VoLL were set to $12.41 \notin k$ Wh for the residential sector, $12.34 \notin k$ Wh for the commercial sector and $1.81 \notin k$ Wh for the industrial sector.

	Input data determination (Java, Eclipse)	Energy system optimisation (GAMS)
RE ³ ASON	 Simulation of energy service demand Appliances Lightning Room comfort Identification of existing technologies Energy generation plants Heating technologies Identification of existing infrastructure (gas, electricity grid, etc.) Calculation of technical and economic renewable energy potential for Biomass, PV and Wind 	 Optimisation of cost for energy system (CO₂ emissions and other target criterias also possible) 4 model years, 72 time slices Time resolution: hourly Spatial resolution: districts Energy imports (electricity, coal, gas, oil), which are prohibited in case of complete energy autonomy Electricity demand has to be covered in every time step Heat demand has to be covered per day, as buildings are considered as natural heat storage units
RE ³ ASON Extensions	 Identification of geothermal potential (borehole temperature and drilling depth) in 8.5 km² resolution District heating designing (Weinand et al. 2019d): Determination of a cost-effective district heating network design for different heat coverages and the associated costs Electricity demands for industrial, commercial and private transport sector (Weinand et al. 2020a) 	 Geothermal plant model equations (Weinand et al. 2019e) Technical Economic Addition of extreme days (Weinand et al. 2019e) Technical model equations for electric vehicles (Weinand et al. 2020a)
		This doctoral thesis

*Figure 8: Overview of RE*³*ASON and the extensions developed in three of the seven studies of this doctoral thesis. Adapted figure from Weinand et al. (2019e).*

4.3.2. Transfer of results

In Weinand et al. (2020a), a methodology is developed to determine the feasibility and costs for complete municipal energy autonomy of all consumption sectors. First, methods for estimating the energy demand and potential for renewable energies are proposed. On this basis, municipalities in which complete energy autonomy is not feasible can be excluded. Subsequently, the RE³ASON model is extended to include the industrial, commercial and personal transport sectors. Based on the municipality typology from Weinand et al (2019a), 15 representative municipalities are selected. The RE³ASON model is then applied to determine the cost-minimal energy system for these 15 preselected municipalities as case studies. On the one hand for the reference case without autonomy and on the other hand for the case with complete municipal energy autonomy. In the reference case, the energy system is optimised without restricting imports and exports. Afterwards, the Levelized Cost of Energy (LCOE) are calculated for both cases and all preselected municipalities (cf. Eq. 4, Grosspietsch et al. (2018)). Thereby the conversion factor for electricity into heat is assumed to be the heat pump's coefficient of performance (3.5) as in Grosspietsch et al. (2018), since the heat load is taken into account for the residential sector.

$$LCOE = \frac{\sum_{y=1}^{Y} \frac{CAPEX_{y} + OPEX_{y}}{(1+r)^{y}}}{\sum_{y=1}^{Y} \frac{E_{m,total,y}}{(1+r)^{y}}}$$
4

The LCOEs are calculated depending on the investments (CAPEX), the operational and maintenance costs (OPEX), the total energy demand $(E_{m,total})$ and the year y. The interest rate r is assumed to be 5%.

A regression is used to transfer the results of the 15 case studies to the entire municipality population. The dependent variable is the difference between LCOEs in the autonomous and in the reference case (Δ LCOE). As independent variables the annual demand of each energy consumption sector of the municipalities and the indicators from the cluster analysis in Weinand et al. (2019a) are used. Those independent variables that correlate with other variables are eliminated. Therefore, for all correlations with an absolute value above 0.9 one variable is excluded.

To avoid an overfitting in the regression, a k-fold cross-validation is applied (Zhang and McCalley 2018). Since the sample is small (n = 15), the leave-one-out cross-validation is used, with k = n = 15. 19 different methods are applied, ranging from linear regression models and support vector machines to Gaussian Process Regression models. From these methods, the model that results in the lowest root mean squared error is selected.

5. Results

The subsections 5.1-5.5 summarise the results of the studies which have been conducted to fulfill the objectives of this doctoral thesis (Weinand et al. 2019a; 2019b; 2019c; 2019d; 2019e; 2020a; 2020b). These results were obtained by applying the methods explained in section 4. Since the next section 6 discusses the thesis in its entirety, the individual studies are only briefly critically assessed in sections 5.1-5.5. The validations and plausibility checks of the methods and results, which were performed in detail within the studies, are also only discussed in a few cases in this framework chapter (e.g. section 5.6). Besides the methodological innovations, Table 2 also provides an overview of the novel applications in all studies.

5.1. Municipality typology

The following results are from the studies Weinand et al. (2019a) and Weinand et al. (2019b) and based on the cluster analysis explained in section 4.1. With the help of the cluster validation methods (cf. section 4.1), an appropriate number of ten clusters could be determined, to which the 11,131 German municipalities are distributed (cf. Figure 9). Due to the high number and differentiation of indicators, clusters overlap with each other for different indicators, but the results also show significant differences between the clusters. Table 3 summarises these distinguishing characteristics for all clusters. For example, Cluster 2 contains all major German cities and most of the other cities in Germany and has a low potential for renewable energies. Cluster 9, on the other hand, describes all German municipalities without inhabitants.

Cluster	Number of	Characteristics
	municipalities	
1	339	Larger towns with highest share of district heating.
2	727	All major German cities with particularly low potential for renewables.
3	1638	Municipalities with highest hydrothermal potential, high income per household as well as low unemployment rate.
4	839	Municipalities with high hydrothermal potential, building age and unemployment rate.
5	5262	"Average" Cluster containing the majority of municipalities. Municipalities with high number of cars per 1,000 inhabitants and very low share of district heating.
6	1370	Municipalities with high building age and high proportion of people over 65 years of age.
7	460	Municipalities with lowest household density, highest number of cars and motor cycles per 1,000 inhabitants, largest share of detached houses and particularly high potential for renewables.
8	388	Municipalities with low building age, lowest proportion of people over 65 years of age and a high hydrothermal potential.
9	75	Rural municipality-free areas without inhabitants and lowest potential for renewables.
10	33	Smallest cluster containing municipalities with high population growth.

Table 3: Summary of the distinguishing characteristics of the ten municipality clusters.



Figure 9: Illustration of all German municipalities with their allocation in the 10 cluster solution (Weinand et al. 2019a). The numbers of municipalities in the clusters are shown in parentheses.

Assigning the municipalities from the three German energy projects "Energy Municipalities", "Bioenergy Villages" and "100% Renewable Energy Regions" to the ten clusters shows that in eight of the ten clusters municipalities are aiming for energy autonomy (in varying degrees). As a result, it is not possible to differentiate between the clusters regarding readiness for such energy projects, which is most likely due to the influence of non-technical factors on the emergence of these initiatives. However, the results of the cluster analysis show that some of the municipalities could be technically more suitable for energy autonomy. For example, Clusters 3, 4, 7 and 8 are characterised by a high potential for renewable energies. The results reduce the effort of subsequent studies, as only a few municipalities from the clusters need to be examined regarding their suitability for energy autonomy to be able to make statements for all municipalities of the cluster. Thereby, the results help to identify municipalities in which already successful measures from other municipalities could be applied, and provide a basis for further energy analyses at the national level.

The methodology used in Weinand et al. (2019a) could be improved for more accurate results in future work. On the one hand, further indicators should be considered, including more indicators from the industrial and commercial sectors as well as indicators relating to the local climate. However, this is challenging due to the lack of available data at this spatial resolution. Furthermore, weights for the indicators could be determined with the help of expert interviews. If the indicators are known, which

have the most considerable influence on the suitability for energy autonomy, these can be weighted more strongly in the cluster analysis and a new set of clusters generated based on these weights.

The collection and preparation of the data in Weinand et al. (2019a) involved a great effort. In addition, the field of energy system modelling is a science that is not yet fully transparent, as models and data are often not published in open access (Pfenninger 2017). To help overcome this hurdle, an open access data set including the 34 socio-energetic indicators from Weinand et al. (2019a) for all 11,131 German municipalities is provided alongside the Data Descriptor article Weinand et al. (2019b). Most of these indicators originate from the cluster analysis described above. In addition to census data such as population density, mobility data such as the number of vehicles and data on the potential of renewables such as wind energy are included. The data set can support in answering a wide range of energy-related research challenges. In a recently published handbook of the European Union (Siragusa et al. 2020), for example, the data set is recommended for reviewing local *Sustainable Development Goals*.

5.2. Contribution of excess heat from biogas plants to energy autonomy

This subsection presents the results of Weinand et al. (2019c), which is based on the original methodology for designing many district heating networks simultaneously (cf. section 4.2). Based on the survey with 600 biogas plants, a mean fraction of 40% excess heat was determined for these plants, which is in agreement with other empirical studies. Extrapolating this fraction to the German biogas plant stock leads to technically feasible CO₂ savings of around 2.5 MtCO₂/a. This involved the simultaneous design of a total of 10,900 district heating networks. Employing the criteria of CO_2 abatement costs and payback period yields about 2 MtCO₂/a below CO₂ abatement costs of 200 €/tCO₂ and 9 years respectively. These relatively high average costs are related to the typically low population density in rural regions where biogas plants are located. The potential CO₂ savings represent about 0.25% of the total German CO₂ emissions in 2016 or around 2.5% of all CO₂ in residential buildings. If a threshold value of 80 €/tCO₂, to reflect the German government's suggested external cost of carbon in 2018, is employed, the carbon reduction potential is about 0.5 MtCO₂. Similarly, a threshold for the expected payback period of 5 years, to reflect an investor's point of view yields potential savings of about 0.75 MtCO₂. These potentials are concentrated in around 3,500 municipalities, where district heating from biogas plants could reduce CO_2 emissions per capita by an average of around 250 kg CO_2/a and cover 12% of the total residential heating demand. In some of these municipalities (cf. Figure 10), large proportions of their heating demand could be economically met (according to the criteria employed here) by this excess heat, hence assisting in the transition to more decentralised autonomous energy systems. On the other hand, if the current price of CO_2 in the EU Emissions Trading System of about 7 €/tCO₂ is taken as a benchmark, the economic fraction of this technical potential saving reduces to 0 tCO₂. Although these results are relatively modest in the overall context of decarbonising the energy system, this study does provide a quantitative basis for decision makers, researchers and energy planners.



*Figure 10: Share of heat demand that can be covered by district heating from biogas in German municipalities (Weinand et al. 2019c). The numbers in brackets represent the number of municipalities to which the shares can be allocated*².

The developed method was applied to Germany as a case study, but could be equally relevant for other countries with high biogas production. Some of these countries (e.g. Czech Republic) have a significantly more widespread district heating network than Germany. Thus, the use of biogas excess heat could lead to similar or even better results for these countries than in the case study for Germany. Apart from a methodology that can be transferred to any country with similar data availability, the study therefore demonstrates that the use of excess heat in biogas plants can be one contribution towards a global energy system decarbonisation.

The employed methodology, whilst adequate for a national estimation of the technical potential and associated costs, has several uncertainties. Most importantly, the shortest birds-eye route from the biogas plant to the centroid of the settlement is used as the required distance for district heating pipelines. Whilst a good estimate for the order of magnitude, this obviously leads to over- and underestimation of the required district heating pipeline length, and therefore the costs. In addition, the developed heuristic does not (necessarily) determine the optimum allocation of heat sources to heat sinks, and may also therefore overestimate the costs. Finally, the focus on residential buildings and the rough consideration of existing district heating supply (but not infrastructure) add additional uncertainties. All of these aspects remain areas for future work.

5.3. Design of municipal district heating networks

The extension of the previously described district heating algorithm for individual municipalities (cf. section 4.2) was part of Weinand et al. (2019d). A comparison of optimisation and heuristic for designing district heating networks in three exemplary municipalities demonstrates the efficiency of the developed heuristic. For municipalities with three, five and seven settlements respectively, the optimisation takes between 500% and 1×10^7 % more time than the heuristic. The resulting deviations in the calculated total investment for the district heating from the results of the optimisation are in all cases below 5%, and in 80% of cases below 0.3%. The efficiency of the heuristic is also demonstrated by the comparison with the Nearest-Neighbour-Heuristic. The latter is not only less efficient, it substantially overestimates the total costs by up to 80% in all cases with less than 100% heat coverage. In addition, the calculated investments in the investigated municipalities ranged from 500 \notin /kW to 1,900 \notin /kW, values which could be validated with investments for existing geothermal district heating networks in Germany.

The developed heuristic consistently yields results within acceptable margins of error of its equivalent combinatorial optimisation problem, is efficient and scales well to other regions or contexts. The methodology would benefit from some further improvements, for example some of the technical aspects such as heat and pressure losses within the district heating network could be modelled more precisely in the heuristic. Furthermore, additional geological and topographical conditions in the municipalities should be taken into account in order to better identify the optimal location of the geothermal plant and the type of network. Additionally, the heuristic should be extended in such a way that the district heating pipelines can also branch off in order to reach several endpoints from one starting point. All of these aspects remain areas for future work. Finally, the results of the heuristic could be compared to the results of the optimisation only for small municipalities (less than 8 settlements). A LP-relaxation of the optimisation problem could help to evaluate the performance of the heuristic also for larger municipalities.

Notwithstanding these shortcomings, the developed method provides a sound basis for decision support for municipal-scale geothermal district heating systems. The heuristic for cost-optimal placement of the geothermal plant (provided as supplementary material of the study) can be extended and should offer useful insights for local planners and authorities when considering the heat source options at their disposal. In addition to supporting the planning of municipal district heating networks, the heuristic can also be used to design district heating networks in holistic energy system optimisations due to the novel possibility of connecting an arbitrary number of buildings to the district heating network.

5.4. Impacts of geothermal plants

In Weinand et al. (2019e), the RE³ASON model was applied for the first time in this thesis to determine the cost-optimal energy system of an autonomous municipality. The focus here is on the residential sector and the industrial, commercial and transport sectors are not considered yet. The main purpose of
this study is to assess the influence of deep geothermal energy systems on the technical feasibility and economic expenditures of municipal energy autonomy. The methodology was explained in sections 4.2-4.3.

The specific research questions addressed in Weinand et al. (2019e) are as follows:

- Could the high costs for completely autonomous municipal energy systems be reduced through the use of geothermal plants?
- Is it sufficient to consider only the electricity generation of the geothermal plant or would the use of the geothermal heat in district heating networks create an additional benefit?

In order to answer these questions, the developed optimisation model was applied to three municipalities from different municipal clusters. Eleven scenarios demonstrated that achieving energy autonomy in the residential sector is (partly) associated with high additional costs. Compared to the scenarios without energy autonomy, total discounted system costs for the period between 2015 and 2030 have increased by at least 4%. Thereby, the utilisation of geothermal plants can significantly reduce the costs for achieving energy autonomy, which answers the first research question above. The electricity generation is preferred to heat generation in geothermal plants, which is related to the high costs for the district heating network. However, the importance of simultaneous modelling of electricity and heat generation in geothermal plants reduce the costs, especially in municipalities with high hydrothermal potential. This provides an answer to the second research question i.e. that in the context of municipal energy system planning it is not sufficient to only consider the electricity side of the plant. Therefore, the installation of geothermal plants could help to decarbonise the energy system through energy autonomy.

Weinand et al. (2019e) have developed a generally-applicable method for the optimal setup of a geothermal plant within or near a residential area and considering both heat and electricity generation. Together with the related contribution for optimally locating the geothermal plant within an existing or new district heating network (Weinand et al. 2019d), the consideration of the heat side represents a significant step forward. Compared to previous studies that focussed on a detailed geothermal plant system setup, typically optimised for power generation, Weinand et al. (2019e) adopt a more holistic approach. Weinand et al. (2019e) and Weinand et al. (2019d) together provide a methodological framework for the economically effective and energetically efficient integration of geothermal plants into local energy systems. In the context of renewable energy system planning this therefore represents an invaluable tool in the context of the energy transition. A validation of the cost and the input determination with data from actual plants shows that the model presented in this work can reasonably be applied to any municipality in Germany and, by employing additional data sources, beyond. It can provide decision support to local energy planners and other relevant stakeholders when considering the renewable energy options at their disposal.

Due to the fact that the employed methodology is intended as an early-stage planning tool, it has several uncertainties, however. Hence the authors emphasize the need for a more detailed energy system planning, especially but not only relating to the geothermal plant, before entering the implementation phase. Most importantly, the costs of geothermal plants are very uncertain and depend on local geological conditions. Whilst the model presented provides a good estimate of the hydrothermal temperatures, the investment can rise due to uncertain incidents. Furthermore, the social acceptance of the communities for deep geothermal energy has to be assessed in future studies.

5.5. Complete energy autonomy in German municipalities

The previously described study (Weinand et al. 2019e) focused on complete energy autonomy in the residential sector of a municipality. However, in order to be separated from the German power grid and thus for national transmission grid planning purposes, the entire municipality would have to become energy autonomous, i.e. all consumption sectors have to be included in the energy system analyses. This was achieved in Weinand et al. (2020a), in which RE³ASON is extended to take these additional sectors into account (cf. section 4.3.1) and which uses a methodology for transferring optimisation results to other municipalities (cf. section 4.3.2).

Germany has been selected as case study, where 6,314 (56%) municipalities were identified, in which complete energy autonomy could be technically feasible. Of these municipalities, 15 were used as case studies, which differ greatly in terms of the indicators used in the following regression analysis. The results of the optimisations showed the influence of individual technologies and measures on the LCOE. Thereby, it became apparent that complete energy autonomy is always associated with a high cost increase. Furthermore, the integration of the industrial and commercial sectors can have a reducing effect on the LCOEs, since fixed costs are distributed across a larger amount of energy. In addition, the flexibility through electric vehicles can moderately reduce LCOEs. Using a stepwise linear regression model (mean absolute percentage error = 12.5%), the results of the optimisations could finally be transferred to the 6,314 municipalities. After the correlation analysis and the exclusion of specific indicators (e.g. industrial electricity demand or technical wind energy potential), the technical bioenergy potential P_B , residential electricity demand $ED_{m,r}$, population density *PD* and technical geothermal potential P_G are selected as features for the regression (cf. Eq. 5). The units are GWh/a for potentials and demands and km⁻² for population density.

$$\Delta LCOE = 0.5924 - 0.0209 \cdot P_B + ED_{m,r} \cdot (-0.0282 + 0.0003 \cdot PD - 0.0013 \cdot P_G) - 0.0012 \cdot PD$$
$$-0.0019 \cdot P_G$$

On average, the additional Δ LCOEs in the autonomous compared to the reference (minimal cost) case, amount to about 0.41 €/kWh. Figure 11 shows the distribution of these additional LCOEs across all German municipalities. When distributing the regression results according to the ten German municipality clusters from Weinand et al. (2019a), the results seem plausible: the highest mean Δ LCOE

is reached in cluster 2 (0.578 \notin /kWh), which mainly contains cities with low renewable energy potential (cf. section 5.1). On the other hand, the lowest mean Δ LCOEs are achieved in clusters 3 (0.350 \notin /kWh), 4 (0.349 \notin /kWh) and 8 (0.379 \notin /kWh), which contain mainly rural municipalities with particularly high potential for renewable energies and especially deep geothermal energy. Apart from energy demand, base load capable bioenergy and deep geothermal energy appear to have the greatest influence on the LCOEs.



Figure 11: Illustration of 6,314 (56%) German municipalities that can become completely autonomous and the associated $\Delta LCOE$ (Weinand et al. 2020a).

As stated above, the Δ LCOE in the 15 case studies result from the comparison of the costs of the autonomous energy systems with an optimised reference system. As the energy systems in reality differ strongly from an optimised system, the real Δ LCOE are probably lower. Assuming that the 15 municipalities in the reference case are not optimised but instead only purchase energy from electricity and gas networks with typical end user prices (Bundesnetzagentur 2019a, 2019b, 2019c) and 2% price increase every 5 years, the Δ LCOE would be reduced by about 0.03 €/kWh on average. This reduction could be even higher as no investments for replacing old energy infrastructures were taken into account in this exemplary calculation. However, in the end, it is obvious that the Δ LCOE in Figure 11 can only serve as a basis for estimating the potential for energy autonomy of individual municipalities in comparison to each other.

Main areas for improving the methodology in Weinand et al. (2020a) include the consideration of grid infrastructures and surplus electricity from neighbouring municipalities, as well as more detailed modelling of industrial demand. The method of calculating and comparing the costs of energy autonomy should be improved to express these costs per municipal end user. In future studies in which the national energy system or transmission grid expansion is planned, the results of this paper can be used as a possible scenario (cf. outlook in section 7).

Section 4.3.1 showed that the RE³ASON optimisation model from Weinand et al. (2020a) was also extended by VoLL to give an example of the possible impacts of load shedding on the energy system and the associated costs. The modified optimisation model was applied to the municipality Prinzenmoor, which was already used in Weinand et al. (2020a) for the detailed sensitivity analyses. As a result, on four extreme days with low wind and solar irradiation, the entire load of industrial and commercial sectors is dropped (a total of 16.9 MWh, which corresponds to 0.5% of the total annual electricity consumption). On the remaining 361 days per year no use is made of the LoL. This changes the energy system compared to the case without LoL: the main energy source changes from deep geothermal energy to wind energy. The total discounted system costs decrease by 54% from 27.7 M€ (or 0.52 €/kWh) to 12.8 M€ (or 0.28 €/kWh) with LoL.

In this case, the LoL has a large impact and could strongly favour the potential or implementation of energy autonomy. However, Prinzenmoor is a very small municipality. Therefore, the influence of LoL should be also investigated in further municipalities in future research. In addition, the modelling of the LoL should be further improved by including maximum energy amounts and maximum length of time intervals for LoL. Furthermore, average nationwide VoLL values should be critically evaluated, as the VoLL could vary greatly depending on the type of household or industrial company.

5.6. Plausibility check of results

During the research on energy autonomy, a comprehensive knowledge on this subject was gained. Thus, a study (Weinand et al. 2020b) has been conducted within this thesis, which reveals research gaps and suggestions for improvement for future studies on municipal energy autonomy. Thereby, the status quo (cf. section 3) and future modelling needs (cf. section 6) for research on decentralised autonomous energy systems is shown. A total of 359 studies were roughly investigated, of which a subset of 123 in detail. The studies were assessed with respect to the characteristics of their methodology and applications, in order to derive common trends and insights.

Some aspects of Weinand et al. (2020b) are presented in the literature review section 3, the discussion in section 6 and the outlook in section 7. However, a compilation of LCOEs, from Weinand et al. (2020b) is discussed in the following since it can be used to check the plausibility of the results of this thesis. Figure 12 shows the LCOEs for 84 articles on decentralised energy autonomy. The studies Weinand et al. (2019e) and Weinand et al. (2020a) are marked by red frames. While in this thesis energy systems

based on 100% renewable energies have been considered, there are also case studies in the literature on energy autonomy including conventional energy technologies. In addition, a security of supply below 100% was considered in several studies. This is equivalent to the *Lost of Load* from section 5.5.

The LCOEs for autonomous energy systems in the 84 case studies amount to 0.42 \$/kWh on average. However, in some studies the resulting costs should be questioned, as they deviate strongly from the average. With 0.39 \$/kWh in Weinand et al. (2019e) and an average of 0.68 \$/kWh in the 15 investigated municipalities of Weinand et al. (2020a), the LCOEs calculated in this thesis are close to the average value in the literature (0.42 \$/kWh). The results from Weinand et al (2020a) are probably above the average, as the 0.68 \$/kWh is the average for 15 optimised municipalities, which were chosen to represent the whole population of 6,314 municipalities. In the other articles, the case studies may have been selected based on their suitability for energy autonomy. Thus the results of this thesis are plausible. If the *Lost of Load* is considered as in section 5.5, the LCOEs (0.28 €/kWh) of an energy autonomous German municipality might even be below the average household electricity price of 0.32 €/kWh (June 2019). However, the calculations with RE³ASON are carried out from a macroeconomic perspective and therefore (in contrast to the household electricity price) no taxes or levies are taken into account.



Figure 12: LCOEs of case studies with energy autonomous municipalities in the literature. The studies are sorted by mean LCOEs, from high to low. The LCOEs were adjusted according to inflation until 2019 and converted into \$/kWh using the average exchange rates in the year of the respective publication. The household electricity price (black dots) in the different countries is shown for comparison. The black dotted line shows the average value (0.41 \$/kWh). The similar figure from Weinand et al. (2020b) is adjusted for this thesis.

6. Critical appraisal

The limitations of the individual studies and methods have already been shown in the subsections of section 5. Therefore, general issues that apply to the entire thesis are discussed in the following.

First, the time resolution of the studies in this thesis is hourly. In non-autonomous energy systems, incorrect dimensioning can be compensated by imports. Thus, the LCOEs of the energy system would change only slightly in case of minor design errors. However, if a completely autonomous energy system is incorrectly designed, a breakdown of the energy supply would result. Therefore, future studies should investigate the influence of a higher time resolution on autonomous energy system planning. In this thesis, an attempt was made to achieve the robustness of the energy system by including extreme days.

Due to the increase of the complexity of the optimisation problem with higher temporal resolution, especially for long-term planning processes, examining a higher resolution is hardly practicable. In these cases, detailed analyses and/or information reduction techniques to generate time typologies and synthetic time slices should be applied. There is an established stream of research focussing on the most suitable/required time resolution for specific research questions, for energy systems with significant renewable generation fractions (e.g. Kotzur et al. (2018) or Nahmmacher et al. (2016)). Studies addressing municipal energy autonomy could also benefit from exploring such approaches. Otherwise there is the above-mentioned risk that systems are incorrectly dimensioned.

Furthermore, the present thesis provides useful information for the economic assessment of the technical potential of energy autonomy. However, there is also a lack of attention paid to non-economic and non-technical criteria in this thesis as well as other studies on energy autonomy. Indeed, economic criteria are arguably the most important, but although they are necessary, they are not sufficient. In Weinand et al. (2020a) it has become apparent that in an industrialised country like Germany there is currently (i.e. with current electricity prices, technology costs and energy-political framework conditions) no economic potential for complete energy autonomy compared to the case with grid connection. Therefore, in future analyses of energy autonomy more criteria than costs need to be considered. There are other important reasons for municipalities to become energy autonomous besides costs, such as increased environmental awareness (cf. section 2).

In addition, this thesis makes use of linear programming from a central planner perspective. It is common practice to leave stakeholder roles outside the scope of energy system analyses and to calculate optimal autonomy transition pathways with a centralized planning approach under the selected objective of technical feasibility and economic viability. However, local energy systems are complex socio-technical systems consisting of different decision-making entities and technological artefacts governed by energy policy in a multi-level institutional space (Koirala et al. 2016). Social relationships among the stakeholders represent a major driver or barrier (Rae and Bradley 2012). In this context, adoption behaviour approaches are useful to understand the types of barriers that exist for new technologies, and

what kind of policies are important to increase diffusion. As a consequence, realising the potential of municipal energy autonomy is not simply a question of technical realities but also a question of individual behaviour and group dynamics. Relevant local stakeholders such as households or communities, energy producers, energy suppliers, service providers, as well as local policy-makers are inter-dependent in the realization of their goals. Future system models need to include the heterogeneous roles different stakeholders play in an existing local environment and the resulting impact their decision making might have. One possible solution could be the extension of techno-economic modelling approaches with the help of socio-economic modelling approaches as agent-based models (Rai and Robinson 2015) or system dynamics models (Selvakkumaran and Ahlgren 2018). The integration of social factors such as the acceptance of deep geothermal plants or wind power plants (cf. for example Tröndle et al. (2019)) could reduce the potential for municipal energy autonomy determined in this thesis.

Neglecting taxes and levies ensured that technologies and measures could be assessed and compared neutrally in this thesis (cf. section 4.3.1). However, it is very challenging in reality to implement the optimal energy systems determined from the macroeconomic perspective of a central planner. For households, for example, the cost-optimal energy system determined from a macroeconomic perspective could be very different from the cost-optimal energy system from a microeconomic perspective, taking into account taxes and levies. Future studies should therefore examine the impact of taking into account taxes and levies on the economic viability of energy autonomy in more detail.

Many different spatial scales (e.g. number of households) have been considered in this thesis, but the optimal spatial size of an energy autonomous region has not yet been identified. This is strongly linked to the question of the optimal degree of centralisation (McKenna 2018). Furthermore, in the present thesis the energy systems of individual municipalities were separately investigated. As demonstrated in Weinand et al. (2020a), however, in the case of complete energy autonomy large energy surpluses can result. The use of these surpluses in neighbouring municipalities could significantly reduce the costs for municipal energy autonomy.

Concerning demand and consumption sectors there is also some potential for improvement. The industry sector as well as the private transportation sector are implemented in a simplified approach in Weinand et al. (2020a). For the former, transferable methods to determine the energy demand and load profile of industries in arbitrary regions could facilitate its implementation. For the latter, a particularly interesting approach would be to optimise the number and use of electric (or fuel cell) vehicles. In general, however, all sectors should be taken into account as in Weinand et al. (2020a), especially when estimating the impact of energy autonomous regions on the surrounding energy system.

Furthermore, validation is challenging in the context of municipal energy autonomy. Often detailed data on the existing energy system is lacking and validation for some hypothetical future scenario is obviously not meaningful. Model design and data assumptions of studies used to gain insights to form the decision making should be transparent and accessible. This not only allows independent review of various stakeholders but also the complete reproducibility of the results (Pfenninger et al. 2014). Hence studies on municipal energy autonomy could increase efforts to publicly release data as in Weinand et al. (2019b) and system models as well as assumptions and results interpretation, in order that diverse affected stakeholders are able to participate in the decision-making process (Pfenninger et al. 2014; DeCarolis et al. 2017).

In addition, the focus on the LCOEs as a benchmark for highly-renewable energy systems in this thesis could provide potentially misleading results. Whilst the LCOE is a good first indicator of the generation costs and allows comparisons across technology, it is noted for neglecting the additional costs of integrating non-dispatchable renewable technologies into the energy system. Three additional cost components should be considered, if the system LCOEs of renewable energy technologies are to be considered (Hirth et al. 2015; Ueckerdt et al. 2013):

• **Profiling costs**, related to the requirement for the dispatchable generation technologies to meet the residual load;

• **Balancing costs**, related to the deviation between forecast and actual non-dispatchable renewable generation; and

• **Grid costs**, related to additional grid reinforcement and extension (at all voltage levels) required to connect renewable generators to the network.

Attempts have been made to consider these cost components in the context of large-scale (e.g. national) energy system analyses (e.g. Brown et al. 2018; Chaudry et al. 2014; Nolden et al. 2013). However, at the regional and municipal scale, they are typically not included. For example, the analyses in the present thesis also did not take network capacities into account. As an improvement, an approach would be conceivable in which the grid is upgraded if a high proportion of renewable energies is installed (as in Morvaj et al. (2017)).

7. Conclusions and outlook

Attention on decentralised autonomous energy systems has increased exponentially in the past three decades, as demonstrated by the absolute number of municipal projects as well as the share of publications in the corpus of scientific literature. This is due to the energy transition and the related environmental awareness as well as the desire of citizens to play an active role in energy supply and to be less dependent on central markets and structures. However, local decision-makers often lack the necessary expertise and therefore need decision support in energy system planning. Thus, this thesis follows the objective to develop novel methods for the technical, economic and environmental assessment of a large number of completely energy autonomous municipalities are disconnected from the gas and electricity grid and supply themselves with energy from renewable energy plants owned and operated by the municipality.

In order to meet the objective, several novel methods of energy system analysis were developed in this thesis as part of seven original research articles. Germany is used as a case study, but the general approach, methods and results are transferable to other contexts. In order to be able to analyse a large number of municipalities (> 1,000), an approach had to be developed that allows to analyse a multitude of energy systems with practicable computational expenses. Therefore, a typology of the energy systems to be investigated was needed. Subsequently, selected energy systems that represent this typology had to be optimised from a techno-economic perspective. In the final step, the results of the optimisations had then to be transferred to all energy systems.

First of all, the 11,131 German municipalities were clustered with regard to their suitability for decentralised energy systems. For this purpose, social indicators such as population density or number of vehicles and technical factors such as generation potential of different renewable energies were employed. Furthermore, since previous studies on municipal energy autonomy with renewable energies often resulted in high total costs due to limited base load potentials and therefore high storage costs, other technologies were sought that could reduce these costs. Deep geothermal energy was identified as a renewable technology that is capable of providing base load and has a high potential in Germany. This technology was implemented in an already existing holistic municipal energy system optimisation model (RE³ASON). This optimisation model is particularly suitable for the investigation of many different municipalities due to the use of public data and the associated applicability to arbitrary municipalities. In order to be able to use the heat in addition to the power generation of the deep geothermal plants, a transferable combinatorial optimisation approach for district heating network design in municipalities was developed in two studies and also integrated into RE³ASON. In a further study, the technical feasibility and economic expenditure of energy autonomy could finally be determined in all 11,131 German municipalities by combining the extended energy system optimisation model with a stepwise linear regression. In a concluding review study, the experience gained over several years on the topic of energy autonomy was used to present suggestions for further improvement of future energy system modelling of decentralised autonomy.

For the case study Germany, ten municipality clusters could be identified, which differ considerably in terms of their characteristics. For example, one municipality cluster includes all major German cities with low renewable energy potentials, while other clusters include more rural municipalities with high renewable energy potentials. During the subsequent energy system optimisations, the results showed that in the case of complete energy autonomy, deep geothermal plants in combination with district heating networks are in fact usually installed. The total costs for municipal energy systems until 2030 could be reduced by up to 50% by using deep geothermal energy and district heating. This underlines the assumption from the literature review that base load technologies are advantageous in energy autonomous systems due to the associated reduced storage costs. However, it is important to mention that deep geothermal energy and district heating networks have not been installed in any reference scenario of municipal energy systems with grid connection. On average, the energy system costs until 2030 in German municipalities increase by about 0.41 €/kWh in the energy autonomous case compared to the optimised reference case. While a technical potential to achieve energy autonomy is present in 56% of the municipalities, there is therefore no economic benefit of complete energy autonomy in an industrialised country like Germany compared to the optimised reference municipal energy system with grid connection. However, this latter statement only applies to an isolated consideration of the municipal system and requires further investigation in the case of an integrated consideration in a national energy system analysis.

The novel methodological approach of this thesis enabled to obtain optimisation results for a high number of energy systems (6,314 municipalities) with practicable computational expenses. In addition to the original data and planning tools published alongside the articles, the findings of this thesis can help to support local decision makers in determining cost-effective municipal energy systems. Thereby, the insights on the impact of different technologies and geographical conditions in municipalities can benefit not only for the planning of complete energy autonomy but also for energy system analysis in general. Furthermore, the results can be utilised for energy system planning at the national level, such as the design of transmission grids. For this purpose, scenarios could be designed in which municipalities with the greatest techno-economic potential for decentralised energy systems become energy autonomous. For example, the assumption could be made that all municipalities with Δ LCOE less than the mean value (0.41 €/kWh) will become autonomous. Then the demand and feed-in from these municipalities could be excluded from the analyses. Furthermore, simultaneous optimisation of transmission grid expansion and selection of autonomous municipalities could be performed to determine the optimal future national energy system.

The original approaches developed in this thesis have been validated and, if this was not possible, at least checked for plausibility. The transferability of the methods for different applications has also been

demonstrated in the various studies of this thesis. For example, the algorithm for designing district heating networks has been extended in a further ongoing study for connecting wind turbines to electricity substations (McKenna et al. 2020).

However, in order to increase the realizability of the case study results identified in this thesis, some methodological extensions are necessary. For example, future research should focus on other perspectives than that of a central planner and other target criteria than costs should be included. In addition, complete autonomous energy systems in particular must be robustly designed, for example by further analysing the Value of Lost Load and whether a security of supply below 100% is acceptable for energy consumers. Furthermore, the temporal resolution should be higher than the usually used hourly resolution. Finally, methodologies should be developed which can involve local stakeholders in the modelling process and thus consider their preferences relating to their future energy system which is also related to the acceptance of technologies like wind power or deep geothermal plants.

Publication bibliography

Alhamwi, A.; Medjroubi, W.; Vogt, T.; Agert, C. (2019): Development of a GIS-based platform for the allocation and optimisation of distributed storage in urban energy systems. In *Applied Energy* 251, p. 113360. DOI: 10.1016/j.apenergy.2019.113360.

Azaza, M.; Wallin, F. (2017): Multi objective particle swarm optimization of hybrid micro-grid system. A case study in Sweden. In *Energy* 123, pp. 108–118. DOI: 10.1016/j.energy.2017.01.149.

BDEW (2017): Erneuerbare Energien und das EEG: Zahlen, Fakten, Grafiken. Available online at https://www.bdew.de/media/documents/Awh_20170710_Erneuerbare-Energien-EEG_2017.pdf, checked on 4/30/2019.

BMWi (2019a): Abkommen von Paris (Paris Agreement). Available online at https://www.bmwi.de/Redaktion/DE/Artikel/Industrie/klimaschutz-abkommen-von-paris.html, checked on 4/29/2019.

BMWi (2019b): Deutsche Klimaschutzpolitik. Available online at https://www.bmwi.de/Redaktion/DE/Artikel/Industrie/klimaschutz-deutsche-klimaschutzpolitik.html, checked on 4/29/2019.

Bonati, A.; Luca, G. de; Fabozzi, S.; Massarotti, N.; Vanoli, L. (2019): The integration of exergy criterion in energy planning analysis for 100% renewable system. In *Energy* 174, pp. 749–767. DOI: 10.1016/j.energy.2019.02.089.

Boon, F. P.; Dieperink, C. (2014): Local civil society based renewable energy organisations in the Netherlands. Exploring the factors that stimulate their emergence and development. In *Energy Policy* 69, pp. 297–307. DOI: 10.1016/j.enpol.2014.01.046.

Brown, T.; Schlachtberger, D.; Kies, A.; Schramm, S.; Greiner, M. (2018): Synergies of sector coupling and transmission reinforcement in a cost-optimised, highly renewable European energy system. In *Energy* 160, pp. 720–739. DOI: 10.1016/j.energy.2018.06.222.

Bundesnetzagentur (2019a): Entwicklung der Gaspreise für Haushaltskunden in Deutschland in den Jahren 2009 bis 2019. Available online at

https://de.statista.com/statistik/daten/studie/168286/umfrage/entwicklung-der-gaspreise-fuer-haushaltskunden-seit-2006/, checked on 3/13/2019.

Bundesnetzagentur (2019b): Strompreise für Gewerbe- und Industriekunden in Deutschland in den Jahren 2009 bis 2019. Available online at

https://de.statista.com/statistik/daten/studie/154902/umfrage/strompreise-fuer-industrie-und-gewerbe-seit-2006/, checked on 3/13/2020.

Bundesnetzagentur (2019c): Strompreise für Haushaltskunden in Deutschland in den Jahren 2009 bis 2019. Available online at https://de.statista.com/statistik/daten/studie/154908/umfrage/strompreise-fuer-haushaltskunden-seit-2006/, checked on 3/13/2019.

Bundesnetzagentur (2019d): Informationen zu Strom- und Gaspreisen für Haushaltskunden. Available online at

https://www.bundesnetzagentur.de/DE/Sachgebiete/ElektrizitaetundGas/Verbraucher/PreiseRechnTari fe/preiseundRechnungen-node.html, checked on 7/13/2019.

CEPA (2018): Study on the estimation of the Value of Lost Load of electricity supply in Europe. Cambridge Economic Policy Associates Ltd. Available online at

https://www.acer.europa.eu/en/Electricity/Infrastructure_and_network%20development/Infrastructure/Documents/CEPA%20study%20on%20the%20Value%20of%20Lost%20Load%20in%20the%20electricity%20supply.pdf, checked on 2/27/2020.

Chaudry, M.; Jenkins, N.; Qadrdan, M.; Wu, J. (2014): Combined gas and electricity network expansion planning. In *Applied Energy* 113, pp. 1171–1187. DOI: 10.1016/j.apenergy.2013.08.071.

Chmiel, Z.; Bhattacharyya, S. C. (2015): Analysis of off-grid electricity system at Isle of Eigg (Scotland). Lessons for developing countries. In *Renewable Energy* 81, pp. 578–588. DOI: 10.1016/j.renene.2015.03.061.

DeCarolis, J.; Daly, H.; Dodds, P.; Keppo, I.; Li, F.; McDowall, W. et al. (2017): Formalizing best practice for energy system optimization modelling. In *Applied Energy* 194, pp. 184–198. DOI: 10.1016/j.apenergy.2017.03.001.

Dorotić, H.; Doračić, B.; Dobravec, V.; Pukšec, T.; Krajačić, G.; Duić, N. (2019): Integration of transport and energy sectors in island communities with 100% intermittent renewable energy sources. In *Renewable and Sustainable Energy Reviews* 99, pp. 109–124. DOI: 10.1016/j.rser.2018.09.033.

Drysdale, D.; Vad Mathiesen, B.; Lund, H. (2019): From Carbon Calculators to Energy System Analysis in Cities. In *Energies* 12 (12), p. 2307. DOI: 10.3390/en12122307.

Engelken, M.; Römer, B.; Drescher, M.; Welpe, I. (2016): Transforming the energy system. Why municipalities strive for energy self-sufficiency. In *Energy Policy* 98, pp. 365–377. DOI: 10.1016/j.enpol.2016.07.049.

eurostat (2020): How are emissions of greenhouse gases by the EU evolving? Available online at https://ec.europa.eu/eurostat/cache/infographs/energy/bloc-4a.html, checked on 4/19/2020.

Eyerer, S.; Schifflechner, C.; Hofbauer, S.; Bauer, W.; Wieland, C.; Spliethoff, H. (2020): Combined heat and power from hydrothermal geothermal resources in Germany. An assessment of the potential. In *Renewable and Sustainable Energy Reviews* 120, p. 109661. DOI: 10.1016/j.rser.2019.109661.

Grosspietsch, D.; Thömmes, P.; Girod, B.; Hoffmann, V. H. (2018): How, When, and Where? Assessing Renewable Energy Self-Sufficiency at the Neighborhood Level. In *Environmental science* & *technology* 52 (4), pp. 2339–2348. DOI: 10.1021/acs.est.7b02686.

Hirth, L.; Ueckerdt, F.; Edenhofer, O. (2015): Integration costs revisited – An economic framework for wind and solar variability. In *Renewable Energy* 74, pp. 925–939. DOI: 10.1016/j.renene.2014.08.065.

IWU (2015). Deutsche Wohngebäudetypologie. Beispielhafte Maßnahmen zur Verbesserung der Energieeffizienz von typischen Wohngebäuden (zweite erweiterte Auflage). Available online at https://www.episcope.eu/downloads/public/docs/brochure/DE_TABULA_TypologyBrochure_IWU.p df, checked on 7/15/2020

Jägemann, C.; Hagspiel, S.; Lindenberger, D. (2013): The economic inefficiency of grid parity. The case of German photovoltaics. EWI Working Paper. Köln (13/19). Available online at http://hdl.handle.net/10419/92970.

Jenssen, T.; König, A.; Eltrop, L. (2014): Bioenergy villages in Germany. Bringing a low carbon energy supply for rural areas into practice. In *Renewable Energy* 61, pp. 74–80. DOI: 10.1016/j.renene.2012.08.014.

Koirala, B. P.; Koliou, E.; Friege, J.; Hakvoort, R. A.; Herder, P. M. (2016): Energetic communities for community energy. A review of key issues and trends shaping integrated community energy systems. In *Renewable and Sustainable Energy Reviews* 56, pp. 722–744. DOI: 10.1016/j.rser.2015.11.080.

Kötter, E.; Schneider, L.; Sehnke, F.; Ohnmeiss, K.; Schröer, R. (2016): The future electric power system. Impact of Power-to-Gas by interacting with other renewable energy components. In *Journal of Energy Storage* 5, pp. 113–119. DOI: 10.1016/j.est.2015.11.012.

Kotzur, L.; Markewitz, P.; Robinius, M.; Stolten, D. (2018): Impact of different time series aggregation methods on optimal energy system design. In *Renewable Energy* 117, pp. 474–487. DOI: 10.1016/j.renene.2017.10.017.

Krajačić, G.; Duić, N.; Carvalho, M. d. G. (2009): H2RES, Energy planning tool for island energy systems – The case of the Island of Mljet☆. In *International Journal of Hydrogen Energy* 34 (16), pp. 7015–7026. DOI: 10.1016/j.ijhydene.2008.12.054.

Mainzer, K.; Fath, K.; McKenna, R.; Stengel, J.; Fichtner, W.; Schultmann, F. (2014): A high-resolution determination of the technical potential for residential-roof-mounted photovoltaic systems in Germany. In *Solar Energy* 105, pp. 715–731. DOI: 10.1016/j.solener.2014.04.015.

Mainzer, K. (2019): Analyse und Optimierung urbaner Energiesysteme - Entwicklung und Anwendung eines übertragbaren Modellierungswerkzeugs zur nachhaltigen Systemgestaltung. Dissertation. Karlsruher Institut für Technologie, Karlsruhe. Institut für Industriebetriebslehre und Industrielle Produktion (IIP). Available online at https://publikationen.bibliothek.kit.edu/1000092481, checked on 3/6/2020.

McKenna, R.; Hollnaicher, S.; Fichtner, W. (2014): Cost-potential curves for onshore wind energy. A high-resolution analysis for Germany. In *Applied Energy* 115, pp. 103–115. DOI: 10.1016/j.apenergy.2013.10.030.

McKenna, R.; Herbes, C.; Fichtner, W. (2015): Energieautarkie. Vorschlag einer Arbeitsdefinition als Grundlage für die Bewertung konkreter Projekte und Szenarien. In *Z Energiewirtsch* 39 (4), pp. 235–252. DOI: 10.1007/s12398-015-0164-1.

McKenna, R.; Bertsch, V.; Mainzer, K.; Fichtner, W. (2018): Combining local preferences with multicriteria decision analysis and linear optimization to develop feasible energy concepts in small communities. In *European Journal of Operational Research* 268 (3), pp. 1092–1110. DOI: 10.1016/j.ejor.2018.01.036.

McKenna, R. (2018): The double-edged sword of decentralized energy autonomy. In *Energy Policy* 113, pp. 747–750. DOI: 10.1016/j.enpol.2017.11.033.

McKenna, R.; Weinand, J. M.; Mulalic, I.; Petrovic, S.; Mainzer, K.; Preis, T.; Moat, H. S. (2020): Improving renewable energy resource assessments by quantifying landscape beauty. In *Working Paper Series in Production and Energy*. DOI: 10.5445/IR/1000118671.

Morvaj, B.; Evins, R.; Carmeliet, J. (2017): Decarbonizing the electricity grid. The impact on urban energy systems, distribution grids and district heating potential. In *Applied Energy* 191, pp. 125–140. DOI: 10.1016/j.apenergy.2017.01.058.

Müller, M. O.; Stämpfli, A.; Dold, U.; Hammer, T. (2011): Energy autarky. A conceptual framework for sustainable regional development. In *Energy Policy* 39 (10), pp. 5800–5810. DOI: 10.1016/j.enpol.2011.04.019.

Nahmmacher, P.; Schmid, E.; Hirth, L.; Knopf, B. (2016): Carpe diem. A novel approach to select representative days for long-term power system modeling. In *Energy* 112, pp. 430–442. DOI: 10.1016/j.energy.2016.06.081.

Neumann, F.; Brown, T. (2019): Heuristics for Transmission Expansion Planning in Low-Carbon Energy System Models. Available online at https://arxiv.org/pdf/1907.10548.pdf, checked on 3/10/2020.

Nolden, C.; Schönfelder, M.; Eßer-Frey, A.; Bertsch, V.; Fichtner, W. (2013): Network constraints in techno-economic energy system models. Towards more accurate modeling of power flows in long-term energy system models. In *Energy Syst* 4 (3), pp. 267–287. DOI: 10.1007/s12667-013-0078-0.

Oldenbroek, V.; Verhoef, L. A.; van Wijk, Ad J.M. (2017): Fuel cell electric vehicle as a power plant. Fully renewable integrated transport and energy system design and analysis for smart city areas. In *International Journal of Hydrogen Energy* 42 (12), pp. 8166–8196. DOI: 10.1016/j.ijhydene.2017.01.155.

Østergaard, P. A.; Lund, H. (2011): A renewable energy system in Frederikshavn using low-temperature geothermal energy for district heating. In *Applied Energy* 88 (2), pp. 479–487. DOI: 10.1016/j.apenergy.2010.03.018.

Peter, S. (2013): Modellierung einer vollständig auf erneuerbaren Energien basierenden Stromerzeugung im Jahr 2050 in autarken, dezentralen Strukturen. Edited by Umweltbundesamt. Available online at

https://www.umweltbundesamt.de/sites/default/files/medien/376/publikationen/climate_change_14_20 13_modellierung_einer_vollstaendig_auf_erneuerbaren_energien.pdf, checked on 3/9/2020.

Peura, P.; Haapanen, A.; Reini, K.; Törmä, H. (2018): Regional impacts of sustainable energy in western Finland. In *Journal of Cleaner Production* 187, pp. 85–97. DOI: 10.1016/j.jclepro.2018.03.194.

Pfenninger, S.; Hawkes, A.; Keirstead, J. (2014): Energy systems modeling for twenty-first century energy challenges. In *Renewable and Sustainable Energy Reviews* 33, pp. 74–86. DOI: 10.1016/j.rser.2014.02.003.

Pfenninger, S. (2017): Energy scientists must show their workings. In *Nature* 542 (7642), p. 393. DOI: 10.1038/542393a.

Rae, C.; Bradley, F. (2012): Energy autonomy in sustainable communities—A review of key issues. In *Renewable and Sustainable Energy Reviews* 16 (9), pp. 6497–6506. DOI: 10.1016/j.rser.2012.08.002.

Rai, V.; Robinson, S. A. (2015): Agent-based modeling of energy technology adoption. Empirical integration of social, behavioral, economic, and environmental factors. In *Environmental Modelling & Software* 70, pp. 163–177. DOI: 10.1016/j.envsoft.2015.04.014.

Sameti, M.; Haghighat, F. (2018): Integration of distributed energy storage into net-zero energy district systems. Optimum design and operation. In *Energy* 153, pp. 575–591. DOI: 10.1016/j.energy.2018.04.064.

Šare, A.; Krajačić, G.; Pukšec, T.; Duić, N. (2015): The integration of renewable energy sources and electric vehicles into the power system of the Dubrovnik region. In *Energ Sustain Soc* 5 (1), p. 17. DOI: 10.1186/s13705-015-0055-7.

Schaber, K.; Steinke, F.; Hamacher, T. (2012): Transmission grid extensions for the integration of variable renewable energies in Europe: Who benefits where? In *Energy Policy* 43, pp. 123-135. DOI: 10.1016/j.enpol.2011.12.040.

Scheffer, K. (2008): Vom Bioenergiedorf zur autonomen Solarenergieregion. In *Solarzeitalter* 20 (4), pp. 23–30.

Schmidt, J.; Schönhart, M.; Biberacher, M.; Guggenberger, T.; Hausl, S.; Kalt, G. et al. (2012): Regional energy autarky. Potentials, costs and consequences for an Austrian region. In *Energy Policy* 47, pp. 211–221. DOI: 10.1016/j.enpol.2012.04.059.

Schröder, T.; Kuckshinrichs, W. (2015): Value of Lost Load. An Efficient Economic Indicator for Power Supply Security? A Literature Review. In *Front. Energy Res.* 3, p. 1378. DOI: 10.3389/fenrg.2015.00055.

Selvakkumaran, S.; Ahlgren, E. (2018): Model-Based Exploration of Co-Creation Efforts. The Case of Solar Photovoltaics (PV) in Skåne, Sweden. In *Sustainability* 10 (11), p. 3905. DOI: 10.3390/su10113905.

Siragusa, A.; Vizcaino, M. P.; Proietti, P.; Lavalle, C. (2020): European Handbook for SDG Voluntary Local Reviews. Edited by European Union. Available online at https://ec.europa.eu/jrc/en/publication/eur-scientific-and-technical-research-reports/european-handbook-sdg-voluntary-local-reviews, checked on 3/11/2020.

trend:research (2017): Eigentürmerstruktur: Erneuerbare Energien. Entwicklung der Akteursvielfalt, Rolle der Energieversorger, Ausblick bis 2020. Available online at https://www.trendresearch.de/studie.php?s=672, checked on 4/30/2019.

Tröndle, T.; Süsser, D.; Lilliestam, J. (2019): Ohne Windenergie keine Energiewende. Die 1000 Meter-Abstandsregelung macht Windenergieausbau unmöglich und stellt damit den Kohleausstieg in Deutschland in Frage. Available online at http://publications.iasspotsdam.de/pubman/faces/viewItemOverviewPage.jsp?itemId=escidoc:4801899, checked on 3/10/2020.

UBA (2020): Erneuerbare Energien in Zahlen. Available online at https://www.umweltbundesamt.de/themen/klima-energie/erneuerbare-energien/erneuerbare-energien-in-zahlen#uberblick, checked on 8/7/2020.

Ueckerdt, F.; Hirth, L.; Luderer, G.; Edenhofer, O. (2013): System LCOE. What are the costs of variable renewables? In *Energy* 63, pp. 61–75. DOI: 10.1016/j.energy.2013.10.072.

Weinand, J. M.; McKenna, R.; Fichtner, W. (2019a): Developing a municipality typology for modelling decentralised energy systems. In *Utilities Policy* 57, pp. 75–96. DOI: 10.1016/j.jup.2019.02.003.

Weinand, J. M.; McKenna, R.; Mainzer, K. (2019b): Spatial high-resolution socio-energetic data for municipal energy system analyses. In *Scientific data* 6 (1), 243. DOI: 10.1038/s41597-019-0233-0.

Weinand, J. M.; McKenna, R.; Karner, K.; Braun, L.; Herbes, C. (2019c): Assessing the potential contribution of excess heat from biogas plants towards decarbonising residential heating. In *Journal of Cleaner Production* 238, 117756. DOI: 10.1016/j.jclepro.2019.117756.

Weinand, J. M.; Kleinebrahm, M.; McKenna, R.; Mainzer, K.; Fichtner, W. (2019d): Developing a combinatorial optimisation approach to design district heating networks based on deep geothermal energy. In *Applied Energy* 251, 113367. DOI: 10.1016/j.apenergy.2019.113367.

Weinand, J. M.; McKenna, R.; Kleinebrahm, M.; Mainzer, K. (2019e): Assessing the contribution of simultaneous heat and power generation from geothermal plants in off-grid municipalities. In *Applied Energy* 255, 113824. DOI: 10.1016/j.apenergy.2019.113824.

Weinand, J. M.; Ried, S.; Kleinebrahm, M.; McKenna, R.; Fichtner, W. (2020a): Identification of Potential Off-Grid Municipalities with 100% Renewable Energy Supply. In *Working Paper Series in Production and Energy*. DOI: 10.5445/IR/1000118013.

Weinand, J. M.; Scheller, F.; McKenna, R. (2020b): Reviewing energy system modelling of decentralized energy autonomy. In *Energy* 203, 117817. DOI: 10.1016/j.energy.2020.117817.

Weinand, J. M. (2020): Reviewing municipal energy system planning in a bibliometric analysis: evolution of the research field between 1991 and 2019. In *Energies* 13 (6), p. 1367. DOI: 10.3390/en13061367.

Woyke, W.; Forero, M. (2014): Methoden zur Bewertung regionaler Energieautarkie. Beitrag auf dem 13. Symposium Energieinnovation. Graz, Österreich.

Zhang, G.; McCalley, J. D. (2018): Estimation of Regulation Reserve Requirement Based on Control Performance Standard. In *IEEE Trans. Power Syst.* 33 (2), pp. 1173–1183. DOI: 10.1109/TPWRS.2017.2734654.

Part B: Articles

Energy 203 (2020) 117817



Review

Reviewing energy system modelling of decentralized energy autonomy



Jann Michael Weinand ^{a, *}, Fabian Scheller ^b, Russell McKenna ^b

^a Chair of Energy Economics, Karlsruhe Institute of Technology, Hertzstraße 16, 76187, Karlsruhe, Germany ^b Energy Systems Analysis, DTU Management, Technical University of Denmark, 2800, Kgs. Lyngby, Denmark

ARTICLE INFO

ABSTRACT

Article history: Received 18 February 2020 Received in revised form 4 May 2020 Accepted 6 May 2020 Available online 11 May 2020

Keywords: Literature review Energy autonomy Off-grid systems 100% renewable Energy system analysis Levelized cost of electricty Municipality Remote area

Research attention on decentralized autonomous energy systems has increased exponentially in the past three decades, as demonstrated by the absolute number of publications and the share of these studies in the corpus of energy system modelling literature. This paper shows the status quo and future modelling needs for research on local autonomous energy systems. A total of 359 studies are investigated, of which a subset of 123 in detail. The studies are assessed with respect to the characteristics of their methodology and applications, in order to derive common trends and insights. Most case studies apply to middleincome countries and only focus on the supply of electricity in the residential sector. Furthermore, many of the studies are comparable regarding objectives and applied methods. Local energy autonomy is associated with high costs, leading to levelized costs of electricity of 0.41 \$/kWh on average. By analysing the studies, many improvements for future studies could be identified: the studies lack an analysis of the impact of autonomous energy systems on surrounding energy systems. In addition, the robust design of autonomous energy systems requires higher time resolutions and extreme conditions. Future research should also develop methodologies to consider local stakeholders and their preferences for energy systems.

© 2020 Elsevier Ltd. All rights reserved.

Contents

1.	Intro	duction
2.	Existi	ing literature reviews
3.	Revie	ew methodology
4	Resul	Its of the literature review
	41	Methodologies and models 7
		411 Perspective 7
		412 Target criterion 8
	42	Suctem boundarias
	4.2.	A21 Spatial conductors and location
		4.2.3 Demand
		4.2.2. Demand
	4.2	4.2.5. Consumption sectors
	4.3.	Feasibility and type of autonomy
	4.4.	Time structure and pathway
		4.4.1. Time horizon
		4.4.2. Time resolution
		4.4.3. Pathway
	4.5.	Technologies
	4.6.	Grid infrastructures
	4.7.	Costs

Corresponding author.

E-mail address: jann.weinand@kit.edu (J.M. Weinand).

https://doi.org/10.1016/j.energy.2020.117817 0360-5442/© 2020 Elsevier Ltd. All rights reserved.

5.	Critical appraisal of energy autonomy studies	. 14
6.	Summary and conclusions	. 15
	Declaration of competing interest	. 15
	Acknowledgements	. 15
	Supplementary data	. 15
	References	. 15

Abbreviations

ABS	Abstract
ar	Article
BBO	Biogeography Based Optimization
CHP	Combined heat and power
Doctype	Type of document, e.g. article or review
EA	Energy autonomy
ES	Energy system
EV	Electric vehicle
HIC	High income county
KEY	Keyword
LCOE	Levelized cost of electricity
LEA	Local energy autonomy
LIC	Low income country
LMIC	Lower middle income country
LPSP	Loss of load probability
MCDA	Multi-criteria-decision analysis
PV	Photovoltaics
RE	Renewable energy
UMIC	Upper middle income country

1. Introduction

Between 1993 and 2017, the percentage of the worldwide population with access to electricity increased from 77% to around 89% [1]. Since 2012, more than 100 million people per year have gained access to electricity. However, it is estimated that even in 2030 about 670 million people will still have no access to electricity [2]. Most people without access live in rural areas (84%) and in sub-Saharan Africa or developing Asia (95%) [3]. Negative examples include the developing countries of Burundi, Chad and Malawi, where less than 15% of the population have access to electricity [1]. Many of the newly electrified regions in developing countries apply off-grid solutions with diesel engines due to long distances to the national grid. Such completely energy autonomous systems are able to meet the energy demands of an entire community without energy imports [4].

Whereas these completely autonomous (i. e. off-grid) energy systems (ESs) exist in developing countries mainly due to cost considerations, there are also efforts by municipalities and regions to become energy autonomous in industrialized countries with complete electrification (i.e. grid-connected). This is due to the energy transition and the related environmental awareness [5] as well as the desire of citizens to play an active role in energy supply and to be less dependent on central markets and structures (e. g. Refs. [6,7]). The majority of municipalities with energy autonomy (EA) aspirations strive for balanced EA and the focus is usually on electrical energy [5]. In this context, ESs achieve balanced EA if they are energy neutral, i.e. the annual locally provided energy exceeds the annual demand [4]. In contrast to a completely energy autonomous solution, imports and exports are possible.

Local energy autonomy (LEA) can take different forms and degrees (Fig. 1). In particular, autonomous ESs need to exhibit the basic criteria as defined by Rae and Braedly [4]. First, the local system is able to generate at least as much energy to meet the demands. Second, the local system allows energy shifting possibilities for times in which there is a temporal mismatch between demand and supply (i.e. through storage or in the case of balanced autonomy through energy infrastructure). Third, the system is capable of operating independently on a *stand-alone* or off-grid basis. Thereby, local autonomy efforts are related to active participation of the community or rather the system components are owned by the community members. In general, autonomy efforts are directly related to the notion of self-governance and community ownership [4]. In this study, EA is defined the same way: it focusses on plants inside the municipality which tend to be operated by the local community, and may also include conventional technologies. However, due to the dependency on fuel transport and the high costs of diesel-based ESs, a supply consisting at least in part of renewable energies (REs) could be worthwhile in these cases. Fig. 1 shows a schematic representation of decentralized (autonomous) energy systems.

As the above examples demonstrate, balanced or completely autonomous ESs are related to different objectives and have different effects on the local setting but possibly also on the overarching system. Due to the increasing relevance of LEA (c.f. section 3), there is a need to elaborate and define transition process aspects



Fig. 1. Schematic representation of a decentralized energy system. Not all technologies found in the reviewed studies are shown. In this review only studies on the municipal level and regional level are included, i.e. no studies on the household or national level. In contrast to the completely energy autonomous case (off-grid), imports from the national level are permitted in the case of balanced energy autonomy.

and successful transition pathways. Therefore, the aim of this study is to review the literature on LEA, i.e. in villages, districts, municipalities and regions, in order to identify the current state of the art and gaps or starting points for future studies. This spatial resolution is chosen since similar conditions apply, for example from a technical point of view (decentralized energy technologies) but also from a social point of view, such as the number and type of stakeholders. Studies on individual buildings and larger regions such as entire nations are therefore excluded. In contrast to the multitude of existing reviews, this paper for the first time shows a comprehensive overview and quantitative analysis of applied methods in EA case studies at the local level (cf. section 2).

This paper is structured as follows: section 2 gives an overview of already published literature reviews about ES analyses with a focus on decentralized EA. Subsequently, section 3 presents the methodology for the literature search. Section 4 then presents and discusses the most important findings from the analysis of the literature, before the studies are critically assessed in section 5. A summary and conclusions are given in section 6.

2. Existing literature reviews

Some review papers have already examined studies on local ES planning for larger (e.g. urban areas) [8,9] and smaller regions (e.g. municipalities/communities) [4,10–16]. In some cases, the focus is on some form of EA [4,10,15,16], or this topic is at least briefly discussed [11,14]. A few other review papers [17–20] discuss EA, but do not focus on the local level as defined in section 1.

Even though there are some undeniable differences regarding

the motives of LEA efforts in developed and developing countries, due to the growing utilization of decentralized RE generators LEA projects also represent a business opportunity in industrialized countries according to the review of Engelken et al. [21]. Despite all of the criticism voiced by Heard et al. [19], the feasibility and viability of such local RE systems have been demonstrated in various studies as shown in the reviews of Brown et al. [17] and Hansen et al. [18].

3

In this context, a strategy to achieve this feasibility requires the discussion of crucial factors regarding EA. Kaundinya et al. [16] present both general success and failure stories of corresponding autonomy projects. According to the authors, a generalized approach to assess the suitability of off-grid and grid-connected systems, based on techno-economic-financial-environmental feasibility does not find adequate coverage in the literature yet. Issues and trends shaping local ESs are conceptually summarized by Keirstead et al. [8] and Koirala et al. [11]. Concerning this matter, local ESs fit very well into the neo-liberal ideas of self-reliance and environmental benefits. Key challenges are the degree and scale of EA, the matching of demand with supply, the importance of socio-economic and political factors and the structural requirements in remote communities [11].

These results go in line with the findings of the review of Rae and Bradley [4]. While taking into account the different drivers of balanced or completely autonomous energy projects, the review investigates environmental, political, economic, technical and social concerns. Even though generation and utilization of RE is



Fig. 2. Development of the number of journal publications for two search queries in Scopus. The bright curve contains the publications on energy system analyses at the local level. For the dark curve, the search query has been adjusted so that the studies also deal with autonomous energy systems. The last search in Scopus was performed on November 20, 2019.

primarily a technical challenge, social and political aspects are the most important factors in its implementation in the community. The conceptual framework of EA by Müller et al. [15] shares a similar focus: the involvement and motivation of administrations and community stakeholders are decisive for the successful transition towards LEA.

Additionally, ensuring stable supply and reliability against all plausible outcomes in RE availability might raise cost and complexity of the systems due to the impacts of worst-case conditions [19]. Distributed energy generation as with local autonomous projects encompasses a wide variety of technologies which tend to be highly sensitive to the deployment context [14]. Thus, it is important to consider the necessary spatial and temporal boundaries of the region or community.

Some of the mentioned aspects have also been posed in terms of future modelling needs. Trans-disciplinary and multi-dimensional features of low-carbon community model approaches are outlined by Nakata et al. [20]: models should consider e.g. the utilization of waste for energy, the inclusion of various sectors, and approaches related to energy-for-development issues in rural areas of developing countries. Dependent on the objectives and constraints, Gamarra and Guerrero [10] point out innovative planning guidelines by reviewing optimization techniques applied to microgrid planning. While the microgrid siting problem of autonomous systems requires robust methodologies, the operation of the autonomous system is only possible with reliable energy management systems. Thereby, stochastic optimization could be one solution for a more realistic estimate [16]. Besides, Scheller and Bruckner [12] present requirements for ES modelling at the municipal level and discuss existing optimization models concerning their fundamental approaches. They provide future modelling needs for successful ES analyses which are also linked with the mentioned challenges of autonomy projects.

In contrast to many other literature reviews, the present paper presents a comprehensive overview and quantitative analysis of applied methods in case studies on energy autonomous systems at the local level. Of the reviews discussed in this section, only [4,10,11,14–16] focus on EA at the local level. Of these six papers, only Kaundinya et al. [16] and Gamarra and Guerrero [10] concentrate on methodological aspects. However, the studies differ from the present review by concentrating on the comparison of grid-connected and off-grid ESs [16] or only on microgrid studies [10]. Furthermore, the studies are from 2009 [16] and 2015 [10], respectively, and since then, the published articles on LEA have more than doubled (cf. section 3). Apart from the general topic, this review also covers new aspects such as a compilation of costs for local energy autonomous systems (cf. section 4.7).

3. Review methodology

The literature on local ES analysis has increased exponentially from 1990 until 2019 (cf. Fig. 2), along with the LEA efforts described in section 1. Scopus¹ was used for the primary literature search, since it covers a wider range of journals [22] as well as more recent sources [23] than other databases like *Web of Science*. The *Initial search* query in Table 1 results in a total of 2,453 studies (cf. Fig. 2). The search query contains the methodology (e. g. ES analysis or simulation), the spatial resolution (e. g. municipality or region) and the restriction that it is a peer-reviewed article. 359 (15%) of the 2,453 studies on local ES analyses deal with autonomous ESs and have also exponentially increased in recent years (*Adjusted search* in Table 1 and Fig. 2).

The increasing importance of the topics could be only related to the generally exponentially increasing number of publications. However, the share of local ES analyses in the field of ES analysis (Energy system analysis search in Table 1) has increased from 8% (1990) to 20% (2019) and that of local energy autonomous systems from 0% to 3%.

The 359 studies about LEA were examined for suitability for this review. 236 studies were excluded for the reasons outlined in Table 2. For 122, most of them were excluded because of an unsuitable spatial resolution (e.g. ES analysis of a single building). A total of 123 studies remained ([24–146]), which were mainly published in the journals Energy and Renewable Energy (cf. Table 3). In addition, Table 4 shows the ten most globally cited

¹ https://www.scopus.com/search/form.uri?display=basic.

 Table 1

 Different search queries for the literature search in Scopus. The abbreviation "TITLE-ABS-KEY" is used to search for the terms in the title, abstract and keywords of the article.

 Included are original research articles (cf. "(DOCTYPE, "ar")" in the search query), which were published between 1990 and 2019.

 Number of

Search name	Search query	Number of studies
Initial search	TITLE-ABS-KEY ("energy system" AND ("simulation" OR "modelling" OR "optimization" OR "analysis") AND ("region" OR "municipalities" OR "municipality" OR "communities" OR "community" OR ("district" AND NOT "district heating") OR "city" OR "cities" OR "town" OR "remote") AND (UMIT-TO (DOCTYPE, "ar"))	2,453
Adjusted search	TTTLE-ABS-KEY ("energy system" AND ("simulation" OR "modelling" OR "optimization" OR "analysis") AND ("region" OR "municipalities" OR "municipality" OR "communities" OR "community" OR ("district" AND NOT "district heating") OR "city" OR "cities" OR "town" OR "remote") AND ("off-grid" OR "off grid" OR ("100%" AND "RE") OR ("100%" AND "renewable") OR "100%- renewable" OR ("energy" AND "autonomy") OR ("energy" AND "autarky") OR ("energy" AND "self-sufficiency") OR ("energy" AND "self-sufficient") OR "energy independent" OR "stand-alone" OR "energy autonomous" OR "island system")) AND (LIMIT-TO (DOCTYPE', ar"))	359
"Energy system analysis" search	TITLE-ABS-KEY("energy system" AND ("simulation" OR "modelling" OR "optimization" OR "analysis")) AND (LIMIT- TO(DOCTYPE,"ar"))	12,368

 Table 2

 Studies that resulted from the adjusted search in Scopus (cf. Table 1) and are not considered in this literature review for the reasons given in the table.

 Fundamental exterior

Texturing adjuster	Number of studies	D.6
Exclusion criterion	Number of studies	References
The study does not consider energy autonomy as defined in section 1 (i. e. at least balanced autonomy has to be analysed)	30	[147-176]
Autonomy is considered in a different context than energy	2	[177,178]
Autonomy is only mentioned as a future target in the paper	3	[179-181]
The spatial resolution of the study does not match our definition of local energy systems (cf. section 1)	122	
Single consumer/households/building	41	[182-222]
Single commercial application	57	
o Agricultural well	2	[223,224]
o Desalination unit	7	[225-231]
o Cellular base station/telecommunication unit	11	[232-242]
o Hospital/healthcare facility	5	[243-247]
o Hotel	5	[248-252]
o Library	1	[253]
o Wireless sensor nodes	1	[254]
o Machinery laboratory	1	[255]
	6	256-261
o Voter registration centre	1	[262]
o Desert safari camp	1	263
o Touristic facility	1	264
o Charging station	1	265
o Mining site	3	[266-268]
o Factory/enterprise	3	269-271
o Refinery	1	272
o Road lighting system	1	[273]
o University facility/school	4	274-277
o Clean water and toilet system	1	[278]
o Wastewater treatment plant	1	279
Large regions	3	280-282
One or several countries	21	[283-303]
Analysis of a single energy plant/technology	35	[304-338]
Aerospace applications	2	[339,340]
Climate analyses	4	[341-344]
Study focusses on control strategies of an energy system	13	345-357
Study introduces a new model without autonomy case study	3	[358-360]
Study develops load profiles for off-grid areas	2	[361.362]
Study focusses on qualitative analysis	15	[363-377]
Analysis of a given 100% renewable system	2	[378,379]
Text language: Korean	2	380,381
Publication not found	1	[382]

 Table 3

 Distribution of the studies among the journals in which they were published. Only those journals are shown which have published five or more studies.

Journal	Number of studies	Share in 123 studies [%]
Energy	16	13
Renewable Energy	14	12
Energy Conversion and Management	9	7
International Journal of Renewable Energy Research	9	7
Applied Energy	7	6
Energies	7	6
Journal of Cleaner Production	6	5
Solar Energy	5	4

6 Table 4

Most relevant articles on local energy autonomy, based on global citations (January 15, 2020).

Article	Title	Journal	Global citations
Ashok 2007 [33]	Optimized model for community-based hybrid energy system.	Renewable Energy	295
Ma et al. 2014 [134]	A feasibility study of a stand-alone hybrid solar-wind-battery system for a remote island.	Applied Energy	257
Kanase-Patil et al. 2010 [72]	Integrated renewable energy systems for off grid rural electrification of remote area.	Renewable Energy	187
Østergaard and Lund 2011 [100]	A renewable energy system in Frederikshavn using low-temperature geothermal energy for district heating.	Applied Energy	180
Ma et al. 2014 [85]	Technical feasibility study on a standalone hybrid solar-wind system with pumped hydro storage for a remote island in Hong Kong.	Renewable Energy	158
Maleki and Askarzadeh 2014 [88]	Optimal sizing of a PV/wind/diesel system with battery storage for electrification to an off-grid remote region: A case study of Rafsanjan, Iran.	Sustainable Energy Technologies and Assessments	102
Haghighat Mamaghani et al. 2016 [60]	Techno-economic feasibility of photovoltaic, wind, diesel and hybrid electrification systems for off-grid rural electrification in Colombia.	Renewable Energy	95
Gupta et al. 2010 [58]	Steady-state modelling of hybrid energy system for off grid electrification of cluster of villages.	Renewable Energy	92
Rohani and Nour 2014 [111]	Techno-economical analysis of stand-alone hybrid renewable power system for Ras Musherib in United Arab Emirates.	Energy	89
Askarzadeh and dos Santos Coelho 2015 [36]	A novel framework for optimization of a grid independent hybrid renewable energy system: A case study of Iran.	Solar Energy	85

Table 5

Overview of the general methodologies applied in the studies.

General methods	Study	Number of studies
Artificial bee swarm	[89,90]	2
optimization		
Biography based optimization	[133]	1
algorithm		
Chaotic search	[128,129]	2
Cukoo search	[115]	1
Discrete harmony search	[47,88,128]	3
Electricity System Cascading	[64]	1
Analysis		
Energy balancing calculation	[70,99,104,122]	4
Firefly algorithm	[114]	1
Flower Pollination algorithm	[113]	1
Fuzzy analytic hierarchy	[98]	1
process		
Genetic algorithm	[69,106]	2
Grey relation analysis	[98]	1
Life cycle cost analysis	[87]	1
Multi-objective particle swarm	[37,125,126]	3
optimization		
Multi-objective crow search	[68]	1
algorithm		
Multi-objective optimization		1
Non-dominated sorting genetic	[125,126]	2
algorithm-ll		10
Uptimization	[30,34,49,54,56,58,59,63,72,74,78,92,98,109,117,124,146]	16
Multi-criteria-decision analysis	[80,92]	2
(MCDA)	[20.01.101]	2
Particle swarm optimization	[36,01,101]	3
Simulation	[128] [34] [30] [31] [32] [32] [32] [42] [42] [42] [45] [52]	1
SHILUALION	[24]= [24] [25] [25] [25] [45] [45,46,48,50] [25] [25,7,60,62,65] [67,71,75]= [77] [78] [86,91,93]= [97] [100,102,103,105,107,108,110,111,116,118] [121] [123,127,130] [132] [134] [145]	69

articles on LEA.

4. Results of the literature review

This section presents and discusses the main findings of the analysis of the 123 studies. First, section 4.1 explains the methods and models used in the studies. Subsequently, section 4.2 shows the system boundaries of the case studies. The type and feasibility of EA under consideration is presented in section 4.3. In the following section 4.4, the temporal resolution is discussed. Section 4.5 then indicates the included technologies before section 4.6 examines the consideration of grid infrastructures. Finally, section 4.7 compares

and discusses the costs for LEA resulting in the studies.

The supplementary Microsoft Excel file in the online appendix contains all the information discussed in this article. Not all aspects could be identified in all studies. In some of these cases, the information in the Excel file are given in parentheses, which indicates assumptions based on detailed analysis of the study. If, in the following sections, the shares in the number of studies do not add up to 123 or the percentages do not add up to 100%, this is due to the fact that not all information could be retrieved from every study.

Model	Study	Number of studies
HOMER or HOMER Pro	[24,25,26,27,28,29,32,35,38,39,41,42,43,45,46,48,50,53,55,57,60,62,65,66,71,75,76,77,80,81,82,83,86,91,93,95,96, 97,105,107,108,110,111,118,119,120,121,127,131,132,134,135,136,137,138,139,140,141,142,144,145]	61
EnergyPLAN	[44,51,52,100,116,123,130]	7
RE ³ ASON	92,124	2
BeWhere/Phasma	[117]	1
FINE/TSAM	[146]	1
FlexiGIS	[30]	1
H ₂ RES	[79]	1
REOM	73	1
ISLA	98	1
LINGO v.10	[72]	1
OSeMOSYS	[56]	1
P ² IONEER	78	1
RegFin	[104]	1

4.1. Methodologies and models

The methods employed in the reviewed literature (cf. Table 5) range from simple energy balancing calculations (e. g. Refs. [70,99,104]) to simulations (e. g. Refs. [79,94,102]), metaheuristics (e. g. genetic algorithm [69], discrete harmony search [88], artificial bee swarm optimization [89,90] or flower pollination algorithm [113]), mixed-integer linear optimizations (e. g. Ref. [30]) and multi-objective optimizations (e. g. Ref. [112]).

When investigating the methods, it is striking that in most cases only simplified calculations are carried out. For example, the number and size of plants is usually predefined and not optimized. Simulations are frequently used (56%, cf. Table 5) and often referred to as optimization. The best examples are the simulation softwares HOMER and EnergyPLAN, which are used in 61 (50%) and 7 (6%) studies respectively (cf. Table 6). Apart from HOMER and EnergyPLAN, other models appear only once or twice or are not specified in the studies (cf. Table 6).

Although most HOMER studies call it optimization, in its core HOMER is a simulation model [383]. The simulation is used to balance energy generation and consumption and calculate the costs. Thousands of scenarios with different parameters can be simulated in a sensitivity analysis. Subsequently, in the so-called optimization, the best solution is selected from among all scenarios depending on the selected criterion (e.g. minimization of costs or fuel usage) [383]. This explorative approach to identifying a pareto front does not necessarily yield the optimal solution. The vast majority of the studies reviewed here in which HOMER is used have a similar structure: first, the economic parameters, the load profile as well as the renewable potentials and the ES under consideration are described for a particular application. The best ES is then usually selected on the basis of costs (97% of cases). These studies therefore typically focus on case studies rather than methodological innovations. The aim of these studies is usually to reduce the diesel consumption of a remote off-grid area. In three studies, newly developed methods were compared with the HOMER model. The results showed that a Biogeography Based Optimization (BBO) algorithm [133], a Genetic Algorithm [69] or the so-called LINGO model [72] perform better than the HOMER model in terms of computing time and minimization of costs. The BBO algorithm, for example, found a better solution than HOMER and reduced the computing time from 15 h to 0.7 h [133]

Despite the weaknesses that HOMER shows in determining an optimal ES, the advantages of the tool should also be appreciated. It is an open access tool that can be used by everyone and does not require a lot of computing power. Therefore, the model is particularly useful for studies on remote areas in countries with low or middle income (75% of the HOMER studies reviewed here) according to the country classification of the World Bank [384] (cf. Table 8).

In contrast to the HOMER model, the EnergyPLAN model was only used for case studies in countries with middle (China [130]) or high (Italy [44], Croatia [51,116], Denmark [52,100], Ireland [123]) income. However, this model is also classified "as a simulation tool rather than an optimization tool" [385] and only includes a dispatch optimization. This means that the user has to specify the technologies and thus has to have a comprehensive understanding of ES analysis (as in HOMER). A step in the right direction was the introduction of HOMER Pro, which is intended to simplify the use of the HOMER software for inexperienced users [386].

Beyond these two models, there are more and more model approaches that generate inputs for optimization models on the basis of publicly available data such as OpenStreetMap.² Examples in the reviewed literature are the FlexiGIS model [30] and the RE³ASON model [92,124]. At least for FlexiGIS an open-source publication in GitHub is planned, according to the authors [30]. These models would enable inexperienced users in the field of ES analysis to determine the energy potentials of a region and optimize the ES. Neither potentials nor technologies and plant sizes would have to be determined before application of the models.

4.1.1. Perspective

In most ES analyses, a *central planner* was used as the perspective. Only in Ramchandran et al. [107], the perspective of a Renewable Energy Service Company was taken instead. On the one hand, these central planner approaches show the macroeconomic optimum. On the other hand, however, these approaches often fall short of realisation: it could be difficult to convince individual homeowners to install the technologies in their homes that are optimal from a macroeconomic but not a business point of view. Therefore, studies of possible incentive systems that could encourage homeowners to implement the macroeconomic EA plan are required. First conclusions could be drawn by comparing the optimal ES at building level from the perspective of the building owner on the one hand and from the macroeconomic perspective on the other.

Promising approaches include for example the combination of optimization approaches with *multi-criteria-decision analysis* (MCDA), as in Refs. [80,92]. These types of analyses do not yet include the perspective of every inhabitant, but at least the perspective of their most important representatives. This could help to strengthen the feasibility of an EA plan.

² https://www.openstreetmap.org/.

Table 7

8

Applied target criteria besides cost in the studies with MCDA or multi-objective optimizations.

Criterion	Study	Number of studies
CO ₂ emissions	[70,80,92,98,112]	5
Land use	[54,70,80,83]	4
Security of supply	[37,68,83,98]	4
Forecast accuracy	[125,126]	2
Renewable share in energy supply	[37,83]	2
Social acceptance	[80,98]	2
Community net imports	[92]	1
Creation of jobs	[80]	1
Ease of installation and operation	[98]	1
Flexibility of the system for future expansion	[98]	1
Gross domestic production	[104]	1
Health issues	[80]	1
Human development index	[80]	1
Noise	[80]	1
Risk of flash floods	[80]	1
Technical efficiency	[80]	1
Technical reliability	[80]	1
Technical lifespan	[80]	1
Technical scalability	[80]	1
Technical maturity	[80]	1
Universal education and gender equality	[80]	1
Water consumption	[54]	1
Water quality	[80]	1

Table 8

Classification of the countries according to The World Bank [384], in which the case studies are conducted in the reviewed literature as well as the share in the total of 123 publications. This country classification is based on the gross national income per capita [384].

Income group [384]	Gross national income per capita [384] [\$]	Countries and studies	Share [%]
Low income	[0; 1,025)	Ethiopia [66,96], Rwanda [144], Tanzania [140], Yemen [31]	4
Lower middle	(1,025; 3,995]	Bangladesh [50,91,136], Cameroon [95,127], Egypt [113], Ghana [26,28], India [29,33,45	33
income		-47,58,59,72,73,77,81,87,93,101,106,107,109,114,115,131,133], Indonesia [32,120], Nigeria [27,137,139], Pakistan	
		[71,75,110], Philippines [98], Timor-Leste [56]	
Upper middle	(3,995; 12,375]	Algeria [43,135], Brazil [118], China [69,80,82-84,130], Colombia [60], Cuba [132], Iran [25,34-36,41,61,63,65,68,88	30
income		–90,125,126,128,129,141], Iraq [138], Malaysia [62,64,119,121], Maldives [55], Mexico [54], Turkey [53]	
High income	(12,375; inf)	Australia [40,57], Austria [117], Canada [38,39,105,122,145], Croatia [51,79,116], Denmark [52,100], Finland [104],	33
		Germany [30,70,78,92,94,124,146], Greece [102,103], Hong Kong [85,86,134], Ireland [123], Italy [44], Japan [143],	
		Korea [76], Oman [24], Saudi Arabia [42,108,142], Scotland [48], Sweden [37], Switzerland [112], United Arab	
		Emirates [97,111], USA (Alaska) [49]	

4.1.2. Target criterion

The adopted perspective is closely linked to a further shortcoming of many of the studies examined: the choice of the target criterion. In 103 (84%) of the studies only costs were minimized. As already shown in section 1, however, costs are not the only reason for EA projects. In a few studies, a different target criterion than costs is applied, but these analyses do not represent an improvement since also only one criterion is considered: technical feasibility [85,116], annual efficiency [102,103] or coal consumption [130]. However, beside the above-mentioned MCDA approaches, other multi-objective approaches represent improvements. In addition to costs, the criteria in Table 7 are also taken into account in the reviewed studies. Possible weightings for the criteria could either be determined on site before the case studies, or taken from surveys such as in Ref. [5].

4.2. System boundaries

This section first presents the spatial resolution and location of the case studies in the reviewed literature (cf. section 4.2.1). Section 4.2.2 then shows that the focus is primarily on the demand product electricity. Finally, the demand sectors considered in the studies are highlighted (cf. section 4.2.3).

4.2.1. Spatial resolution and location

Table 8 classifies the case studies of the reviewed papers according to the income classification of The World Bank [384]. Most of the studies were conducted in the lower middle income country (LMIC) India (21), upper middle income countries (UMIC) Iran (17) and China (6) as well as the high income country (HIC) Germany (7). Whilst the case studies in India and Iran focused mainly on the ESs of remote areas without grid connection, in Germany a complete electrification already exists. The case studies on complete autonomous ESs in HICs are therefore more about isolating communities from the transmission grid. These studies are linked to the question whether the ES transformation should be achieved through decentralized or centralised expansion of RE sources. In Ref. [387], for example, the decentralized expansion is evaluated as more cost-effective for the German case due to higher required transmission grid expansion costs in the centralised case. The possible impacts of autonomous communities on the surrounding ES (cf. section 4.3) therefore plays a very important role especially in the HIC studies.

Whilst the low income countries (LIC), the LMIC and the UMIC mainly consider remote rural areas as case studies with 100%, 92% and 74% of the studies, respectively, these remote areas account for only 23% of the studies on HIC. Instead, studies on HIC also often investigate EA for islands (23%), cities (20%), regions (17%) and municipalities (14%). This is also reflected by the number of residents examined in the case studies: in the studies on LIC, LMIC and

UMIC case studies with a maximum of 4,750 inhabitants are analysed; in the HIC studies, case studies with up to 640,000 inhabitants are investigated. The exact area or city names of the case studies as well as the number of households and inhabitants examined can be found in the supplementary Excel file.

In this context, it is noticeable that the complexity of the applied methodology adapts to the size of the considered region. This means, the smaller a region is chosen, the more details can be included in the analysis. For example, in Waenn et al. [123] the operation of the ES with 640,000 inhabitants is determined with the help of a less complex simulation (EnergyPLAN). The largest case study in which an ES is designed with the help of an optimization is in Schmidt et al. [117]: a large region with 21,000 inhabitants. However, in Schmidt et al. only two time slices are considered during the optimization (cf. section 4.4.2) to reduce the model complexity.

In [60,76,104,106,111,124,144] different spatial scales were compared as case studies, i.e. the number of households or inhabitants was varied. However, these studies do not yet provide indications about the optimal size of energy autonomous ESs.

4.2.2. Demand

In all the studies reviewed, the electricity demand of the ES is included. Heating or cooling demand, on the other hand, is only considered in 30 (24%) or 13 (11%) of the studies, respectively. As already indicated in section 1 for EA projects, this also demonstrates the focus on electricity in the literature. In most cases, the demand is based on time series that have been determined or collected beforehand. However, there are also examples of EA case studies such as Refs. [30,92,124], in which the demand and load profiles are determined automatically on the basis of publicly available data. The electricity and heat generation technologies used in the studies are presented in section 4.5.

In addition to electricity and heat, other demand products such as food are also indirectly taken into account, for example through land-use competition as in Schmidt et al. [117]. However, only direct demand products are discussed in the present literature review. This includes the demand for water considered in ten studies. In Refs. [28,46,47,63,93] this is considered by the electricity demand of a water pump, e. g. for an agricultural well. In Refs. [67,74,79,89] the ES contains a desalination unit for water distillation. In Fuentes-Cortés et al. [54] the water-energy nexus is considered in the analvsis, which means that water demand in the energy supply is taken into account. In this case the water demand includes fresh water for households, water used for regulating the temperature of the thermal demand as well as water needed as by-product in the combined heat-and-power (CHP) units. Water consumption is included in Ref. [54] alongside costs and land use in the multiobjective function of the optimization model. Therefore [54], in particular shows a suitable way to consider water demand in future studies about autonomy. The studies [47,54,74,79] are the only examples which consider all three types of demand (electricity, heat and water).

4.2.3. Consumption sectors

Among the consumption sectors, mainly the residential sector is considered (102 studies; 83%), followed by the commercial sector (55; 45%), industrial sector (23; 19%) and transportation sector (11; 9%). The ES is usually designed for all considered sectors. By contrast, in Bagheri et al. [38] the residential, commercial and industrial sectors are examined in separate analyses. Thereby, the ES for the industrial sector shows the lowest *levelized cost of electricity* (LCOE) in the autonomous case with 100% RE. Whilst the commercial sector with schools and hospitals also is important in studies about remote areas, larger industries and the transport

sector are considered almost exclusively in case studies for municipalities, cities, islands or larger regions. In Refs. [72,73], industries are also considered in remote villages. However, they are referred to as *rural industries*, which probably corresponds more to the commercial sector of HICs in terms of demand structure. An interesting point is that in the cases where heat and industry were regarded, only a balanced autonomy is part of the analysis. This is probably due to the fact that, for example, high-temperature heat in industry can only be generated with specific RE plants and, in the completely autonomous case, would be associated with excessively high costs.

For residential, commercial and industrial sectors, the demand is usually known in advance in the studies. However, for the consideration of the transport sector several different approaches are applied. In Refs. [70,100,104,123] a fixed fuel demand for traditional vehicles is covered. In Refs. [51,52,116], electric vehicles (EVs) are considered within the EnergyPLAN model. In Dorotić et al. [51] all vehicles and ferries on the island of Korčula in Croatia are replaced by electrically powered alternatives. The EVs not only serve as batteries, but can also be used for vehicle-to-grid, i.e. feeding electricity from the EV battery into the grid. Šare et al. [116] analyse three scenarios for the municipality Dubrovnik in Croatia with different EV penetrations in 2020, 2030 and 2050. Krajačić et al. [79] and Oldenbroek et al. [99], on the other hand, included fuel cell vehicles in their ES analyses. In some scenarios in Ref. [79] the transport load is covered 100% by renewable hydrogen. None of the studies optimizes the number of electric or fuel cell vehicles.

4.3. Feasibility and type of autonomy

In the reviewed literature, studies on completely autonomous ESs predominate with 110 (89%), whereas balanced LEA is only considered in 14 (11%) studies. The only study that analyses both cases seems to be Sameti and Haghighat [112], in which a net-zero energy district is investigated in three scenarios with grid connection and one as a stand-alone variant without grid connection.

Generally EA is feasible in the case studies. The only exception is the study by Alhamwi et al. [30] who do not obtain a feasible solution in their ES model and come to the conclusion that an off-grid city (165,000 inhabitants) is economically and technically not practicable. However, there are also other examples which do not come to a favourable result for LEA. Krajačić et al. [79] find that the cost of electricity for a 100% renewable island is up to 15 times higher than the current (2009) electricity price. Furthermore, in Oldenbroek et al. [99] a 100% renewable supply can only be achieved if 20% of the vehicle fleet are fuel cell vehicles. Also Šare et al. [116] come to the conclusion that large storage capacities are necessary for a 100% renewable supply. Jenssen et al. [70] show that the available biomass potentials of a model municipality are sufficient for 100% power and heat supply, but not to replace transport fuel. All these examples have in common that they examine bigger regions, cities and islands as case studies in high-income countries in Europe.

In the studies on completely autonomous ESs, uncertainties due to disconnection from the grid infrastructures should play a very important role, since a non-optimal design of the ES cannot be compensated by imports. Therefore it is even more important to design these ESs robustly. There are several appropriate approaches in the studies. For example, in 39 (32%) studies a possible security of supply below 100% is implemented as a *loss of power supply probability* (LPSP). In most cases ([34-36,45,47,58,59,63, 65,69,71-73,81,82,84,85,89-91,95-97,101,109,113,114,127,129]), the LPSP is modelled as a fixed value or results from other fixed

the LPSP is modelled as a fixed value or results from other fixed values. Other studies ([37,61,68,83,98,106,115]) in which the LPSP is

associated with weightings or penalty costs and thus integrated into the objective function of an optimization represent an improvement. In Hakimi et al. [61] different penalty costs were assumed for the residential, commercial and industrial sector. In future studies, the so-called value of loss load could be a suitable estimation of penalty costs. In Shivakumar et al. [388], for example, the value of loss load was calculated for households in all European Union member states. This data set with penalty costs based on the same methodology could make results of studies more comparable. As expected, the LPSP are rarely considered in HICs, as the inhabitants are accustomed to high security of supply. Four case studies in Canada [145], Sweden [37] and Hong Kong [85,134] are the only examples. However, for autonomous systems these LPSP become more relevant.

Further studies try to robustly design off-grid ESs by taking extreme conditions into account. In Petrakopoulou et al. [102,103], the plants of the ES are over-dimensioned and complementary technologies are used. In addition, the optimization model in Weinand et al. [124] considers extreme days on which demand is particularly high and no wind or solar radiation is present.

4.4. Time structure and pathway

In this section the time horizon (section 4.4.1), the chosen temporal resolution (section 4.4.2) and the pathway for the ES transition (section 4.4.3) are demonstrated.

4.4.1. Time horizon

The time horizon in the case studies is usually chosen between 15 and 25 years, which represents an appropriate choice for estimating total discounted system costs or LCOEs for an ES. However, there are also variations upwards and downwards: Adamarola et el [28], and Drysdale et al. [52] even consider 35 and 45 years respectively. Jenssen et al. [70], Moeller et al. [94], Oldenbrock et al. [99], Østergaard and Lund [100] as well as Šare et al. [116] consider one year whereas Kandil et al. [74] use only a time horizon of 24 h. For the latter study, a time horizon of 24 h could be too short, even though only the operating costs of an autonomous ES are determined. At least an extreme day should have been considered for this analysis.

4.4.2. Time resolution

The time resolution of models is of particular importance in studies on EA. This is especially true for completely autonomous ESs (cf. section 4.3). Non-optimal design of balanced energy autonomous systems could be compensated by imports from surrounding energy infrastructures. This does not apply for complete autonomy. Therefore, a particularly critical assessment is made when off-grid ESs are designed on the basis of an annual energy balance, as it seems in Stephen et al. [122]. Stephen et al. [122] investigate the residential and commercial energy supply for an off-grid Canadian aboriginal community. There are also other examples with a very rough time resolution, but these studies only consider balanced autonomy: Jenssen et al. [70] and Peura et al. [104] also conduct an annual balancing of energy (i.e. one time step) whereas the optimization model of Schmidt et al. [117] is based on two seasons (winter/summer) per year (i.e. two time steps).

In almost all studies (91, 74%) the time resolution is set to hours. There is only one study with a higher time resolution, namely Kötter et al. [78] with 15-min time steps. Kötter et al. investigate the balanced EA of a region consisting of 17 sub-regions in Germany. However, it is not clear how many of the 15-min time steps are used in the analysis. The robustness of results on completely autonomous ESs based on models with hourly resolution must at least be questioned. In these cases it is even more important to consider the methods explained in the previous section, such as LPSP or extreme conditions. In addition, ESs based on base-load capable technologies such as biomass can be considered more robust than those based only on volatile energy such as wind or photovoltaics (PV) (more on this in section 4.5). Usually all hours of a year are considered in the investigations with hourly resolution (59 of 91, 65%).

Overall it seems, however, that the number of time steps decreases with the complexity of a model, presumably in order to avoid computing time problems: the RE³ASON model based on public data uses only 288 [92] or 432 [124] time slices and the multi-objective optimizations of Fuentes-Cortés and Ponce-Ortega [54] or Yazdanpanah Jahromi et al. [125] use only 96 and 744 time slices respectively. Another example is the optimization with *multitier targets* (e. g. scenarios with different demands) according to the *World Bank Global Tracking Framework* by Fuso Nerini et al. [56], which comprises only 18 time steps per year. This is a general problem of ES analyses. However, as mentioned above, the number of time steps is more crucial in ES analyses including complete EA.

4.4.3. Pathway

EA projects are always associated with the objective that the ES will become energy autonomous in the medium to long-term future. This means that there will be a transition over several years. However, in almost all reviewed studies (115, 94%), overnight is chosen as the pathway, i.e. the new ES replaces the old one immediateley and not during several years. This would correspond to an inaccurate calculation of total discounted system costs or LCOEs, as demands and costs may change during the considered time horizon. Only in Dorotić et al. [51], Drysdale et al. [52], Fuso Nerini et al. [56], Krajačić et al. [79], McKenna et al. [92] and Weinand et al. [124] was the pathway modelled as a transition. Dorotić et al. [51] seem to simulate at least every second year in EnergyPLAN from 2011 to 2030. The CO2 emissions of the system are decreasing and the REs share is increasing until they reach their minimum (0% CO2 emissions) or maximum (100% RE share) values in 2030. Drysdale et al. [52] also use the EnergyPLAN model. However, they seem to simulate only two years, 2016 and 2050. Fuso Nerini et al. [56] apply the system optimization model OSe-MOSYS for a case study village in Timor Leste. The authors seem to optimize every year from 2010 until 2030. However, as mentioned above, for each year only 18 time steps are considered (six per day and three seasons per year), i.e. 360 time steps in total. Thereby the demand changes during the time horizon. For example, it is assumed, that the households reach the target tier in 2025. The target tier would be one of five tiers: for example the households would get access to general lighting, air circulation and television in tier-2 or small appliances in tier-3. In the H2RES model in Krajačić et al. [79] every fifth year from 2005 until 2015 is simulated. The same applies to the RE³ASON model in McKenna et al. [92] and Weinand et al. [124] (time horizon from 2015 until 2030). In the two latter studies, however, the method is a mixed-integer linear optimization: all four years are optimized simultaneously, i.e. it is decided when which plant or measure will be installed. By considering the existing infrastructure (e.g. already installed PV modules), as in the RE³ASON model, models are enabled to consider a transition pathway.

4.5. Technologies

As already discussed in section 4.1, many case studies on energy autonomous remote rural areas deal with the reduction of diesel and the increase of REs in the system. As shown in Table 9, diesel, therefore, is the most frequently considered in the studies after PV, wind and stationary batteries. A total of 73 studies (59%) consider

55

Table 9

Classification of the technologies included in the reviewed literature as well as the frequency of their consideration.

Category	Technology	No. of studies
Renewable electricity generation technologies	PV	117 (95%)
	Wind (onshore)	85 (69%)
	Biomass CHP	39 (32%)
	Hydropower plant	21 (17%)
	Concentrated solar power	3 (2%)
	Deep geothermal plant	2 (2%)
Heating/cooling technologies	District heating/cooling	10 (8%)
	Heat pump	7 (8%)
	Solar thermal collector	6 (5%)
	Electric heater	6 (5%)
Storage technologies	Stationary battery	93 (76%)
	Hydrogen with fuel cell	18 (11%)
	Thermal	8 (7%)
	Pumped-hydro	4 (3%)
	Power-to-Gas (methanisation)	2 (2%)
Transport technologies	Electric vehicle (modelled with battery)	3 (2%)
	Fuel cell vehicle	3 (2%)
Conventional generation technologies	Diesel generator	63 (51%)
-	Gas fired CHP	6 (5%)
	Gas turbine plant	6 (5%)

conventional generation technologies such as diesel generators and gas fired CHP/turbine plants in their ES analyses.

When classifying biomass CHP, hydropower plants, deep geothermal plants as well as conventional generation technologies as baseload-capable, 26 studies (21%) remain, in which only volatile generation technologies are considered. In 16 of these 26 studies, no long-term storage options such as hydrogen storage, pumped-hydro-storage or power-to-gas are considered. In such cases, it is essential to take account of uncertainties. In ten of the 16 studies [34–36,65,69,84,90,114,115,134], these uncertainties are at least addressed via LPSP and in another study [102] by including extreme conditions. Even more than in other studies, the results of the completely autonomous ESs in Al-Shetwi et al. [31], Khan et al. [76], Kim et al. [76] and Mas'ud [137] must therefore be questioned, in which only volatile energy technologies and no uncertainties are considered.

The fact that so few studies examine heating or cooling technologies (cf. Table 9) is related to the focus on electricity in the studies (cf. section 4.2.2). In addition, technologies that do not belong to the standard technologies such as PV or wind are primarily investigated in case studies in HIC in Europe. For example, the technologies deep geothermal energy, power-to-gas or district heating are analysed primarily in Germany (deep geothermal energy: [124]; power-to-gas: [78,94]; district heating: [70,124]) or Denmark (deep geothermal energy (only heat): [100]; district heating: [52,100]), while unconventional vehicles such as EVs [51,52,116] or fuel cell vehicles [79,99,141] are examined primarily in case studies in Croatia. This suggests that the studies on remote rural areas are primarily concerned with the electrification of the area and not with the choice of optimal energy technologies. On the other hand, technologies such as deep geothermal energy (despite high potential in e.g. India or Sub-Saharan Africa [389]) are not relevant for these rather small regions (see section 4.2.1) due to high fixed costs [124].

In summary, the studies on LEA investigate a wide range of technologies. However, for a robust design of an energy autonomous system based on REs, the combination of fluctuating and nonfluctuating generation technologies as well as different storage technologies could be advantageous. Some of these technologies that could be beneficial in a completely autonomous case, such as seasonal heat storage, have not yet been analysed. In general, the more diverse the technologies under consideration, the more economically or environmentally sustainable the ES could be designed. On the other hand, the complexity and computing time of ES models increases with the number of technologies. In any case, work still needs to be done in which a very broad range of technologies is considered and the optimum technologies for EA are identified. Based on the results of the reviewed studies, no definite trend towards the most economic technologies for achieving EA can yet be identified (cf. section 4.7).

4.6. Grid infrastructures

Grid infrastructures are rarely modelled in the studies. Heating grids are implemented only in Refs. [92,112,124], the electricity grid only in Refs. [30,92,94,124]. In Refs. [26,77] at least the costs for setting up the distribution network are taken into account. It is interesting to note that all four case studies that consider the electricity grid are located in Germany. In Refs. [92,124] the electricity and heating network is only represented in a simplified way by energy flows between districts. However, Weinand et al. [124] also contains a transferable approach for designing district heating networks in arbitrary municipalities. District heating systems are also designed in Sameti and Haghighat [112]. While in Ref. [124] the district heating network is modelled top-down for entire municipalities, Ref. [112] is better suited as a bottom-up application for districts for which exact building locations and energy demands are known. Therefore, depending on the application, the two studies offer possible approaches for future analyses.

In Moeller et al. [94], the capacities and connections of the electric transmission network between German regions are modelled. The analysis also examines whether the transmission capacities are sufficient, depending on the share of REs. In the FlexiGIS model in Alhamwi et al. [30], OpenStreetMap is used to obtain data on lines and substations of the distribution network in order to determine the optimal placement of a battery storage in an urban area. Unfortunately, power grid data is not yet completely included in OpenStreetMap and therefore this method is not useable for every case study.

Electricity grids are only considered in a simplified way in the papers. A promising approach for future studies could be the one of Morvaj et al. [156], in which the distribution network is modelled according to a linearized AC power flow approach. Of interest is the implementation of a binary modelling variable, which determines



Fig. 3. LCOEs of the energy autonomous case studies in the literature. The studies are sorted by mean LCOEs, from high to low. The LCOEs were adjusted according to inflation until 2019 [391] and converted into \$/kWh using the average exchange rates [392-394] in the year of the respective publication. The household electricity price (black dots) in the different countries is shown for comparison [395]. The black dotted line shows the average value (0.41 \$/kWh).

Table 10

Mean levelized cost of electricity in the 83 studies depending on the characteristics of the energy systems. The LCOEs were adjusted according to inflation until 2019 [391] and converted into \$/kWh using the average exchange rates [392-394] in the year of the respective publication.

Characteristics of energy system	Number of studies	Mean LCOE [\$/kWh]	Highest outlier [\$/kWh] and respective study	Mean LCOE without highest outlier [\$/kWh]
All energy systems	83	0.41	3.27 (Askari and Ameri [34])	0.38
Including conventional energy	48	0.37	0.95 (Bekele and Palm [66])	0.36
Including conventional energy (security of supply	< 5	0.61	2.01 (Shezan et al. [121])	0.26
100%)				
100% RE	37	0.41	1.57 (Li [82])	0.38
100% RE (security of supply < 100%)	18	0.43	3.27 (Askari and Ameri [34])	0.27

whether the distribution grid needs to be upgraded, depending on the amount and type of REs added to the ES. In the case of an upgrade, the expansion of REs would involve additional costs. However, the study uses available grid data of the *IEEE European Low Voltage Test Feeder case* [390]. Since this grid data is not available for arbitrary case studies, the grid capacities would have to be estimated.

4.7. Costs

Section 4.1 has already shown that the ESs in the literature were mostly designed on the basis of cost minimization. Therefore, a comparison of these costs is reasonable. In 83 (68%) of the 123 studies, the LCOEs for autonomous ESs were stated (cf. Fig. 3). For Fig. 3, the LCOEs from the studies were adjusted according to inflation until 2019 [391] and converted into \$/kWh using the average exchange rates [392–394] in the year of the respective publication. As all but one of the 83 studies consider the residential sector, the household electricity price (import from grid) in the different countries is shown for comparison [395]. For eight countries (e.g. Ethiopia [66] or Yemen [31]), the household electricity price could not be found.

The mean LCOEs amount to 0.41 \$/kWh (black dotted line in Fig. 3 and Table 10). Consequently, the costs in the studies of Khan et al. [75] and Hosseini et al. [65] are nearly average. The LCOEs in an ES with 100% RE (case studies with 100% RE as well as with 100% RE and a security of supply < 100% in Fig. 3 and Table 10) are on average 0.42 \$/kWh (0.37 \$/kWh without the outlier in Askari and Ameri [34]) whereas in an ES with conventional energy 0.39 \$/kWh (0.36 \$/kWh without the outlier in Shezan et al. [121]) is achieved. As expected, in studies considering both cases, the LCOEs of 100% RE systems are higher than of conventional ESs. Likewise, the LCOEs decrease if cases with security of supply below 100% are considered. Furthermore, the household electricity prices are lower than the LCOEs for the autonomous system for almost every study. The only exceptions include case studies in countries with aboveaverage electricity prices (Australia [40], Croatia [51] and Germany [78,94]), or the two studies with the lowest LCOEs [106,109] in Fig. 3.

18 out of the first 20 upward outliers in Fig. 3 apply the HOMER model (for the other two, the applied model cannot be found in the article), i.e. a non-optimal design of the ES could be responsible for the high LCOEs (cf. section 3.1). In the following, the three studies from Fig. 3 with the highest upward outliers in the LCOEs are discussed (up to the study Li [82] with 1.57 \$/kWh on average). The highest LCOEs in the study by Askari and Ameri in 2009 [34] are caused by the high inflation in Iran between 2009 and 2019 (+364%). As a result, the costs are adjusted from originally 0.75 \$/kWh to 3.27 \$/kWh. The study by Shezan et al. [121] with the high LCOEs of 2.01 \$/kWh needs further investigation, especially since an unmet load of 0.01% is considered here, which should reduce the

LCOEs. Unfortunately, by analysing the study it is not really possible to determine how the HOMER model achieves the high LCOEs for the most economic ES with PV, wind, diesel generator and stationary battery. However, surprisingly, a figure in the study shows more realistic HOMER results with LCOEs of 0.62 \$\/kWh, but these LCOEs are not discussed further in the text. In Li [82] the predefined design of the ES seems to lead to the high LCOEs of 1.54 \$\%LWh. In fact, an ES with 500 kW of PV and 9.1 MWh of stationary batteries is assumed for 100 households in China. This ES appears to be oversized, which again demonstrates the need for a good understanding of the ES when designing ESs with HOMER (cf. section 4.1). In addition, the PV-battery system is compared with a PV-battery-fuel cell system, but the capacities of PV and battery are not changed. Thus, it is obvious that the PV-battery system leads to lower costs.

The reason for the low LCOEs for energy autonomous systems of the downward outliers is more difficult to determine and would require an in-depth analysis in a separate study. In the case study in Rajanna and Saini [106], for example, there is great potential for baseload hydropower and bioenergy, which could be related to the low LCOEs of 0.07 \$/kWh. Examining further studies for reasons related to LCOE would be beyond the scope of this literature review. However, the detailed table in the supplementary Excel file could be used in further studies to investigate the dependencies of the LCOEs on the characteristics of the studies, e.g. through cluster or regression analyses. The supplementary Excel File and Fig. 3 are useful as a basis for evaluating future studies about LEA. If, for example, the LCOEs deviate as much from the average of 0.41 \$/kWh in future studies as in Li [82], the applied methods and results have to be further investigated.

Since most of the studies calculate LCOEs, these figures are very suitable for comparing the results of LEA studies. However, another cost parameter that is particularly relevant for the inhabitants who have to pay for the costs of the autonomous ES is rarely shown in the studies: total costs per inhabitant. These are only shown in Jenssen et al. [70] (1.4–2.3 k\$), Schmidt et al. [117] (220 \$/a more than in the reference scenario without autonomy) and Weinand et al. [124] (21.0–54.8 k\$). However, the cost per inhabitant should be included in all future studies in order to assess the feasibility of the EA project.

In some studies, the costs of the completely autonomous ES are compared with the costs of grid connection [33,87,97,101,111,131]. In all examples, the grid connection scenario turns out to be less economical. This is due to the fact that these studies only consider remote areas that are far away from the nearest grid connection point. The break-even points for the distance from which the network connection would be worthwhile are calculated in Refs. [87,101,111,131]. This is also related to the question of the optimal degree of centralisation as well as the optimal size of energy autonomous municipalities (cf. section 5).

5. Critical appraisal of energy autonomy studies

14

In this section, some of the key findings of section 4 are subjected to further critical evaluation. Firstly, though it is difficult to generalize across all 123 reviewed studies, some emerging trends may be highlighted:

- Mostly conventional/established technologies are analysed, with less attention paid to emerging but potentially gamechanging technologies such as deep geothermal energy and fuel cell vehicles;
- The sectoral focus is on residential, with much less consideration of industrial and transportation sectors;
- Network infrastructure is rarely considered, including electricity, gas and heat/cooling;
- Only a minority of studies account for the existing infrastructure as well as the transition from this state to some improved future state along a pathway;
- Most studies focus on complete EA, with some (12%) dealing with balanced EA.

The time resolution of most studies is hourly, with one study going into more detail at the 15-min level. For long-term planning purposes, the hourly resolution is suitable, but it should also be combined with more detailed analyses and/or information reduction techniques to generate time typologies and synthetic time slices. There is an established stream of research focussing on the most suitable/required time resolution for specific research questions, for ESs with significant renewable generation fractions (e.g. Refs. [396,397]). Studies addressing LEA could also benefit from exploring such approaches. Otherwise there is a danger that systems are incorrectly dimensioned and are inadequate to ensure supply security in times of peak demand, which for autonomous/ off-grid systems is potentially critical.

There is also a lack of attention paid to non-economic and nontechnical criteria in studies of FA at the local scale. Indeed, economic criteria are arguably the most important, but although they are necessary, they are not sufficient. In section 1, among others, it has already been highlighted that certain areas are very far from the national grid and therefore a stand-alone system is appropriate. In this case, the focus on costs as a target criterion is justified, however, the comparison of complete autonomy with the grid connection case should always be demonstrated. Otherwise, in future analyses of EA more criteria than costs need to be considered. There are other important reasons for municipalities to become energy autonomous besides costs, such as increased environmental awareness (cf. section 1). Furthermore, a comparison with the electricity price in section 4.7 shows that EA usually leads to significantly higher costs, and thus from a cost perspective there is no potential for EA.

In terms of the modelling approaches employed for highly renewable, autonomous ESs there is a clear dominance of linear programming (i.e. optimization) and simulation (i.e. dispatch rules for energy balancing), from a central planner perspective. It is encouraging to find that many researchers are also capturing the non-economic criteria such as health, noise, water and acceptance issues (cf. Table 7). However, these contributions are still in the minority of those reviewed here, and the overwhelming majority do not satisfactorily reflect the true complexity encountered in real-world energy transitions. It is common practice to leave stakeholder roles outside the scope of the studies or models and to calculate optimal autonomy transition pathways with a centralised planning approach under the selected objective of technical feasibility and economic viability. However, local ESs are complex sociotechnical systems consisting of different decision-making entities and technological artefacts governed by energy policy in a multilevel institutional space [11]. Social relationships among the stakeholders represent a major driver or barrier as also stated by previous reviews [4]. In this context, adoption behaviour approaches are useful to understand the types of barriers that exist for new technologies, and what kind of policies are important to increase diffusion. As a consequence, realising the potential of LEA is not simply a question of technical realities but also a question of individual behaviour and group dynamics. Relevant local stakeholders as households or communities, energy producers, energy suppliers, service providers, as well as local policy-makers are inter-dependent in the realisation of their goals. Future system models need to include the heterogeneous roles different stakeholders play in an existing local environment and the resulting impact their decision making might have. One possible solution could be the extension of the presented techno-economic modelling approaches with the help of socio-economic modelling approaches as agent-based models [398,399] or system dynamics models [400].

Many different spatial scales (e.g. number of households) have been considered in the case studies (cf. section 4.2.1). Whilst the optimal spatial size of an energy autonomous region has not yet been identified, it is interesting to note that the mean number of households and inhabitants in energy-autonomous regions is 340 and 18,200, respectively (based on the 56 and 49 articles containing this information, respectively). This is strongly linked to the question of the optimal degree of centralisation [401]. Concerning demand and consumption sectors there is also some potential for improvement (cf. section 4.2). The demand product water, for example, which is strongly linked to energy, or the consumption sector industry is only very rarely taken into account. For the latter, transferable methods to determine the energy demand and load profile of industries in arbitrary regions could facilitate its implementation. The transport sector is also almost completely ignored in studies about LEA. A particularly interesting approach would be to optimize the number and use of electric or fuel cell vehicles. In general, however, all sectors should be taken into account, especially when estimating the impact of one or many energy autonomous regions on the surrounding ES.

Validation is challenging in the context of local ESs, which might explain why very few existing studies attempt to do this. Often detailed data on the existing ES is lacking and validation for some hypothetical future scenario is obviously not meaningful. Model design and data assumptions of studies used to gain insights to form the decision making should be transparent and accessible. This not only allows independent review of various stakeholders but also the complete reproducibility of the results [402]. Whilst there is a strong trend towards open-source models and data within the wider ES modelling field [403], of the reviewed models, none appear to be fully open source, with HOMER and EnergyPLAN only being open access. Hence LEA studies could increase efforts to publicly release data and system models as well as assumptions and results interpretation, in order that diverse affected stakeholders are able to participate in the decision-making process [402,404].

In the future, the increasing availability of large(r) amounts of (more) open data should facilitate the implementation of datadriven methodologies. These approaches are already providing helpful insights into different aspects of energy production and demand. For example, machine learning approaches are capable of capturing non-linear and complicated relationships, and could be implemented more often [405]. In future EA case studies, such approaches could be used to detect existing renewable energy plants [406], predict renewable energy generation [407] or predict energy demands [408,409]. Data availability can therefore not only facilitate the validation of results, but also create novel insights through the application of innovative methods. Improved data quality should help to increase the accuracy of energy system models for decentralized energy systems.

In addition, the focus on the LCOEs as a benchmark for highly-RE systems could provide potentially misleading results. Whilst the LCOE is a good first indicator of the generation costs and allows comparisons across technology, it is noted for neglecting the additional costs of integrating non-dispatchable renewable technologies into the ES. Three additional cost components should be considered, if the true *system LCOEs* of RE technologies are to be considered [410,411]:

- Profiling costs, related to the requirement for the dispatchable generation technologies to meet the residual load;
- **Balancing costs**, related to the deviation between forecast and actual non-dispatchable renewable generation; and
- Grid costs, related to addition grid reinforcement and extension (at all voltage levels) required to connect renewable generators to the network.

Attempts have been made to consider these cost components in the context of large-scale (e.g. national) ES analyses (e.g. Refs. [412–414]). However, at the regional and municipal scale, as demonstrated by this review, they are typically not included. This is despite the fact that, when considering balanced autonomy, these effects on the surrounding ES are of particular importance. Balanced EA and the associated increasing feed-in by renewables could make network expansion even more essential and also make new allocation systems for grid fees necessary [401]. The result could be economic inefficiencies compared to the established system of centralised generation, transmission and distribution [415]. However, possible ES impacts were not considered in any of the 14 studies on balanced autonomy. Hence there is a need for further research to address these and the above mentioned deficits.

6. Summary and conclusions

Research attention on decentralized autonomous energy systems has increased exponentially in the past three decades, as demonstrated by the absolute number of publications and the share of these studies in the corpus of energy system modelling literature. This paper shows the status quo and future modelling needs for research on local autonomous energy systems. A total of 359 studies are roughly investigated, of which a subset of 123 in detail. The studies are assessed with respect to the characteristics of their methodology and applications, in order to derive common trends and insights.

The results show that most case studies were conducted in the middle-income countries India, Iran and China as well as the highincome country Germany. In the middle-income country studies, mostly remote rural areas without an electricity network connection are considered, whereas in high-income countries the case studies are much more diverse and also include cities and islands. In addition, most studies only focus on the residential sector and the supply of electricity. A wide range of technologies has already been covered in the literature, including less common technologies such as power-to-gas and fue cell vehicles. However, the network infrastructure is rarely considered. The levelized costs of electricity for local autonomous energy systems in 83 case studies amount to 0.41 \$/kWh on average. Thereby, studies are identified in which the resulting costs should be questioned, as they deviate strongly from the average.

In terms of the employed methodology, most of the reviewed literature reports an optimization or simulation approach, with a central planner perspective. They typically employ a time resolution of 1 h, but for some studies also increase this to 15-min resolution. Whilst it is commendable that some of the studies also consider non-economic criteria such as social and environmental aspects, neither the system-level impacts nor the diverse stakeholders are included in most works. Furthermore, there is a general lack of transparency across most reviewed literature, meaning that neither open data nor open models are widely applied to local energy systems.

Hence, future research should focus on the following methodological innovations. Other perspectives than that of a central planner and other target criteria than costs should be included. This could contribute to the realizability of the case study results. System impacts of many local autonomous energy systems have not yet been investigated, which could make new distribution systems and grid fees necessary. Complete autonomous energy systems in particular must be robustly designed, for example by analysing the value of lost load and whether a security of supply below 100% is acceptable for consumers of the case study, preferably using penalty costs for unmet load in the target function. In addition, extreme conditions such as extreme days with low solar radiation or wind should be considered and the temporal resolution should be higher than the usually used hourly resolution. Finally, methodologies should be developed which can involve local stakeholders in the modelling process and thus consider their preferences relating to their future energy system.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Acknowledgements

Fabian Scheller receives funding from the European Union's Horizon 2020 research and innovation programme under the Marie Skłodowska-Curie grant agreement no. 713683 (COFUNDfellowsDTU). Russell McKenna and Fabian Scheller kindly acknowledge the financial support of the FlexSUS Project (Project nbr. 91352), which has received funding in the framework of the joint programming initiative ERA-Net Smart Energy Systems' focus initiative Integrated, Regional Energy Systems, with support from the European Union's Horizon 2020 research and innovation programme under grant agreement No 775970. Russell McKenna also gratefully acknowledges the support of the Smart City Accelerator project. The usual disclaimer applies.

Appendix A. Supplementary data

Supplementary data to this article can be found online at https://doi.org/10.1016/j.energy.2020.117817.

References

- [1] International Energy Agency. Energy access database [October 22, 2019];
- Available from: https://www.iea.org/energyaccess/database/.
 International Energy Agency. Energy access outlook 2017: from poverty to prosperity [October 22, 2019]. Available from: https://www.iea.org/ publications/freepublications/publication/WEO2017SpecialReport_
- EnergyAccessOutlook.pdf. [3] The World Bank. Access to electricity (% of population). [October 22, 2019]; will be formy hereally the support of the second second
- Available from: https://data.worldbank.org/indicator/eg.EL.Accs.Zs.
 [4] Rae C, Bradley F. Energy autonomy in sustainable communities—a review of key issues. Renew Sustain Energy Rev 2012;16(9):6497–506. https://doi.org/ 10.1016/j.rser.2012.08.002
- [5] Engelken M, Römer B, Drescher M, Welpe I. Transforming the energy system: why municipalities strive for energy self-sufficiency. Energy Pol 2016;98:

16

- 365-77. https://doi.org/10.1016/j.enpol.2016.07.049.
 [6] Boon FP, Dieperink C Local civil society based renewable energy organisations in The Netherlands: exploring the factors that stimulate their emergence and development. Energy Pol 2014;69:297-307. https://doi.org/10.1016/j.enpol.2014.01.046.
 [7] Volz R Bedeutung und Potenziale von Energiegenossenschaften in Deutschland eine empirische Aufbereitung. Inf Raumentwickl 515-524;9(10).
 [8] Keirsteal L lennings M Sivalumar A A raview of urban energies.
- [8] Keirstead J, Jennings M, Sivakumar A. A review of urban energy system
- Reinstead J.; Jennings M., Straktinar A. A review of undar energy System models: approaches, challenges and opportunities. Renew Sustain Energy Rev 2012;16(6):3847-66. https://doi.org/10.1016/j.rser.2012.02.047.
 Mirakyan A, Guio R de. Integrated energy planning in cities and territories: a review of methods and tools. Renew Sustain Energy Rev 2013;22:289-97. https://doi.org/10.1016/j.rser.2013.01.033.
 Gamarra C, Guerrero JM. Computational optimization techniques applied to provide and a service a service a service research energy Rev 2014;2, 244.

- Gamaria C, Guerrero JM, Computational optimization techniques applied to microgrids planning: a review. Renew Sustain Energy Rev 2015;48:413-24. https://doi.org/10.1016/j.rser.2015.04.025.
 Koirala BP, Koliou E, Friege J, Hakvoort RA, Herder PM. Energetic commu-nities for community energy: a review of key issues and trends shaping integrated community energy systems. Renew Sustain Energy Rev 2016;56: 722-44. https://doi.org/10.1016/j.rser.2015.11.080.
 Scheller F, Bruckner T. Energy system optimization at the municipal level: an application of producing appropriate and challoneer. Renew Entry Review Review.
- Scheller F, Bruckher I. Energy System optimization at the municipal leVel: an analysis of modeling approaches and challenges. Renew Sustain Energy Rev 2019;105:444-61. https://doi.org/10.1016/j.rser.2019.02.005.
 Huang Z, Yu H, Peng Z, Zhao M. Methods and tools for community energy planning: a review. Renew Sustain Energy Rev 2015;42:1335-48. https://doi.org/10.1016/j.rser.2014.11.042.
 Allan G, Eromenko I, Gilmartin M, Kockar I, McGregor P. The economics of distributed approxy for generation. J Interview. Review Sustain Energy Rev 2015;42:1335-48. https://doi.org/10.1016/j.rser.2014.11.042.
- [14] Alian G., Erômenko I, Gimartin M, Kokar J, McGregor P. Ine economics of distributed energy generation: a literature review. Renew Sustain Energy Rev 2015;42:543–56. https://doi.org/10.1016/j.rser.2014.07.064.
 [15] Müller MO, Stämpfli A, Dold U, Hammer T. Energy autarky: a conceptual framework for sustainable regional development. Energy Pol 2011;39(10): 5800–10. https://doi.org/10.1016/j.enpol.2011.04.019.
 [16] Kaundinya DP, Balachandra P, Ravindranath NHL Grid-connected versus
- stand-alone energy systems for decentralized power—a review of literature. Renew Sustain Energy Rev 2009;13(8):2041–50. https://doi.org/10.1016/
- Renew Sustain Energy Rev 2009;13(8):2041–50. https://doi.org/10.1016/ j.rser.2009.02.002.
 [17] Brown TW, Bischof-Niemz T, Blok K, Breyer C, Lund H, Mathiesen BV. Response to Burden of proof: a comprehensive review of the feasibility of 100% renewable-electricity systems'. Renew Sustain Energy Rev 2018;92: 834–47. https://doi.org/10.1016/j.rser.2018.04.113.
- 834-47. https://doi.org/10.1016/j.rser2.2018.04.113.
 [18] Hansen K, Breyer C, Lund H. Status and perspectives on 100% renewable energy systems. Energy 2019;175:471-80. https://doi.org/10.1016/j.energy.2019.03.092.
 [19] Heard BP, Brook BW, Wigley TML, Bradshaw CJA. Burden of proof: a comprehensive review of the feasibility of 100% renewable-electricity systems. Renew Sustain Energy Rev 2017;76:1122-33. https://doi.org/10.1016/j.energy1201203114 er.2017.03.114
- Nakata T, Silva D, Rodionov M. Application of energy system models for designing a low-carbon society. Prog Energy Combust Sci 2011;37(4): 462–502. https://doi.org/10.1016/j.pces.2010.08.001.
 Engelken M, Römer B, Drescher M, Welpe IM, Picot A. Comparing drivers,
- barriers, and opportunities of business models for renewable energies: a review. Renew Sustain Energy Rev 2016;60:795-809. https://doi.org/ 0 1016/i rser 2015 12 163
- 10.1016/j.rser.2015.12.163.
 Falagas ME, Pitsouni EI, Malietzis GA, Pappas G. Comparison of PubMed, Scopus, Web of science, and google scholar: strengths and weaknesses. Faseb J 2008;22(2):338–42. https://doi.org/10.1096/fj.07-9492LSF.
- [23] Chappin EIL, Ligtvoet A. Transition and transformation: a bibliometric anal-
- 5567036.2019.1656308
- [25] Abnavi MD, Mohammadshafie N, Rosen MA, Dabbaghian A, Fazelpour F. Techno-economic feasibility analysis of stand-alone hybrid wind/photvol-tai/disel/battery system for the electrification of remote rural areas: case study Persian Gulf Coast-Iran. Environ Prog Sustain Energy 2019;38(5): 13172. https://doi.org/10.1002/ep.13172
- [26] Adaramola MS, Agelin-Chaab M, Paul SS. Analysis of hybrid energy systems
- Adaramola MS, Agelin-Chaab M, Paul SS, Analysis of hybrid energy systems for application in southerm Ghana. Energy Convers Manag 2014;88:284–95. https://doi.org/10.1016/j.enconman.2014.08.029. Adaramola MS, Oyewola OM, Paul SS. Technical and economic assessment of hybrid energy systems in south-west Nigeria. Energy Explor Exploit 2012;30(4):533–52.
- [28] Adaramola MS, Quansah DA, Agelin-Chaab M, Paul SS, Multipurpose [26] Hudrahnoba Hos, Qualitatio De, Reenfechaar Mr. 1991. Status St. Multipurpose renewable energy resources based hybrid energy system for remote com-munity in northern Ghana. Sustainable Energy Technologies and Assess-ments 2017;22:161–70. https://doi.org/10.1016/j.seta.2017.02.011.
 [29] Afzal A, Mohibullah M, Kumar Sharma V. Optimal hybrid renewable energy
- systems for energy security: a comparative study. Int J Sustain Energy 2010;29(1):48-58. https://doi.org/10.1080/14786460903337241.
- [30] Alhamwi A. Mediroubi W. Vogt T. Agert C. Development of a GIS-based

platform for the allocation and optimisation of distributed storage in urban energy systems. Appl Energy 2019;251:113360. https://doi.org/10.1016/ j.apenergy.2019.113360. [31] Al-Shetwi AQ, Sujod MZ, Al Tarabsheh A, Altawil IA. Design and economic

- Al-ShetWi AQ, Sujod MZ, AI Tarabsheh A, Altawil IA. Design and economic evaluation of electrification of small villages in rural area in Yemen using stand-alone PV system. Int J Renew Energy Resour 2016;6(1):289–98. Arefin SS, Das N. Optimized hybrid wind-diesel energy system with feasi-bility analysis. Technol Econ Smart Grids Sustain Energy 2017;2(1):63. https://doi.org/10.1007/s40866-017-0025-6.
- Ashok S. Optimised model for community-based hybrid energy system [33] Renew Energy renene.2006.04.008. 2007;32(7):1155-64. https://doi.org/10.1016/ Renew
- j.renene.2006.04.008. Askari IB, Ameri M. Optimal sizing of photovoltaic—battery power systems in a remote region in Kerman, Iran. Proc IME J Power Energy 2009;223(5): 563-70. https://doi.org/10.1243/09576509JPE717. Askari IB, Ameri M. Techno-economic feasibility analysis of stand-alone renewable energy systems (PV/bat, wind/bat and hybrid PV/wind/bat) in kerman, Iran. Energy Sources B Energy Econ Plann 2012;7(1):45-60. https:// doi.org/10.1080/15567240903330384. [35]
- [36] Askarzadeh A, dos Santos Coelho L. A novel framework for optimization of a grid independent hybrid renewable energy system: a case study of Iran. Sol Energy 2015; 112:383-96. https://doi.org/10.1016/j.solener.2014.12.013.
- Energy 2015; 112:33 95. https://doi.org/10.1016/j.solener.2014.12.013.
 [37] Azaza M, Wallin F. Multi objective particle swarm optimization of hybrid micro-grid system: a case study in Sweden. Energy 2017;123:108-18. https://doi.org/10.1016/j.energy.2017.01.149.
 [38] Bagheri M, Delbari SH, Pakzadmanesh M, Kennedy CA. City-integrated renewable energy design for low-carbon and climate-resilient communities. Appl Energy 2019;239:1212-25. https://doi.org/10.1016/j.japenergy.2019.02.031.
 [39] Denergy J, Shirard M, Bredue E, Konzelu GA optimel classing of hybrid
- [39] Bagheri M, Shirzadi N, Bazdar E, Kennedy CA. Optimal planning of hybrid renewable energy infrastructure for urban sustainability: green Vancouver. Renew Sustain Energy Rev 2018;95:254–64. https://doi.org/10.1016/ isupr2010.02002
- [40] Balaji V, Gurgenci H. Search for optimum renewable mix for Australian off-grid power generation. Energy 2019;175:1234–45. https://doi.org/10.1016/ j.energy.2019.03.089.
- Jenergy.2019.03.089.
 [41] Baneshi M, Hadianfard F. Techno-economic feasibility of hybrid diesel/PV/ wind/battery electricity generation systems for non-residential large elec-tricity consumers under southern Iran climate conditions. Energy Convers Manag 2016;127:233-44. https://doi.org/10.1016/j.encoman.2016.09.008.
 [42] Baseer MA, Alqahtani A, Rehman S, Techno-economic design and evaluation
- Dasce Mr., Ingenianty, Refinanty, Refine Construction (using a network of hybrid energy systems for residential communities: case study of jubail industrial city. J Clean Prod 2019;237:117806. https://doi.org/10.1016/ ii.dorsr.2019.117806 [43] Bentouba S, Bourouis M. Feasibility study of a wind-photovoltaic hybrid
- power generation system for a remote area in the extreme south of Algeria. Appl Therm Eng 2016;99:713–9. https://doi.org/10.1016/ japplthermalerg.2015.12.014. Bonati A, Luca G de, Fabozzi S, Massarotti N, Vanoli L. The integration of exergy criterion in energy planning analysis for 100% renewable system. Energy 2019;174:749–67. https://doi.org/10.1016/j.energy.2019.02.089. Castellanos JG, Walker M, Poggio D, Pourkashanian M, Nimmo W. Modelling
- [44]
- [45] [45] Castenanos Jo, Warker M, Poggio D, Pourkashaman M, Nimino W. Modelming an off-grid integrated renewable energy system for rural electrification in India using photovoltaics and anaerobic digestion. Renew Energy 2015;74: 390-8. https://doi.org/10.1016/j.trenene.2014.08.055.
 [46] Chauhan A, Saini RP. Techno-economic optimization based approach for energy management of a stand-alone integrated renewable energy system for remote areas of India. Energy 2016;94:138-56. https://doi.org/10.1016/ isoncerr.2015.10.126.
- energy.2015.10.13
- [47] Chauhan A, Saini RP. Size optimization and demand response of a stand-alone integrated renewable energy system. Energy 2017;124:59-73. https://doi.org/10.1016/j.energy.2017.02.049.
 [48] Chmiel Z, Bhattacharyya SC. Analysis of off-grid electricity system at isle of eigg (scotland): lessons for developing countries. Renew Energy 2015;81: 578-88. https://doi.org/10.1016/j.renene.2015.03.061.
 [49] Clark W, Isherwood W. Distributed generation: remote power systems with dynamic d terrers techeologing. Energy B04(2314):1572-89. https://
- [49] Clark W, Michael W, Distributed generation: reinforce power systems with advanced storage technologies. Energy Pol 2004;32(14):1573–89. https:// doi.org/10.1016/S0301-4215(03)00017-X.
 [50] Das BK, Hoque N, Mandal S, Pal TK, Raihan MA. A techno-economic feasibility
- of a stand-alone hybrid power generation for remote area application in Bangladesh. Energy 2017;134:775–88. https://doi.org/10.1016/ Bangladesh. Ene j.energy.2017.06.024
- Dorotic H, Doračić B, Dobravec V, Pukšec T, Krajačić G, Duić N. Integration of transport and energy sectors in island communities with 100% intermittent renewable energy sources. Renew Sustain Energy Rev 2019;99:109–24. https://doi.org/10.1016/j.rser.2018.09.033.
 Drysdale D, Vad Mathiesen B, Lund H, From carbon calculators to energy
- system Analysis in cities. Energies 2019;12(12):2307. https://doi.org 10.3390/en12122307
- [53] DUTSUIR B, Gokcol C, Umut I, Ucar E, Kocabey S. Techno-economic evaluation of a hybrid PV—wind power generation system. Int J Green Energy 2013;10(2):117–36. https://doi.org/10.1080/15435075.2011.641192.
- supply systems for low-income communities involving multiple-objectives. Energy Convers Manag 2017;151:43—52. https://doi.org/10.1016/

- conman.2017.08.050 [55] Fulhu M, Mohamed M, Krumdieck S. Voluntary demand participation (VDP) for security of essential energy activities in remote communities with case study in Maldives. Energy for Sustainable Development 2019;49:27–38. https://doi.org/10.1016/j.esd.2019.01.002.
- [56] Fuso Nerini F, Dargaville R, Hovells M, Bazilian M. Estimating the cost of energy access: the case of the village of Suro Crait in Timor Leste. Energy 2015;79:385–97. https://doi.org/10.1016/j.energy.2014.11.025.
 [57] Ganguly P, Kalam A, Zayegh A. Fuzzy logic-based energy management sys-
- tem of stand-alone renewable energy system for a remote area power system. Aust J Electr Electron Eng 2019;16(1):21–32. https://doi.org/10.1080/
- 2019.1588091 Gupta A, Saini RP, Sharma MP. Steady-state modelling of hybrid energy system for off grid electrification of cluster of villages. Renew Energy 2010;35(2):520–35. https://doi.org/10.1016/j.renene.2009.06.014. Gupta A, Saini RP, Sharma MP. Modelling of hybrid energy system—Part III: [58]
- [59]
- Gupta A, Saini KP, Sharma MP. Modelling of hybrid energy system—Part III: case study with simulation results. Renew Energy 2011;36(2):474–81. https://doi.org/10.1016/j.renene.2009.04.036. Haghighat Mamaghani A, Avella Escandon SA, Najafi B, Shirazi A, Rinaldi F. Techno-economic feasibility of photovoltaic, wind, diesel and hybrid elec-trification systems for off-grid rural electrification in Colombia. Renew En-ure 2010;2020, 202. https://doi.org/10.1026/0006 ergy 2016:97:293-305. https://doi.org/10.1016/j.renene.2016.05.086
- [61] Hakimi SM, Moghaddas-Tafreshi SM, HassanzadehFard H. Optimal sizing of reliable hybrid renewable energy system considered various load types. J Renew Sustain Energy 2011;3(6):62701. https://doi.org/10.1063/ doi:10.1016/j.
- [62] Halabi LM, Mekhilef S. Flexible hybrid renewable energy system design for a typical remote village located in tropical climate. J Clean Prod 2018;177: 908—24. https://doi.org/10.1016/j.jclepro.2017.12.248.
- Hod 24. https://doi.org/10.1010/j.telp10.2017.12.246.
 Heydari A, Askarzadeh A. Techno-economic analysis of a PV/biomass/fuel cell energy system considering different fuel cell system initial capital costs. Sol Energy 2016;133:409 20. https://doi.org/10.1016/j.solener.2016.04.018.
 Ho WS, Hashim H, Hassim MH, Muis ZA, Shamsuddin NLM, Design of [63]
- [64] Ho distributed energy system through electric system cascade analysis (ESCA). Appl Energy 2012;99:309–15. https://doi.org/10.1016/ ppl Energy apenergy.2012.04.016.
- Japenegy 2012 USAND, Hosseini SJ, Moazzami M, Shahinzadeh H. Optimal sizing of an isolated hybrid wind/PV/battery system with considering loss of power supply probability. Majlesi Journal of Electrical Engineering 2017;11(3):63–9. [65]
- [66]
- [67]
- Bekele G, Palm B. Feasibility study for a standalone solar—wind-based hybrid energy system for application in Ethiopia. Appl Energy 2010;87(2):487–95. https://doi.org/10.1016/j.apenergy.2009.06.006.
 Iaquaniello G, Montanari W, Salladini A. Standalone CSP-DG system for electrification of remote areas and desalinated water supply. Sol Energy 2017;157:1056–63. https://doi.org/10.1016/j.solener.2017.09.026.
 Jamshidi M, Askarzadeh A. Techno-economic analysis and size optimization of an off-grid hybrid photovoltaic, fuel cell and diesel generator system. Sustainable Cities and Society 2019;44:310–20. https://doi.org/10.1016/ j.scs.2018.10.021. [68] 2018 10 021
- Jaced MS, Song A, Ma T. Techno-economic assessment of a stand-alone hybrid solar-wind-battery system for a remote island using genetic algo-rithm. Energy 2019;176:704–17. https://doi.org/10.1016/ [69] thm. Energy energy.2019.03.131.
- Jenergy.2019.03.131.
 [70] Jenssen T, König A, Eltrop L. Bioenergy villages in Germany: bringing a low carbon energy supply for rural areas into practice. Renew Energy 2014;61: 74–80. https://doi.org/10.1016/j.renene.2012.08.014.
 [71] Kamran M, Asghar R, Mudasars M, Ahmed SR, Fazal MR, Abid MI, et al. Designing and optimization of stand-alone hybrid renewable energy system
- for rural areas of Punjab, Pakistan. Int J Renew Energy Resour 2018;8(4):
- [72] Kanase-Patil AB, Saini RP, Sharma MP. Integrated renewable energy systems Kanase-rati Ao, Sain Ar, Shain AY, Shaina MF, Integrated relevable energy systems for off grid rural electrification of remote area. Renew Energy 2010;35(6): 1342–9. https://doi.org/10.1016/j.renene.2009.10.005. Kanase-Patil AB, Saini RP, Sharma MP, Development of REOM model based
- [73] on seasonally varying load profile for hilly remote areas of Uttarakhand state India. Energy 2011;36(9):5690–702. https://doi.org/10.1016/ ergy.2011.06.057. ìn
- Kandil MS, Farghal SA, El-Alfy AE. Optimum operation of an autonomou energy system suitable for new communities in developing countries. Ele [74]
- [75] Khan AN, Akhter P, Mufti GM. Techno-economic evaluation of the central-[75] Khan AW, Akhter P, Multi GM. Techno-economic evaluation of the central-ized hybrid renewable energy systems for off-grid rural electrification. In-ternational Journal of Smart Home 2016;10(5):61-8. https://doi.org/ 10.14257/ijsh.2016.10.5.07.
 [76] Kim H, Baek S, Choi K, Kim D, Lee S, Kim D, et al. Comparative analysis of on-and off-grid electrification: the case of two south Korean islands. Sustain-ability 2016;8(4):350. https://doi.org/10.3390/su8040350.
 [77] Kobayakawa T, Kandpal TC. Optimal resource integration in a decentralized consultation.
- energy system: assessment of the existing system and simulation for its expansion. Energy for Sustainable Development 2016;34:20–9. https://doi.org/10.1016/j.esd.2016.06.006.
- https://doi.org/10.1016/j.esd.2016.06.006. [78] Kötter E, Schneider L, Sehnke F, Ohnmeiss K, Schröer R. The future electric power system: impact of Power-to-Gas by interacting with other renewable energy components. Journal of Energy Storage 2016;5:113–9. https:// doi.org/10.1016/j.est.2015.11.012.

- [79] Krajačić G, Duić N, Carvalho MdG. H2RES, Energy planning tool for island energy systems – the case of the Island of Mijet*. Int J Hydrogen Energy 2009;34(16):7015–26. https://doi.org/10.1016/j.ijhydene.2008.12.054. Kumar A, Singh AR, Deng Y, He X, Kumar P, Bansal RC. Integrated assessment
- [80] Kumar A, Singu AK, Deng T, Per A, Kumar T, Jansa KK, Integrated assessment of a sustainable microgrid for a remote village in hilly region. Energy Convers Manag 2019;180:442–72. https://doi.org/10.1016/j.enconman.2018.10.084. Kumaravel S, Ashok S. An optimal stand-alone biomass/solar-PV/piico-hydel hybrid energy system for remote rural area electrification of isolated village
- in western-ghats region of India. Int J Green Energy 2012;9(5):398-408. https://doi.org/10.1080/15435075.2011.621487.
- https://doi.org/10.1080/15436/5.2011.b2148/.
 [82] Li C. Techno-economic study of off-grid hybrid photovoltaic/battery and photovoltaic/battery/fuel cell power systems in Kunming, China. Energy Sources, Part A: recovery, Utilization, and Environmental Effects 2018;4(13):1588–604. https://doi.org/10.1080/15567036.2018.1549134.
 [83] Lu J, Wang W, Zhang Y, Cheng S. Multi-objective optimal design of stand-
- Lu J, Wang W, Zhang Y, Cheng S. Multi-objective optimal design of stand-alone hybrid energy system using entropy weight method based on HOM-ER. Energies 2017;10(10):1664. https://doi.org/10.3390/en10101664. Ma T, Javed MS. Integrated sizing of hybrid PV-wind-battery system for remote island considering the saturation of each renewable energy resource. Energy Convers Manag 2019;182:178–90. https://doi.org/10.1016/ incomment 2019.130670. [84] Convers Mar. man.2018.12.059.
- [85] Ma T. Yang H. Lu L. Peng J. Technical feasibility study on a standalone hybrid solar-wind system with pumped hydro storage for a remote island in Hong Kong. Renew Energy 2014;69:7–15. https://doi.org/10.1016/ j.renene.2014.03.028.
- Ma T, Yang H, Lu L Study on stand-alone power supply options for an iso-lated community. Int J Electr Power Energy Syst 2015;65:1–11. https:// [86] doi.org/10.1016/j.ijepes.2014.09.023.
 [87] Mahapatra S, Dasappa S. Rural electrification: optimising the choice between
- decentralised renewable energy sources and grid extension. Energy for Sustainable Development 2012;16(2):146–54. https://doi.org/10.1016/
- [88] Maleki A, Askarzadeh A. Optimal sizing of a PV/wind/diesel system with battery storage for electrification to an off-grid remote region: a case study of Rafsanjan, Iran. Sustainable Energy Technologies and Assessments 2014;7: 147-53. https://doi.org/10.1016/j.seta.2014.04.005.
- Har-Job, https://doi.org/10.1016/j.sea.2014.04.005.
 Maleki A, Pourfayaz F, Ahmadi MH. Design of a cost-effective wind/photo-voltaic/hydrogen energy system for supplying a desalination unit by a heuristic approach. Sol Energy 2016;139:666–75. https://doi.org/10.1016/ er 2016/09/028
- [90] Maleki A. Pourfavaz F. Hafeznia H. Rosen MA. A novel framework for optimal hotovoltaic size and location in remote areas using a hybrid method: a case study of eastern Iran. Energy Convers Manag 2017;153:129–43. https:// doi.org/10.1016/j.enconman.2017.09.061.
- doi.org/10.1016/j.enconman.2017.09.061. Mandal S, Das BK, Hoque N. Optimum sizing of a stand-alone hybrid energy system for rural electrification in Bangladesh. J Clean Prod 2018;200:12–27. [91]
- system for rural electrification in Bangladesh. J Clean Prod 2018;200:12–27. https://doi.org/10.1016/j.jclepr.2018.07.257.
 [92] McKenna R, Bertsch V, Mainzer K, Fichtner W. Combining local preferences with multi-criteria decision analysis and linear optimization to develop feasible energy concepts in small communities. Eur J Oper Res 2018;268(3): 1092–110. https://doi.org/10.1016/j.ejor.2018.01.036.
 [93] Mishra R, Singh S. Sustainable energy plan for a village in Punjab for self energy generation. Int J Renew Energy 2013;3(3):640–6.
 [94] Moeller C, Meiss J, Mueller B, Hlusiak M, Breyer C, Kastner M, et al. Trans-
- forming the electricity generation of the Berlin–Brandenburg region, Ger-many. Renew Energy 2014;72:39–50. https://doi.org/10.1016/ irenene.2014.06.042. e.201-
- Muh E, Tabet F. Comparative analysis of hybrid renewable energy systems [95] for off-grid applications in Southern Cameroons. Renew Energy 2019;135: 41-54. https://doi.org/10.1016/j.renene.2018.11.105.
- 41–54. https://doi.org/10.1016/j.renene.2018.11.105.
 [96] Nigussie T. Bogale W. Bekele F, Dribssa E. Feasibility study for power generation using off- grid energy system from micro hydro-PV-diesel generator-battery for rural area of Ethiopia: the case of Melkey Hera village, Western Ethiopia. AIMS Energy 2017;5(4):667–90. https://doi.org/10.3934/energy.2017.4.667.
- [97]
- energy.2017.4.667. Nour M, Rohani G. Prospect of stand-alone PV-diesel hybrid power system for rural electrification in UAE. Int J Renew Energy Resour 2014;4(3):749–58. Ocon JD, Cruz SMM, Castro MT, Aviso KB, Tan RR, Promentilla MAB. Optimal multi-criteria selection of hybrid energy systems for off-grid electrification. Chamical Engineering Transactions 2018:70:367–72.
- Chemical Engineering Transactions 2018;70:367–72. [99] Oldenbroek V, Verhoef LA, van Wijk AJM. Fuel cell electric vehicle as a power [99] Oldenbroek V, Verhoel LA, Van Wijk Ajik. Fuel cell electric Venicle as a power plant: fully renewable integrated transport and energy system design and analysis for smart city areas. Int J Hydrogen Energy 2017;42(12):8166-96. https://doi.org/10.1016/j.ijhydene.2017.01.155.
 [100] Østergaard PA, Lund H. A renewable energy system in Frederikshavn using low-temperature geothermal energy for district heating. Appl Energy 2011;88(2):470-87. https://doi.org/10.1016/j.jenergy.2010.03.018.
 [101] Patel AM, Singal SK. Economic analysis of integrated renewable energy united for the state of the state of the state of the state of energiable energy.
- stem for electrification of remote rural area having scattered population
- System for electrication or femole valuat area naving scattered population. Int J Renew Energy Resour 2018;8(1):524–39.
 Petrakopoulou F, Robinson A, Loizidou M. Exergetic analysis and dynamic simulation of a solar-wind power plant with electricity storage and hydrogen generation. J Clean Prod 2016;113:450–8. https://doi.org/10.1016/
- [103] Petrakonoulou F Robinson A Loizidou M Simulation and analysis of a stand

alone solar-wind and pumped-storage hydropower plant. Energy 2016;96

- 676-83. https://doi.org/10.1016/j.energy.2015.12.049.
 [104] Peura P, Haapanen A, Reini K, Törmä H. Regional impacts of sustainable energy in western Finland. J Clean Prod 2018;187:85-97. https://doi.org/10.1016/j.jclepro.2018.03.194.
- 10.1016/j.jclepro.2018.03.194.
 Rahman MM, Khan MM-U-H, Ullah MA, Zhang X, Kumar A. A hybrid renewable energy system for a North American off-grid community. Energy 2016;97:151-60. https://doi.org/10.1016/j.energy.2015.12.105.
 Rajanna S, Saini RP. Modeling of integrated renewable energy system for
- [100] Rajama S, Saim KP. Modeling of megrated renewable energy system to electrification of a remote area in India. Renew Energy 2016;90:175–87. https://doi.org/10.1016/j.renene.2015.12.067.
 [107] Ramchandran N, Pai R, Parihar AKS. Feasibility assessment of Anchor-Business-Community model for off-grid rural electrification in India. Renew Energy 2016;97:197–209. https://doi.org/10.1016/ j.renene.2016.05.036.
 [109] Rank MAM, Unoder A, Brouchartze UBPLI, Parformene applicit of hybrid
- [108] Ramli MAM, Hiendro A, Bouchekara HREH. Performance analysis of hybrid PV/diesel energy system in western region of Saudi arabia. Int J Photoenergy 2014;2014(12):1–10. https://doi.org/10.1155/2014(56251.
 Ray A, Jana K, De S. Polygeneration for an off-grid Indian village: optimization by economic and reliability analysis. Appl Therm Eng 2017;116:182–96.
- https://doi.org/10.1016/j.applthermaleng.2016.11.020. [110] Rehman SU, Rehman S, Qazi MU, Shoaib M, Lashin A. Feasibility study of hybrid energy system for off-grid rural electrification in southern Pakistan. Energy Explor Exploit 2016;34(3):468-82. https://doi.org/10.1177/ 0144598716630176. hybr. Energy E
- 0144598716630176. [111] Rohani G, Nour M. Techno-economical analysis of stand-alone hybrid renewable power system for Ras Musherib in United Arab Emirates. Energy 2014;64:828-41. https://doi.org/10.1016/j.energy.2013.10065. [12] Sameti M, Haghighat F. Integration of distributed energy storage into net-
- Samen M, Hagnigha F. Integration of distributed energy storage into net-zero energy district systems: optimum design and operation. Energy 2018;153:575-91. https://doi.org/10.1016/j.energy.2018.04.064. Samy MM, Barakat S, Ramadan HS. A flower pollination optimization algo-rithm for an off-grid PV-Fuel cell hybrid renewable system. Int J Hydrogen Energy 2019;42(4):2141-52. https://doi.org/10.1016/ j.jijhydene.2018.05.127.
- Junyache. 2016.05.127.
 Standard M. S. St
- [115] Sanajaooa singin 5, Pernantez E. Modeling, Size optimization and sensitivity analysis of a remove hybrid renewable energy system. Energy 2018;143: 719–31. https://doi.org/10.1016/j.energy.2017.11.053.
 [116] Šare A, Krajačić G, Pukšec T, Duić N. The integration of renewable energy sources and electric vehicles into the power system of the Dubrovnik region. Energ Sustain Soc 2015;5(1):17. https://doi.org/10.1186/s13705-015-0055-
- [117] Schmidt J, Schönhart M, Biberacher M, Guggenberger T, Hausl S, Kalt G, et al. Regional energy autarky: potentials, costs and consequences for an Austrian region. Energy Pol 2012;47:211–21. https://doi.org/10.1016/ jenpol.2012.04.059. region.
- j.enpol.2012.04.059. Sepulveda TT, Martinez L Optimization of a hybrid energy system for an isolated community in Brazil. Int J Renew Energy Resour 2016;6(4): [118] 476-81.
- 14/6-81.
 [119] Shezan SA. Optimization and assessment of an off-grid photo-voltaic-diesel-battery hybrid sustainable energy system for remote residential applications. Environ Prog Sustainable Energy 2019 2017;32:100. https://doi.org/10.1002/ep.13340.
 [120] Shezan SKA, Al-Mamoon A, Ping HW. Performance investigation of an advanced hybrid renewable energy system in Indonesia. Environ Prog Sustainable energy 2019 2012;24:104.

- advanced hybrid renewable energy system in Indonesia. Environ Prog Sustain Energy 2018;37(4):1242–32. https://doi.org/10.1002/ep.12790.
 Shezan SA, Julai S, Kibria MA, Ullah KR, Saidur R, Chong WT, et al. Performance analysis of an off-grid wind-PV (photovoltaic)-disel-battery hybrid energy system feasible for remote areas. J Clean Prod 2016;125:121–32. https://doi.org/10.1016/j.jclepro.2016.03.014.
 Stephen JD, Mabee WE, Pribowo A, Pledger S, Hart R, Tallio S, et al. Biomass for residential and commercial heating in a remote Canadian aborginal community. Renew Energy 2016;86:563–75. https://doi.org/10.1016/j.jrenene.2015.08.048.
- 1JETERE 2013-06-043. [123] Waenn A, Connolly D, Ó Gallachóir B. Investigating 100% renewable energy supply at regional level using scenario analysis. International Journal of Sustainable Energy Planning and Management 2014;3:21–32. https:// doi.org/10.5278/IJSEPM.2014.3.3.
- [124] Weinand JM, McKenna R, Kleinebrahm M, Mainzer K. Assessing the contri-[125] Vechania ya, interna a, technolamin a, wanaki K. Pascasing and contribution of simultaneous heat and power generation from geothermal plants in off-grid municipalities. Appl Energy 2019;255:113824. https://doi.org/ 10.1016/j.apenergy.2019.113824.
 [125] Yazdanpanah Jahromi MA, Farahat S, Barakati SM. Optimal size and cost
- analysis of stand-alone hybrid wind/photovoltaic power-generation sys-tems. Civ Eng Environ Syst 2013;31(4):283–303. https://doi.org/10.1080/ 0286608.2013.853752
- [126] Vazdanah-Jahromi MA, Barakati S-M, Farahat S. An efficient sizing method with suitable energy management strategy for hybrid renewable energy systems. Int. Trans. Electr. Energ. Syst. 2014;24(10):1473–92. https://
- [127] Yimen N, Hamandjoda O, Meva'a L, Ndzana B, Nganhou J. Analyzing of a photovoltaic/wind/biogas/pumped-hydro off-grid hybrid system for rural

electrification in sub-saharan africa—case study of djoundé in northern Cameroon. Energies 2018;11(10):2644. https://doi.org/10.3390/ Cameroon. en11102644.

- [128] Zhang W, Maleki A, Rosen MA, Liu J. Optimization with a simulated Zhang W, Maleki A, Rosen MA, Liu J. Optimization with a simulated annealing algorithm of a hybrid system for renewable energy including battery and hydrogen storage. Energy 2018;163:191–207. https://doi.org/ 10.1016/j.energy.2018.08.112.
 Zhang W, Maleki A, Rosen MA, Liu J. Sizing a stand-alone solar-wind-hydrogen energy system using weather forecasting and a hybrid search optimization algorithm. Energy Conversion and Management 2019;180: 609–21. https://doi.org/10.1016/j.encomman.2018.08.102.
- [129]
- Dos 21. https://doi.org/10.1016/j.encomman.2018.08.102. Zhao G, Guerrero JM, Jiang K, Chen S. Energy modelling towards low carbon development of Beijing in 2030. Energy 2017;121:107–13. https://doi.org/ 10.1016/j.energy.2017.01.019. Lal DK, Dash BB, Akella AK. Optimization of PV/Wind/Micro-Hydro/diesel [130]
- [131]
- Lat DK, Dash BB, AReita AK. Optimization of PV/Wind/Micro-Hydro/dieset hybrid power system in homer for the study area. International Journal on Electrical Engineering and Informatics 2011;3(3):307-25. Galvez GH, Probst O, Lastres O, Rodríguez AN, Ugás AJ, Durán EA, et al. Optimization of autonomous hybrid systems with hydrogen storage: life cycle assessment. International Journal of Energy Research 2012;36(6): 270-67. https://doi.org/10.1000/jer.10200 [132] 49-63. https://doi.org/10.1002/er.1830.
- [133] Kumar R. Gupta RA. Bansal AK. Economic analysis and power management of Kumar K, Gupta KA, Bansal AK. Economic analysis and power management of a stand-alone wind/photovoltaic hybrid energy system using biogeography based optimization algorithm. Swarm and Evolutionary Computation 2013;8:33–43. https://doi.org/10.1016/j.swevo.2012.08.002. Ma T, Yang H, Lu L A feasibility study of a stand-alone hybrid solar-wind-battery system for a remote island. Applied Energy 2014;121:149–58. https://doi.org/10.1016/j.apenergy.2014.01.090. Baghdadi F, Mohammedi K, Diaf S, Behar O. Feasibility study and energy purposition of a remote island.
- [134]
- [135] Bagnadai F, Mohammedi K, Diar S, Benar O. Feasibility study and energy conversion analysis of stand-alone hybrid renewable energy system. Energy Conversion and Management 2015;105:471–9. https://doi.org/10.1016/ j.encomman.2015.07.051.
- [136] ternational Journal of Renewable Energy Research 2015;5(4):1220-9.
- ternational Journal of Renewable Energy Research 2015;5(4):1220-9.
 [137] Mas'Ud AA. The application of homer optimization software to investigate the prospects of hybrid renewable energy system in rural communities of sokoto in Nigeria. International Journal of Electrical and Computer Engi-neering 2017;7(2):596-603. https://doi.org/10.11591/jiceev7i2.pp596-603.
 [138] Aziz AS, bin Tajuddin MFN, bin Adzman MR, Ramli MAM. Feasibility analysis of FV/diesel/battery hybrid energy system using multi-year module. Inter-national Journal of Renewable Energy Research 2018;8(4):1980-93.
 [139] Bashir N, Modu B. Techno-economic analysis of off-grid renewable energy systems for rural electrification in Northeastern Nigeria. International Jour-nal of Renewable Energy Research 2018;8(3):1217-28.
 [140] Khan MA, Zeb K, Sathishkumar P, Himanshu L, Sinivasa Rao S, Gopi CVVM, et al. A novel off-grid optimal hybrid energy system for rural electrification

- Khan MA, Zeb K, Salinshkumar P, Himanshui L, ShinVasa Kao S, Gopi CVVM, et al. A novel off-grid optimal hybrid nergy system for rural electrification of Tanzania using a closed loop cooled solar system. Energies 2018;11(4). https://doi.org/10.3390/en11040905.
 Akhtari MR, Baneshi M. Techno-economic assessment and optimization of a hybrid renewable co-supply of electricity, heat and hydrogen system to enhance performance by recovering excess electricity for a large energy consumer. Energy Conversion and Management 2019;188:131–41. https:// doi.org/10.1015/j.necomen.2010.0167 [141] doi.org/10.1016/i.enconman.2019.03.067
- Awan AB. Performance analysis and optimization of a hybrid renewable energy system for sustainable NEOM city in Saudi Arabia. Journal of Renewable and Sustainable Energy 2019;11(2). https://doi.org/10.1063/ [142]
- [143] Farzaneh H. Design of a hybrid renewable energy system based on super-
- Farzanen H. Design of a hybrid renewable energy system based on super-critical water gasification of biomass for off-grid power supply in Fukushima. Energies 2019; 12(14). https://doi.org/10.3390/en12142709. Gambino V. Citto RD, Cherubini P, Tacconelli C, Micangeli A, Giglioli R. Methodology for the energy need assessment to effectively design and deploy mini-grids for rural electrification. Energies 2019; 12(3). https:// doi.org/10.3390/en12030574.
- doi.org/10.399/en1203074.
 [145] Kalthanthrige R, Rajapakse AD, Lamothe C, Mosallat F. Optimal sizing and performance evaluation of a hybrid renewable energy system for an off-grid power system in northern Canada. Technology and Economics of Smart Grids and Sustainable Energy 2019;4(1). https://doi.org/10.1007/s40866-
- Kannengießer T, Hoffmann M, Kotzur L, Stenzel P, Schuetz F, Peters K, et al. [146]
- Kannengeeser I, Hoimann M, Kotzur L, Stenzel P, Schuetz F, Peters K, et al. Reducing computational load for mixed integer linear programming: an example for a district and an island energy system. Energies 2019;12(14). https://doi.org/10.3390/en12142825. Muselli M, Notton G, Poggi P, Louche A. Computer-aided analysis of the integration of renewable-energy systems in remote areas using a geographical-information system. Applied Energy 1999;63(3):141–60. https://doi.org/10.1016/S0306-2619(99)00027-6.
- https://doi.org/10.1019/30300-0019(39)0009-00.
 Kaldellis JK. Parametrical investigation of the wind-hydro electricity pro-duction solution for Aegean Archipelago. Energy Conversion and Manage-ment 2002;43(16):2097–113. https://doi.org/10.1016/S0196-8904(01) [148]
- [149] Nystedt Å, Shemeikka J, Klobut K. Case analyses of heat trading between buildings connected by a district heating network. Energy Conversion and Management 2006;47(20):3652–8. https://doi.org/10.1016/ Management
- enconman.2006.02.030. [150] Liu H-Y, Wu S-D. An assessment on the planning and construction of an
- Lu H-7, wu S-D. An assessment on the planning and construction of an island renewable energy system a case study of Kimmen Island. Renewable Energy 2010;35(12):2723-31. https://doi.org/10.1016/j.renene.2010.04.021. Brandoni C, Polonara F. Technical and economic aspects of municipal energy planning. International Journal of Sustainable Development and Planning 2012;7(2):221-36. https://doi.org/10.2495/SDP-V7-N2-221-236. Prezz-Romero M, Lozano-Tello A, Romero-Cadaval E, Martins J. Community and residential energy storage in smart grids. JEIP Advances in Information and Communic stion. Zebendergy 2012;30:4315-23. [151]
- [152]
- and residential energy storage in small glub. In: Advances in information and Communication Technology 2013;394:315-22.
 [153] Vick BD, Moss TA. Adding concentrated solar power plants to wind farms to achieve a good utility electrical load match. Solar Energy 2013;92:298-312. https://doi.org/10.1016/j.solener.2013.03.007.
 [154] Raghav Chakravarthy ES, Bhargavi A, Parkavi Kathirvelu K, Balasubramanian R. Analysis and simulation of isolated wind dises! hybrid resture witten ABNI lowered of Engineering and Apriled Sciences 2014/07/1
- power system. ARPN Journal of Engineering and Applied Sciences 2014;9(7) 056-63
- [155] Moret S, Peduzzi E, Gerber L, Maréchal F. Integration of deep geothermal energy and woody biomass conversion pathways in urban systems. Energy Conversion and Management 2016;129:305–18. https://doi.org/10.1016/ particle.com/partic 2016.09.079
- [156] Morvai B. Evins R. Carmeliet I. Decarbonizing the electricity grid: the impact [156] MorVaj B, Evins K, Carmeier J. Decaroonizing the electricity grin: the impact on urban energy systems, distribution grids and district heating potential. Applied Energy 2017;191:125-40. https://doi.org/10.1016/ j.apenergy.2017.01.058.
 [157] Paiho S, Hoang H, Hukkalainen M. Energy and emission analyses of solar assisted local energy solutions with seasonal heat storage in a Finnish case district. Renewable Energy 2017;107:147-55. https://doi.org/10.1016/ ircneng.2017.02.03
- e.2017.02.003
- [158] Prasanna A, Dorer V, Vetterli N. Optimisation of a district energy system with a low temperature network. Energy 2017;137:632–48. https://doi.org/ 10.1016/j.energy.2017.03.137.
 [159] Robertson E, Galloway S. Multienergy vector modelling of a scottish energy
- Robertson E, Ganoway S. Multienergy vector modeling of a scottas inenergy system: transitions and technology implications. Proceedings of the Insti-tution of Mechanical Engineers, Part A: Journal of Power and Energy 2017;231(6):580–9. https://doi.org/10.1177/0957650917711820.
 Yazdanie M, Densing M, Wokaun A. Cost optimal urban energy systems planning in the context of national energy policies: a case study for the city of Basel. Energy Policy 2017;110:176–90. https://doi.org/10.1016/ j.enpol.2017.08.009.
- [160]
- [161] Scheller F. Krone J. Kühne S. Bruckner T. Provoking residential demand [101] Scherd T, Nohe J, Rome S, Florent F. Towong Concurrent denaming response through variable electricity tariffs – a model-based assessment for municipal energy utilities. Technology and Economics of Smart Grids and Sustainable Energy 2018;3(1). https://doi.org/10.1007/s40866-018-0045-x.
 [162] Burton CA, Ryan C, Rismanchi B, Candy S. Urban shared energy systems and
- behaviour change simulating a common pooled resource problem. Smart and Sustainable Built Environment 2019. https://doi.org/10.1108/SASBE-01 019-001
- [163] Perera ATD, Nik VM, Wickramasinghe PU, Scartezzini J-L. Redefining energy system flexibility for distributed energy system design. Applied Energy 2019;253. https://doi.org/10.1016/j.japenergy.2019.113572.
 [164] Razmjoo A, Shirmohammadi R, Davarpanah A, Pourfayaz F, Aslani A. Stand-
- [164] Razhigo A, Siminohamiadi K, Davapaniai A, Fouriyaz P, Asiani A. Stand-alone hybrid energy systems for remote area power generation. Energy Reports 2019;5:231–41. https://doi.org/10.1016/j.egyr.2019.01.010.
 [165] Teah HS, Yang Q, Onuki M, Teah HY. Incorporating external effects into project sustainability assessments: the case of a green campus initiative based on a solar PV system. Sustainability (Switzerland) 2019;11(20). https://doi.org/10.3300/su11205786 https://doi.org/10.3390/su11205786.
- [166] Toboso-Chavero S, Nadal A, Petit-Boix A, Pons O, Villalba G, Gabarrell X, et al. Towards productive cities: environmental assessment of the food-energy-water nexus of the urban roof mosaic. Journal of Industrial Ecology 2019;23(4):767–80. https://doi.org/10.1111/jiec.18229. Weinand JM, McKenna R, Fichtner W. Developing a municipality typology for
- [167] modelling decentralised energy systems. Utilities Policy 2019;57:75 https://doi.org/10.1016/j.jup.2019.02.003.
- [168] Calvert K. Mabee W. More solar farms or more bioenergy crops? Mapping
- 0.1186/s13705-017-0136-x
- [170] Fan S, Li Z, Wang J, Piao L, Ai Q. Cooperative economic scheduling for multiple energy hubs: a bargaining game theoretic perspective. IEEE Access 2018;6:27777–89. https://doi.org/10.1109/ACCESS.2018.2839108.
 [171] Meschede H. Increased utilisation of renewable energies through demand
- response in the water supply sector a case study. Energy 2019;175:810—7. https://doi.org/10.1016/j.energy.2019.03.137.
- [172]
- https://doi.org/10.1016/j.energy.2019.03.137.
 Facchinetti E, Rohrbach B, van der Wel G, Bollinger A. Monetary value of a district's flexibility on the spotand reserve electricity markets. Buildings 2018;8(12). https://doi.org/10.3390/buildings8120181.
 Perera ATD, Coccolo S, Scartezzini J-L, Mauree D. Quantifying the impact of urban climate by extending the boundaries of urban energy system modeling. Applied Energy 2018;222:847–60. https://doi.org/10.1016/j.incengru/2018.04.004 [173] energy 2018 04 004

- [174] Gardumi F, Welsch M, Howells M, Colombo E. Representation of balancing options for variable renewables in long-term energy system models: an application to OSeMOSYS. Energies 2019;12(12). https://doi.org/10.3390/
- en12122366.
 [175] Hirvonen J, ur Rehman H, Sirén K. Techno-economic optimization and analysis of a high latitude solar district heating system with seasonal storage, considering different community sizes. Solar Energy 2018;162:472–88. https://doi.org/10.1016/j.solener.2018.01.052.
 [176] Yazdanie M, Densing M, Wokaun A. The role of decentralized generation and
- storage technologies in future energy systems planning for a rural agglom-eration in Switzerland. Energy Policy 2016;96:432-45. https://doi.org/
- eration in Switzerland. Energy Policy 2016;96:432–45. https://doi.org/ 10.1016/j.enpol.2016.06.010.
 [177] Amil MLC Developing energy statistics. Comparison of Galician energy balance with other autonomous regions [La elaboración de estadísticas energéticas. Comparación del balance energético gallego y de otras comu-nidades autónomas]. Revista Galega de Economia 2010;19(1):1-16.
 [178] Regueiro Ferreira RM, Sánchez Sellero MC. Correspondences analysis appli-cation data on employment generation in the electrical system of Spain [Aplicación da análise de correspondencias sobre os datos de xeración de emprego no sistema eléctrico de España]. Revista Galega de Economia 2014/23(1):27-43.
- [179] Lund H. Möller B. Mathiesen BV. Dyrelund A. The role of district heating in future renewable energy systems. Energy 2010;35(3):1381-90. http: doi.org/10.1016/j.energy.2009.11.023.
- Intuit enewate energy systems integy 2010,5(3):1501-50. https:// doiorg/10.1016/j.energy.2009.11.023.
 Petrović S, Karlsson K, Ringkøbing-Skjern energy atlas for analysis of heat saving potentials in building stock. Energy 2016;110:166–77. https:// 10.1016/j.energy.2016.04.046.
- [181] Jaccard M, Murphy R, Zuehlke B, Braglewicz M. Cities and greenhouse gas
- Jacchi M, Mirky M, Zachie G, Diegertse H, Chao May Goura (2019) 144. https:// doi.org/10.1016/j.enpol.2019.07.011. Byrne J, Shen B, Wallace W. The economics of sustainable energy for rural development: a study of renewable energy in rural China. Energy Policy Development: a study of renewable energy in rural China. [182]
- [183] Vosen SR, Keller JO. Hybrid energy storage systems for stand-alone electric Vosen SK, keller JO, Hybrid energy storage systems for stand-alone electric power systems: optimization of system performance and cost through control strategies, International Journal of Hydrogen Energy 1999;24(12): 1139–56. https://doi.org/10.1016/S0360-3199(98)00175-X. Santarelli M, Macagno S. Hydrogen as an energy carrier in stand-alone ap-plications based on PV and PV-micro-hydro systems. Energy 2004;29(8): 1150–92. https://doi.org/10.1016/B.jourg.2004.02005.
- [184]
- 1159-82. https://doi.org/10.1016/j.energy.2004.02.023. [185] Khan MJ, Iqbal MT. Pre-feasibility study of stand-alone hybrid energy systems for applications in Newfoundland. Renewable Energy 2005;30(6): 835–54. https://doi.org/10.1016/j.renene.2004.09.001.
- Obara S. Operating schedule of a combined energy network system with fuel cell. International Journal of Energy Research 2006;30(13):1055–73. https:// org/10.1002/er.1203.
- [187] Kaldeliis JK. An integrated model for performance simulation of hybrid wind-diesel systems. Renewable Energy 2007;32(9):1544-64. https:// doi.org/10.1016/j.renene.2006.07.004.
- , Notton G, Belhamel M, Haddadi M, Louche A. Design and technoeconomical optimization for hybrid PV/wind system under various meteo-rological conditions. Applied Energy 2008;85(10):968-87. https://doi.org/
- 10.1016/j.apenergy.2008.02.012. [189] Palmer DJ, Sachs GD, Sembler WJ. A solar-hydrogen fuel-cell home and reamen (b), bathor (b), bethore (b), resolution (c), resolutio
- Applied Energy 2009;86(7-8):1024-30. https://doi.org/10.1016/
- Applied Energy 2009;86(7–8):1024–30. https://doi.org/10.1016/ j.apenergy.2008.10.015.
 [191] Bentouba S, Slimani A, Boucherit MS, Bourouis M, Coronas A. Analysis of photovollaic power system in remote area in Adrar south of Algeria. Inter-national Review of Mechanical Engineering 2010;4(4):460–5.
 [192] Colella WG, schneider SH, Kammen DM, Jhunjhunwala A, Teo N. Optimizing of the densure field engineering 2010;4(4):460–5.
- the design and deployment of stationary combined heat and power fuel cell systems for minimum costs and emissions - Part I: model design. Journal of Fuel Cell Science and Technology 2011;8(2). https://doi.org/10.1115/
- [193] Colella WG, Schneider SH, Kammen DM, Jhunjhunwala A, Teo N. Optimizing the design and deployment of stationary combined heat and power fuel cell systems for minimum costs and emissions - Part II: model results. Journal of Fuel Cell Science and Technology 2011;8(2). https://doi.org/10.1115/

- 1.4001757.
 [194] Kyriakarakos G, Dounis AI, Rozakis S, Arvanitis KG, Papadakis G. Poly-generation microgrids: a viable solution in remote areas for supplying po-wer, potable water and hydrogen as transportation fuel. Applied Energy 2011;88(12):4517-26. https://doi.org/10.1016/j.apenergy.2011.05.038.
 [195] Adaramola MS. Feasibility study of off-grid hybrid energy systems for ap-plications in Ondo state Nigeria. Journal of Engineering and Applied Sciences 2012;7(1):72-8. https://doi.org/10.1016.3923/jeas.2012.72.78.
 [196] Shiroudi A, Rashidi R, Gharehpetian GB, Mousavifar SA, Akbari Foroud A. Case study: simulation and optimization of photovoltaic-wind-battery hybrid energy system in Taleghan-Iran using homer software. Journal of Renewable and Sustainable Energy 2012;4(5). https://doi.org/10.1063/ 1.4754440. 1 4754440

- [197] Anayochukwu AV, Ndubueze NA. The potential of stand-alone PV/wind hybrid energy systems for power supply to remote rural areas in Nigeria. International Journal of Energy for a Clean Environment 2013;14(1):1-20. https://doi.org/10.1615/InterfEnerCleanEnv.2012004922.
 [198] Rawat R, Chandel SS. Simulation and optimization of solar photovoltaic-wind stand alone hybrid system in hilly terrain of India. International Journal of Renewable Energy Research 2013;3(3):595-604.
 [199] Kusakana K. Techno-economic analysis of off-grid hydrokinetic-based hybrid energy systems for onsbore/emote area in South Africa. Energy 2014;58:
- energy systems for onshore/remote area in South Africa. Energy 2014;68: 947-57. https://doi.org/10.1016/j.energy.2014.01.100. [200] Lacko R, Drobnič B, Mori M, Sekavčnik M, Vidmar M. Stand-alone renewable
- [200] Lacko K, Drobnik B, Moli M, Sekavenik M, Vidinal M, Sahid-alone tenewable combined heat and power system with hydrogen technologies for household application. Energy 2014;77:164–70. https://doi.org/10.1016/ j.energy.2014.05.110.
 [201] Lacko K, Drobnik B, Sekavenik M, Mori M. Hydrogen energy system with
- [201] Lako K, Drobnic B, SekaVnik M, Mort M. Hydrogen energy system with renewables for isolated households: the optimal system design, numerical analysis and experimental evaluation. Energy and Buildings 2014;80: 106-13. https://doi.org/10.1016/j.enbuild.2014.04.009.
 [202] Muda WMW, Fong CA, Al-Kayiem HH. ZSSEACA. Simulation-based method to evaluate a PV/wind hybrid renewable energy system in Terengganu. WIT Transactions on Ecology and the Environment 2014;186:91-102. https:// doi.org/10.2495/EXIS140081
- loi.org/10.2495/ESUS140081
- [203] Saheb D, Koussa M, Hadji S. Technical and economical study of a stand-alone vind energy system for remote rural area electrification in Algeria enewable Energy and Power Quality Journal 2014;1(12):638–43. https:// wind
- doi.org/10.24084/repqi12.439.
 Yazdanpanah M-A. Modeling and sizing optimization of hybrid photovoltaic/ wind power generation system. Journal of Industrial Engineering Interna-tional 2014;10(1). https://doi.org/10.1007/s40092-014-0049-7.
- Inonai 2014;10(1). https://doi.org/10.1007/360082/014-00497.
 [205] Alnejaili T, Drid S, Mehdi D, Chrifi-Alaoui L, Belarbi R, Hamdouni A. Dynamic control and advanced load management of a stand-alone hybrid renewable power system for remote housing. Energy Conversion and Management 2015;105:377-92. https://doi.org/10.1016/j.encomana.2015.07.080.
 [206] Ani VA. Feasibility analysis and simulation of a stand-alone photovoltaic.
- [200] Am VA. Feasibility analysis and simulation of a stand-alone photovoltatic energy system for electricity generation and environmental sustainability equivalent to 650VA fuel-powered generator popularly known as "I pass my neighbour. Frontiers in Energy Research 2015;3(SEP). https://doi.org/ 10.3389/fenrg.2015.00038.
 [207] Hong GW, Abe N, Baclay JR M, Arciaga L. Assessing users' performance to environmental environment in the consisting of the distribution of statements.
- sustain off-grid renewable energy systems: the capacity and willingness approach. Energy for Sustainable Development 2015;28:102-14. https:// 1016 sd 2015 07 004
- [208] Kim J-H, Kim J-F. Analysis of photovoltaic applications in zero energy building cases of IEA SHC/EBC Task 40/Annex 52. Sustainability (Switzerland) 2015;7(7):8782–800. https://doi.org/10.3390/srJ7078782.
 [209] Sharifishourabi M, Alimoradiyan H, Atikol U. Modeling of hybrid renewable

- [209] Sharifishourabi M, Alimoradiyan H, Atikol U. Modeling of hybrid renewable energy system: the case study of Istanbul, Turkey, Journal of Thermal Engineering 2016;2(6):990-4. https://doi.org/10.1818/6jtte.10645.
 [210] McKenna R, Merkel E, Fichtner W. Energy autonomy in residential buildings: a techno-economic model-based analysis of the scale effects. Applied Energy 2017;189:800-15. https://doi.org/10.1016/j.apenergy.2016.03.062.
 [211] Sagani A, Vrettakos G, Dedoussis V. Viability assessment of a combined hybrid electricity and heat system for remote household applications. Solar Energy 2017;151:33-47. https://doi.org/10.1016/j.solener.2017.05.011.
 [212] Duman AC, Giller G. Techno-economic analysis of df-grid PV/wind/fleel cell hybrid system combinations with a comparison of regularly and seasonally occupied households. Sustainable Cities and Society 2018;42:107-26. https://doi.org/10.1016/j.scs.2018.06.029.
 [213] Ferrari L, Bianchini A, Galli G, Ferrara G, Carnevale EA. Influence of actual component characteristics on the optimal energy mix of a photovoltaic-
- [213] Ferrari L, Bianchini A, Galli G, Ferrara G, Carnevale EA. Influence of actual component characteristics on the optimal energy mix of a photovoltaic-wind-diesel hybrid system for a remote off-grid application. Journal of Cleaner Production 2018;178:206–19. https://doi.org/10.1016/ j.jclepro.2018.01.032.
 [214] Maleki A. Modeling and optimum design of an off-grid PV/WT/FC/diesel hybrid system considering different fuel prices. International Journal of Low-Carbon Technologies 2018;13(2):140–7. https://doi.org/10.1093/ijilct/ ctv006.
- [215] Rey-Hernández JM, Yousif C, Gatt D, Velasco-Gómez E, San José-Alonso J, Rey-Martínez FJ. Modelling the long-term effect of climate change on a zero energy and carbon dioxide building through energy efficiency and renew-ables. Energy and Buildings 2018;174:85–96. https://doi.org/10.1016/ uild.2018.06.006
- [216] Franchini G, Brumana G, Perdichizzi A. Monitored performance of the first energy+ autonomous building in Dubai. Energy and Buildings 2019;205. https://doi.org/10.1016/j.enbuild.2019.109545.
 [217] Leary J, Schaube P, Clementi L. Rural electrification with household wind
- systems in remote high wind regions. Energy for Sustainable Development
- 2019;52:154-75. https://doi.org/10.1016/j.ed.2019.07.008.
 [218] Maouedj R, Benmedjahed M, Saba D, Mamemri A, Barbaoui B, Bezari S. Experimental analysis of a stand-alone wind-photovoltaic hybrid system in the Sahara Desert. European Journal of Electrical Engineering 2019;21(1):
- 93–7. https://doi.org/10.18280/ejee.210114.
 [219] Pawale TP, Chakrasali RI, Motekar RS, Halabhavi SB. Power reliability of hybrid non-conventional off-grid generation systems using multilevel inverter. International Journal of Recent Technology and Engineering

- 2019;8(1):3450-5.
 [220] Obara S, Watanabe S, Rengarajan B. Operation planning of an independent microgrid for cold regions by the distribution of fuel cells and water electrolyzers using a genetic algorithm. International Journal of Hydrogen Energy 2011;36(22):14295-308. https://doi.org/10.1016/
- Jappagene.2011.08.004.
 Shah KK, Mundada AS, Pearce JM. Performance of U.S. hybrid distributed energy systems: solar photovoltaic, battery and combined heat and power.
 Energy Conversion and Management 2015;105:71–80. https://doi.org/ [221] 10.1016/i.enconman.2015.07.048
- [222] Assaf J. Shabani B. A novel hybrid renewable solar energy solution for [222] Assai J. Shadam B. An novel nyotu renewable solar energy solution to continuous heat and power supply to standalone-alone applications with ultimate reliability and cost effectiveness. Renewable Energy 2019;138: 509-20. https://doi.org/10.1016/j.renene.2019.01.099.
 [223] Heydari A. Askarzadeh A. Optimization of a biomass-based photovoltaic
- Heydan A, Askarzaden A. Optimization of a biomass-based photovortaic power plant for an off-grid application subject to loss of power supply probability concept. Applied Energy 2016;165:601–11. https://doi.org/ 10.1016/j.apenergy.2015.12.095. Khattab NM, Badr MA, El Shenawy, et alSharawy HH, Shalaby MS. Feasibility of hybrid renewable energy water pumping system for a small farm in Egypt. International Journal of Applied Engineering Research 2016;11(11): 7406–14
- [224]
- [225] Habali SM, Saleh IA. Design of stand-alone brackish water desalination wind energy system for Jordan. Solar Energy 1994;52(6):525–32. https://doi.org/ 10.1016/0038-0922(94)90660-2.
- [226] Karimi L, Abkar I , Aghajani M, Ghassemi A. Technical feasibility comparison of off-grid PV-EDR and PV-RO desalination systems via their energy con-sumption. Separation and Purification Technology 2015;151:82-94. https:// g/10.1016/i.sennur.2015.07.023
- Maleki A Khajeh MC, Rosen MA. Weather forecasting for optimization of a hybrid solar-wind-powered reverse osmosis water desalination system using a novel optimizer approach. Energy 2016;114:1120-34. https:// doi.org/10.1016/j.energy.2016.06.134.
 [228] Astolfi M, Mazzola S, Silva P, Macchi E. A synergic integration of desalination
- Astolin M, Mazzola S, Silva F, MacChi E, A synergic integration of desaination and solar energy systems in stand-alone microgrids. Desalination 2017;419: 169–80. https://doi.org/10.1016/j.desal.2017.05.025. Kyriakarakos G, Dounis AI, Arvanitis KG, Papadakis G. Design of a Fuzzy Cognitive Maps variable-load energy management system for autonomous PV-reverse osmosis desalination systems: a simulation survey. Applied En-erg. 2012;187:E75. 94. https://doi.org/10.1016/j.10177
- ergy 2017;187:575-84. https://doi.org/10.1016/j.apenergy.2016.11.077. [230] Wu B, Maleki A, Pourfayaz F, Rosen MA. Optimal design of stand-alone reverse osmissi desalination friven by a photovoltatic and diesel generator hybrid system. Solar Energy 2018;163:91–103. https://doi.org/10.1016/ j.solener.2018.01.016.
- jsolener.2018.01.016.
 [231] Rezk H, Sayed ET, Al-Dhaifallah M, Obaid M, El-Sayed AHM, Abdelkareem MA, et al. Fuel cell as an effective energy storage in reverse osmosis desalination plant powered by photovoltaic system. Energy 2019;175:423-33. https://doi.org/10.1016/j.energy.2019.02.167.
 [232] McGowan JG, Manwell JF, Avelar C, Warner CL. Hybrid wind/PV/diesel hybrid power systems modeling and south American applications. Renewable En-ergy 1906;9(1-4):836-47.
 [233] Gómez G, Martínez G, Gálvez JL, Gila R, Cuevas R, Maellas J, et al. Optimi-zation of the photovolt is: bufferson cumple victure of a ted a bone armota.
- Solar Energy 2013;97:460–73. https://doi.org/10.1016/j.jst.400
- olar Energy solener.2013.08.016.
- JSOEner.2013.08.016.
 Alsharif MH, Nordin R, Ismail M. Energy optimisation of hybrid off-grid system for remote telecommunication base station deployment in Malaysia. Eurasip Journal on Wireless Communications and Networking 2015;2015(1):1-15. https://doi.org/10.1186/s1363-015-0284-7.
 Goel S, Ali SM. Hybrid energy systems for off-grid remote telecom tower in Oddate. Index. Internet Descent 2015;20(2):116-2. [235]
- [236] Odisha, India, International Journal of Ambient Energy 2015;36(3):116-22. https://doi.org/10.1080/01430750.2013.823110.
- https://doi.org/10.1080/01430/50.2013.823110.
 Olatomiwa L, Mekhile S, Huda ASN, Sanusi K. Techno-economic analysis of hybrid PV-diesel-battery and PV-wind-diesel-battery power systems for mobile BTS: the way forward for rural development. Energy Science and Engineering 2015;3(4):271-85. https://doi.org/10.1002/ese3.71.
 Scamman D, Newborough M, Bustamante H. Hybrid hydrogen-battery sys-[237]
- [238] tems for renewable off-grid telecom power. International Journal of Hydrogen Energy 2015;40(40):13876-87. https://doi.org/10.1016/ Hydrogen Energy i.iihydene.2015.08.071.
- Jajhydene.2015.08.01. Alsharif MH, Kim J. Hybrid off-grid SPV/WTG power system for remote cellular base stations towards green and sustainable cellular networks in South Korea. Energies 2017;10(1). https://doi.org/10.3390/en10010009. Yeshalem MT, Khan B. Design of an off-grid hybrid PV/wind power system
- [240] Yeshalem MI, Khan B. Design of an off-grid hybrid PV/wind power system for remote mobile base station: a case study. AIMS Energy 2017;5(1): 96–112. https://doi.org/10.3934/energy.2017.1.96. Alsharif MH, Kim J, Kim JH. Energy optimization strategies for Eco-friendly cellular base stations. Energies 2018;11(6). https://doi.org/10.3390/ en11061500.
- [242] Murugesan C. Marimuthu CN. Cost optimization of PV-diesel systems in nanogrid using LJ cuckoo search and its application in mobile towers. Mobile

- Netw Appl 2019;24(2):340-9. https://doi.org/10.1007/s11036-018-1046-7.
 [243] El Asmar T. React: self-sufficient renewable energy air-conditioning system for Mediterranean countries. Desalination 2008;220(1-3):600-11. https:// doi.org/10.1016/i.desal.2007.04.061.
- Grossland AF, Anuta OH, Wade NS. A socio-technical approach to increasing the battery lifetime of off-grid photovoltaic systems applied to a case study in Rwanda. Renewable Energy 2015;83:30–40. https://doi.org/10.1016/ increase.2016.01.0000 [244]
- [245] Olatomiwa L, Blanchard R, Mekhilef S, Akinyele D. Hybrid renewable energy [245] Olatoliniwa I, Bainchalu K, Mekniel S, Aknifele D, Nybhi Felewabie energy supply for rural bealthcare facilities: an approach to quality healthcare de-livery. Sustainable Energy Technologies and Assessments 2018;30:121–38. https://doi.org/10.1016/j.seta.2018.09.007.
 [246] Babatunde OM, Adedoja OS, Babatunde DE, Denwigwe IH. Off-grid hybrid renewable energy system for rural healthcare centers: a case study in Nigeria. Energy Science and Engineering 2019;7(3):676–93. https://doi.org/ 10.0001/ene2314
- 1002/ese3 314
- [247] Raghuwanshi SS, Arva R, Design and economic analysis of a stand-alone hybrid photovoltaic energy system for remote healthcare centre. Interna-tional Journal of Sustainable Engineering 2019. https://doi.org/10.1080.
- [248] Nfah EM, Ngundam JM. Feasibility of pico-hydro and photovoltaic hybrid
- [246] Nah Ew, Nguhuan JM, Peasibility of phC-hydro and photowida hydrid power systems for remote villages in Cameroon. Renewable Energy 2009;34(6):1445-50. https://doi.org/10.1016/j.renene.2008.10.019.
 [249] Aagreh Y, Al-Chzawi A. Feasibility of utilizing renewable energy systems for a small hotel in Ajloun city, Jordan. Applied Energy 2013;103:25-31. https:// doi.org/10.1016/j.apenergy.2012.10.008.
 [250] Shezan SA, Saidur R, Ullah KR, Hossain A, Chong WT, Julai S. Feasibility problemic of a hydrid eff orid und DC hydro energy energy for the organization.
- Snezan SA, Sahur K, Ulian KK, Hossain A, Chong WL, Julai S. Feasibility analysis of a hybrid off-grid wind-DcD-battery energy system for the eco-tourism remote areas. Clean Technologies and Environmental Policy 2015;17(8):2417–30. https://doi.org/10.1007/s10098-015-0983-0. Okonkwo EC, Okwose CF, Abbasoglu S. Techno-economic analysis of the potential utilization of a hybrid PV-wind turbine system for commercial buildings in Jordan. International Journal of Renewable Energy Research 2017;7(2):908–14
- [251] :908 -14.
- 2017;7(2):908-14.
 [252] Jung TY, Kim D, Moon J, Lim S. A scenario analysis of solar photovoltaic grid parity in the Maldives: the case of Malahini resort Sustainability (Switzerland) 2018;10(11). https://doi.org/10.3390/su10114045.
 [253] Galvão JR, Moreira L, Gaspar G, Vindeirinho S, Leitão S. Energy system retrofit in a public services building. Management of Environmental Quality: An International Journal 2017;28(3):302-14. https://doi.org/10.1108/MEQ-02-2014 0029 014-0028
- [254] Li N, Liu C, Li Y. Research on a long-duration solar-powered wireless sensor node energy system. Gaojishu Tongxin/Chinese High Technology Letters 2010;20(11):1201-5. https://doi.org/10.3772/jissn.1002-0470.2010.11.018.
 [255] Karakoulidis K, Mavridis K, Bandekas DV, Adoniadis P, Potolias C, Vordos N.
- [253] Katakohudes K, Mavinis K, Dahresa K, Jahresa K, 0.1016/i.renene.2011.07.032.
- [257] Carroquino J, Dufo-López R, Bernal-Agustín JL. Sizing of off-grid renewable
- earcy systems for drip irrigation in Mediterranean crops. Renewable En-ergy 2015;76:566–74. https://doi.org/10.1016/j.reneme.2014.11.069. Mudasser M, Yirido EK, Corscadden K. Cost-benefit analysis of grid-connected wind-biogas hybrid energy production, by turbine capacity and [258] ite. Renewable Energy 2015;80:573-82. https://doi.org/10.1016/ renene.2015.02.055. site.
- Boussetta M. Elbachtiri R. Elhammoumi K. Khanfara M. Optimal sizing of grid-connected PV-Wind system case study: agricultural farm in Morocco. Journal of Theoretical and Applied Information Technology 2016;86(2): [259]
- 196–206. [260] Nacer T, Hamidat A, Nadjemi O. A comprehensive method to assess the feasibility of renewable energy on Algerian dairy farms. Journal of Cleaner Production 2016;112:3631–42. https://doi.org/10.1016/ j.jclepro.2015.06.101.
- ebbagh T, Kelaiaia R, Zaatri A, Bechara T, Abdelouahed L. Investigation of the [261] use of a central unique renewable energy system versus distributed units for crop irrigation. Clean Technologies and Environmental Policy 2018;20(10):
- (2005) Comparison Clean Technologies and Environmental Policy 2018; 20(10): 2365–73. https://doi.org/10.1007/s10098-018-1599-y.
 (262) Anayochukwu AV, Onyeka AE. Feasibility study and simulation of optimal power system for off-grid voter registration centres. International Journal of Renewable Energy Research 2014;4(3):647–54.
 (263) Keshtkaran E, Fardoun AA, Noura H. Employing hydrogen fuel cell in hybrid standard and the probability of the proba
- energy systems for stand-alone, off-grid remote sites in UAE. Renewable Energy and Power Quality Journal 2014;1(12):811-7. https://doi.org/ 24084/repgi12.499
- [10.2404][pt]12.425.
 [10.2404][pt]12.425.
- electric vehicle charging station model. Energies 2015;8(11):13265-83. https://doi.org/10.3390/en81112368.

- [266] Amusat OO, Shearing PR, Fraga ES. Optimal integrated energy systems design incorporating variable renewable energy sources. Computers and Chemical Engineering 2016;95:21–37. https://doi.org/10.1016/ j.compchemeng.2016.08.007.
- Ansong M. Mensah LD. Adaramola MS. Techno-economic analysis of a hybrid [267] Autoring M, Mensan LD, Autorinola MS, Techno-econine analysis of a hybrid system to power a mine in an off-grid area in Ghana. Sustainable Energy Technologies and Assessments 2017;23:48–56. https://doi.org/10.1016/
- [268] Amusat OO, Shearing PR, Fraga ES. Optimal design of hybrid energy systems
- Amidsi Oo, Sheading FK, Fidga ES, Opfinial design of hybrid energy systems incorporating stochastic renewable resources fluctuations. Journal of Energy Storage 2018;15:379–99. https://doi.org/10.1016/j.est.2017.12.003. Diab F, Lan H, Zhang L, Ali S. An environmentally friendly factory in Egypt based on hybrid photovoltaic/wind/diesel/battery system. Journal of Cleaner Production 2016;112:3884–94. https://doi.org/10.1016/ j.jclepro.2015.07.008. [269]
- [270] Fabrizio E, Branciforti V, Costantino A, Filippi M, Barbero S, Tecco G, et al. Monitoring and managing of a micro-smart grid for renewable sources exploitation in an agro-industrial site. Sustainable Cities and Society 2017;28:88–100. https://doi.org/10.1016/j.scs.2016.08.026. Roth A, Boix M, Gerbaud V, Montastruc L, Etur P. A flexible metamodel ar-
- chitecture for optimal design of Hybrid Renewable Energy Systems (HRES) case study of a stand-alone HRES for a factory in tropical island. Journal of Cleaner r Production 2019;223:214-25. https://doi.org/10.1016/
- Jclepro.2019.03.095.
 [272] Khosravi A, Koury RNN, Machado L, Pabon JJG. Energy, exergy and economic analysis of a hybrid renewable energy with hydrogen storage system. Energy 2018;148:1087-102. https://doi.org/10.1016/j.energy.2018.02.008.
 [273] Duman AC, Giller O. Techno-economic analysis of off-grid photovoltaic LED
- road lighting systems: a case study for northern, central and southern re gions of Turkey. Building and Environment 2019;156:89–98. https://doi.org/ 10.1016/j.buildepy.2019.04.005
- Diemuodeke EO, Oko COC. Optimum configuration and design of a photo voltaic-diesel-battery hybrid energy system for a facility in University of Port Harcourt, Nigeria. International Journal of Ambient Energy 2016;37(1): 2-9. https://doi.org/10.1080/01430750.2013.866906.
- [275] Kumar A, Biswas A. Techno-economic optimization of a stand-alone photo-voltaic-battery renewable energy system for low load factor situation- a comparison between optimization algorithms. International Journal of En-gineering, Transactions A: Basics 2017;30(10):1555–64. https://doi.org/ 10.5829/ije.2017.30.10a.17.
- Aziz AS, Tajuddin MFNB, Adzman MRB, Feasibility analysis of PV/Wind/ [276]
- A212 AS, Tajuddin MFNB, Ad2man MKB, Feasibility analysis of PV/Windy Battery hybrid power generation: a case study. International Journal of Renewable Energy Research 2018;8(2):661–71. Salisu S, Mustafa MW, Mohammed OO, Mustapha M, Jumani TA. Techno-economic feasibility analysis of an offgrid hybrid energy system for rural electrification in Nigeria. International Journal of Renewable Energy Records 2010(2):242–26 Research 2019:9(1):261-70.
- [278] Goel S, Sharna R. Feasibility study of hybrid energy system for off-grid rural water supply and sanitation system in Odisha, India. International Journal of Ambient Energy 2016;37(3):314–20. https://doi.org/10.1080/ 01430750.2014.962089.
- [279] Di Fraia S, Macaluso A, Massarotti N, Vanoli L. Energy, exergy and economic analysis of a novel geothermal energy system for wastewater and sludge treatment. Energy Conversion and Management 2019;195:533-47. https:// /10.1016 an 2019 05 039
- [260] Haas J. Cebulla F. Nowak W. Rahmann C. Palma-Behnke R. A multi-service approach for planning the optimal mix of energy storage technologies in a fully-renewable power supply. Energy Conversion and Management 2018;178:355-66. https://doi.org/10.1016/j.enconman.2018.09.087.
- Waite M, Modi V. Impact of deep wind power penetration on variability at load centers. Applied Energy 2019;235:1048–60. https://doi.org/10.1016/ japenergy.2018.11.021. Kasperowicz R, Pinczyński M, Khabdullin A. Modeling the power of renew-[281]
- able energy sources in the context of classical electricity system trans-formation. Journal of International Studies 2017;10(3):264-72. https://
- [283] Karlsson K, Meibom P. Optimal investment paths of future renewable based energy system-Staing the optimal investment paths for future renewable based energy system-Staing the optimisation model Balmorel. International Jour-nal of Hydrogen Energy 2008;33(7):1777–87. https://doi.org/10.1016/
- [284] Vaillancourt K, Labriet M, Loulou R, Waaub J-P. The role of nuclear energy in long-term climate scenarios: an analysis with the World-TIMES model. En ergy Policy i.enpol.2008.01.015. 2008;36(7):2296-307. https://doi.org/10.1016/
- Jenpol.2008.01.015. Goodbody C, Walsh E, McDonnell KP, Owende P. Regional integration of renewable energy systems in Ireland the role of hybrid energy systems for small communities. International Journal of Electrical Power and Energy Systems 2013;44(1):713–20. https://doi.org/10.1016/j.ijepes.2013.08.012.
- [286]
- Systems 2013;44(1):713–20. https://doi.org/10.1016/j.ijepes.2012.08.012. Breyer C, Bogdanov D, Komoto K, Ehara T, Song J, Encbish N, North-east asian super grid: renewable energy mix and economics. Japanese Journal of Applied Physics 2015;54(8). https://doi.org/10.7567/JJAP.44.08K[01. Bogdanov D, Breyer C. North-East Asian Super Grid for 100% renewable energy supply: optimal mix of energy technologies for electricity, gas and heat supply options. Energy Conversion and Management 2016;112: 176–90. https://doi.org/10.1016/j.enconman.2016.01.019.

- I.M. Weinand et al. / Energy 203 (2020) 117817
- [288] Dominković DF, Bačeković I, Ćosić B, Krajačić G, Pukšec T, Duić N, et al. Zero
- [288] Dominkovic DF, Bačekovic I, Cosič B, Krajačić G, Pukšer T, Duić N, et al. Zero carbon energy system of south east Europe in 2050. Applied Energy 2016;184:1517-28. https://doi.org/10.1016/ji.apenergy.2016.03.046.
 [289] Frew BA, Becker S, Dvorak MJ, Andresen GB, Jacobson MZ. Flexibility mechanisms and pathways to a highly renewable US electricity future. Energy 2016;101:65-78. https://doi.org/10.1016/ji.energy.2016.01.079.
 [290] Aghalosseini A, Bogdanov D, Breyer C. A techno-economic study of an entirely renewable energy-based power suphy for North America for 2030 conditions. Energies 2017;10(8). https://doi.org/10.13390/en10081171.
 [201] Breur C, Bordhavu D, Culvari A, Astphoresini A, Brokya U, SUS, Kockingen O.
- [291] Breyer C, Bogdanov DD, Gulagi A, Aghahosseini A, Barbosa LSNS, Koskinen O, et al. On the role of solar photovoltaics in global energy transition scenarios. Progress in Photovoltaics: Research and Applications 2017;25(8):727-45. https://doi.org/10.1002/pip.2885.
 [292] Gulagi A, Bogdanov D, Breyer C A cost optimized fully sustainable power system for Southeast Asia and the Pacific Rim. Energies 2017;10(5). https://doi.org/10.2002/pip.2885.
- org/10.3390/en10050583
- doi.org/10.3390/en1005083.
 [293] Gulagi A, Bogdanov D, Fasihi M, Breyer C. Can Australia power the energy-hungry asia with renewable energy? Sustainability (Switzerland) 2017;9(2). https://doi.org/10.3390/su9020233.
 [294] Gulagi A, Choudhary P, Bogdanov D, Breyer C. Electricity system based on 100% renewable energy for India and SAARC. PLoS ONE 2017;12(7). https:// doi.org/10.1371/journal.com/page/101611.
- 100% renewable energy for India and SAARC. PLoS ONE 2017;12(7). https:// doi.org/10.1371/journal.pone.0180611.
 [295] Hess D, Wetzel M, Cao K-K. Representing node-internal transmission and distribution grids in energy system models. Renewable Energy 2018;119: 874-90. https://doi.org/10.1016/j.irenene.2017.10.041.
 [296] Ikäheimo J, Kiviluoma J, Weiss R, Holttinen H. Power-to-ammonia in future North European 100 % renewable power and heat system. International Journal of Hydrogen Energy 2018;43(36):17295-308. https://doi.org/ 10.1016/j.ijhydene.2018.06.121.
 [207] Oruma D, Metkering A, Bandara D, Bantaro G, Bathura Da Sallin and Sallin S. S
- [297] Oyewo SA, Sqahaosseini A, Bogdanov D, Breyer C. Pathways to a fully sus-tainable electricity supply for Nigeria in the mid-term future. Energy Con-version and Management 2018;178:44–64. https://doi.org/10.1016/ isureprep.2010.0026
- [298] Bogdanov D, Farfan J, Sadovskaia K, Aghahosseini A, Child M, Gulagi A, et al. Radical transformation pathway towards sustainable electricity via evolu-tionary steps. Nature Communications 2019;10(1). https://doi.org/10.1038/ s41467-019-08855-1
- [299] Bogdanov D, Toktarova A, Breyer C. Transition towards 100% renewable power and heat supply for energy intensive economies and severe conti-nental climate conditions: case for Kazakhstan. Applied Energy 2019;253.
- nental climate conditions: case for Kazakhstan. Applied Energy 2019;253. https://doi.org/10.1016/j.apenergy.2019.113606.
 Breyer C, Fasihi M, Aghahosseini A. Carbon dioxide direct air capture for effective climate change mitigation based on renewable electricity: a new type of energy system sector coupling. Mitigation and Adaptation Strategies for Global Change. 2019. https://doi.org/10.1007/s11027-019-9847-y.
 Breyer C, Khalli S. Bogdanov D. Solar photovoltaic capacity demand for a sustainable transport sector to fulfil the Paris Agreement by 2050. Progress in Photovoltaics: Research and Annitications 2010-97(11)/978-90. https://
- Sustainable transport sector to fullin the Parts Agreement by 2050. Progress in Photovoltaics: Research and Applications 2019;27(11):978-89. https:// doi.org/10.1002/pip.3114.
 [302] Child M, Kemfert C, Bogdanov D, Breyer C. Flexible electricity generation, grid exchange and storage for the transition to a 100% renewable energy system in Europe. Renewable Energy 2019;139:80-101. https://doi.org/ 10.1016/j.renene.2019.02.077.
- 10.1016/j.renene.2019.02.077.
 Moner-Girona M, Bódis K, Morrissey J, Kougias I, Hankins M, Huld T, et al. Decentralized rural electrification in Kenya: speeding up universal energy access. Energy for Sustainable Development 2019;52:128–46. https:// doi.org/10.1016/j.ed.2019.07.009.
 Vanhanen JP, Kauranen PS, Lund PD, Manninen LM. Simulation of solar
- hydrogen energy systems. Solar Energy 1994;53(3):267-78. https://doi.org/ 10.1016/0038-092X(94)90633-5.
- 10.1016/0038-092X(94)90633-5.
 [305] Notton G, Muselli M, Poggi P, Louche A. Decentralized wind energy systems providing small electrical loads in remote areas. International Journal of Energy Research 2001;25(2):141-64. https://doi.org/10.1002/er.670.
 [306] Tsoutsos TD, Zoulias EL, Lymberopoulos N, Glöckner R. Analysis of the barriers for the hydrogen energy technology in stand-alone power systems. Wind Engineering 2004;28(5):615-20. https://doi.org/10.1260/ 10952404
- 0309524043028055.
 [307] Khan MJ, Iqbal MT. Dynamic modeling and simulation of a small wind-fuel cell hybrid energy system. Renewable Energy 2005;30(3):421–39. https://doi.org/10.1016/j.renene.2004.05.013.
 [308] McKeough P, Solantausta Y, Kyllönen H, Faaij A, Hamelinck C, Wagener M, et al. Techno-economic analysis of biotrade chains. Upgraded biofuels from Russia and Canada to the Netherlands 2005;2312:3–40. VTT Tiedotteita Valtion Terknillinen Tutkimuskeskus.
 [309] Hrayshat ES. Wind resource assessment of the Jordanian southern region. Renewable Energy 2007;32(11):1948–60. https://doi.org/10.1016/j.renene.2006.11.008.
 [310] Matsunaka M, Shido S, Kondo K, Mivamaru H, Murata L Burpun calculation.
- [310] Matsunaka M, Shido S, Kondo K, Mivamaru H, Murata I, Burnup calculation of fusion-fission hybrid energy system with thorium cycle. Fusion Engi-neering and Design 2007;82(15–24):2779–85. https://doi.org/10.1016, j.fusengdes.2007.03.047.
- [311] Aghajani S, Joneidi IA, Kalantar M, Mortezapour V. Modeling and simulation of a PV/FC/UC hybrid energy system for stand alone applications. Interna-tional Review on Modelling and Simulations 2010;3(1):82–9.
- [312] Hammar L, Ehnberg J, Mavume A, Francisco F, Molander S. Simplified site

screening method for micro tidal current turbines applied in Mozambique. Renewable Energy 2012;44:414–22. https://doi.org/10.1016/ Renewable Energy j.renene.2012.02.010.

- j.renene.2012.02.010.
 [313] Jain S, Jiang J, Huang X, Stevandic S. Modeling of fuel-cell-based power supply system for grid interface. IEEE Transactions on Industry Applications 2012;48(4):1142–53. https://doi.org/10.1109/TIA.2012.2199454.
 [314] Moussa MF, Dessouky VG, Hakansson A. JLCHMHRJ, Design and control of a diode champed multilevel wind energy system using a stand-alone AC-DC-AC converter, Smart Innovation, Systems and Technologies 2013;22: 797–812. https://doi.org/10.1007/978-3-642-36645-1_71.
 [315] Gehring D. Wave catcher provides momentum for alternative energy. Sea Technology 2014;55(8):41-7.
 [316] Labangiri M. Namuellikh O. Cadacher A. Catcher, Market M. Schwarz, Sandra M. Schwarz, Sandr
- [316] Jahangiri M, Nematollahi O, Sedaghat A, Saghafian M. Techno-economical assessment of renewable energies integrated with fuel cell for off grid electrification: a case study for developing countries. Journal of Renewable
- [317]
- electrification: a case study for developing countries, journal of kenewable and Sustainable Energy 2015;7(2). https://doi.org/10.1063/14981852.
 Manfrida G, Secchi R. Performance prediction of a small-size adiabatic compressed-air energy storage system. International Journal of Thermody-namics 2015;18(2):111-9. https://doi.org/10.5541/jiot.5000071710.
 Stockley T, Thanapalan K, Bowkett M, Williams J. Enhanced OCV prediction mechanism for a stand-alone PV-lithium ion renewable energy system. Systems Science and Control Engineering 2015;3(1):524-34. https://doi.org/
- Systems Science and Control Engineering 2015;3(1):524-34. https://doi.org/ 10.1080/21642583.2015.1111780.
 Ursu V, Bartha S. Wind conditions modeling for small wind turbines. UPB Scientific Bulletin, Series C: Electrical Engineering and Computer Science 2015;77(2):277-88.
 Assou R, Rekioua T, Rekioua D, Tourzi A, Application of nonlinear predictive
- Austor R, Rekloud J, Rekloud J, Holliz R Application for limited predictive control for charging the battery using wind energy with permanent magnet synchronous generator. International Journal of Hydrogen Energy 2016;41(45):20964-73. https://doi.org/10.1016/j.ijhydene.2016.05.249.
 Parkhurst KM, Saffron CM, Miller RO. An energy analysis comparing biomass torrefaction in depots to wind with natural gas combustion for electricity generation. Applied Energy 2016;179:171-81. https://doi.org/10.1016/j. ispancerr.2016.06.121
- [321] apenergy.2016.05.121
- [322] Barberis S. Roncallo F. Traverso A. Towards innovative district energy management: a case study with stochastic renewable generators. Entrepre-neurship and Sustainability Issues 2017;4(3):294–309. https://doi.org/ 10.9770/jesi.2017.4.35(5.
- [323] Bordin C, Anuta HO, Crossland A, Gutierrez IL, Dent CJ, Vigo D. A linear programming approach for battery degradation analysis and optimization in offgrid power systems with solar energy integration. Renewable Energy 2017;101:417–30. https://doi.org/10.1016/j.renene.2016.08.066. [324] Oliveira Costa Souza Rosa C de, Costa KA, da Silva Christo E, Bertahone PB.
- Complementarity of hydro, photovoltaic, and wind power in Rio de Janeiro State. Sustainability (Switzerland) 2017;9(7). https://doi.org/10.3390/ [325] Parra D. Norman SA. Walker GS. Gillott M. Optimum community energy
- Parta D, Norman SA, Walker GS, Gillott M, Optimum community energy storage for renewable energy and demand load management. Applied En-ergy 2017;200:358–69. https://doi.org/10.1016/j.apenergy.2017.05.048. Schweiger G, Larsson P-O, Magnusson F, Lauenburg P, Velut S. District heating and cooling systems framework for Modelica-based simulation and dynamic optimization. Energy 2017;137:566–78. https://doi.org/ 10.1016/j.energy.2017.05.115.
- [327] Amiri S. Weinberger G. Increased cogeneration of renewable electricity Hindi S, Weinewale G. Interset Ogenetation of Fenewale electricity through energy cooperation in a Swedish district heating system - a case study. Renewable Energy 2018;116:366–77. https://doi.org/10.1016/ j.renene.2017.10.003.
- [328] A computationally efficient Li-ion electrochemical battery model for long-
- A computationally efficient Li-Ion electrochemical battery model for long-term analysis of stand-alone renewable energy systems. Journal of Energy Storage 2018;17:93–101. https://doi.org/10.1016/j.est.2018.02.015. Khraiwish Dalabeeh AS. Optimizing the combinations of sites, turbine types and cells types of a hybrid power system for remote sites in Jordan. Journal of Engineering and Applied Sciences 2018;13(16):6659–67. https://doi.org/ [329] 10.3923/ieasci.2018.60 59.6667
- U.Sy23/je38c1.2018b659b667.
 U.X. Chalvatzis KJ, Stephanides P. Innovative energy islands: life-cycle cost-benefit analysis for battery energy storage. Sustainability (Switzerland) 2018;10(10). https://doi.org/10.3390/su10103371.
 Sanni M. Mapping dynamics of low-carbon energy innovation for small hydropower technology in Nigeria. African Journal of Science, Technology, Innovation and Development 2018;10(5):569–78. https://doi.org/10.1080/ 20042362 2017.1002657. 0421338 2017 1380587
- [332] Almeshaiei E. Al-Habaibeh A. Shakmak B. Rapid evaluation of micro-scale Amiestialet E, Ar-habitet A, Shakhak B, Kapiu evaluation 01 mitro-scale photovoltaic solar energy systems using empirical methods combined with deep learning neural networks to support systems' manufacturers. Journal of Cleaner Production 2019. https://doi.org/10.1016/j.jclepro.2019.118788. Gan LK, Echenique Subiabre EJP. A realistic laboratory development of an
- [333] isolated wind-battery system. Renewable Energy 2019;136:645-56. https:// loi.org/10.1016/i.ren 2019.01.024
- Gadegesin AO, Sun Y, Nwulu NI. Techno-economic analysis of storage degradation effect on levelised cost of hybrid energy storage systems. Sus-tainable Energy Technologies and Assessments 2019;36. https://doi.org/ 10.1016/j.eco.2010.0106756 10.1016/j.seta. 2019.10053
- [335] Ghorbanzadeh M, Astaneh M, Golzar F. Long-term degradation based anal-ysis for lithium-ion batteries in off-grid wind-battery renewable energy

J.M. Weinand et al. / Energy 203 (2020) 117817

systems. Energy 2019;166:1194-206. https://doi.org/10.1016

- jenergy.2018.10120. [336] Nedjalkov A, Meyer J, Göken H, Reimer MV, Schade W. Blueprint and implementation of rural stand-alone power grids with second-life lithium ion vehicle traction battery systems for resilient energy supply of tropical or remote regions. Materials 2019;12(16). https://doi.org/10.3390/ remote
- [337] Tong K, Fang A, Yu H, Li Y, Shi L, Wang Y, et al. Estimating the potential for industrial waste heat reutilization in urban district energy systems: method
- Industrial wate near returnation in two Chinese provinces. Environmental Research Letters 2017;12(12). https://doi.org/10.1088/1748-9326/aa8a17.
 Ma T, Yang H, Lu L Feasibility study and economic analysis of pumped hydro storage and battery storage for a renewable energy powered island. Energy Conversion and Management 2014;79:387-97. https://doi.org/10.1016/ j.encomman.2013.12.047.
 Mar B, Scalewich JB, une Decempendicing in congrupting mathematical and hydrogeneous and hyd
- [339] Muñoz JR, Spakovsky MR von. Decomposition in energy system synthesis design optimization for stationary and aerospace applications. Journal of Aircraft 2003;40(1):35-42.
- [340] 10.1109/IESTPE.2016.2615763
- [34] Eitzen ZA, Xu K-M, Wong T. Statistical analyses of satellite cloud object data from CERES. Part V: relationships between physical properties of marine boundary layer clouds. Journal of Climate 2008;21(24):6668–88. https:// doi.org/10.1175/2008(Cl/1300) doi.org/10.1175/2008JCLI2307.1. [342] Fasullo JT, Trenberth KE. The annual cycle of the energy budget. Part I: global
- mean and land-ocean exchanges. Journal of Climate 2008;21(10):2297–312. https://doi.org/10.1175/2007JCL11935.1.
- https://doi.org/10.1175/2007/IC11935.1.
 [343] Fasulo JT, Trenberth KE. The annual cycle of the energy budget. Part II: meridional structures and poleward transports. Journal of Climate 2008;21(10):2313-25. https://doi.org/10.1175/2007/IC11936.1.
 [344] Chatziagorakis P, Ziogou C, Elmasides C, Sirakoulis GC, Karafyllidis I, Andreadis I, et al. Enhancement of hybrid renewable energy systems control with a semicidar to transfer forenerging. The area of Olivio.
- with neural networks applied to weather forecasting: the case of Olvio. Neural Computing and Applications 2016;27(5):1093–118. https://doi.org/ 10.1007/s00521-015-2175-6.
- [345] Fargli HM, Fahmy FH, El-Sayed MAH. Artificial intelligence techniques controlling pv-wind powered rural zone in Egypt. Renewable Energy and Power Quality Journal 2009;1(7):102–10. https://doi.org/10.24084/ Power pgi07.257
- Sabari Nathan L, Ashok Kumar L, Vaikundaselvan B, Ravi Krishna S. Modeling of control and power management of a hybrid wind-solar system using optimization technique. International Review on Modelling and Simulations [346]
- [347] Malla SG, Bhende CN. Voltage control of stand-alone wind and solar energy
- Maila SG, Bhende CN, Voltage Control of stand-alone wind and solar energy system. International Journal of Electrical Power and Energy Systems 2014;56:361–73. https://doi.org/10.1016/j.ijepes.2013.11.030. Bogaraj T, Kanakaraj J, Chelladuraj J. Modeling and simulation of stand-alone hybrid power system with fuzzy MPPT for remote load application. Archives of Electrical Engineering 2015;64(3):487–504. https://doi.org/10.2478/aee-eol.co.002 [348] 015-0037
- [349] Prakash SL, Arutchelvi M, Jesudaivan AS, Autonomous PV-arrav excited [349] Prakash SL, Arutchelvi M, Jesudaiyan AS. Autonomous PV-array excited wind-driven induction generator for off-grid application in India. IEEE Journal of Emerging and Selected Topics in Power Electronics 2016;4(4): 1259-69. https://doi.org/10.1109/JESTPE.2016.2579678.
 [350] Prinsloo G, Mammoli A, Dolson R. Discrete coorgeneration optimization with storage capacity decision support for dynamic hybrid solar combined heat
- storage capacity decision support for dynamic hyporta solar combined near and power systems in isolated rural villages. Energy 2016;116:1051–64. https://doi.org/10.1016/j.energy.2016.10.026.
 [351] Islar M, Brogaard S, Lemberg-Pedersen M. Feasibility of energy justice: exploring national and local efforts for energy development in Nepal. Energy Policy 2017;105:668–76. https://doi.org/10.1016/j.enpol.2017.03.004.
 [352] Prinsloo G, Mammoli A, Dobson R. Customer domain supply and load co-ordinizion a core for sympt villages and tracective corpticip in such off grid
- ordination: a case for smart villages and transactive control in rural off-grid microgrids. Energy 2017:135:430-41. https://doi.org/10.1016 ия. Епе 2017.06.106
- Jenergy.2017.06.106 (353) Roumila Z, Rekioua D, Rekioua T. Energy management based fuzzy logic controller of hybrid system wind/photovoltaic/diesel with storage battery. International Journal of Hydrogen Energy 2017;42(30):19525–35. https:// doi.org/10.1016/j.ijhydene.2017.06.006.
- doi.org/10.1016/j.ijhydene.2017.06.006.
 [354] Prinsloo G, Dobson R, Mammoli A. Synthesis of an intelligent rural village microgrid control strategy based on smartgrid multi-agent modelling and transactive energy management principles. Energy 2018;147:263-78.
 [355] Zaibi M, Champenois G, Roboam X, Belhadj J, Sareni B. Smart power management of a hybrid photovoltaic/wind stand-alone system coupling battery storage and hydraulic network. Mathematics and Computers in Simulation 2018;416:101.
- storage and hydraulic network. Mathematics and Computers in simulation 2018;146:210-28. https://doi.org/10.1016/j.matcom.2016.08.009.
 [356] Mosaad MI, Abed El-Raouf MO, Al-Ahmar MA, Bendary FM. Optimal PI controller of DVR to enhance the performance of hybrid power system feeding a remote area in Egypt Sustainable Cities and Society 2019;47. https://doi.org/10.1016/j.scs.2019.101469.
- [357] Swart AJ. Cost-effective remote energy monitoring using the ESP8266 NodeMCU. International Journal of Recent Technology and Engineering

2019;7(6):974-9

- [358] Alhamwi A, Medjroubi W, Vogt T, Agert C. GIS-based urban energy systems models and tools: introducing a model for the optimisation of flexibilisation
- models and tools: introducing a model for the optimisation of flexibilisation technologies in urban areas. Applied Energy 2017;19:11–9. https://doi.org/10.1016/j.apenergy.2017.01.048.
 [359] Alhamwi A, Medjroubi W, Vogt T, Agert C. Modelling urban energy requirements using open source data and models. Applied Energy 2018;231: 1100–8. https://doi.org/10.1016/j.apenergy.2018.09.164.
 [360] Böckl B, Greiml M, Leitner L, Pichler P, Kriechbaum L, Kienberger T. HyFloW-a hybrid load flow-modelling framework to evaluate the effects of energy storage and sector coupling on the electrical load flows. Energies 2019;12(5). https://doi.org/10.3390/en12050956.
 [361] Prinsloo GJ. Scoping exercise to determine load profile archetype reference shapes for solar co-generation models in isolated off-grid rural African villages. J energy South Afr 2016;27(3):11. https://doi.org/10.17159/2413-3051/2016/v27/33175.
- 051/2016/v27i3a1375
- [362] Lombardi F. Balderrama S. Quoilin S. Colombo F. Generating high-resolution multi-energy load profiles for remote areas with an open-source stochastic model. Energy 2019;177:433–44. https://doi.org/10.1016 Energy
- j.energy.2019.04.097. [363] Das A, Balakrishnan V. Grid-connectivity of remote isolated islands-A proposition in Indian context, Journal of Renewable and Sustainable En-
- proposition in Indian context. Journal of Renewable and Sustainable Energy 2012;4(4). https://doi.org/10.1063/1.4738591.
 Hauber J, Ruppert-Winkel C. Moving towards energy self-sufficiency based on renewables: comparative case studies on the emergence of regional processes of socio-technical change in Germany. Sustainability (Switzerland) 2012;4(4):491-530. https://doi.org/10.3390/su4040491.
 Späth P. Understanding the social dynamics of energy regions-the importance of discourse analysis. Sustainability (Switzerland) 2012;4(6):1256-73. https://doi.org/10.3300/su4051256.
- ttns://doi.org/10.3390/su4061256
- https://doi.org/10.390/sta061256.
 Bourennani F, Rahnamayan S, Naterer GF. Optimal design methods for hybrid renewable energy systems. International Journal of Green Energy 2015;12(2):148–59. https://doi.org/10.1080/15435075.2014.888999.
 Hecher M, Vilsmaier U, Akhavan R, Binder CR. An integrative analysis of
- [367]
- districts. Building and Environment 2017;118:362-76. https://doi.org 10.1016/i.buildenv.2017.03.037.
- [370] Mikkola M, Jussila A, Ryynänen T. Collaboration in regional energy-efficiency development. Green Energy and Technology 2017;55–66. https://doi.org/ 0.100 44899 978-3-319
- [371] Young J. Brans M. Analysis of factors affecting a shift in a local energy system
- [371] Young J, Brans M, Analysis Of factors allecting a smit in a local energy system towards 100% renewable energy community. Journal of Cleaner Production 2017;169:117–24. https://doi.org/10.1016/j.jclepro.2017.08.023.
 [372] Katre A, Tozzi A. Assessing the sustainability of decentralized renewable energy systems: a comprehensive framework with analytical methods. Sustainability (Switzerland) 2018;10(4). https://doi.org/10.3390/ cu10041058 su10041058
- [373] Sareen S. Baillie D. Kleinwächter J. Transitions to future energy systems: Janet 1, Janue 7, Heinwachter 7, Hanshous John and Statianability (Switzerland) John Andrew 1, Jona 1, John All Charles 1, John All Statianability (Switzerland) Bertheau P, Dionisio J, Jütte C, Aquino C. Challenges for implementing renewable energy in a cooperative-driven off-grid system in the Philippines.
- Environmental Innovation and Societal Transitions 2019. https://doi.org/ 0.1016/i.eist.2019.03.002.
- 10.1016/j.eist.2019.03.002. Heikkurinen P, Lozanoska J, Tosi P. Activities of degrowth and political change. Journal of Cleaner Production 2019;211:555–65. https://doi.org/ 10.1016/j.jclepro.2018.11.119. Kalkbrenner BJ. Residential Vs. community battery storage systems con-sumer preferences in Germany. Energy Policy 2019;129:1355–63. https:// doi.org/10.1016/j.orge01019.03.041 [375]
- [376] oi.org/10.1016/i.enpol.2019.03.041
- [377] Mazzone A. Decentralised energy systems and sustainable livelihoods, what are the links? Evidence from two isolated villages of the Brazilian Amazon. Energy and Buildings 2019;186:138-46. https://doi.org/10.1016/ j.enbuild.2019.01.027.
- [378] Østergaard PA. Comparing electricity, heat and biogas storages' impacts on Obsergaard PA. Comparing electricity, near and biogas storages impacts on renewable energy integration. Energy 2012;37(1):255–62. https://doi.org/ 10.1016/j.energy.2011.11.039.
 Akizu O, Bueno G, Barcena I, Kurt E, Topaloğlu N, Lopez-Guede JM. Contri-butions of bottom-up energy transitions in Germany: a case study analysis. Energies 2018;11(4). https://doi.org/10.3390/en11040849.
 Kim K, Kim J. The optimal design and economic evaluation of a stand-alone PES approxy extem for excitantial design and economic evaluation of a stand-alone
- [380] RES energy system for residential, agricultural and commercial sectors. Korean Chemical Engineering Research 2016;54(4):470–8. https://doi.org/ 10.9713/kcer.2016.54.470.
- Lee D, Han S, Kim J. Economic and environmental assessment of a renewable stand-alone energy supply system using multi-objective optimization. Korean Chemical Engineering Research 2017;55(3):332-40. https://doi.org/ 10.9713/kcer.2017.55.3.332
- [382] Braun M. Girma Z. Techno economic assessment and ontimization study of

I.M. Weinand et al. / Energy 203 (2020) 117817

hybrid power system using homer software for electrification of rural district in Ethiopia. International Journal of Renewable Energy Research 2013;3(3):

- [383] HOMER Energy, HOMER Pro: simulation, optimization, and sensitivity
- [383] HOMER Energy. HOMER Pro: simulation, optimization, and sensitivity analysis [October 23, 2019]; Available from: https://www.homerenergy. com/products/pro/docs/latest/solving_problems_with_homer.html.
 [384] The World Bank World Bank country and lending groups [October 28, 2019]; Available from: https://datahejndeskworldbank.org/knowledgebase/ articles/906519-world-bank-country-and-lending-groups.
- a LUCe3/2005/19-WOHd-balk-Coult y-and-relating splotps.
 [38] Department of Development and Planning. Introduction: introduction to EnergyPLAN [October 25, 2019]; Available from: https://www.energyplan. eu/training/introduction/.
 [386] Walker M. HOMER Optimizer™, a faster path to finding least-cost microgrid options [January 15, 2019]; Available from: https://microgrid.news.com/ homer-optimizer-afster-path-to-finding-least-cost-microgrid-options].
 [387] Reiner Lemoine Institut Vergleich und Optimierung von zentral und dezentral_orientieten_auchs/und/den_zu/ einer_Strowersorgnum_aus/
- dezentral orientierten Vulgedri und Opinietung Stromversorgung aus Erneuerbaren Energien in Deutschland [October 31, 2019]; Available from: https://www.bvmw.de/fileadmin/pdf-archiv/Studie_zur_dezentralen_
- [388] Shivakumar A, Welsch M, Taliotis C, Jakšić D, Baričević T, Howells M, et al. Valuing blackouts and lost leisure: estimating electricity interruption costs for households across the European Union. Energy Research & Social Science 2017;34:39–48. https://doi.org/10.016/j.erss.2017.05.010.
- 2017;34:39–48. https://doi.org/10.1016/j.erss.2017.05.010. Limberger J. Boxem T. Pluymaekers M. Bruhn D. Manzella A. Calcagno P. et al. Geothermal energy in deep aquifers: a global assessment of the resource base for direct heat utilization. Renewable and Sustainable Energy Reviews [389]
- 2018;82:961-75. https://doi.org/10.1016/j.rser.2017.09.084. [390] IEEE Distribution Test Feeders [January 29, 2020]; Available from: https://
- [391] Eglitis L. WorldData. [January 08, 2020]; Available from: https://www
- worlddatamo).
 [392] Reserve Bank of India. Exchange rate of the Indian rupee vis-à-vis the sdr, US dollar, Pound Sterling, D.M./Euro and Japanese Yen (Financial Year-Annual Average and end-Year Rates). [January 08, 2020]; Available from: https:// www.rbi.org.injScripts/PublicationsView.aspx?id=19131.
- [393] Statistica Euro (EUR) to U.S. dollar (USD) annual average exchange rate from 1999 to 2019 [January 08, 2020]; Available from: https://www.statista.com/ statistics/12794/euro-to-u-s-dollar-annual-average-exchange-rate/.
 [394] PoundSterling. The Canadian dollar to U.S. Dollar historical exchange rates
- conversion page for 2019. [January 08, 2020]; Available from: https://w poundsterlinglive.com/best-exchange-rates/best-canadian-dollar-to-usollar-histor -2019
- [395] GlobalPetrolPrices. Electricity prices [January 13, 2020]; Available from:
- https://www.globalpetrolprices.com/electricity_prices/. [396] Kotzur I, Markewitz P, Robinius M, Stolten D. Impact of different time series
- [396] Kolzuri, Markewitz P, Kobmits M, Stoten D. Impact of different time series aggregation methods on optimal energy system design. Renewable Energy 2018;117:474–87. https://doi.org/10.1016/j.renene.2017.10.017.
 [397] Nahmmacher P, Schmid E, Hirth L, Knopf B. Carpe diem: a novel approach to select representative days for long-term power system modeling. Energy 2016;112:430–42. https://doi.org/10.1016/j.energy.2016.06.081.
 [398] Rai V, Robinson SA. Agent-based modeling of energy technology adoption:
- empirical integration of social, behavioral, economic, and environmental factors. Environmental Modelling & Software 2015;70:163-77. https:// doi.org/10.1016/j.envs0ft.2015.04.014. [399] Scheller F, Johanning S, Bruckner T. A review of designing empirically

grounded agent-based models of innovation diffusion: development process, conceptual foundation and research agenda [February 05, 2020]; Available from: https://www.econstor.eu/handle/10419/191981. Selvakkumaran S, Ahlgren E. Model-based exploration of Co-creation efforts: the case of solar photovoltaics (PV) in skåne, Sweden. Sustainability 2018;10(11):3905. https://doi.org/10.3390/su10113905. McKenna R. The double-edged sword of decentralized energy autonomy. Energy Policy 2018;113:747-50. https://doi.org/10.1016/ j.enpol.2017.11.033.

- [400]
- [401]
- [402]
- [404]
- j.enpol.2017.11.033.
 Pfenninger S, Hawkes A, Keirstead J. Energy systems modeling for twenty-first century energy challenges. Renewable and Sustainable Energy Re-views 2014;33:74–86. https://doi.org/10.1016/j.rser.2014.02.003.
 Pfenninger S, Hirth I, Schlecht I, Schmid E, Wiese F, Brown T, et al. Opening the black box of energy modelling: strategies and lessons learned. Energy Strategy Reviews 2018;19:63–71. https://doi.org/10.1016/j.esr.2017.12.002.
 Decarolis J, Daly H, Dodds P, Keppo I, Li F, McDowall W, et al. Formalizing best practice for energy system optimization modelling. Applied Energy 2017;194:184–98. https://doi.org/10.1016/j.apenergy.2017.03.01.
 Fan C, Xiao F, Yan C, Liu C, Li Z, Wang J. A novel methodology to explain and evaluate data-driven building energy performance models based on inter-pretable machine learning. Applied Energy 2019;235:1551–60. https:// doi.org/10.1016/j.apenergy.2018.11.081.
 Mainzer K, Killinger S, McKenna R, Fichtner W, Assessment of rooftop [405]
- [doi:10.100/aptency3.00.011001.
 [doi:10.1001.
 [doi:10.1 doi.org/10.1016/j.solener.2017.06.065. Feng C, Cui M, Hodge B-M, Zhang J. A data-driven multi-model methodology
- [407]
- Feirg C, Cui M, Houge E-M, Zhang J, A data-driven indiver internoter internoted internotogy with deep feature selection for short-term wind forecasting. Applied Energy 2017;190:1245–57. https://doi.org/10.1016/j.apenergy.2017.01.043.
 Amasyali K, El-Gohary NM. A review of data-driven building energy con-sumption prediction studies. Renewable and Sustainable Energy Reviews 2018;81:1192–205. https://doi.org/10.1016/j.rser.2017.04.095.
 Mohan N, Soman KP, Sachin Kumar S. A data-driven strategy for short-term. electric logic force string wind wapping mode algorithm. [408]
- [409] electric load forecasting using dynamic mode decomposition model. Applied Energy 2018;232:229-44. https://doi.org/10.1016/j.apenergy.2018.09.190. Hirth L, Ueckerdt F, Edenhofer O. Integration costs revisited – an economic
- [410] [410] Infinit, Jeckeller P, Definiter G. Integration Ossi Tevated a an economic framework for wind and solar variability. Renewable Energy 2015;74: 925-39. https://doi.org/10.1016/j.renene.2014.08.065.
 [411] Ueckerdt F, Hirth L, Luderer G, Edenhofer O. System LCOE: what are the costs of variable renewables? Energy 2013;63:61-75. https://doi.org/10.1016/
- energy.2013.10.072
- [412] Brown T. Schlachtberger D. Kies A. Schramm S. Greiner M. Synergies of sector biowith schadungereich, schadung, sc
- [413] Chaudry M., Jenkins N., Qadrdan M., Wu J. Combined gas and electricity network expansion planning. Applied Energy 2014;113:1171-87. https://doi.org/10.1016/j.apenergy.2013.08.071.
 [414] Nolden C., Schönfelder M., Eßer-Frey A., Bertsch V., Fichtner W. Network constraints in techno-economic energy system models: towards more accurate modeling of power flows in long-term energy system models. Energy System 2014;10:2072.
- Syst 2013;4(3):267–87. https://doi.org/10.1007/s12667-013-0078-0. Jägemann C, Hagspiel S, Lindenberger D. The economic inefficiency of grid parity: the case of German photovoltaics: EWI Working. Paper. [October 29, 2019]; Available from: https://www.econstor.eu/bitstream/10419/92970/1/ 775175242.pdf. [415]

Utilities Policy 57 (2019) 75-96



Contents lists available at ScienceDirect
Utilities Policy



journal homepage: www.elsevier.com/locate/jup

Developing a municipality typology for modelling decentralised energy systems



J.M. Weinand^{a,*}, R. McKenna^b, W. Fichtner^a

^a Chair of Energy Economics, Institute for Industrial Production (IIP), Karlsruhe Institute for Technology (KIT), Hertzstraße 16, 76187, Karlsruhe, Germany
^b DTU Management Engineering, Technical University of Denmark, Denmark

ARTICLE INFO

ABSTRACT

Keywords: Cluster analysis Decentralised energy system Socio-energetic indicators Energy autonomy One solution to the large-scale integration of renewable energies could be decentralised autonomous municipal energy systems. Several studies have already analysed the technical and economic implications of the mainly decentralised future energy system, but most are restricted in their insights by limited temporal and spatial resolution. This study employs a cluster analysis to develop a municipality typology in order to analyse the techno-economic suitability of municipalities for autonomous energy systems. A total of 34 socio-technical in-dicators are employed at the municipal level, with a particular focus on the sectors of Private Households and Transport, and the potentials for decentralised renewable energies. Selected quantitative cluster validation methods are employed alongside qualitative criteria to determine the optimal number of clusters. This results in a total of ten clusters, which show a large variation as well as some overlap with respect to specific indicators. An analysis of the municipalities from three German renewable energy projects "Energy Municipalities", "Bioenergy Villages" and "100% Renewable Energy Regions" shows that in eight of the ten clusters municipalities are aiming for energy autonomy trojects, however, especially if the degree of social acceptance and engagement for such projects is to be considered. To answer the more techno-economical part of this question, future work will employ the developed clusters in the context of an energy system optimisation. Insights gained at the municipal level will then be qualitatively transferred to the national context to assess the implications for the whole energy system.

1. Introduction

Ambitious national targets in energy policy are leading to a radical change in the energy industry, which is particularly marked by the expansion of renewable energies. Germany already generated 30% of electricity with renewable energy technologies in 2016 (Statistisches Bundesamt, 2017a), including around 50 GW of wind (on- and off-shore), about 7 GW of bioenergy and 40 GW photovoltaic (PV) plants (BMWi, 2016), of which around 98% are connected to the low voltage distribution networks (Wirth, 2016). Municipalities are often referred to as the driving force behind the energy transition since renewable energies alongside energy efficiency are exploited on a decentralised basis due to their characteristics. Hence the characteristics of the energy system are changing towards a more decentralised structure, which also applies to the owners and operators of energy plants. In Germany, private individuals are increasingly investing in renewable energy systems or forming so-called citizen-energy cooperatives for this

purpose. In fact, the majority of renewable plants in Germany are owned and operated by private individuals, farmers and communities (Klaus Novy Institut e.V. & trend:research 2011). This development is based on various socio-economic motives: among other things, citizens have the desire to play an active role in energy supply and to be more independent of central markets and structures (e.g. Boon and Dieperink, 2014; Volz, 2012).

In this context, the concept of municipal energy autonomy (Deutschle et al., 2015; Rae and Bradley, 2012; McKenna et al., 2014b, 2015, 2017b) has become established, which is employed here to also include energy autarky (Müller et al., 2011), self-sufficiency (Deutschle et al., 2015; Balcombe et al., 2015) and integrated community energy systems (Koirala et al., 2016). Alone the number of terms for this concept illustrates the diversity within the literature, which also extends to its definition. Three rough distinctions can be made between complete energy autonomy (i.e. off-grid), net or balanced energy autonomy, whereby local generation equals or exceeds demand on an

^{*} Corresponding author. E-mail address: jann.weinand@kit.edu (J.M. Weinand).

https://doi.org/10.1016/j.jup.2019.02.003 Received 8 January 2019; Accepted 11 February 2019 0957-1787/ © 2019 Elsevier Ltd. All rights reserved.

annual basis, and a tendency towards higher energy autonomy through decentralised renewables (McKenna et al., 2015). The extensive survey of Engelken et al. (2016) shows that the overwhelming number of municipalities with energy autonomy aspirations strive for the state of balanced energy autonomy and that the focus is usually on electrical energy.

The feasibility of municipal energy autonomy has been investigated in several case studies. In Scheffer (2008), a rural model region with 10,000 inhabitants and agriculture as well as trade and commerce, but without large-scale industry, is considered. The suitability of a rural settlement structure for energy autonomy is also investigated in Peter (2013), who shows that renewable energies could cover the electricity requirements of the "example village" with 3850 inhabitants, but with immense storage costs. Jenssen et al. (2014) conclude that complete energy autonomy in an "average" German municipality is technically attainable through the "bioenergy village" approach, albeit at high costs. Schmidt et al. (2012) examine the advantages and disadvantages of energy autonomy compared to conventional energy supply in Sauwald, Austria, Woyke and Forero (2014) evaluate complete energy autonomy in Pellworm, a municipality with 1100 inhabitants, which has already been a model location for the construction of renewable energies. Although the supply of energy exceeds the demand, complete energy autonomy is not possible with the current energy system in Pellworm due to grid constraints. Finally, the study by Bur (2012) examines the Marston Vale region in the UK, which would have to import heat energy and fuel in particular, while a large proportion of the demand for electricity could be met by energy supplied by the region itself.

Despite some general conclusions from these studies, such as a tendency to focus on balanced energy autonomy and electricity in more rural municipalities, there is until now no general framework within which to assess the feasibility of energy autonomy for a specific municipality. In addition, the high spatial and temporal resolutions required to satisfactorily model decentralised energy systems with large fractions of renewable energies make approaches to information reduction indispensable. This paper goes some way towards filling these gaps by developing a typology of Germany's 11,131 municipalities to support the selection of municipalities for future decentralised energy autonomy projects. With the help of a cluster analysis, these municipalities are divided into homogeneous clusters according to socio-energetic indicators. The objective is to identify municipalities where energy autonomy aspirations could make technical and economic sense, and thereby to support the transferal of successful projects to other municipalities within the same cluster. In addition, a foundation for energy system models is developed which enables large-scale modelling of decentralised energy systems without the requirement for high spatial resolutions, which is often a central limitation in such models at the national scale and above (Keles et al., 2017). Finally, representatives of municipalities can be encouraged to initiate energy autonomy projects themselves if they have already been successfully implemented in a similar municipality.

The paper is structured as follows. Section 2 presents a literature review and more clearly locates this paper in context. Section 3 then presents the methodology, before section 4 presents and section 5 discusses the results. The paper closes with a summary and conclusions in section 6.

2. Literature review

Several areas of energy research are relevant to this contribution, including those relating to the analysis of decentralised and centralised energy systems, the field of urban morphology, and the application of cluster analysis to energy systems in order to reduce information quantity whilst retaining quality.

Characterising and contrasting centralised and decentralised energy systems is a relevant area of research for this paper because it strongly relates to the suitability of decentralised energy systems to become energy autonomous. Examples of contributions in this area include Funcke and Bauknecht (2016), who develop typologies for both of these types of energy infrastructure, by focussing on infrastructure location and operation. Further, Schmid et al. (2016) analyse the actor types, motives and conceivable roles within today's centralised and tomorrow's decentralised energy systems from the perspectives of technology, actors and institutions. Others raise the question of the optimal "degree of centralisation" (Zentralisierungsgrad), first coined by Jensch (1989), i. e. the level at which decentralised energy systems should be aggregated and balanced (Bauknecht et al., 2015). Currently, most energy autonomous regions rely on the overarching centralised energy system for their flexibility and controllability (Funcke and Bauknecht, 2016). For example, Wimmer et al. (2014) compare centralised with decentralised wind expansion scenarios, concluding that the overall flexibility requirements are similar in both cases. Reiner Lemoine Institut (2013) finds that a decentralised renewables expansion would be economically favourable, largely due to higher required network expansion costs in the centralised case. Others reach the opposite conclusion, however, that centralised and hybrid energy systems are more economically efficient than purely decentralised ones (acatech, 2016). Although it is clear that a completely renewable energy supply based on decentralised, autonomous regions does not seem economical due to very large storage requirements (Peter, 2013), there is no clear consensus about the optimal degree of centralisation. Especially the related question of the technical feasibility of decentralised energy autonomy is addressed in this paper whilst the micro- and macro economic assessment is left to future work.

Urban morphology is the second relevant research area. It focusses on the form of the urban environment, including building types, ages and forms, and (amongst other things) its implications for the energy system. The field is well established, as demonstrated by the earlier contribution of Steemers (2003), who analysed the relationship between urban morphology and energy use in buildings and transport, the two main sectors (other than industry) that are relevant for urban planning. Also Ratti et al. (2005) explored the effects of urban textures on building energy consumption with digital elevation models, with case studies in three European cities. Similar methods were also more recently employed in the LSECities project (Rode et al., 2014a, 2014b), which analysed the effects of different types of urban forms on heat energy demand and derived generalised insights into these relationships in larger European cities. In the context of her PhD thesis, Miller (2013) approaches the connection between urban form and building energy use with a multi-scale approach and using the Metro Vancouver region in Canada as an example. All of these studies demonstrate the diversity amongst the urban building stock, leading to a substantial variation in heat demand. Others within this field have examined the relationship between solar energy potential and urban morphology in London, concluding that by optimising combinations of eight variables of urban form the solar irradiation of roofs and facades could be increased by around 9% and 45% (Sarralde et al., 2015). More recently, Jrquizo et al. (2017) explored different urban morphology metrics and their impact on energy consumption in four districts of Newcastle, UK. In a more detailed analysis, Hargreaves et al. (2017) investigate the most cost-effective decarbonisation options for regions with different urban forms in a UK context, showing for example how low-density urban areas are more suited to exploit ground-source heat pumps. Summarising, then, the field of urban morphology offers insights into the connection between energy demand and urban structures, but does not provide a transferable typology for the whole decentralised energy system.

The third and most relevant research field for this paper is that of cluster analysis. Despite being a common method in energy studies more widely, it has not often been employed in the analysis of decentralised energy systems. One example is Chévez et al. (2017), who examine the single region "Great La Plata" in Argentina at the

administrative level. The region is clustered into eight census area types with a k-means cluster analysis according to the consumption of electrical energy and other socioeconomic variables. The most important result is that electricity consumption increases strongly with the household sizes, which could, for example, support the construction of distribution networks. In addition, Unternährer et al. (2017) cluster 6224 buildings not yet connected to the local heating network at the administrative level. Depending on indicators such as the demand for space heating and domestic hot water, as well as georeferenced drilling costs for deep geothermal energy, the cluster analysis results in 16 clusters. Clusters and typologies have often been applied at the district scale, in identifying the most cost-effective low carbon energy solution for different types of districts (Hargreaves et al., 2017; McKenna et al., 2016, 2017a; Su et al., 2017), as well as at the building scale, for example in the context of residential heat demand studies (McKenna al., 2016, 2017a). In addition, Marquant et al. (2017) present a holistic approach for the optimisation of multi-scale distributed energy systems, by employing clusters of similar buildings at the district level.

There are some examples of applications of cluster analysis at higher levels of spatial aggregation. For example, Kaundinya et al. (2013) employ a k-medoid clustering method to divide a region in India into clusters of villages for supply with decentralised biomass power plants, and the value of k is chosen to minimise the total system costs. For Austria, Bramreiter et al. (2016) divide all of the 82 Austrian "Climate and Energy Model Regions (CEMs)", which aim for energy autonomy, into three clusters by ten indicators (e. g. population density, employment figures, energy consumption). In a subsequent step, all other Austrian municipalities are examined by cluster analysis, with the aim of identifying municipalities with characteristics similar to those of the CEMs. It is shown that large parts of Austria could also become CEMs and thus have the potential to become energy self-sufficient, at least on an annual basis. In the study of Requia et al. (2017), all 5570 municipalities in Brazil are divided into five clusters. However, the analysis does not focus on socio-energetic indicators, but on six types of pollutant emissions in the Transport sector such as CO_2 and NO_x . To transfer the results to energy systems of municipalities, indicators for the other consumption sectors Private Households. Industry and Commercial would have to be included in the cluster analysis. The investigations are not always limited to one country. For example, Noiva et al. (2016) investigate 142 cities, spread across all continents. Indicators for the analysis of the cities divided into six clusters are the parameters of supply and demand for water.

Another relevant example in the present case is the PhD dissertation of Wall (2016), who conducted a cluster analysis with the German district-free cities ("kreisfreie Städte") as objects and based on 41 socioenergetic indicators. The cluster analysis in Wall (2016) differs from this study not only in the choice of indicators but also in the choice of the research objects. The survey objects are not the municipalities, but only the 107 district-free cities in Germany. Other studies have employed cluster analysis to German regions, but most of these neither have a high spatial resolution nor focus on energy aspects. For example, in Kronthaler (2003) Germany was divided into 97 regions, which were then assigned to ten clusters in a cluster analysis. The study looked at 13 socioeconomic indicators, including employment figures and investment in industry. The research showed that the economic power of the regions in eastern Germany is still significantly lower than that of the western German regions. Heinritz (2000) also came to a similar conclusion, by evaluating the economic strength of the 441 districts in Germany, and dividing the districts into five clusters by socio-economic indicators such as gross domestic product per inhabitant. In three other studies, German municipalities are investigated, but none of the studies considers all 11.131 municipalities. Gevler et al. (2008) only analyse 240 municipalities in the core region of Central Germany. The delimitation into six clusters is based on local development trends. These include 16 indicators such as the development of the settlement and traffic area or business tax revenue per inhabitant. In Schultz and Utilities Policy 57 (2019) 75–96

Brandt (2016) 2916 of the 11,131 German municipalities are divided into nine clusters by demographic indicators (among other things, the "share of single-person households" or "share of under-18s"). Finally, the 1102 municipalities in the federal state of Baden-Württemberg are investigated in Statistisches Landesamt Baden-Württemberg (2009). The goal was not to place the municipalities in clusters but to identify the two municipalities that are closest to each other. Indicators such as population density or cars per 1000 inhabitants were used. Hence, although several German regions have been analysed with cluster analysis, a classification with energy indicators has not yet been made at the municipal level. This is the research gap addressed in this paper, as outlined in the following section.

3. Methodology

This section describes the data collection and standardisation (cf. section 3.1) as well as the execution of the factor analysis (cf. section 3.2) and cluster analysis (cf. section 3.3). The vast majority of cluster analyses evaluated in section 2 perform a hierarchically agglomerative cluster analysis with the Ward algorithm. 17 of the 23 analyses evaluated in Wall (2016) also apply hierarchically agglomerative cluster analysis. Hierarchical cluster analysis generates high-quality clusters and is, therefore, also used in this paper. To support the traceability of the cluster analysis, the most important information according to Bacher et al. (2010) is listed in Table 1.

3.1. Data collection and standardisation

Many of the indicators used in the studies mentioned in section 2 are also used in the cluster analysis presented in this paper, as well as newly selected indicators. This study uses the indicators in a comprehensive analysis and for the first time clusters all 11,131 municipalities in Germany. The 59 indicators used in the cluster analysis include data on the energy consumption sectors Private Households, Transport, Industry and Commercial as well as data to estimate the potential for renewable energies (see Table 2). The indicators whose data is only available at the district level are shown in italics in Table 2. In the following, the "X" values in brackets are used as abbreviations for the indicators. For the last three groups of indicators in the Private Household sector, the specific allocations of the "X" values will be described later in the text. Only the indicators used in the final analysis are assigned to "X" values. The question of why not all indicators are used is answered in section 3.1.5. A complete list of all indicators and their references is given in Table 8 in the Appendix.

27 of the 59 indicators were also used by Wall (2016) in his analysis of district-free cities. In the following, the reasons for selecting the additional indicators are explained.

3.1.1. Indicators of the consumption sector private households

Private households account for 26% of Germany's final energy consumption and should not be neglected in the energetic classification of municipalities. The majority of the final energy (69%) is used in households for space heating (Umweltbundesamt & BMWi, 2017).

3.1.1.1. Share of heating types. For the shares of heating types, the available data have been grouped into three groups:

Table 1

	Overview of	f the :	most	important	aspects	of	traceability	of a	cluster	analysis.	
--	-------------	---------	------	-----------	---------	----	--------------	------	---------	-----------	--

Object s	11,131 German municipalities
Variables/Indicators	59 Indicators (see section 3.1)
Algorithm	Ward
Cluster analysis method	Hierarchical-agglomerative
Criteria used to determine the	26 different methods and elbow criteria
number of clusters	(see section 3.3.2)
Software used	R

Table 2

Overview of the indicators used in the cluster analysis. Italics means that the data of the indicators were only available at the district level.

	and means and me and or me ma	icadore incre entry analabre at are area	
Consumption sector Private Households (29)	Consumption sector Transport (11)	Consumption sector Industry and Commercial (12)	Potential for renewable energies (7)
Population development between 2010 and 2015 (X1) [%]	Number of motor vehicles per 1000 inhabitants (X27)	Share of employment in the industrial sector [%]	Achievable hydrothermal temperature (X32) [°C]
Living space per person (X2) [m ²]	Number of cars per 1000 inhabitants (X28)	Share of employment in the commercial sector [%]	Necessary hydrothermal drilling depth (X33) [m]
Share of single-person households (X3) [%]	Share of diesel vehicles [%]	Energy productivity of manufacturing industry [€/GJ]	Technical PV potential per inhabitant (X34) [kWh/y]
Average household size (X4) [Persons]	Share of petrol vehicles [%]	Energy intensity of manufacturing industry [MJ/€]	Technical PV potential per km² (X35) [MWh/y]
Household density (X5) [Housholds per km ²]	Share of gas vehicles [%]	Productivity level of manufacturing industry [€/GJ]	Technical wind potential per inhabitant (X36) [MWh/y]
Share of owner-occupied apartments (X6) [%]	Share of hybrid vehicles [%]	Specific energy consumption of manufacturing industry [MJ/€]	Technical wind potential per km² (X37) [MWh/y]
Income per household (X7) [k€]	Share of electric vehicles [%]	Share of industrial sales tax [%]	Share of forest and
Share of over 65-year-olds (X8) [%]	Share of other vehicle types [%]	Share of commercial sales tax [%]	agricultural land (X38) [%]
Unemployment rate (X9) [%]	Population density (X29) [Inhabitants per km²]	Development of employment share in the industrial sector [%]	-
Share of settlement and traffic area (X10) [%]	Share of 18-64-year-olds (X30) [%]	Development of employment share in the commercial sector [%]	-
Heating days	Share of commuters in the workforce [%]	Development of energy intensity in the manufacturing sector [%]	-
Heating degree days	_	Number of manufacturing enterprises	_
Degree day number	_	per 1000 nousenoids (X31)	
Share of heating types (3 indicators) (X11-X13) [%]	-		
Share of building age class (9 indicators) (X14-X22) [%]	-		
Share of building type (4 indicators) (X23-X26) [%]	-		

1) Share of buildings with heating systems based on district heating (X11)

 Share of buildings with heating systems not based on district heating (X12)

3) Share of buildings without heating system (X13)

This segmentation allows conclusions to be drawn as to whether and to what extent there is a district heating network, a gas network or both in the municipalities. The existing infrastructures influence the selection decision of technologies which are suitable in the municipalities. As an example, power-to-heat plants and power-to-gas plants offer great opportunities for future flexibility in power generation. However, to store the energy from these plants, various networks are required, such as a district heating network for power-to-heat plants or a gas network for power-to-gas plants (Böttger et al., 2014). Furthermore, district heating systems are suitable for the integration of heat from renewable energies such as geothermal plants (Durst, 2015).

3.1.1.2 Shares of building age classes. The insulation condition of the building envelope has a significant influence on the space heating requirement in buildings (Braun, 2010). The building age class influences the insulation condition of the building envelope and is, therefore, an essential indicator for estimating the heat demand (Schuler et al., 2000). In the cluster analysis applied here, the building ages were divided into nine groups (see Table 8 in the Appendix).

3.1.1.3. Shares of building types. The type of building also has a significant influence on the demand for space heating in private households (Wei et al., 2014). Shipworth et al. (2010), for example, showed that the operating hours of the heating system in English homes are statistically dependent on the type of building. The biggest difference was found between detached houses, in which the heating is on for much longer, and terraced houses. This study distinguishes between detached houses (X23), semi-detached houses (X24), terraced houses (X25) and "other types of buildings" (X26).

3.1.2. Indicators of the consumption sector transport

For the indicators representing the Transport sector, the shares of hybrid, gas and other vehicles in the vehicle stock have been added (compared to Wall, 2016). Hybrid vehicles also include an internal combustion engine in addition to the electric motor. The combustion engine can compensate for the disadvantage of the limited range of electric vehicles (Høyer, 2008). The number of gas vehicles in Germany is around 100,000, and they can contribute to a significant reduction in pollutants and, in some cases, CO_2 emissions. If biomethane or synthetic methane is added to the fuel, gas vehicles can be as climate-friendly as electric vehicles (BWwi, 2016).

3.1.3. Indicators of the consumption sector Industry and Commercial

The Industry consumption sector accounts for almost 50% of the electricity supplied in Germany (Javied et al., 2016). Most of the data from the Industry consumption sector are only available for the

manufacturing sector. These data are suitable for estimating the energy consumption of industry, as manufacturing accounts for the largest share of energy consumption (27.4% of Germany's total primary energy demand) (Umweltbundesamt, 2016).

3.1.3.1. Number of manufacturing enterprises per 1000 households. The number of manufacturing enterprises is the only indicator of this sector provided at the municipal level. The indicator is based on 1000 households to compare the values for the different municipalities.

3.1.4. Indicators of the potential for renewable energies

In most practical examples of municipal energy autonomy, renewable energies are used to establish a sustainable energy system (Schmidt et al., 2012). Therefore, the potentials of renewable energies in a region are important indicators. The potentials of renewable energies applied in the cluster analysis are explained below.

3.1.4.1. Achievable hydrothermal temperature. In Germany, an increase in deep geothermal power stations is expected by 2030 (installed capacity in 2030: 850 MW_{cl}) (Hechler and Bredel-Schürmann, 2011). From 2003 to 2017, the annual supply of thermal energy by deep geothermal energy plants has increased from 60 GWh_{cl} to 1.3 TWh_{cl} , the supply of electrical energy has increased from $0 GWh_{cl}$ to 155 GWh_{cl} (Agemar et al., 2014, 2018 Umweltbundesamt, 2018). In this study, the focus is on hydrothermal systems, because petrothermal systems are not yet used in Germany (Hechler and Bredel-Schürmann, 2011). Electricity from geothermal energy currently receives a subsidy of 25.2 ε -cents/ kWh (Deutscher Bundestag, 2017).

Hydrothermal power plants have two main advantages: on the one hand, unlike many other renewable energy plants, they are capable of providing energy as base load. On the other hand, they show the lowest emissions of pollutants after hydroelectric power plants during the life cycle of the plant (Purkus and Barth, 2011). At the municipal level, several geothermal power plants are already being used to supply local and district heating (Hechler and Bredel-Schürmann, 2011). Therefore, the use of this technology should also be considered in future energy autonomy efforts.

An important indicator for estimating the economic potential of a geothermal plant is the achievable hydrothermal temperature. Hydrothermal temperatures above 110 °C are required for the economical operation of a geothermal plant to generate electricity (Agemar et al., 2014). Fig. 1 shows that the achievable hydrothermal temperatures strongly depend on the region. This means that municipalities have different hydrothermal potentials. Therefore, the indicators for



Fig. 1. Achievable average hydrothermal temperature (°C) at a depth of up to 5000 m in German municipalities according to Agemar (2017).

hydrothermal energy are included in the cluster analysis.

3.1.4.2. Necessary hydrothermal drilling depth. The depth of drilling to the water reservoirs mainly determines the amount of investment for a geothermal plant. The depth of the well depends on the local temperature gradient. In Germany, the average temperature gradient is 32 K/km, in some regions (Upper Rhine valley) up to 100 K/km are reached (Agemar et al., 2014).

3.1.4.3. Technical photovoltaic (PV) potential per inhabitant/per km². The data from Mainzer et al. (2014) were used to estimate the PV potential in municipalities. However, this data must be made comparable for the cluster analysis. The indicator is therefore determined by dividing the PV potential in kWh by the number of inhabitants. This means that the technical PV potential per inhabitant can now be used for each municipality. However, this indicator does not allow a statement to be made about the potential of the PV systems in the relation to the area. For this purpose, the energy density of the PV systems is determined by dividing the PV potential in MWh by the area in km².

3.1.4.4. Technical wind potential per inhabitant/per km². The indicators for the technical wind potential in MWh were determined in analogy to the technical PV potential per inhabitant and per km². The data from McKenna et al. (2014a) was used for this. This data is available at postcode level and could be assigned to the municipalities using the geoinformation system QGIS.

3.1.4.5. Share of forest and agricultural land in total area. Land areas are required for the construction and operation of many technologies based on renewable energies such as wind power plants, ground-mounted photovoltaics and biogas plants (Marx Gómez et al., 2014). While wind power plants may only be built in certain areas such as forests or agricultural land, biogas plants require entire areas for the cultivation of energy crops (mainly maize) (Liker-Jans et al., 2017; McKenna et al., 2014a). The proportion of woodland and agricultural land in the total area of the municipalities can, therefore, be used as an indicator to estimate the potential for these renewable energies.

3.1.5. Data in the investigation

In the final study, only 38 of the 59 indicators were used. The indicators used are marked in Table 2 with a compounded X-value. The first cluster analysis with all indicators showed that there was too much dependence on the indicators based on district data for them to accurately represent values for municipalities. For this reason, the indicators shown in italics in Table 2 were excluded from further analysis. In addition, the indicator "Share of commuters in employment" could not be included in the study, as the data only exist for 2014 and are incomplete.

Furthermore, one indicator for each of the proportions of heating types, building age classes and building types were eliminated for a reason described below. Many calculation steps of a factor and cluster analysis require a positive semi-definite data matrix (Lorenzo-Seva and Ferrando, 2006). A symmetrical matrix is positive semi-definite if all eigenvalues are nonnegative (Zhang, 2011). In this study, the matrices were not positive semi-definite. This problem was solved by eliminating linearly-dependent variables. Since the proportions of the indicator groups heating types, building age classes and building types can be added up to 100% in each case, an indicator value can always be calculated with the other indicator values. Therefore, X13, X22 and X26 are deleted from the dataset. More information about positive semi-definite matrices can be found in Zhang (2011).

3.1.6. Standardisation

Once the data is complete, it must be standardised for factor analysis. Standardisation serves to make the indicators comparable in their

range of values (Milligan and Cooper, 1988). This prevents indicators with larger values from being weighted more strongly. Many studies use the Z-transformation to standardise the data. The standardised values Z are calculated using the original indicator value X, the arithmetic mean X and the standard deviation s (Heyde, 1990):

$$Z = \frac{X - \bar{X}}{s}$$
(1)

However, Milligan & Cooper (1988) showed that this traditional Zvalue method leads to poorer results in cluster analyses than other standardisation methods. In most cases, the Z-value method works well only with normally distributed data (Office for National Statistics, 2015). The following calculation has proved to be the best method, which is also used in the present study (Milligan and Cooper, 1988):

$$Z = \frac{X - \min(X)}{\max(X) - \min(X)}$$
(2)

3.2. Exploratory factor analysis

An exploratory factor analysis serves to examine the data and reduce the number of (required) indicators. The j-th factor F_j can be determined using the k indicators $X_i, X_2, ..., X_k$ and the weights or factor loadings W_j :

$$F_j = W_{j1}X_1 + W_{j2}X_2 + \dots + W_{jk}X_k \tag{3}$$

The larger the factor load W_{ji} , the stronger the value of the factor F_j is determined by the indicator X_i (Aljandali, 2017). The R-function "fa" from the package "psych" is used here for factor analysis (Revelle, 2017). Factor analysis was conducted following the steps proposed in Osborne (2014), which are explained below.

3.2.1. Selection of the extraction method

An extraction method is used to investigate the correlation between all indicators with the aim of extracting the latent variables. A latent variable, here a factor, is a variable that cannot be measured directly but is the basis of the observed variables. If the data is predominantly normally distributed, then the maximum likelihood method is best suited as an extraction method, if it is not normally distributed, the principal axis factor method should be used (Osborne, 2014). Fig. 7 in the Appendix shows the distributions of the standardised indicators. The data were checked for normal distribution with the Kolmogorov-Smirnov test since the Shapiro-Wilk test is only suitable for data records with up to 5000 datasets (Shapiro and Wilk, 1965; Lopes, 2011). The pvalues were smaller than 2.2 ± 10^{-16} , so all data series are not normally distributed. Therefore, the principal axis factor method.

However, in the factor analysis using the principal axis factor method, so-called Heywood cases occurred. A Heywood case occurs when variances are negative, or correlations (in this case some factor loadings) are greater than one. Due to the Heywood cases, the solution of the factor analysis is inadmissible. In addition, the causes of Heywood cases are difficult to distinguish (Dillon et al., 1987). With the recommended extraction method in Revelle (2017), the "minimum residual" method, almost the same result was obtained as with the principal axis factor method, since only one indicator was assigned to a different factor. However, Heywood cases also occurred when using this extraction method. The Heywood cases were not discussed in Os (2014), so no other method was recommended for this case. Revelle (2017) states that in contrast to other methods, the "Minimum Rank Factor Analysis" (MRFA) does not include Heywood cases. Therefore, the MRFA method is selected below as the extraction method. The MRFA method is described in Lorenzo-Seva and Ferrando (2006) as the only method that calculates the part of the variance explained by each factor. This is also the only difference between this extraction method and the "minimum residual" method (Shapiro and Berge, 2002).

Utilities Policy 57 (2019) 75–96

3.2.2. Selection of the number of factors

In his study, Osborne (2014) indicates that no criterion for selecting the number of factors is better than another, the suitability of the criteria varies depending on the case. Therefore, several methods should be used. In this paper, the Kaiser criterion from Kaiser (1960) combined with a "Scree-Plot" and the "Parallel Analysis" from Hom (1965) are applied. Ten factors are recommended with the Kaiser method, and nine with the Parallel Analysis (cf. Fig. 8 in the Appendix). In the following, ten factors are assumed according to the Kaiser criterion (cf. curve "Eigenvalues > 0" in Fig. 8 in the Appendix).

3.2.3. Selection of the rotation method

The rotation was invented shortly after the factor analysis to facilitate the interpretation of the results of the factor analysis (Osborne, 2014). The goal is a simple structure in which each indicator describes as few factors as possible (or "loads onto them"). In addition, rotation creates groups of factors containing related indicators (Yong and Pearce, 2013). This analysis uses the "Varimax" method, which is widely used in practice, to maximise the variance of factor loadings and minimise the number of factors (Eckstein, 2016).

3.2.4. Results of the factor analysis

Table 3 shows the allocation of the indicators to the factors resulting from the factor analysis with the extraction method MRFA. The indicator X31 is the only one of the indicators not described by the factors, as its factor loading is very low for each factor. This means that X31 is no longer included in the further analysis. Fig. 9 in the Appendix shows the size of the factor loadings of all remaining indicators for each factor. The results can be assessed as plausible since each factor describes a specific issue (see column "Factor name" in Table 3). Fig. 10 in the Appendix also shows a correlation diagram of the indicator values. As an example, a high correlation between X29, X5 and X10 is shown there. These indicators are therefore all assigned to Factor 1 (see Table 3).

3.3. Cluster analysis

As already described above, high quality clusters are generated with hierarchical agglomerative cluster analysis. However, this method requires high computing times (Bouguettaya et al., 2015). In the cluster analysis implemented here, the high computing times were mainly due to the complex determination of the number of clusters. Similar to Wall (2016), the results of the factor analysis were used as input for the cluster analysis.

3.3.1. Ward algorithm

The clusters can be classified using distance metrics. To determine the distance matrix, the distance or similarity between all objects is determined (Johnson, 1967). The Ward algorithm is the only method among the agglomerative cluster methods that is based on the classical sum of squares and determines groups, minimising dispersion within the groups at each step. The sum of the squares is determined with the help of the distance matrix (Murtagh and Legendre, 2014). In this study, the distance matrix is calculated using the Euclidean distance, since it should be the basis for the Ward method (Miyamoto et al., 2015).

To use the Ward algorithm, the R function "hclust" has been executed (Müllner, 2016). Within this function, two different algorithms Ward1 or Ward2 can be selected. Murtagh and Legendre (2014) showed that only the algorithm Ward2 minimises the Ward criterion and should, therefore, be used. For more information about the mathematical differences of Ward1 and Ward2, the authors refer to Murtagh and Legendre (2014). The difference d^2 of two clusters R and Q is calculated with the help of the cluster foci \bar{x} using the following equation (Gentle et al., 1991):

Utilities Policy 57 (2019) 75–96

Table 3

Assignment of the indicators with their factor loadings to the ten factors and naming of the factors.

Factor	Indicators	Abbre-vations	Factor loading	Factor name
1	1) Household density	X5	0.917	Area factor (all indicators refer to the area of
	Share of settlement and transport area	X10	0.918	the municipality)
	3) Population density	X29	0.934	
	4) Technical PV potential per km ²	X35	0.921	
	5) Share of forest and agricultural area	X38	-0.768	
2	1) Income per household	X7	0.464	East/West Factor (this factor reflects the
	2) Unemployment rate	X9	-0.503	inequalities between West and East Germany)
	Share of buildings built before 1919	X14	-0.629	
	Share of buildings built between 1919 and 1949	X15	-0.791	
	Share of buildings built between 1960 and 1969	X17	0.587	
	Share of buildings built between 1970 and 1979	X18	0.791	
	7) Share of buildings built between 1980 and 1989	X19	0.560	
3	1) Achievable hydrothermal temperature	X32	0.949	Hydrothermal factor
	Necessary hydrothermal drilling depth	X33	0.937	
4	1) Number of motor vehicles per 1000 inhabitants	X27	0.857	Traffic factor
	2) Number of cars per 1000 inhabitants	X28	0.882	
5	 Share of over 65-year-olds 	X8	-0.726	Age factor
	2) Share of buildings built between 1990 and 1999	X20	0.737	
	3) Share of buildings built between 2000 and 2005	X21	0.529	
	4) Share of 18-64-year-olds	X30	0.626	
6	 Share of buildings with heating systems based on district heating 	X11	-0.939	Heating system factor
	2) Share of buildings with heating systems not based on district heating	X12	0.934	
7	1) Population development between 2010 and 2015	X1	0.459	Population Factor (all indicators depend on
	2) Living space per person	X2	-0.660	population size)
	3) Average household size	X4	0.811	
	4) Technical PV potential per person	X34	-0.555	
8	1) Share of buildings built between 1950 and 1959	X16	0.891	-
9	1) Share of single-person households	X3	0.436	Building factor
	2) Share of owner-occupied apartments	X6	-0.550	-
	3) Share of detached houses	X23	-0.863	
	4) Share of semi-detached houses	X24	0.663	
	5) Share of terraced houses	X25	0.719	
10	1) Technical wind potential per inhabitant	X36	0.823	Wind factor
	2) Technical wind potential per km ²	X37	0.754	

 $d^{2}(R, Q) = \frac{2|R| |Q|}{|R| + |Q|} \bar{x}(R) - \bar{x}(Q)^{2}.$

(4)

3.3.2. Determining the number of clusters

In hierarchical agglomerative cluster analysis, the number of clusters is not known in advance but must be determined using suitable methods (Salvador and Chan, 2004). The more clusters selected, the more similar the objects within the clusters are. At the same time, the clusters are more difficult to distinguish between each other as the number of clusters increases.

In some studies such as Wall (2016) or Yang et al. (2017), the number of clusters is estimated using the common but often inaccurate "elbow" method. Alternatively, the R-function "NbClust" from Charrad et al. (2014) offers 30 methods for determining the optimal number of clusters. None of the criteria studied so far can predict the optimal number of clusters in any case (Albatineh and Niewiadomska-Bugaj, 2011). Therefore, all 30 methods were implemented. More information about the mathematical description of the methods can be found in Charrad et al. (2014). The results of the procedures in the context of this study are shown in Table 9 in the Appendix. Only 26 of the 30 methods are listed in the table since the computationally intensive methods such as "gamma" had to be aborted after almost two months of computing time. As can be seen in Table 9 in the Appendix, the 26 methods yielded quite different values for the number of clusters. It is therefore necessary to examine more closely whether the methods should be used at all in this particular case. 22 of the 30 procedures are already explained and evaluated in Milligan and Cooper (1985). For Example, the "ch" procedure of Calinski and Harabasz (1974) was rated as the best procedure. However, Islam et al. (2016), showed that "ch" is poor with a high number of clusters and usually prefers - as in this study - a 2 cluster solution. Almost all 26 methods have poor functionality with a high number of clusters (cf. Table 9 in the Appendix). The only algorithm for which a good functionality with high cluster numbers could be found in the relevant literature is "duda", which suggests ten clusters in this study. However, it should be further examined whether the ten clusters represent the optimal number of clusters in this study.

Therefore, several cluster solutions with different numbers of clusters are compared to determine an appropriate number of clusters. Table 4 shows how the structure of the clusters changes from five clusters up to 15. For example, Cluster 1 from the 5 cluster solution divides into two further clusters at 14 cluster solution.

Fig. 2 shows the course of the within-cluster sum of squares as a function of the number of clusters. The within-cluster sum of squares describes the squared distance of an object to the cluster centre, i. e. how similar the object is to the other objects of the group (Anderson, 2001). The smaller the within-cluster sum of squares is, the more similar the objects in the clusters are.

The aforementioned elbow method is based on the within-cluster sum of the squares, as shown in Fig. 2, where the elbow represents the point of decreasing marginal returns. This means that right behind the elbow, with an increase in the number of clusters, the increase in information is very small. However, the region of the elbow is often not as clearly visible, as in Fig. 2 (Kodinariya and Makwana, 2013), so this method alone could not be used. The elbow method is after all only a heuristic one (Tibshirani et al., 2001). The elbow could be between 5 and 20 clusters in the area delimited by black dotted lines. The 10 cluster solution proposed by the "duda" method (see Table 9 in the Appendix), is also in this area (red dotted lines).

Therefore, the clusters need to be analysed further. It emerged that the new clusters formed in the 11 cluster solution differed significantly less from each other than the clusters formed in the previous steps. The

Development of cluster composition for solutions with 5–15 clusters.



upper diagram of Fig. 3 illustrates the deviation in the mean values of all indicators for the two new clusters in the 11 cluster solution. The values have been scaled to values between 0 and 1 to improve the comparability. The two new clusters are clusters 5 with 3927 municipalities and 8 with 1335 municipalities, as the cluster numbers change in each step (cf. Table 4). The diagram shows that the mean values for each indicator are approximately the same.

Table 4

This means that a further separation of the clusters from ten clusters onwards creates only a low added value. As a comparison, the curves of the mean values of the two newly created clusters in the 10 cluster solution are shown in the bottom diagram of Fig. 3. In this case, the mean values vary significantly, so the number of clusters should be increased from nine to ten. In the following, ten clusters will be selected as the appropriate number of clusters, since this number can be justified by the "duda" method, the elbow method and further analysis.

4. Results of the cluster analysis

Fig. 4 shows all German municipalities with a colour assignment to the clusters of the 10 cluster solution. The broader outlines separate the 16 federal states in Germany. Especially in Rhineland-Palatinate and Schleswig-Holstein, some municipalities seem to be dark to black. This is due to the small size of the municipalities; in Rhineland-Palatinate, the municipalities have by far the smallest size. Due to the poor visibility of these municipalities, the map is magnified in Figs. 11–13 in the Appendix.

The mean values of all 34 indicators were determined for all clusters and every single cluster (see Table 5). The different colours are chosen to distinguish between the sectors Private Households (blue/red) and Transport (yellow) as well as potential for renewable energies (green).

The following description of the clusters is based on the mean values in Table 5. To help classify the clusters, the proportions of municipalities per cluster are assigned to the seven municipality types of the BBSR typology in Fig. 5 (BBSR, 2015). The criteria for classifying the municipalities are the population and the central function of the municipality. The evaluation of the central function is based on the central place theory of Christaller (1980). A municipality is defined as a rural municipality if either the population is less than 5000 inhabitants or if the municipality has no basic central function. The cities in the BBSR typology are classified according to population size with the lower limits of 5,000, 10,000, 20,000, 50,000, 100,000 and 500,000 inhabitants.

Cluster 1 contains an above-average number of larger towns (see Fig. 5). This cluster is characterised by the highest share of district heating systems by far. This is obvious since district heating networks are particularly suitable in towns and conurbations with high heat demand densities (Connolly et al., 2014). The high proportion of over 65-year-olds is also typical of German cities (Lauf et al., 2016). The population density is above average, while vehicles per 1000 inhabitants are the second lowest. The potential for renewable energies is below average except for the mediocre wind power potential.

The largest share of cities is in **Cluster 2** (see. Fig. 5). In this cluster, the rural municipalities account for the smallest share compared to the other clusters, and the cities from the larger small town to the larger cities take the highest share. Fig. 4 shows that the largest cities in Germany, such as Berlin, Hamburg, Munich and Cologne, are all part of this cluster. For this reason, the indicators household density, population density as well as the shares of terraced houses and semi-detached houses are particularly high in this cluster, and the share of detached nouses is particularly low. Furthermore, buildings built between 1950 and 1979 dominate the municipalities in this cluster. This is due to the destruction of many cities during the Second World War. In the city of Dresden, for example, large areas of prefabricated concrete slab buildings were created in the 1970s due to a shortage of housing (Wurm et al., 2009). The number of vehicles per 1000 inhabitants is the lowest, as there are more transport alternatives in cities and the average









Cluster 4 Cluster 8

Fig. 3. The mean values of the two newly formed clusters in the 11 cluster solution (upper figure) and the 10 cluster solution (bottom figure) over the 34 indicators.

distances travelled are shorter because of the high population density (Woldeamanuel et al., 2009). Due to the high building density, the technical PV potential per km² is the highest here. On the other hand, the technical PV potential per inhabitant is the lowest after Cluster 10 due to the high population density. As expected, the proportion of forest and agricultural land in this cluster is the smallest, so that the technical wind power potential is also very low. The geothermal potential is below average.

In **Cluster 3**, the hydrothermal potential is very high; an average hydrothermal temperature of 90 °C at a depth of 2400 m can be used in the municipalities. Fig. 4 also shows that the municipalities of this cluster are predominantly located in the three large German hydro-thermal regions "North German Basin", "Upper Rhine Graben" and "South German Molasse Basin" (Agenar et al., 2014). The potentials for the other renewable energies are average. Furthermore, there are more modern detached houses in the municipalities of the cluster, the income per household is high, and the unemployment rate is particularly low. From this cluster onwards, the share of rural municipalities in each cluster is more than 45%, and larger cities from the small midtown onwards are only represented to a small extent (see, Fig. 5).

In Fig. 4, Cluster 4 is represented by dark green coloured municipalities and occupies a large, almost continuous area. A closer look reveals that the western border of the area corresponds to the border of the former German Democratic Republic (GDR). The municipalities from Cluster 6 and 8 are also predominantly located in the territory of the former GDR. Cluster 4 is characterised by a high proportion of old houses, and the proportion of buildings built between 1919 and 1949 reaches its maximum here. Buildings built between 1970 and 1989 are very scarce in these municipalities. Also, the unemployment rate is particularly high. In line with this, the population in these municipalities has been declining the most in recent years, and income per household is the lowest. The sharp decline in the population is due to the growing childlessness in eastern Germany since German reunification (Bernardi and Keim, 2017). At the same time, population density and average household size are the lowest in this cluster. These two latter indicators also determine the high value of photovoltaic potential per inhabitant. On the other hand, the photovoltaic potential per km² in this cluster is the second lowest after cluster 9, due to the small share of settlement areas in the total area. In contrast to this, the wind power potential in this cluster is the second highest. In addition, the municipalities of the cluster could exploit the second highest hydrothermal potential in Germany, but this would require drilling 300 m deeper on average than in Cluster 3.

With 47% of all German municipalities, **Cluster 5** contains the largest number of municipalities. In contrast to Cluster 1, district heating systems are the least widespread in this case, whereas the proportion of heating types that are not based on district heating is the highest. The number of cars and motor vehicles per 1000 inhabitants is also very high. The cluster has the second lowest hydrothermal potential. In addition, the potential for photovoltaics in this cluster is only mediocre, and the potential for wind power is low. Based on the indicators selected in this study, Cluster 5 represents the "average" municipalities in Germany.

Like Cluster 4, **Cluster 6** is also characterised by a high building age because of its location in eastern Germany. Due to the increasing childlessness, the population is declining and the proportion of people over 65 years of age is steadily increasing. The values of the indicators representing the transport sector are rather average. In contrast to the low to average wind power and photovoltaic potential, the hydrothermal potential in this cluster is particularly low.

Cluster 7 contains almost exclusively rural municipalities and small towns (see Fig. 5). The proportion of apartments occupied by the owner and the living space per person are at their maximum, while at the same time the household density is minimal. Due to the low density of households and population, the number of cars and motor vehicles per 1000 inhabitants reaches its maximum here. In addition, the detached



Fig. 4. Illustration of all German municipalities with their allocation in the 10 cluster solution. The numbers of municipalities in the clusters are in parentheses.

houses reach the largest share in this cluster. Furthermore, this cluster has the highest potential for renewable energies, despite its very low geothermal potential. The high living space per person and the low density of households mean that the highest photovoltaic and wind power potential per km² is at its maximum, as most of the municipalities in the cluster are located in Northern Germany and thus in areas with high wind speeds and have a high proportion of forest and agricultural land. The building age in Cluster 8 is unusually low, although these

municipalities are mainly located in Eastern Germany. This can be explained by an example: in the description of Clusters 4 and 6, the decline in population in eastern Germany has already been discussed. Although this development applies to all the new federal states, the decline in Brandenburg between 1990 and 2008 was significantly lower. This was mainly due to new settlements in the surrounding area of Berlin, the so-called "commuter belt" (Jesse et al., 2014). Municipalities from Cluster 8 almost exclusively form this commuter belt (cf. Fig. 4). Due to the rising rent in Berlin, more and more young families

 Table 5

 Mean values of the indicators X1-X38 for the ten Clusters and all Clusters.

Indicator	X1	X2	X3	X4	X5	X6	X7	X8	X9	X10	X11	X12
Mean value	-1.0	51.2	27.1	2.4	81.1	63.0	34.2	20.1	3.7	13.0	2.7	96.3
Cluster 1	-1.3	47.6	31.2	2.3	118.9	51.4	30.3	21.4	5.6	16.6	20.0	78.9
Cluster 2	1.1	46.7	34.1	2.3	417.5	49.9	37.8	21.1	4.4	34.9	3.8	95.8
Cluster 3	1.1	50.5	25.8	2.5	71.3	63.3	41.3	18.8	2.9	12.7	2.1	97.2
Cluster 4	-4.3	49.5	27.4	2.2	23.5	61.7	24.9	20.2	6.9	6.7	3.3	95.9
Cluster 5	-0.8	53.3	26.7	2.4	59.9	65.2	35.7	19.7	3.0	12.3	1.5	97.4
Cluster 6	-3.7	47.2	27.2	2.3	54.6	62.4	26.2	22.5	4.7	10.1	2.5	96.5
Cluster 7	-2.1	58.0	24.3	2.4	21.4	69.7	32.8	20.2	3.3	7.7	1.9	96.9
Cluster 8	-0.9	48.6	23.4	2.4	40.7	65.9	33.9	17.3	3.7	11.4	4.9	94.6
Cluster 9	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	3.9	2.0	90.7
Cluster 10	33.5	21.3	28.3	6.2	37.7	62.5	57.6	20.9	4.6	9.5	2.9	96.5

Indicator	X14	X15	X16	X17	X18	X19	X20	X21	X23	X24	X25
Mean value	18.9	11.4	8.5	11.3	13,4	11.1	14.6	7.5	77.2	10.2	7.8
Cluster 1	17.5	14.0	9.0	10.9	12.3	10.0	14.6	8.1	68.0	12.3	15.1
Cluster 2	10.0	10.9	12.1	16.4	16.3	11.8	12.0	6.8	58.3	17.5	20.8
Cluster 3	14.3	7.6	7.6	12.4	16.8	12.6	15.9	8.6	76.7	11.5	6.2
Cluster 4	28.1	20.5	11.8	5.4	5.1	7.0	13.4	6.6	72.6	15.7	7.6
Cluster 5	15.2	8.9	9.0	13.5	15.7	12.4	14.4	7.5	80.9	8.5	6.2
Cluster 6	35.2	19.9	5.4	4.7	6.5	8.5	12.6	5.0	75.8	9.1	9.7
Cluster 7	28.5	9.4	6.4	11.4	12.8	8.8	12.2	7.6	85.2	4.7	2.9
Cluster 8	18.6	10.9	4.0	4.3	6.4	8.2	29.2	13.6	78.6	10.5	6.7
Cluster 9	15.4	8.3	8.5	13.2	14.8	11.0	12.8	6.5	74.0	9.4	6.0
Cluster 10	26.0	16.6	8.2	10.4	10.3	8.8	12.0	5.4	79.5	11.4	5.1

Indicator	X27	X28	X29	X30	X32	X33	X34	X35	X36	X37	X38
Mean value	832.7	634.5	183.3	62.8	29.6	842.8	2482.7	398.4	41.4	2000.9	83.6
Cluster 1	785.7	598.5	273.0	62.4	44.5	1342.1	2173.0	501.4	44.9	2062.8	77.2
Cluster 2	697.2	588.3	928.8	62.0	21.6	552.4	1987.0	1703.7	1.1	605.7	60.9
Cluster 3	855.0	649.7	171.0	62.4	90.8	2438.4	2390.4	389.8	33.0	2327.2	83.7
Cluster 4	819.9	618.2	33.5	65.2	88.1	2740.0	2755.9	88.0	120.4	2759.8	88.2
Cluster 5	860.6	652.5	141.4	62.5	5.6	156.9	2570.9	342.4	24.0	1737.4	84.8
Cluster 6	800.5	615.0	114.7	63.6	3.4	106.9	2477.5	260.0	34.1	1704.4	86.7
Cluster 7	957.1	661.5	41.7	61.3	25.3	789.4	2818.0	113.7	222.6	5863.6	89.9
Cluster 8	819.9	645.4	104.5	66.7	69.8	2086.4	2486.1	247.2	32.0	1897.8	83.8
Cluster 9	0.0	0.0	0.0	0.0	19.4	592.5	0.0	7.0	0.0	320.1	90.0
Cluster 10	711.9	560.1	86.3	63.4	41.5	1297.3	1285.5	211.1	46.8	2123.2	84.1



Fig. 5. Classification of municipalities according to the BBSR municipality typology.

⁸⁵

are moving into the commuter belt. This also explains the maximum proportion of 18-64-year-olds and the minimum proportion of 65 yearolds in this cluster (Bünger, 2017). The proportion of cars per 1000 inhabitants is also above average here, presumably because most people have to drive to work in the city. A closer examination of the red municipalities shown in Fig. 4 reveals that most of the municipalities are located in the surrounding area of major cities in clusters 1 and 2. Thus, the conclusions mentioned above on the Berlin "commuter belt" can also be transferred to the other municipalities in Cluster 8. Also, this cluster has the third highest geothermal potential, while the potential of the other renewable energies is below average.

Cluster 9 contains all areas in which there are no inhabitants. Therefore, all indicators that depend on the population have a value of zero. These areas are municipality-free (in German: "gemeindefrei"), and therefore 100% of them are rural municipalities (see Fig. 5). Settlement and traffic area is present in these municipalities, because of roads leading through these areas. However, this indicator has the smallest value here. At the same time, the proportion of forest and agricultural land reaches its highest level. It is interesting to note that the technical wind power potential per km2 is nevertheless at its minimum in this cluster. The reason for this could be, among other things, nature reserves in which no wind turbines may be installed. The technical photovoltaic potential in this cluster also approaches zero since only a few buildings are located here. Despite the buildings, no residents are assigned to these areas, as the buildings in the municipal areas belong to military training areas or similar (Goderbauer, 2016). This cluster has the lowest potential for renewable energies, as the geothermal potential is also below average.

With only 33 municipalities, Cluster 10 represents the smallest cluster in this study. This cluster is characterised by the highest population growth between 2010 and 2015. Due to the largest average household size by far, the income per household is also reaching its maximum value and the technical PV potential per inhabitant its minimum value. In addition, the number of vehicles per 1000 inhabitants in this cluster is below average. The cluster must be evaluated as an outlier since many of the characteristics of this cluster are due to the high population growth. The population figures from 2015 have been used in the calculation of many indicators to establish a uniform reference. However, the most recent household data are available for 2014 and have only been roughly updated since the last survey in 2011. As a result, the high population growth leads to, among other things, high values for the average household size, as the number of households is no longer up to date. This cluster, therefore, includes outliers. Nevertheless, the heterogeneity and independence of the cluster can be justified by the significantly higher population growth as in the other clusters. The distinguishing characteristics of the ten clusters are summarised in Table 6

The data and results of the cluster analysis can be made available upon request.



Fig. 6. Violin Plot of indicator X11 in % for the ten clusters.

5. Discussion

5.1. Critical appraisal of the methodology

Wall (2016) shows that factor analysis is an important step ahead of cluster analysis. However, most studies describe cluster analyses without prior factor analysis. For this reason, the cluster analysis was repeated again without the factor analysis. The results were worse than those of the cluster analysis with the values from the factor analysis. For example, with the factor values, the 75 municipality-free areas without population (Cluster 9) were already divided into a cluster in the 6 cluster solution (cf. Table 4). In the analysis with the raw data, these municipalities were not separated, at least up to the 20 cluster solution.

Whilst cluster analysis provides a good basis for transferring the results of energy autonomy studies to other municipalities, results cannot always be completely transferred and an examination of the individual case will be necessary. This is illustrated by the following example: Fig. 6 shows the violin plot of the indicator "Share of buildings with heating systems based on district heating (X11)" for all clusters. In a violin plot, the density trace and the box plot are combined into one diagram (for more information see: Hintze and Nelson, 1998). The red plus signs indicate the position of the mean value and the green boxes indicate the position of the median.

Cluster 1 is characterised by a high average share of district heating. However, Fig. 6 shows that this cluster also contains a few municipalities with very low shares of district heating. These municipalities are then more similar to the cluster focus in the other indicators.

 Table 6

 Summary of the distinguishing characteristics of the ten clusters.

Cluster	Number of municipalities	Characteristics
1	339	Larger towns with highest share of district heating.
2	727	All major German cities with particularly low potential for renewables.
3	1638	Municipalities with highest hydrothermal potential, high income per household as well as low unemployment rate.
4	839	Municipalities with high hydrothermal potential, building age and unemployment rate.
5	5262	"Average" Cluster containing the majority of municipalities. Municipalities with high number of cars per 1000 inhabitants and very low
		share of district heating.
6	1370	Municipalities with high building age and high proportion of people over 65 years of age.
7	460	Municipalities with lowest household density, highest number of cars and motor cycles per 1000 inhabitants, largest share of detached
		houses and particularly high potential for renewables.
8	388	Municipalities with low building age, lowest proportion of people over 65 years of age and a high hydrothermal potential.
9	75	Rural municipality-free areas with no inhabitants and lowest potential for renewables.
10	33	Smallest cluster containing municipalities with high population growth.

Table 7

Comparison of the model, exa	mple and average municipalities/1	egions from relevant literature	e with the average municipal	ity from this study.	The cells c	ontaining
"n/a" show indicators that we	ere not specified in the sources.					

Municipality from	Number of inhabitants	Number of buildings	Average household size [people]	Share of settlements and traffic areas [%]	Population density [Inhabitants/km²]	Number of vehicles per 1000 inhabitants
Jenssen et al. (2014)	3000	800	2.2	n/a	n/a	n/a
Scheffer (2008)	10,000	n/a	n/a	n/a	n/a	630
Peter (2013)	3850	1224	3.1	8	106	n/a
Burgess et al. (2012)	25,550	n/a	2.4	8	310	n/a
Schmidt et al. (2012)	20,619	n/a	n/a	< 11	68	n/a
Woyke and Forero (2014)	1100	n/a	n/a	n/a	n/a	n/a
Average munici-pality (cf. Table 5)	7380	1670	2.4	13	183	830

The following explains in more detail why it is necessary to carry out a cluster analysis to appropriately transfer the results of energy system analyses in municipalities by comparing the model or average municipalities from the studies described in Section 1 with the mean values of all municipalities in this study (cf. Table 7).

Table 7 shows that the average municipality differs from the municipalities/regions of the studies of Jenssen et al. (2014), Scheffer (2008) and Peter (2013). A comparison of the values reveals that none of the surveyed municipalities represents an average municipality in Germany. Even though it was not the intention to select an average municipality in Germany in some of the studies, the results are difficult to transfer to other municipalities or regions. Rather, the choice of municipalities and data appears to be influenced by many other than technical factors in some cases. For example, Scheffer (2008) tried to use the indicator values to describe a rural municipality. In the classification of German municipalities in BBSR (2015), the municipality would be placed in the category "Smaller Small Town". In addition, only a few data on the municipalities are described in the studies. This could give the impression that the results from Jenssen et al. (2014), for example, can be transferred to municipalities with 3000 inhabitants, 800 buildings and a household size of 2.2 persons. Instead, a transfer ability depends on how precisely the municipality is represented, i.e. how many indicators are used to describe the municipality. Since the cluster analysis conducted in our study uses considerably more indicators to describe the municipalities, the result can be used as a basis for transferring appropriate energy systems to other municipalities.

5.2. Suitability of municipalities for energy autonomy

To investigate the suitability of individual municipalities and clusters for energy autonomy, precise calculations must be performed. Nevertheless, in this section an attempt is made to determine an initial assessment of this suitability by analysing the clusters in which municipalities are already aiming for energy autonomy. For this, 165 municipalities from the energy projects "Energy Municipalities", "Bioenergy Villages" and "100% Renewable Energy Regions" are assigned to the ten clusters. These projects aim to achieve the goal of an autonomous energy supply in the municipalities. However, the municipalities have defined different objectives in the projects about autonomous supply, including "100% heat", "100% electricity" or "100% renewable energies" (McKenna et al., 2014b). Some of the municipalities take part in several of the projects mentioned above. Districts involved in the projects were not included in the analysis. The result of the assignment is shown in Table 8.

First of all, it is noticeable that no municipalities from Clusters 9 and 10 participate in the energy projects. This fact is quickly explained since there is no population in Cluster 9 and Cluster 10 is very small and is also more of an outlier cluster. The first two clusters, on the other hand, have the largest proportion of municipalities that are members of the energy projects. As shown above, these clusters contain most of the cities (see Fig. 5). One reason for the high proportions in these clusters could be the existence of a critical mass of innovators (cf. Deutschle et al., 2015), but such aspects could not be included in this analysis. On the other hand, however, achieving the goal of energy autonomy is all the more difficult, the more inhabitants a municipality has. While in rural municipalities the focus is often on the expansion of renewable energies, development in large cities depends to a large extent on development outside the city borders. Discussions in major cities are mainly focused on increasing energy efficiency, creating smart grids and providing storage capacity. For example, the City of Munich aims to halve per capita emissions by 2030 (Gailing et al., 2013). In the other clusters, the municipalities' participation in energy projects is not so pronounced. However, in each of the clusters 3 to 8, at least two municipalities are involved in energy projects. This is likely to be due to the non-technical reasons mentioned above rather than a suitability of the municipalities per se. If the potential for renewable energies is used as a basis for the assessment, the municipalities from Cluster 3 and Cluster 7 could be particularly suitable for energy autonomy: Cluster 3, among other things, due to its high hydrothermal potential and the associated potential base-load energy supply; Cluster 7 because of the highest potential for renewable energies.

6. Summary and conclusion

In the context of the trend towards decentralised energy systems, both high temporal and spatial resolutions are required in order to adequately consider their interactions with the centralised system. This is a central challenge in energy modelling, as compromises must inevitably be made between model resolution, scope and computational feasibility. This paper makes a significant contribution to complexity reduction in this area by clustering the 11,131 German municipalities using 34 pre-identified socio-energetic indicators, mainly based on freely available data relating to the consumption sectors of Private Households and Transport, as well as indicators relating to the

Т	а	b	ŀ	e
-	u	v		-

Cluster	1	2	3	4	5	6	7	8	9	10
Number	25	21	20	6	76	12	2	3	0	0
Fraction in the cluster	7.4%	2.9%	1.2%	0.7%	1.4%	0.9%	0.4%	0.8%	0%	0%
Example	Jühnde	Munich	Furth	Barth	Brilon	Jena	Hürup	РІеβ	-	-

potentials for renewable energies. The method involves two main steps, namely a factor analysis and a cluster analysis. For the former, different methods are weighed against each other, and the most effective methods for allocating the indicators to factors are chosen. Selected cluster validation methods are then used to determine an appropriate number of 10 clusters to which the 11.131 municipalities are distributed. Due to the high number and differentiation of indicators, clusters overlap with each other for different indicators, but the results also show significant differences between the clusters. For example, Cluster 2 contains all major German cities and most of the other cities in Germany and has a low potential for renewable energies. Cluster 9, on the other hand, describes all German municipalities in which there are no inhabitants.

The methodology used in this study could be improved for more accurate results in future work. On the one hand, other indicators should be included in the study, including, if possible, indicators from the Industry and Commercial sector as well as indicators relating to the local climate. However, this is challenging due to the lack of available data at this spatial resolution. If available, data should also refer to the same year, as some of the results might be distorted because of different reference years, as shown by the average household size in cluster 10. Furthermore, weights for the indicators should be determined with the help of expert interviews. If it is known which indicators have the most considerable influence on the suitability for energy autonomy, these can be weighted more strongly in the cluster analysis and a new set of clusters generated based on these weights. In addition, the employed cluster methodology should be scrutinised more closely. Although the selection methods can be adequately justified in this study, others (e.g. Chicco, 2012) have shown that the Ward algorithm is not always the best choice for cluster analysis. Further work is also required to analyse the economic effects of municipal energy autonomy on the overarching energy system (for a discussion see Jägemann et al., 2013, McKenna, 2018).

A comparison of the average municipality from the dataset used here with the average municipalities from other energy autonomy studies is difficult due to a lack of data at the level of detail employed

A. Appendix

Table 8

Indicators for cluster analysis and the associated units and references

Utilities Policy 57 (2019) 75-96

here. Based on the available data in these studies, a comparison shows few similarities, which means the results of the studies relating to their transferability to other municipalities should be questioned. Assigning the municipalities from the three German energy projects "Energy Municipalities", "Bioenergy Villages" and "100% Renewable Energy Regions" to the 10 clusters further shows that in eight of the ten clusters municipalities are aiming for energy autonomy (in varying degrees). As a result, it is not possible to differentiate between the clusters regarding readiness for such energy projects, which is most likely due to the influence of non-technical factors on the emergence of these initiatives. However, the results of the cluster analysis show that some of the municipalities could be technically more suitable for energy autonomy, for example Cluster 7 is characterised by a high potential for renewable energies. A comparison of the ten clusters with the average municipality from the data set also demonstrates their benefit, with a large variation across clusters in terms of energy demand structure, renewables potentials and overall size. The results therefore reduce the effort of subsequent studies, as only a few municipalities from the clusters need to be examined regarding their suitability for energy autonomy to be able to make statements for all municipalities of the cluster. However, this study also makes it clear that not every result can be transferred to all the other municipalities within a cluster, instead an individual examination is required for each municipality. Nevertheless, the results help to identify municipalities in which already successful measures from other municipalities could be applied, and provide a basis for further energy analyses at the national level.

Acknowledgements

The authors gratefully acknowledge funding by the German Federal Ministry of Education and Research (BMBF) within the Kopernikus Project ENSURE 'New ENergy grid StructURes for the German Energiewende' (funding reference: FKZ 03SFK1N0) as well as the financial support of the PhD College "Energy and Resource Efficiency" (ENRES), from the Federal State of Baden-Wuerttemberg.

Indicator	References
Indicators of the Consumption Sector Private Households	
Population development between 2010 and 2015 [%]	(Statistisches Bundesamt, 2011a; 2017b) and own calculation
Living space per person [m ²]	(Statistisches Bundesamt, 2017b; 2014d) and own calculation
Share of single-person households in total number of households [%]	(Statistisches Bundesamt, 2014e) and own calculation
Average household size [Number of persons]	(Statistisches Bundesamt, 2014e; 2017b) and own calculation
Household density [Households per km ²]	(Statistisches Bundesamt, 2017b; 2014e) and own calculation
Share of owner-occupied apartments in total number of apartments [%]	(Statistisches Bundesamt, 2014d) and own calculation
Income per household [k€]	(Statistisches Bundesamt, 2014e; 2011b) and own calculation
Share of over 65-year-olds in total population [%]	(Statistisches Bundesamt, 2014a) and own calculation
Unemployment rate [%]	(Statistisches Bundesamt, 2017b; 2014a, 2016a) and own calculation
Share of settlement and traffic area in total area [%]	(Statistisches Bundesamt, 2016b) and own calculation
Heating days (long-term average)	Institut für Wohnen und Umwelt (2017)
Heating degree days (long-term average)	Institut für Wohnen und Umwelt (2017)
Degree day number (long-term average)	Institut für Wohnen und Umwelt (2017)
Share of buildings with heating systems based on district heating [%]	(Statistisches Bundesamt, 2014d) and own calculation
Share of buildings with heating systems not based on district heating [%]	(Statistisches Bundesamt, 2014d) and own calculation
Share of buildings without heating system [%]	(Statistisches Bundesamt, 2014d) and own calculation
Share of buildings built before 1919 in total building stock (X14) [%]	(Statistisches Bundesamt, 2014d) and own calculation
Share of buildings built between 1919 and 1949 in total building stock (X15) [%]	(Statistisches Bundesamt, 2014d) and own calculation
Share of buildings built between 1950 and 1959 in total building stock (X16) [%]	(Statistisches Bundesamt, 2014d) and own calculation
Share of buildings built between 1960 and 1969 in total building stock (X17) [%]	(Statistisches Bundesamt, 2014d) and own calculation
Share of buildings built between 1970 and 1979 in total building stock (X18) [%]	(Statistisches Bundesamt, 2014d) and own calculation
Share of buildings built between 1980 and 1989 in total building stock (X19) [%]	(Statistisches Bundesamt, 2014d) and own calculation
Share of buildings built between 1990 and 1999 in total housing stock (X20) [%]	(Statistisches Bundesamt, 2014d) and own calculation
Share of buildings built between 2000 and 2005 in total building stock (X21) [%]	(Statistisches Bundesamt, 2014d) and own calculation
Share of buildings built from 2006 onward in total building stock (X22) [%]	(Statistisches Bundesamt, 2014d) and own calculation
	(continued on next name

ed on next page;

Utilities Policy 57 (2019) 75-96

Indicator	References
Share of detached houses in total building stock (X23) [%]	(Statistisches Bundesamt, 2014d) and own calculation
Share of semi-detached houses in total building stock (X24) [%]	(Statistisches Bundesamt, 2014d) and own calculation
Share of terraced houses in total building stock (X25) [%]	(Statistisches Bundesamt, 2014d) and own calculation
Share of "other building types" in total building stock (X26) [%]	(Statistisches Bundesamt, 2014d) and own calculation
Indicators of the Transport consumption sector	
Number of motor vehicles per 1000 inhabitants	(Statistisches Bundesamt, 2017b; Kraftfahrt-Bundesamt, 2017a) and own calculation
Number of cars per 1000 inhabitants	(Statistisches Bundesamt, 2017b; Kraftfahrt-Bundesamt, 2017a) and own calculation
Share of diesel vehicles in vehicle fleet [%]	(Kraftfahrt-Bundesamt, 2017b) and own calculation
Share of petrol vehicles in vehicle fleet [%]	(Kraftfahrt-Bundesamt, 2017b) and own calculation
Share of gas vehicles in vehicle fleet [%]	(Kraftfahrt-Bundesamt, 2017b) and own calculation
Share of hybrid vehicles in vehicle fleet [%]	(Kraftfahrt-Bundesamt, 2017b) and own calculation
Share of electric vehicles in vehicle fleet [%]	(Kraftfahrt-Bundesamt, 2017b) and own calculation
Share of "other vehicle type" in vehicle fleet [%]	(Kraftfahrt-Bundesamt, 2017b) and own calculation
Population density [Inhabitants per km ²]	(Statistisches Bundesamt, 2017b) and own calculation
Share of 18-64-year-olds in the total population [%]	(Statistisches Bundesamt, 2014a) and own calculation
Share of commuters in total workforce [%]	(Statistisches Bundesamt, 2014f) and own calculation
Indicators of the Consumption Sector Industry and Commercial	
Share of employment in the industrial sector [%]	(Statistisches Bundesamt, 2015a) and own calculation
Share of employment in the commercial sector [%]	(Statistisches Bundesamt, 2015a) and own calculation
Energy productivity of manufacturing industry [€/GJ]	(Statistisches Bundesamt, 2014b; 2014c) and own calculation
Energy intensity of manufacturing industry [MJ/€]	(Statistisches Bundesamt, 2014b; 2014c) and own calculation
Productivity level of manufacturing industry [€/GJ]	(Statistisches Bundesamt, 2014b; 2014c) and own calculation
Specific energy consumption of manufacturing industry [MJ/€]	(Statistisches Bundesamt, 2014b; 2014c) and own calculation
Share of industrial sales tax in total sales tax [%]	(Statistisches Bundesamt, 2014g) and own calculation
Share of commercial sales tax in total sales tax [%]	(Statistisches Bundesamt, 2014g) and own calculation
Development of employment share in the industrial sector [%]	(Statistisches Bundesamt, 2015a, 2000) and own calculation
Development of employment share in the commercial sector [%]	(Statistisches Bundesamt, 2015a, 2000) and own calculation
Development of energy intensity in manufacturing industry from 2003 to 2014 [%]	(Statistisches Bundesamt, 2014b; 2014c, 2003a; 2003b) and own calculation
Number of manufacturing enterprises per 1000 households	(Statistisches Bundesamt, 2015b) and own calculation
Indicators of the potential for renewable energies	
Achievable hydrothermal temperature [°C]	(Agemar 2017) and own calculation
Necessary hydrothermal drilling depth [m]	Own calculation
Technical PV potential per inhabitant [kWh/y]	(Mainzer et al., 2014; Statistisches Bundesamt, 2017b)
Technical PV potential per km ² [MWh/y]	(Mainzer et al., 2014; Statistisches Bundesamt, 2017b)
Technical wind potential per inhabitant [MWh/y]	(McKenna et al., 2014a; Statistisches Bundesamt, 2017b) and own calculation
Technical wind potential per km ² [MWh/y]	(McKenna et al., 2014a; Statistisches Bundesamt, 2017b) and own calculation
Share of forest and agricultural land in total area [%]	(Statistisches Bundesamt, 2016b) and own calculation

Table 9 Number of clusters resulting from 26 different procedures and evaluation of the procedures. A high number of clusters in the "evaluation of the procedure" column means a number of more than 4 clusters.

Index	Nu-mber	Evaluation of the procedure
"ch" (Calinski and Harabasz, 1974)	2	Poor with a high number of clusters. Often prefers 2 cluster solutions (Islam et al., 2016; Arbelaitz et al., 2013;
		Vendramin et al., 2010).
"duda" (Duda and Hart, 1973)	10	Good with a high number of clusters (Milligan and Cooper, 1985; Islam et al., 2016). Second best procedure in (Milligan
"neoudet?" (Dude and Hort, 1072)	10	and Cooper, 1965.
"aipder" (Hubert and Levin, 1975)	4	-
childex (Hubert and Levill, 1976)	0	Determines the optimum number of clusters ± 1 with a probability of only 50% (Islam et al., 2016). Poor with a high number of clusters (Arbelaitz et al., 2013).
"beale" (Beale, 1969)	2	Poor with a high number of clusters (Arbelaitz et al., 2013).
"ptbiserial" (Milligan, 1980, 1981)	10	A high number of clusters is often underestimated (Milligan and Cooper, 1985).
"db" (Davies and Bouldin, 1979)	10	In (Arbelaitz et al., 2013) the third-best index with a high number of clusters, but low success rate with a high number
		of clusters (Milligan and Cooper, 1985; Arbelaitz et al., 2013).
"frey" (Frey and van Groenewoud, 1972)	1	Result contradicts the cluster idea because of the number of clusters < 2 . The Number of clusters is rather
		underestimated with a high number of clusters (Milligan and Cooper, 1985).
"hartigan" (Hartigan, 1975)	5	Works well with a small number of indicators (Tibshirani et al., 2001; Albatineh and Niewiadomska-Bugaj, 2011). Poor
0		with a high number of clusters (Milligan and Cooper, 1985).
"ratkowsky" (Ratkowsky and Lance, 1978)	8	Poor with a high number of clusters (Milligan and Gooper, 1985).
"scott" (Scott and Symons, 1971)	3	Poor with a high number of clusters (Millican and Cooper, 1985).
"marriot" (Marriott, 1971)	7	Poor with a high number of clusters (Millican and Cooper, 1985).
"ball" (Ball and Hall, 1965)	3	Poor with a high number of clusters (Millican and Cooper, 1985).
"trcovw" (Milligan and Cooper, 1985)	3	Poor with a high number of clusters (Milligan and Cooper, 1985).
"tracew" (Milligan and Cooper, 1985)	5	Poor with a high number of clusters (Millican and Cooper, 1985).
"friedman" (Friedman and Rubin, 1967)	3	Poor with a high number of clusters (Milligan and Cooper, 1985).
"mcclain" (McClain and Rao, 1975)	2	Poor with a high number of clusters (Millican and Cooper, 1985).
"rubin" (Friedman and Rubin, 1967)	8	Poor with a high number of clusters (Millican and Cooper, 1985).
"kl" (Krzanowski and Lai, 1988)	3	Identifies only 40-50% of the clusters (Albatineh and Niewiadomska-Bugai, 2011; Islam et al., 2016).
"silhouette" (Rousseeuw, 1987)	3	Poor with a high number of clusters (Islam et al., 2016; Arbelaitz et al., 2013).
"gap" (Tibshirani et al., 2001)	2	Poor with a high number of clusters (Islam et al., 2016).
"dindex" (Lebart et al., 2002)	5	_
	-	

(continued on next page)

Utilities Policy 57 (2019) 75-96

Table 9 (continued)





Fig. 7. Distributions of indicator values.



Fig. 8. Results in determining the number of factors.

Utilities Policy 57 (2019) 75-96



Fig. 9. Factor loadings in each of the ten factors.



Fig. 10. Correlation matrix of the indicator values.



Fig. 11. Illustration of the northern German municipalities with their allocation in the 10 cluster solution.



Fig. 13. Illustration of the southern German municipalities with their allocation in the ten cluster solution

References

- acatech, 2016. Flexibility concepts for the German power supply 2050. Ensuring stability in the age of renewable energies. Available online at: https://www.leopoldina.org/ uploads/tx_leopublication/2016_02_Stellungnahme_Flexibility_concepts.pdf checked on 4/4/2017
- Agemar, Thorsten, Weber, Josef, Schulz, Rüdiger, 2014. Deep geothermal energy pro-duction in Germany. Energies 7 (7), 4397–4416. https://doi.org/10.3390/
- Agemar, T., 2017. Geothermische Informationssystem für Deutschland. Available online www.geotis.de/homepage/project.php?loc=de_DE_checked_on_8/14, 2017.
- Agemar, Thorsten, Weber, Josef, Moek, Inga S., 2018. Assessment and public reporting of Agenar, Horstei, Weber, Josef, Mues, Inga S., 2016. Assessment and puolic reporting of geothermal resources in Germany: review and outlook. Energies 11 (2), 332–349. https://doi.org/10.3390/en11020332.
 Albatineh, Ahmed N., Niewiadomska-Bugaj, Magdalena, 2011. MCS. a method for finding the number of clusters. J. Classif. 28 (2), 184–209. https://doi.org/10.1007/s00357-
- Aliandali, Abdulkader, 2017, Factor Analysis, In: Aliandali, Abdulkader (Ed.), Multivariate Methods and Forecasting with IBM SPSS® Statistics. Chan, s.L. Springer International Publishing (Statistics and Econometrics for Finance), pp
- Anderson, Marti J., 2001. A new method for non-parametric multivariate analysis of variance. Austral Ecol. 26 (1), 32–46. https://doi.org/10.1111/j.1442-9993.2001
- Arbelaitz, Olatz, Gurrutxaga, Ibai, Muguerza, Javier, Pérez, Jesús M., Perona, Iñigo, 2013. Anderanz, Oraz, Gurruxaga, Ibai, Muguerza, Javier, Ferez, Jesus M., Feroita, Inigo, 2013 An extensive comparative study of cluster validity indices. Pattern Recogn. 46 (1), 243–256. https://doi.org/10.1016/j.patcog.2012.07.021. Bacher, Johann, Pöge, Andreas, Wenzig, Knut, 2010. Clusteranalyse.
- Anwendungsorientierte Einführung in Klassifikationsverfahren. 3., erg., vollst überarb, und neu sestaltete Aufl. Öldenbours, München,
- Balcombe, Paul, Rigby, Dan, Azapagie, Adisa, 2015. Energy self-sufficiency, grid demand variability and consumer costs. Integrating solar PV, Stirling engine CHP and battery storage. Appl. Energy 155, 393–408. https://doi.org/10.1016/j.apenergy.2015.06. 017.
- Ball, G.H., Hall, D.J., 1965. ISODATA: a Novel Method of Data Analysis and Pattern Classification. NTIS, Springfield Available online at: http://worldcatlibraries.org
- Classification. NTIS, springfield Available online at: http://worldcatlibraries.org/ wcpa/cel.c77143991. knecht, D., Vogel, M., Funcke, S., 2015. Energiewende zentral oder dezentral? In: Diskussionspapier im Rahmen der Wissenschaftlichen Koordination des BMBF Förderprogramms "Umwelt- und Gesellschaftsverträgliche Transformation des Energiesystems". Öko-Institut. Freiburg.
- BBSR, 2015. Referenz Gemeinden und Gemeindeverbände, Stadt- und Gemeindetvp. R, 2015. Netereinz demeinden und Genernkortsmassen aus dem Berner Available online at: http://www.bbsr.bund.de/BBSR/DE/Raumbeobachtung/ Raumabgrenzungen/StadtGemeindetyp/downloadangebote.html?nn = 443182 Raumabgenzaugen/stadtseineinkerg/vorminoadaugevorminister checked on 8/29/2017. Beale, E.M., 1969. Euclidean Cluster Analysis. Scientific Control Systems Ltd Available Beale, E.M., 1969. Euclidean Cluster Analysis. Scientific Control Systems Ltd Available
- online at: http://worldcatlibraries.org/wcpa/oclc/500100942 checked on 8/14/ 2017.
- 2017.
 Bernardi, L., Keim, S., 2017. Childless at age 30. A qualitative study of the life course plans of working women in east and west Germany. In: Kreyenfeld, Michaela, Konietzka, Dirk (Eds.), Childlessness in Europe. Contexts, Causes, and Consequences. Springer Open, Berlin, pp. 253–267.
 BMW, 2016. Emeuerbare Energien in Zahlen: Nationale und internationale Entwicklung im Jahr 2015. BMW, Berlin.
- Boon, Frank Pieter, Dieperink, Carel, 2014. Local civil society based renewable energy organisations in The Netherlands. Exploring the factors that stimulate their emer-gence and development. Energy Policy 69, 297–307. https://doi.org/10.1016/j. Böttger, Diana, Götz, Mario, Lehr, Nelly, Kondziella, Hendrik, Bruckner, Thomas, 2014.
- Potential of the power-to-heat technology in district heating grids in Germany. Energy Procedia 46, 246–253. https://doi.org/10.1016/j.egypro.2014.01.179.
- Bouguettaya, Athman, Yu, Oj, Liu, Xumin, Zhou, Xiangya To, Di Oj, Egy JD, 2014, OJ, 1179.
 Bouguettaya, Athman, Yu, Oj, Liu, Xumin, Zhou, Xiangmin, Song, Andy, 2015. Efficient agglomerative hierarchical clustering. Expert Syst. Appl. 42 (5), 2785–2797. https:// doi.org/10.1016/j.eswa.2014.09.054.
 Brametiter, Rafael, Truger, Barbara, Schinko, Thomas, Bechar-Friedl, Birgit, 2016.
- Identification of economic and energy framework conditions of the Austrian climate and energy model regions. In: LINKS Working Paper 1.1, Available online at: http:// pure.liasa.aci/14590/1/LINKS Working Paper 1.1 (inal.pdf checked on 8/2/2017.
- Braun, Frauk G., 2010. Determinants of household's space heating type. A discrete choice analysis for German households. Energy Policy 38 (10), 5493–5503. https://doi.org/ /j.enpol. Bünger, Reinhardt, 2017. Arbeiten in Berlin - wohnen in Brandenburg. In: Der
- Tagesspiegel, 5/29/2017, Available online at: http://www.tagesspie wirtschaft/immobilien/immowelt-studie-arbeiten-in-berlin-wohnen-in /19855150.html checked on 9/5/2017.
- Burgess, Paul J., Rivas Casado, Monica, Gavu, Jerry, Mead, Andrew, Cockerill, Tim, Lord, Richard, et al., 2012. A framework for reviewing the trade-offs between, renewable energy, food, feed and wood production at a local level. Renew. Sustain. Energy Rev. 16 (1), 129-142, http: rg/10.1016/i.rser.2011.07.142
- Calinski, T., Harabasz, J., 1974. A dendrite method for cluster analysis. Commun. Stat
- Canaski, F., Hatakaski, S., 1974. A dealth endot for closed analysis. Commun. Stat. Theor. Methods 3 (1), 1–27. https://doi.org/10.1080/03610027408827101.
 Charrad, Malika, Ghazzali, Nadia, Boiteau, Véronique, Niknafs, Azam, 2014. NbClust. An R package for determining the relevant number of clusters in a data set. J. Stat. Software 61 (6). https://doi.org/10.18637/jss.v061.i0

- Chévez, Pedro, Barbero, Dante, Martini, Irene, Discoli, Carlos, 2017. Application of the krease clustering method for the detection and analysis of areas of homogeneous residential electricity consumption at the Great La Plata region, Buenos Aires, Argentina. Sustain. Cities Soc. 32, 115–129. https://doi.org/10.1016/j.scs.2017.0
- Chicco, Gianfranco, 2012. Overview and performance assessment of the clustering methods for electrical load pattern grouping. Energy 42 (1), 68–80. https://doi
- 10.1016/j.energy.2011.12.031. Christaller, Walter, 1980. Die zentralen Orte in Süddeutschland. E. ökonomisch-ge Constanter, Water, 1990. Die Zufraher Orte in Suddeutschnach. E. okolomischgeo-graphische Untersuchung über d. Gestermäßigkeit d. Verbreitung u. Entwicklung d. Siedlungen mit städtischen Funktionen. 3., unveränd. Aufi., reprograf. Nachdr. d. 1. Aufi., Jenna 1933. Wissenschaftliche Buchges, Darmstadt. Connolly, D., Lund, H., Mathiesen, B.V., Wemer, S., Möller, B., Persson, U., et al., 2014.
- Heat Roadmap Europe. Combining district heating with heat savings to decarbonise the EU energy system. Energy Policy 65, 475–489. https://doi.org/10.1016/j.enpol.
- Davies, David L, Bouldin, Donald W., 1979. A cluster separation measure. IEEE Trans. Pattern Anal. Mach. Intell. PAMI 1 (2), 224–227. https://doi.org/10.1109/TPAMI.
- Deutscher Bundestag, 2017. Gesetz für den Ausbau erneuerbarer Energien
- Deutscher Bundestag, 2017. Gesetz nur den Ausbau erneuernarer Energien. (ErneuerhareEnergien-Gesetz EEG 2017). EEG 2017. Available online at: https:// www.gesetze-im-internet.de/eeg.2014/EEG_2017.pdf checked on 8/14/2017. Deutschle, Jürgen, Hauser, Wolfgang, Sonnberger, Marco, Tomaschek, Jan, Brodecki, Lukasz, Fahl, Uirche, 2015. Energie-Autonomie in Theorie und Praxis. Z. Energiewintschaft 39 (3), 151–162. https://doi.org/10.1007/s12398-015-01606-015.
- Dillon, William R., Kumar, Ajith, Mulani, Narendra, 1987. Offending estimates in cov ariance structure analysis. Comments on the causes of and solutions to Heywoo cases. Psychol. Bull. 101 (1), 126–135. https://doi.org/10.1037/0033-2909.101
- Duda, Richard O., Hart, Peter E., 1973. Pattern Classification and Scene Analysis. Wiley e publication, New York NY (A Wiley Interscience publication, New York NY u.a. Dunn, J.C., 1974. Well-separated clusters and optimal fuzzy partitions. J. Cybern. 4 (1),
- 95-104 htt 10 1080
- 35-104. https://doi.org/10.1060/01597/2740540059. Durst, Anna, 2015. In: BINE Informationsdienst (14/2015) (Ed.), Fernwärme effizient Nutzen, Available online at: http://www.bine.info/fileadmin/content/ Publikationen/Projekt-Infos/2015/Projekt_14-2015/ProjektInfo_1415_Internetx.pd
- PHDIMATUORED FUGATIONS (2015) FUGATION (201 Coblex Wieshed
- springer Gabler, Wiesbaden. Engelken, Maximilian, Römer, Benedikt, Drescher, Marcus, Welpe, Isabell, 2016. Transforming the energy system. Why municipalities strive for energy self-suffi-ciency. Energy Policy 98, 365–377. https://doi.org/10.1016/j.enpol.2016.07.049. Frey, T., van Groenewoud, H., 1972. A cluster Analysis of the D 2 matrix of white spruce
- stands in saskatchewan based on the maximum-minimum principle. J. Ecol. 60 (3), 873. ht (doi.org/10.2307
- 873. https://doi.org/10.2007/2285671. Friedman, H.P., Rubin, J., 1967. On some invariant criteria for grouping data. J. Am. Stat. Assoc. 62 (320), 1159–1178. https://doi.org/10.1080/01621459.1967.105500923. Funcke, Simon, Bauknecht, Dierk, 2016. Typology of centralised and decentralised visions for electricity infrastructure. Util. Pol. 40, 67–74. https://doi.org/10.1016/j.jup.

- for electricity infrastructure. Util. Pol. 40, 67–74. https://doi.org/10.1016/j.jup. 2016.03.005.
 Galling, Ludger, Hueesker, Frank, Kern, Kristine, Röhring, Andreas, 2013. Die räumliche Gestaltung der Energiewende zwischen Zentralität und Dezentralität. Explorative Anwendung einer Forschungsheuristik. In: IRS Working Paper 151.
 Gentle, J.E., Kaufman, L., Rousseuw, P.J., 1991. Finding groups in data. An introduction to chuster Analysis. Biometrics 47 (2), 788. https://doi.org/10.2007/2532178.
 Geyler, Stefan, Wamer, Barbara, Brandl, Anja, Kuntze, Martina, KoleMi, Forschungsverbund (Eds.), 2008. Leipzig, Leipzig (02).
 Godenbauer, Sebastian, 2016. Daten von Deutschland. In: Godenbauer, Sebastian (Ed.), Mathematische Optimierung der Wahlkreiseintellung für die Deutsche Bundestagswahl. Modelle und Algorithmen für eine bessere Beachtung der gesetzlichen Vorgaben. Springer Spektnum (BestMasters). Wiesbaden, pp. 121–143.
 Halkidi, M., Vazirgiannis, M., Batistakis, Y., 2000. Quality scheme Assessment in the clustering process. In: Zighed, Djamel A., Jan, Komorowski, Jan, Zytkow (Eds.), Principles of Data Mining and Knowledge Discover, 416 European Conference, PKDD 2000 Lyon, France, September 13-16, 2000 Proceedings. Springer (Lecture Notes in Computer Science, 1910), Berlin, Heidelberg, pp. 265–276.
 Halkidi, M., Vazirgiannis, M., 2001. Clustering validity assessment: finding the optimal conference, NED 2000 Lyon, France, September 13-16, 2000 Proceedings. Springer (Lecture Notes in Computer Science, 1910), Berlin, Heidelberg, pp. 265–276.
- Computer Science, 1910, bernin, Heichleurg, pp. 205–27.0.
 Halkkili, M., Vazirgiannis, M., 2001. Clustering validity assessment: finding the optimal partitioning of a data set. In: Cercone, Nick, Lin, Tsau Y., Wu, Xindong (Eds.), 2001 IEEE International Conference on Data Mining. Proceedings: 29 November-2 December, 2001, San Jose, California. 2001 IEEE International Conference on Data Mining. San Jose, CA, USA, 29 Nov-2 Dec. 2001. IEEE Computer Society, IEEE International Conference on Data Mining. IEEE Computer Society, Los Alamitos, Calif. are, 127, 104. Calif, pp. 187-194.
- Hargreaves, Anthony, Cheng, Vicky, Deshmukh, Sandip, Leach, Matthew, Steemers, Koen, 2017. Forecasting how residential urban form affects the regional carbon savings and costs of retrofitting and de centralised energy supply. Appl. Energy 186, 549–561. https://doi.org/10.1016/00.0016 https://doi.org/10.1016/j.apenergy.2016.02.095. Hartigan, John A., 1975. Clustering Algorithms. Wiley, New York NY u.a (Wiley series in
- robability and mathematical statistics; Applied probability and statistics Hechler, H., Bredel-Schürmann, Stefan, 2011. In: ASUE Arbeitsgemeinschaft für spar-
- amen und unwelt/reundlichen Energieverbrauch e.V. Berlin (Ed.), Tiefe Geothermie, Available online at: http://asue.de/sites/default/files/asue/th umwelt_klimaschutz/2011/broschueren/07_05_11_asue-tiefe-geothermie.pdf

93

Utilities Policy 57 (2019) 75-96

checked on 8/14/2017.

- 000. Nationalatlas Bundesrepublik Deutschland. Leibniz-Institut He für Länderkunde Elsevier Spektrum Akadem Verl München
- Heyde, Thoms P. E. auf der, 1990. Analyzing chemical data in more than two dimen-sions. A tutorial on factor and cluster analysis. J. Chem. Educ. 67 (6), 461. https://
- www.mg/tic.tic2l/cmt0/p401.
 Hintzey, Jerry L., Nelson, Ray D., 1998. Violin plots. A box plot-density trace synergism. Am. Statistician 52 (2), 181–184. https://doi.org/10.1080/00031305.1998.
- John L., 1965. A rationale and test for the number of factors in factor analysis. Horr
- Psychonetrika 30 (2), 179–185. https://doi.org/10.1007/BF02289447. Høyer, Karl Georg, 2008. The history of alternative fuels in transportation. The case of electric and hybrid cars. Util. Pol. 16 (2), 63-71. https://doi.org/10.1016/j.jup
- Hubert, Lawrence J., Levin, Joel R., 1976. A general statistical framework for assessing categorical clustering in free recall. Psychol. Bull. 83 (6), 1072–1080. https://doi.
- Hubert, Lawrence, Arabie, Phipps, 1985. Comparing partitions. J. Classif. 2 (1), 193-218.
- Institut für Wohnen und Umwelt, 2017. Gradtagszahlen in deutschland. Available online Deute chland.xls checked on 9/7/2017.
- Grandgezaniery Deutschaltexts Greekee 01977/2017.
 Islam, Md. Atiqui, Alizadeh, Behrooz Z., van den Heuvel, Edwin, R., Bruggeman, Richard, Cahn, Wiepke, Haan, Lieuwe de, et al., 2016. A comparison of indices for identifying the number of clusters in hierarchical clustering. A study on cognition in schizophrenia patients. Commun. Stat: Case Stud., Data Anal. Appl. 1 (2), 98–113. https://
- doi.org/10.1088/23737484.2015.1103670. emann, C., Hagspiel, S., Lindenberger, D., 2013. The economic inefficiency of parity: the case of German photovoltaics. In: EWI Working Paper No 13/19.
- Javied, Tallal, Rackow, Tobias, Stankalla, Roland, Sterk, Christian, Franke, Jörg, 2016. A study on electric energy consumption of manufacturing companies in the German industry with the focus on electric drives. Procedia CIRP 41, 318-322. https://doi.
- and a starty with the four of the four Energiewirtschaftliche Tagesfr. 39 (4), 190–195. Jenssen, Till, König, Andreas, Eltrop, Ludger, 2014. Bioenergy villages in Germany.
- Bringing a low carbon energy supply for rural areas into practice. Renew. Energy 61, 74-80. https://doi.org/10.1016/j.renene.2012.08.014.
 se, Eckhard, Schubert, Thomas, Thieme, Tom (Eds.), 2014. Politik in Sachsen. Springer
- Johnson, Stephen C., 1967. Hierarchical clustering schemes. Psychometrika 32 (3),
- 241–254. https://doi.org/10.1007/BF02289588. Kaiser, Henry F., 1960. The application of electronic computers to factor Analysis. Educ.
- Kaiser, neiny F., 1900. The application of electronic computers to factor haurysis. Educ Psychol. Neas. 20 (1), 141–151. https://doi.org/10.1177/001316446002000116. Kaundinya, Deepak Paramashivan, Balachandra, P., Ravindranath, N.H., Ashok, Veilumuthu, 2013. A GIS (geographical information system)-based spatial data mining approach for optimal location and capacity planning of distributed biomass power generation facilities. A case study of Turnkur district, India. Energy 52, 77–88. https://doi.org/10.1016/j.computer.0012.02.011
- bittps://doi.org/10.1016/j.energy.2013.02.011.
 Keles, D., Jochem, P., McKenna, R., Ruppert, M., Fichtner, W., 2017. Meeting the modeling needs of future energy systems. Energy Technol. 5 (7), 1007–1025. https://doi.org/10.1002/ente.201600607.
 Klaus Novy Institut e V.; trendresearch, 2011. "Marktakteure: Emeuerbare-Energien-
- Kaus Novy institut e.v.; trendresearch, 2011. "Warktakteure: Emeuerbare-Energien-Anlagen in der Stromerzeugung". Köln. Kodinariya, T., Makwana, P., 2013. Review on determining number of cluster in K-means clustering. Int. J. Adv. Res. Comput. Sci. Manag. Stud. 1 (6), 90–95.
- chistering, Int. J. Adv. Kes. Comput. Sci. Manag. Stud. 1 (6), 90–95.
 Korlaha, Binod Prasad, Kölou, Elta, Friege, Jonas, Hakvoort, Rudi A., Herder, Paulien M., 2016. Energetic communities for community energy. A review of key issues and trends shaping integrated community energy systems. Renew. Sustain. Energy Rev. 56, 722–744. https://doi.org/10.1016/j.rser.2015.11.080.
 Kraftfahrt-Bundesamt, 2017a. In: Kraftfahrt-Bundesamt (Ed.), Fahrzeugzulassungen (FZ).
- Bestand an Krafttanrzeugen

 Available online at: https://www.kba.de/SharedLocg/item______

 Fahrzeuge/F2/2017/fz3 2017_pdf.jdf.jsessionid =

 001266C8B0663B938AA41410071E0075.live21303?_blob = publicationFile&v=3

 001266C8B0663B938AA41410071E0075.live21303?_blob

 Bestand an Kraftfahrzeugen und Kraftfahrzeuganhängern nach Gemeinden
- checked on 9/7/2017.
 Kraftfahrt-Bundesamt, 2017b. In: Kraftfahrt-Bundesamt (Ed.), Fahrzeugzulassungen (FZ).
- Bestand an Kraftfahrzeugen und Kraftfahrzeuganhängern nach Zulassungsbezirken, online at: https://www.kba.de/SharedDocs/Publikationen/DE/Statistik/ 2/FZ/2017/fz1_2017_pdf.pdf?_blob=publicationFile&v=3 checked on 9/ Available online at: htt 7/2017
- //2017. Kronthaler, F., 2003. A Study of the Competitiveness of Regions Based on a Cluster Analysis: the Example of East Germany. Institut für Wirtschaftsforschung Halle Available online at: http://www.iwh-halle.de/fileadmin/user_upload/publ iwh_discussion_papers/179.pdf checked on 11/17/2017.
- Krzanowski, W.J., Lai, Y.T., 1988. A criterion for determining the number of groups in a data set using sum-of-squares clustering. Biometrics 44 (1), 23. https://doi.org/10.
- Lauf, Steffen, Haase, Dagmar, Kleinschmit, Birgit, 2016. The effects of growth, shrinkage, population aging and preference shifts on urban development—a spatial scenario analysis of Berlin, Germany. Land Use Pol. 52, 240–254. https://doi.org/10.1016/j. landusepol.2015.12.017.
- Iandusepol.2015.12.017. Lebart, Ludovic, Morineau, Alain, Piron, Marie, 2002. Statistique exploratoire multi-dimensionnelle. 3e éd., nouveau tirage révisé. Dunod (Sciences Sup. Mathématiques) dimensionnelle. 3e éd. 2e et 3e cycles), Paris.

- Utilities Policy 57 (2019) 75-96
- Lopes, Raul H.C., 2011. Kolmogorov-smirnov test. In: Lovric, Miodrag (Ed.), International Encyclopedia of Statistical Science. Springer Berlin Heidelberg, Berlin, Heidelberg, 718 720
- Lorenzo-Seva, Urbano, Ferrando, Pere J., 2006. FACTOR. A computer program to fit the exploratory factor analysis model. Behav. Res. Methods 38 (1), 88–91. https://doi.
- Lüker-Jans, Nicola, Simmering, Dietmar, Otte, Annette, 2017. The impact of biogas plants on regional dynamics of permanent grassland and maize area-the example of Hesse, Germany (2005-2010). Agric. Ecosyst. Environ. 241, 24-38. https://doi.org/10.
- Introjuget.2017.02.053. Inter, Kai, Fain, Karoline, McKenna, Russell, Stengel, Julian, Fichtner, Wolf, Schultmann, Frank, 2014. A high-resolution determination of the technical poten for residential-roof-mounted photovoltaic systems in Germany. Sol. Energy 105, programmer and provide the systems in Germany. Sol. Energy 105, programmer and photovoltaic systems in Germany. Sol. Energy 105, programmer and photovoltaic systems in Germany. Sol. Energy 105, programmer and photovoltaic systems in Germany. Sol. Energy 105, programmer and photovoltaic systems in Germany. Sol. Energy 105, programmer and photovoltaic systems in Germany. Sol. Energy 105, programmer and photovoltaic systems in Germany. Sol. Energy 105, programmer and photovoltaic systems in Germany. Sol. Energy 105, programmer and photovoltaic systems in Germany. Sol. Energy 105, programmer and photovoltaic systems in Germany. Sol. Energy 105, programmer and photovoltaic systems in Germany. Sol. Energy 105, programmer and photovoltaic systems in Germany. Sol. Energy 105, programmer and photovoltaic systems in Germany. Sol. Energy 105, programmer and photovoltaic systems in Germany. Sol. Energy 105, programmer and photovoltaic systems in Germany. Sol. Energy 105, programmer and photovoltaic systems in Germany. Sol. Energy 105, programmer and photovoltaic systems in Germany. Sol. Energy 105, programmer and photovoltaic systems in Germany. Sol. Energy 105, programmer and photovoltaic systems in Germany. Sol. Energy 105, programmer and photovoltaic systems in Germany. Sol. Energy 105, programmer and photovoltaic systems in Germany. Sol. Energy 105, programmer and photovoltaic systems in Germany. Sol. Energy 105, programmer and photovoltaic systems in Germany. Sol. Energy 105, programmer and photovoltaic systems in Germany. Sol. Photov tential 715-731 htt 1016/i.solener.2014.04.015 doi ore/1
- Virgenzia Marchan, 2017 (2017) (20
- v35-v35. https://doi.org/10.016/j.apenergy.2017.09.057. Marriott, F.H.C., 1971. Practical problems in a method of cluster Analysis. Biometrics 27 (3), 501. https://doi.org/10.2307/2528592. Information and communication technology for energy efficiency. In: Marx Gómez, Jorge,
- Information and communication technology for energy efficiency. In: Marx Gómez, Jorge, Somenschein, Michael, Vogel, Ute, Winter, Andreas, Rapp, Barbara, Giesen, Nils (Eds.), Conference proceedings, Envirolnio 2014 28th International Conference on Informatics for Environmental Protection; 10th 12th September, 2014, Oldenburg, Germany. Envirolnify, International Conference on Informatics for Environmental Protection. Oldenburg: BIS-Verl. der Carl-von-Ossietzky-Univ, Available online at: http://nbn-resolving.de/umrnhorde.gbv:715-oops-20009 checked on 11/17/2017. McClain, John O., Rao, Vithala R, 1975. CLUSTISZ: a program to test for the quality of clustering of a set of objects. J. Market, Res. 4 (12), 456-460.
- McKenna, R., Hollnaicher, S., Fichtner, W., 2014a. Cost-potential curves for onshore wind energy. A high-resolution analysis for Germany. Appl. Energy 115, 103–115. https://
- McKenna, Russell, Jäger, Tobias, Fichtner, Wolf, 2014b. Energieautarkie ausgewählte Ansätze und Praxiserfahrungen im deutschsprachigen Raum. uwf 22 (4), 241–247. https://doi.org/10.1007/s00550-014-0339-v.
- https://doi.org/10.100//s00530-014-03399.
 McKenna, Russell, Herbes, Carsten, Fichtner, Wolf, 2015. Energieautarkie. Vorschlag einer Arbeitsdefinition als Grundlage für die Bewertung konkreter Projekte und Szenarten. Z. Energiewirtschaft 39 (4), 235–252. https://doi.org/10.1007/s12390
- McKenna, R., Hofmann, L., Merkel, E., Fichtner, W., Strachan, N., 2016. Analysing so-cioeconomic diversity and scaling effects on residential electricity load profiles in the context of low carbon technology uptake. Energy Policy 97, 13–26. https://doi.org/ bab.pdf.
- Kenna, R., Djapic, P., Weinand, J., Fichtner, W., Strbac, G., 2017a. Assessing the im-plications of socioeconomic diversity for low carbon technology uptake in electrical distribution networks. Appl. Energy. https://doi.org/10.1016/j.apenergy.2017.07
- McKenna, Russell, Merkel, Erik, Fichtner, Wolf, 2017b. Energy autonomy in residential Braxenia, Russei, meiner, Link, Freihner, wolf, 20170. Energy autonomy in residential buildings. A techno-economic model-based analysis of the scale effects. Appl. Energy 189, 800–815. https://doi.org/10.1016/j.apenergy.2016.03.062.
 McKenna, R., 2018. The double-edged sword of decentralized energy autonomy. Energy Policy 113, 747–750.
- Forsy and J. O. Form and Building Energy: Quantifying Relationships Using a Multi-Scale Approach. PhD Thesis in Resource Management and Environmental Studies at the Faculty of Graduate Studies. Vancouver. Miller, N
- Milligan, Genn W., 1980. An examination of the effect of six types of error perturbation on fifteen clustering algorithms. Psychometrika 45 (3), 325–342. https://doi.org/10.
- Milligan, Glenn W., 1981. A Monte Carlo study of thirty internal criterion measures for ster analysis. Psychometrika 46 (2), 187–199. ht
- Milligan, Glenn W., Cooper, Martha C., 1985. An examination of procedures for de-termining the number of clusters in a data set. Psychometrika 50 (2), 159–179.
- https://doi.org/10.1007/BF02294245. Milligan, Glenn W., Cooper, Martha C., 1988. A study of standardization of variables in
- Igan, Glenn W., Cooper, Martha C., 1988. A study of standardization of variables in cluster analysis. J. Classif 5 (2), 181–204. https://doi.org/10.1007/Rb01897163. amoto, Sadaaki, Abe, Ryosuke, Endo, Yasunori, Takeshita, Jun-ichi, 2015. Ward method of hierarchical clustering for non-Euclidean similarity measures. In: Köppen, Mario (Ed.), Proceedings of the 2015 Seventh International Conference of Soft Computing and Pattern Recognition (SoCPAR 2015). Fukuoka, Japan, November 13-15, 2015. 2015 7th International Conference of Soft Computing and Pattern Recognities (COPD). Education. Longer Indexing and Pattern Descenting (COPD). Education. Longer Meditica Intelligence Descenter Labor. Miyam m (a) 2015. 2015 for international Conference of Soft Computing and Pattern Recognition (SoCPaR). Fukuoka, Japan. Machine Intelligence Research Labs; International Conference of Soft Computing and Pattern Recognition; SoCPaR. IEEE, Discontentier, M.J. and O.A. (2016).
- Müller, Matthias Otto, Stämpfli, Adrian, Dold, Ursula, Hammer, Thomas, 2011. Energy autarky. A conceptual framework for sustainable regional development. Energy Policy 39 (10), 5800-5810. https /doi.org/10.1016/j.enpol.2011.04.019
- Müllner, Daniel, 2016. Package "fast cluster". Available online at: htt org/web/packages/fastcluster/fastcluster.pdf checked on 8/16/2 Munier, Daniel, 2016. Package: asscusser: A vanable online at: https://eat.reproject.org/web/packages/fastcluster/fastcluster.pdf thecked on 8/16/2017.
 Murtagh, Fionn, Legendre, Pierre, 2014. Ward's hierarchical agglomerative clustering method. Which algorithms implement ward's criterion? J. Classif. 31 (3), 274–295.
- https://doi.org/10.1007/s00357-014-9161-z. Noiva, Karen, Fernández, John E., Wescoat, James L., 2016. Cluster analysis of urban
- (Nova, Kalci, Pernankez, Jonit J., (Rescen, Janies E., 2010. Cluster analysis of manning, water supply and demand. Toward large-scale comparative sustainability planning. Sustain. Citles Soc. 27, 484–496. https://doi.org/10.1016/j.scs.2016.06.003. Office for National Statistics, 2015. Methodology Note for the 2011 Area Classification for Output Areas. checked on 8/16/2017.

- Osborne, Jason W., 2014. Best Practices in Exploratory Factor Analysis. [Louisville, Ky.]: [CreateSpace Independent Publishing Platform]. Peter, Stefan, 2013. In: Umweltbundesamt (Ed.), Modellierung einer vollständig auf er-
- igung im Jahr 2050 in autarka euerbaren Energien basierenden Stron
- Purkus, Alexandra, Barth, Volker, 2011. Geothermal power production in future electricity markets-a scenario analysis for Germany. Energy Policy 39 (1), 349-357. oi.org/10.1016/j.enpol.2010.10.003.
- Res, Callum, Bradley, Flora 2012. Energy autonomy in sustainable communities—a review of key issues. Renew. Sustain. Energy Rev. 16 (9), 6497–6506. https://doi.org/ 10.1016/j.resr.2012.08.002.
 Ratkowsky, D.A., Lance, G.N., 1978. A Criterion for determining the number of groups in
- sification. Aust. Comput. J. 10 (3), 115-117.

Ratti, Carlo, Baker, Nick, Steemers, Koen, 2005. Energy consumption and urban texture.

- at: https://www.bvmw.de/fileadmin /ende.pdf.pdf checked on 11/8/2017
- Requia, Weeberb J., Roig, Henrique L., Koutrakis, Petros, Adams, Matthew D., 2017. Requia, Weeberb J., Roig, Henrique L., Koutrakis, Petros, Adams, Matthew D., 2017. Modeling spatial patterns of traffic emissions across S570 municipal districts in Brazil. J. Clean. Prod. 148, 845–853. https://doi.org/10.1016/j.jclepro.2017.02.010. Revelle, William, 2017. Package 'psych'. Available online at: https://cran.r-project.org/ web/packages/psych/psych.pdf checked on 8/16/2017. Rode, Philipp, Keim, Christian, Robazza, Guido, Viejo, Pablo, Schoffeld, James, 2014a. Christian and Checked and Schoffeld, James, 2014a.

- Rode, Philipp, Keim, Christian, Robazza, Guido, Viejo, Pablo, Schofield, James, 2014a. Cities and energy. Urban morphology and residential heat-energy demand. Environ. Plan. B Plan. Des. 41 (1), 138–162. https://doi.org/10.1068/b39065.Rode, P., Burdett, R., Robazza, G., Schofield, J., Keim, C., Avci, N., Dimitrova, S., Garcia, P.V., Murshed, S.M., Bahu, J.-M., 2014b. Cities and Energy: Urban Morphology and Heat Energy Demand. Final Report. LSE, IEEE, London February 2014.Rousseuw, Peter J., 1987. Silhouettes. A graphical aid to the interpretation and vali-dation of cluster analysis. J. Comput. Appl. Math. 20, 53–65. https://doi.org/10. 1016/0377-0427(87)90125-7.
- dation of cluster analysis. J. Comput. Appl. Math. 20, 53–65. https://doi.org/10. 1016/0377.0427(87)90125-7.
 Salvador, S., Chan, P., 2004. Determining the number of clusters/segments in hierarchical clustering/segmentation algorithms. In: 16th IEEE International Conference on Tools with Artificial Intelligence. Proceedings: ICTAI 2004: 15-17 November, 2004, Boca Raton, Florida, pp. 576–584 16th IEEE International Conference on Tools with Artificial Intelligence. Boca Raton, FL, USA, 15-17 Nov. 2004. International Conference on Tools for Artificial Intelligence; IEEE Computer Society. Los Alamitos, Collé. IEEE Computer Society. alif: IFFF Co muter Societa
- Sarralde, Juan José, Quinn, David James, Wiesmann, Daniel, Steemers, Koen, 2015. Solar energy and urban morphology. Scenarios for increasing the renewable energy po-tential of neighbourhoods in London. Renew. Energy 73, 10–17. https://doi.org/10. 2010/10.1016/10016/10.1016/10.1016/10.1016/10.1016/10.1016/10.1016/10.1016/10.1016/10.1016/10.1016/10.1016/10.1016/10016/10016/10016/10016/10016/10016/10016/10016/10016/10016/10016/10000/10000/10000/10000/10000/10000/10000/10000/10000/10000/10000/1000 1016/j.renene.2014.06.028 Scheffer, K., 2008. Vom Bioenergiedorf zur autonomen Solarenergieregion. Solarzeitalter
- 23-30
- Schmid, Eva, Knopf, Brigitte, Pechan, Anna, 2016. Putting an energy system transfor-mation into practice. The case of the German Energiewende. Energy Res. Soc. Sci. 11, 263-275. https://doi.org/10.1016/j.ers.2015.11.002.Schmidt, J., Schönhart, M., Biberacher, M., Guggenberger, T., Hausl, S., Kalt, G., et al.,
- Schmidt, J., Schönnart, M., Biberacher, M., Gugenberger, L., Haust, S., Kalt, V., et al., 2012. Regional energy autarky. Potentials, costs and consequences for an Austrian region. Energy Policy 47, 211–221. https://doi.org/10.1016/j.enpol.2012.04.059.
 Schuler, Andreas, Weber, Christoph, Fahl, Ulrich, 2000. Energy consumption for space heating of West-German households. Empirical evidence, scenario projections and policy implications. Energy Policy 28 (12), 877–894. https://doi.org/10.1016/
- SU301-4215(U)000/44, Schultz, Annett, Brandt, 2016. Demographietypisiserung für Kommunen. Methodisches Vorgehen und empirische Befunde Berichtigahr 2013. Berteismann Stiftung. Bochum. Available online att. https://www.wegweiser-kommune.de/documents/ 10184/10615/Methodik_Clusterung.pdf/05a1b137.7dbf-4bf4-828d-9a097a4f2805 checked on 8/2/2017. Scott, A.J., Symons, M.J., 1971. Clustering methods based on likelihood ratio criteria.
- Biometrics 27 (2), 387. https://doi.org/10.2307/2529003. Shapiro, S.S., Wilk, M.B., 1965. An analysis of variance test for normality (complete
- Shapiro, S.S., Wilk, M.B., 1965. An analysis of variance test tor normality (complete samples). Biometrika 52 (3/4), 591. https://doi.org/10.2307/2333700.
 Shapiro, Alexander, Berge, Jos M. F. ten, 2002. Statistical inference of minimum rank factor analysis. Psychometrika 67 (1), 79–94. https://doi.org/10.1007/BF02294710.
 Shipworth, Michelle, Firth, Steven K., Gentry, Michael L., Wright, Andrew J., Shipworth, David T., Lomas, Kevin J., 2010. Central heating thermostat settings and timing. Building demographics. Build. Res. Inf. 38 (1), 50-69. https://doi.
- 09613210903263007. Statistisches Bundesamt, 2000. In: Statistische Ämter des Bundes und der Länder (Ed.), Erwerbstätige nach Wirtschaftsbereichen Jahreschurchschnitt (WZ2008) regionale Tiefe: Kreise und krfr. Städte, Available online at: https://www.regionalstatistik.de/ genesis/online/datagisesionid=DEEIDOC2D10898099008854D066648AC.reg2? operation=abruftabelleAbrufen&selectionname=638-61-484.levelindex=0& relid = 1504792196820 windex = 13 checked on 9/7/2017
- Revende 1504792196820800628 13 Greekel of 977/2017.
 Statistisches Bundessunt, 2003a. In: Statistische Amter des Bundes und der Länder (Ed.), Bruttoinlandsprochikt/Bruttowertschöpfung (WZ 2008) Jahressumme regionale Tiefe: Kreise und krft: Städte, Available online at: https://www.regionalstatistik.de/ genesis/online/data;jsestonid = DEE1DDC2D1089B9F9908854D96E648AC.reg2 checked on 9/7/2017.
- Statistisches Bundesamt, 2003b. In: Statistisches Bundesamt (Ed.). Energieverwendung der Betriebe des Verarbeitenden Gewerbes sowie des Bergbaus und der Gewinnung von Steinen und Erden - Jahressumme - regionale Tiefe: Kreise und krfr. Städte, Available online at:

data;jsessionid=DEE1DDC2D1089B9F9908854D96E648AC.reg2?operation= abruftabelleAbrufen&selectionname = 060-41-4&levelindex = 0&levelid =

- abruttabelleAbruttenkselectionname = 000-41.48teVellndex = 0&teVelld= 1504/97108820&index = 1 checked on 9/7/2017. Statistisches Bundesamt, 2011a. In: Statistische Ännter des Bundes und der Länder (Ed.), Gemeindeverzeichnis Gebietsstand: 31.12.2010 (Jahr). Alle politisch selbständigen Gemeinden in Deutschland nach Fläche, Bevölkerung, Bevölkerungsdichte und der Postleitzahl des Verwaltungssitzes der Gemeinder, Wiesbaden. Available online at: https://genurg.detatiet.do///Gebierung/deseinender// destatis.de/DE/ZahlenFakten/LaenderRegionen/Region nis/Administrativ/Archiv/GVAuszugJ/31122010 Auszug GV
- Gemeindeverzeichnis/Administrativ/Archiv/GVAuszugJ/31122010_Auszug_GV. html checked on 9/7/2017. Statistisches Bundesamt, 2011b. In: Statistischen Ämter des Bundes und der Länder (Ed.), "Lohn- und Einkommensteuer Jahressumme -regionale Tiefe: Gemeinden, Samt-/ Verbandsgemeinden", Available online at: https://www.regionalstatistik.de/ genesis/online/dataijsessionid = 62/264E1/497FA378AD49A9004D9A6DF6.reg2? operation = statistikAbritabellen&levelindex = 0&levelid = 1504787362480& for data de data de data G2 (2017)
- operation = statistikAbruitabellen&levelindex = 0&levelid = 150478/362480& index = 2 checked on 97/2017. Statistisches Bundesamt, 2014a. In: Statistische Ämter des Bundes und der Länder (Ed.), Bevölkerung, Wiesbaden. Available online at: https://www.regionalstatistik.de/ genesis/online/dataigesionid = 62c84E17497FA378A049A0004D9A6DF6.reg2? operation = statistikAbruitabellen&levelindex = 0&levelid = 1504787709962& ndex = 2 checked on 9/7/2017.
- index = 2 checked on 9/7/2017. Statistisches Bundesamt, 2014b. In: Statistische Ämter des Bundes und der Länder (Ed.), Bruttoinlandsprodukt/Bruttowertschöpfung (WZ 2008) Jahressumme regionale Tiefe: Kreise und krfr. Städte, Available online at: https://www.regionalstatistik.de/ genesis/online/dataigseisolind = DEIE DDC2D1089809F908854D065648AC.reg2? operation = previous&levelindex = 2&levelid = 1504791473323&levelid = 1504791439317 (step = 1 checked on 9/7/2017.
- 1504/9143931/Rstep=1 checked on 9///2017. Statistisches Bundesamt, 2014c. In: Statistische Amter des Bundes und der Länder (Ed.), Energieverwendung der Betriebe des Verarbeitenden Gewerbes sowie des Bergbaus und der Gewinnung von Steinen und Erden Jahressumme regionale Tiefe: Kreise und krift Kädte, Avaulable online alt. https://www.regionalstatistik.de/genesis/ online/data;jsessionid=DEE1DDC2D1089B9F9908854D90E648AC.reg2? operation = abruftabelleAbrufen&selectionname = 060-41-4&levelindex = 08
- operation = abruitabelleAbruiensselectionname = 00-41.48keVelindex = 0& levelid = 1504791708820&index = 1 checked on 9/7/2017. Statistisches Bundesamt, 2014d. In: Statistische Ämter des Bundes und der Länder (Ed.), Gebäude und Wohnungen, Wiesbaden. Available online at: https://www. regionalstatistik.de/genesis/online/datajsessionid= 62C84E17497FA378AD49A0904D9AbDF6.reg2?operation = statistikenVerzeichnisNexStep8levelindex=0&levelid=1504786087209&index=3 0 metricula 0.4766.bit = 0.677(0017)
- & structure level = 3 checked on 9/7/2017. estructurezvet=3 criected on 977/2017. Statistische Stundesamt, 2014e. In: Statistische Ämter des Bundes und der Länder (Ed.), Haushalte und Familien, Wiesbaden. Available online at: https://www. regionalstatistik.de/genesis/online checked on 977/2017. Statistisches Bundesamt, 2014f. In: Statistische Ämter des Bundes und der Länder (Ed.),
- Statistisches Bundesami, 2014: In: Statistische Amter des Bundes und der Lander (Ed.), Sozialversicherungspflichtig Beschäftigte am Arbeits- und Wohnner sowie Ein- und Auspendler über Gemeindegrenzen-Stichtag 30.06. regionale Tiefe: Gemeinden, Available online at: https://www.regionalstattikt.de/genesis/online/ datajæssionid=DEEID022D108896P9008854D066648AC.reg2?operation= abruftabelleAbruften8æelectionname=25414-58levelinde=18levelid= 1564700893028indes=5 checked on 9/7/2017. Statistisches Bundesamt, 2014g. In: Statistische Amter des Bundes und der Länder (Ed.),
- Unsatzsever, Umsatzseverpflichtige, steverbarer Umsatz für Lieferungen und Leistungen nach Wirtschaftsabschnitten Jahr regionale Tiefer. Kreise und krri-Stäche. Available online at: https://www.regionalstatistik.de/genesis/online/ data;jsessionid=DEEIDDC2D108989F9908854D96E648AC.reg2?operation= -31-4&levelindex = 0&levelid = ruftabelleAbrufen&selectionname 1504791907481&index = 7 checked on 9/7/2017.
- Isol4/9190/481Randex = / checked on 9///2017. Statistisches Bundesant, 2015a. In: Statistische Anter des Bundes und der Länder (Ed.), Erwerbstätige nach Wirtschaftsbereichen- Jahresdurchschnitt (WZ2008) regionale Tiefe: Kreise und krfr. Städte, Available online at: https://www.regionalstatistic.de/ genesis/online/dartagisesionid = DEEI DDC2D1080809F9008854D066648AC.reg2/ operation = abmitabelleAbmien&selectionname = 638-61.4klevelindex = 1& https://www.regionalstatistic.de/
- operation = abruitabelleAbruiensselectionname = 6.38-61-48keVelindex = 18 levelid = 156/4791245508index = 3 checked on 9/7/2017. Statistisches Bundesamt, 2015b. In: Statistische Ännter des Bundes und der Länder (Ed.), Verarbeitendes Gewerbe, Bergbau und Gewinnung von Steinen und Erden: Betriebe, Beschäftigte Stichtag: 30.09. regionale Tiefe: Gemeinden, Samt -/ Verbandsgemeinden, Available online at: https://www.regionalstatistik.de/genesis/ online/datarjsessionid = DEEIDDC2D10898959088543096664484.cm.eg.2? operation = abruitabelleAbruien8xelectionname = 001-03-58levelindex = 08. levelid=16047032347298iedbase = 3 develord en 0/27017 relid = 1504792334773 windex = 2 checked on 9/7/2017.
- 1504788083861&index = 2 checked on 9/7/2017.
- & level index = 1 & level id = uftabelleAbrufen&selectionname = 449 1504788598036&index = 2 checked on 9/7/2017.
- Statistisches Bundesamt, 2017a. Bruttostromerzargung in Deutschland für 2014 bis 2016. Available online at: https://www.destatis.de/DE/ZahlenFakten/ Wirtschaftsbereiche/Energie/Erzeugung/Tabellen/Bruttostromerzeugung.html checked on 9/29/2017.

- Statistisches Bundesamt, 2017b. In: Statistische Ämter des Bundes und der Länder (Ed.), Gemeindeverzeichnis - Gebietsstand: 30.06.2017 (2. Quartal). Alle politisch selbständigen Gemeinden (mit Gemeindeverband) in Deutschland nach Fläche, sensatangen Gemeinden (um Gemeindeverband) in Deutschnaht hach Patche, Bevölkerung (auf Grundlage des Zensus 2011), Bevölkerungsdichte und der Postleitzahl des Verwaltungssitzes der Gemeinde, Wiesbaden. Available online at: https://www.destatis.de/DE/ZahlenFakten/LaenderRegionen/Regionales/ ugQ/AuszugGV2QAktuell.html strativ/Archiv/GVAuszi
- checked on 9/7/2017. cnecked of 9///2017.
 Statistisches Landesamt Baden-Württemberg, 2009. Weit auseinander und doch bei-sammen (06/2009) In: Statistisches Monatsheft Baden-Württemberg, pp. 51–53.
 Available online at: https://www.statistik.bww.de/Service/Veroeff/Monatshefte/ PDF/Beitrag00 06 11.pdf checked on 8/2/2017.
 Steemers, Koen, 2003. Energy and the city. Density, buildings and transport. Energy Beitle 01 20 21 de use (doit not build on buildings of the programmer for the programmer).
- Bulld. 35 (1), 3–14. https://doi.org/10.1016/S0378-7788(02)00075-0.
 Su, Shi, Lu, Hai, Zhang, Liang, Alanne, Karl, Yu, Zitao, 2017. Solar energy utilization patterns for different district typologies using multi-tobjective optimization. A comparative study in China. Sol. Energy 155, 246–258. https://doi.org/10.1016/j.
- solener.2017.00.036. Tibbirnai, Robert, Walther, Guenther, Hastie, Trevor, 2001. Estimating the number of clusters in a data set via the gap statistic. J. Roy. Stat. Soc. B 63 (2), 411–423. https:// doi.org/10.1111/1467.9686.00929. /10.1111
- unveltbundesamt, 2016. Branchenabilangiger Energieverbrauch des verarbeitenden Gewerbes. Available online at: https://www.unweltbundesamt.de/daten/indust branchenabianegiger-energieverbrauch-des/textpart-1_checked on \$/11/2017. Umweltbundesamt; BMWI, 2017. In: Bundesministerium für Wirtschaft und Energie
- (BMWi) (Ed.), Energieeffizienz in Zahlen, Available online at: http://www.bmwi.
- (BMW) (Ed.), Energeeffizienz in Zahlen, Available online at: http://www.tmwi. de/Redation/DE/Publikationen/Energie/energieeffizienz-in-zahlen.pdf?_blob = publicationFile&v=10 checked on 11/2/2017. Umweltbundesamt, 2018. Emeuerbare Energie in Deutschland. Daten zur Entwicklung im Jahr 2017. Available online at: https://www.umweltbundesamt.de/sites/default/ files/medien/376/publikationen/180315_uba.hg.eeinzahlen_2018_bf.pdf checked on 6/20/2018. 6/28/2018
- Unternährer, Jérémy, Moret, Stefano, Joost, Stéphane, Maréchal, François, 2017. Spatial clustering for district heating integration in urban energy systems. Application to geothermal energy. Appl. Energy 190, 749–763. https://doi.org/10.1016/j.apenerg 2016.12.136.
- Urquizo, Javier, Calderón, Carlos, James, Philip, 2017. Metrics of urban morphology and
- Orquizo, Javier, Canderon, Carlos, Joanes, Philip, 2017. Metrics of urban morphology and their impact on energy consumption. A case study in the United Kingdom. Energy Res. Soc. Sci. 32, 193–206. https://doi.org/10.1016/j.erss.2017.03.011.
 Vendramin, Lucas, Campello, Ricardo J.G.B., Hruschka, Eduardo R, 2010. Relative clustering validity criteria. A comparative overview. Stat. Anal. Data Min. 3. https:// doi.org/10.1002/sam.10080.

Utilities Policy 57 (2019) 75-96

- Volz, R., 2012. Bedeutung und Potenziale von Energiegenossenschaften in Deutschland.
- Volz, R., 2012. Bedeutung und Potenziale von Energiegenossenschaften in Deutschland. Inf. Raumentwickl. 9 (10), 515–524.
 Wall, Wilhelm, 2016. Energetisch vergleichbare Städtegruppen eine gesamtheitliche Clusteranalyse und Clusterauswahl deutscher kreisfreier Städte auf Basis der ty-pischen Verbrauchssektoren und sozio-energetischer Indikatoren. Bochum: Selbstverlag des Lehnstuhls Energiesysteme und Energiewirtschaft.
 Wei, Shen, Jones, Rory, Wilde, Pieter de, 2014. Dirtving factors for occupant-controlled space heating in residential buildings. Energy Build. 70, 36–44. https://doi.org/10. 1016/j.enbuild.2013.11.001.
 Wimmer, D., Heinemann, C., Bauknecht, D., 2014. Die Auswirkung räumlich verteilter Windstromproduktion auf den Flexibilitätsbedarf im deutschen Stromsystem. Energiewirtschaftliche Tagesfr. 64 (12), 32–35.
 Wirth, H., 2016. Recent Facts about Photovoltatics in Germany. Fraunhofer ISE, Freiburg , Accessed date: 14 October 2016.
- Accessed date: 14 October 2016. Accessed date: 14 October 2016.
 Woldeamanuel, Mintesnot G., Cyganski, Rita, Schulz, Angelika, Justen, Andreas, 2009.
 Variation of households' car ownership across time. Application of a panel data model. Transportation 36 (4), 371–387. https://doi.org/10.1007/s11116-009-
- Woyke, Wolfgang, Forero, Mario, 2014. Methoden zur Bewertung regionaler Energieautarkie. In: Beitrag auf dem 13. Symposium Energieinnovation. Graz,
- Österreich.
- Osterreich. Wurm, Michael, Taubenbock, Hannes, Roth, Achim, Dech, Stefan, 2009. Urban struc-turing using multisensoral remote sensing data: by the example of the German cities Cologne and Dresden. In: Joint Urban Remote Sensing Event, 2009. 20 22 May 2009. Shanghai, China ; Idte 5H GRS/SIPSF Workshop on Data Pusion and Remote Sensing over Urban Areas (URBAN 2009) and the 7th International Urban Remote Sensing Over Orban Areas (ORDAN 2009) and the Ath International Orban Reinfore Sensing Conference (UBS 2009)]. 2009 Joint Urban Remote Sensing Event. Shangbai, China. Geoscience and Remote Sensing Society; International Society for Photogrammetry and Remote Sensing; Joint Urban Remote Sensing Event; Urban Remote Sensing Joint Event; GRSS/ISPRS Workshop on Data Fusion and Remote Sensing over Urban Areas; URBAN; International Urban Remote Sensing Conference; URB_URE_REMORTMENT In a 1 a 2 a 3 URS. IEEE, Piscataway, NJ, pp. 1-8. Yang, Junjing, Ning, Chao, Deb, Chirag, Zhang, Fan, Cheong, David, Lee, Siew Eang,
- Yang, Junjing, Ning, Chao, Deb, Chirag, Zhang, Fan, Cheong, David, Lee, Siew Eang, et al., 2017. Is-Shape clustering algorithm for building energy usage patterns analysis and forecasting model accuracy improvement. Energy Build. 146, 27–37. https://doi. org/10.1016/j.enbuild.2017.03.071.Yong, An Cie, Pearce, Sean, 2013. A beginner's guide to factor Analysis. Focusing on exploratory factor Analysis. Tutorials Quant. Methods Psychol. 09 (2), 79–94.Zhang, Fuzhen, 2011. Positive semidefinite matrices. In: Zhang, Fuzhen (Ed.), Matrix Theory. Basic Results and Techniques, 2. ed. Springer Science + Business Media LLC (Universitest), New York, NY, pp. 199–252.

91

SCIENTIFIC DATA

DATA DESCRIPTOR

OPEN Spatial high-resolution socioenergetic data for municipal energy system analyses

Jann M. Weinand^{1*}, Russell McKenna² & Kai Mainzer¹

In the context of the energy transition, municipalities are increasingly attempting to exploit renewable energies. Socio-energetic data are required as input for municipal energy system analyses. This Data Descriptor provides a compilation of 40 indicators for all 11,131 German municipalities. In addition to census data such as population density, mobility data such as the number of vehicles and data on the potential of renewables such as wind energy are included. Most of the data set also contains public data, the allocation of which to municipalities was an extensive task. The data set can support in addressing a wide range of energy-related research challenges. A municipality typology has already been developed with the data, and the resulting municipality grouping is also included in the data set.

Background & Summary

National targets in energy policy are leading to a radical change in the energy sector. The associated expansion of renewable energies is mainly decentralised, which also applies to the owners and operators of energy plants: private individuals increasingly invest in renewable energy systems or form so-called citizen-energy cooperatives¹. More and more municipalities are striving to exploit renewable decentralised energy generation. Thereby they

In the course of the growing interest in renewable energy systems, an increasing number of energy system analyses for the development of climate protection plans are conducted at the municipal level. However, many municipalities lack both the financial resources and the expertise to determine the potential for renewables or develop effective climate protection plans3. These municipalities would benefit from studies on their suitability for decentralised energy systems.

In addition, a growing number of energy system models are used whose input values are based on public data^{4,5}. The availability of data can therefore also support the development of energy system models

We recently grouped the 11,131 German municipalities with regard to their suitability for decentralised energy⁶. The cluster analysis included 38 socio-energetic indicators which comprised data on the energy consumption sectors "Private Households" and "Transport" as well as data to estimate the potential for renewable energies. Among the data on renewable energies are spatial high-resolution photovoltaic⁷, wind⁸ and hydrother-mal⁶ potentials. The hereby-published dataset contains all indicators and the resulting cluster composition for all German municipalities.

The dataset⁹ enables energy researchers to conduct studies at municipal and national levels without having to obtain and synthesize a large amount of data. For example, the cluster composition can help to transfer results from energy system analyses of individual municipalities to other, similar municipalities.

Methods

In the following, the various methods for determining and allocating the data to the municipalities are described. A distinction is made between data on census, mobility and renewable energies as well as cluster data (cf. Online-only Table 1).

Census data. Despite the fact that the census data are public, consolidating the data is an extensive task⁶. This is related on the one hand to different identifiers for the municipalities in the various census data tables. On the other hand, the number of municipalities in Germany is constantly reducing as several municipalities are merged into one. In addition, the data tables are rarely complete and data are missing for individual municipalities.

¹Chair of Energy Economics, Karlsruhe Institute of Technology, Karlsruhe, Germany. ²DTU Management, Technical University of Denmark, Lyngby, Denmark. *email: jann.weinand@kit.edu

SCIENTIFIC DATA (2019) 6:243 | https://doi.org/10.1038/s41597-019-0233-0



Fig. 1 Procedure for consolidating the census data.

Figure 1 shows the procedure for consolidating the data for one census table. The data were assigned to the municipalities from the municipal register of 2017 (https://www.statistikportal.de/de/produkte/gemeindeverzeichnis). Some census data tables, however, do not exist for 2017. Apart from the table from 2011, which was needed for the population development, the oldest table used is from 2014. In these cases, municipalities are indicated which no longer existed in 2017 as they were merged into one municipality. Then the values in the older tables were combined and assigned to the newly established municipality. If not all municipalities could be assigned a value after these steps, other identification numbers than the municipality key were applied. In case this was not sufficient, the names of the municipalities were used to assign the data. However, there are many municipalities with the same name in Germany. Therefore, attention was paid to the unambiguity of the names, e.g. by combining them with numbers from the municipality key. In a few cases, data could still not be assigned after these steps (for less than 2% of municipalities). Then the data was manually collected, e.g. via web searches.

The population density in the data set is related to the municipal area. Population density in relation to settlement areas would be an even more relevant indicator for energy system analyses as this data could be used to estimate costs for district heating networks. This more accurate data is published in a parallel study¹⁰.

Mobility data. The mobility data in the municipalities in Germany are published in PDF format¹¹. The delivery of the data in a processable format such as csv is subject to a fee. Therefore, the data was transferred from the PDF to a csv chart. Subsequently, the data could be allocated according to the same procedure as for the census data (cf. Fig. 1).

Renewable energy data. In earlier studies, potentials for electricity generation from photovoltaics, wind and geothermal energy were determined. The methods are explained in more detail below.

Technical photovoltaic potential. A high-resolution determination of the technical potential of residential roof-mounted photovoltaic systems for each municipality in Germany was presented by Mainzer *et al.*⁷. The method for calculating these potentials consists of two stages (see Fig. 2): first, the usable roof area in each municipality is calculated using statistical data such as the number and type of residential buildings as well as statistical data on roof geometries and the usable share of the roofs for photovoltaic systems. Next, the calculated roof area together with assumptions on the distribution of inclination and azimuth angles is combined with each municipalities' solar radiation data as well as the relative irradiation for specific inclination and azimuth angles to calculate the geographical potential. Combined with the technical PV plant efficiency, the technical potential for each municipality can then be inferred. Since the calculation method is described in detail in Mainzer *et al.*⁷, all assumptions can easily be adjusted to the readers own preferences.

Non-residential buildings could be considered with our newer, more detailed methods¹², which, however, could not yet be applied to the whole of Germany due to the higher resource demand (calculation time and storage capacity) of these methods.

Technical wind potential. In another study, the potentials for onshore wind in Germany were calculated⁸. The applied method employs multiple data sources for land use categories, annual average wind speeds and techno-economic wind turbine data for several hundred plants as shown in Fig. 3. By excluding unsuitable areas, such as areas with a gradient above 20°, urban areas and natural parks, and inserting a buffer area around these as well as residential, commercial and industrial areas, the technically feasible area for onshore wind energy was determined. These remaining areas are associated with suitability factors based on empirical values and annual

SCIENTIFIC DATA | (2019) 6:243 | https://doi.org/10.1038/s41597-019-0233-0

/ww.nature.com/scientificdata/







Fig. 3 Methodology for the assessment of technical wind energy potentials.

average wind speed data at 1 km² resolution. The final step, and one key innovation of the study, involved matching a wind turbine to the polygon types (combination of wind speed and land use category) based on the lowest levelized costs of electricity (LCOE). The results represent the lowest cost realization of the technical potential for onshore wind in Germany, based on the then (2013) state of the art in turbine technology. The technical onshore wind potential data was aggregated to postcode level in McKenna *et al.*⁸. With the help

The technical onshore wind potential data was aggregated to postcode level in McKenna *et al.*⁵. With the help of the geo-information system QGIS, the postcode areas were intersected with the administrative municipal areas from 2017. In this way, the wind potentials could be divided among German municipalities.

SCIENTIFIC DATA | (2019) 6:243 | https://doi.org/10.1038/s41597-019-0233-0

/ww.nature.com/scientificdata/





Achievable hydrothermal temperature. Figure 4 shows that the achievable hydrothermal temperatures in Germany strongly depend on the region. This means that municipalities have different hydrothermal potentials. In the GeotIS project, the hydrothermal temperatures in Germany were determined (cf. left part of Fig. 4)¹³. With the help of a raster contained in the GeotIS tool, the temperatures could be transferred manually to a spatial resolution of $8.5 \, \mathrm{km}^2$. Subsequently, achievable hydrothermal temperatures could be assigned to the municipalities. Thereby, the temperature in a municipality and closest to the centre of the municipality was assigned to the municipality. Since on the one hand the transfer of 57,535 data points from the GeotIS map was very time-consuming and on the other hand different assignment methods to the municipalities could be chosen, a data table for the hydrothermal temperatures depending on the precise coordinates is also provided⁹. These coordinates represent the intersections of the grid shown in the left part of Fig. 4.

Cluster data. In Weinand *et al.*⁶, indicators from Online-only Table 1 were first standardised to values between zero and one to ensure that all indicators have the same weight in the cluster analysis. Subsequently, a factor analysis was used to filter out the indicators that were not relevant for the further steps. Then the municipalities were grouped using a hierarchical cluster analysis. Thereby, 26 cluster validation criteria were applied to determine an appropriate number of clusters. The ten resulting clusters can also be found in the provided data set. Among these clusters are municipality groups with a high potential for renewable energies (Cluster 3, 4, 7 and 8), with a high proportion of district heating (Cluster 1) or cities with a high population density (Cluster 2).

Data Records

The municipality data summarised in Online-only Table 1 are available as an xls file in an online repository⁹. More than 400,000 data entries can be clearly assigned and identified on the basis of the column headings. The data for the hydrothermal temperatures are available as point coordinates in 8.5 km^2 resolution in a csv file⁹. The first column contains the latitudes, the first row the longitudes. The hydrothermal temperature is in the cell where latitudes and longitudes intersect. The entries are numbers from 0 to 5, where 0 means that there is no hydrothermal temperature is between $40\,^{\circ}\text{C}$ and $60\,^{\circ}\text{C}$, for 2 between $60\,^{\circ}\text{C}$ and $100\,^{\circ}\text{C}$, for 3 between $100\,^{\circ}\text{C}$ and $130\,^{\circ}\text{C}$, for 4 between $130\,^{\circ}\text{C}$ and $160\,^{\circ}\text{C}$ and $160\,^{\circ}\text{C}$ and $190\,^{\circ}\text{C}$.

Technical Validation

When validating the data, the heterogeneity of the "concept of municipality" as used by the different federal states becomes apparent. For example, Baden-Wuerttemberg, which accounts for about 13% of the population, has only 75 municipalities with less than 1000 inhabitants. The neighbouring Rhineland Palatinate, which accounts for 5% of the population, yet counts 1624 municipalities with less than 1000 inhabitants (https://www.statistikportal. de/de/produkte/gemeindeverzeichnis). In an international context, the "concept of municipality" is even more heterogeneous. The data set thus provides high-resolution data, but the system boundary of a municipality is not necessarily the most suitable one for energy system analyses.

Census and mobility data. The census and mobility data are official data from German authorities and are therefore assumed to be accurate. Nevertheless, we have checked these data for anomalous values. For example, in 75 municipalities the population is zero. These municipalities are "municipality-free areas", i.e. municipalities

SCIENTIFIC DATA (2019) 6:243 | https://doi.org/10.1038/s41597-019-0233-0

without inhabitants. These are usually municipalities with nature reserves or military stations (for more details please refer to Weinand et al.6).

Technical photovoltaic potential. A validation of the technical PV potential has been performed for two intermediate results, using cadastral data from the federal state of Baden-Württemberg, which comprises about 10% of the area and 13% of the residential buildings in Germany. The validation shows that the number of residential buildings from the Baden-Württemberg cadastral building data differs by just 1.65% from the publicly available statistical data. It also shows that the assumed residential building sizes as well as the total ground floor area for Baden-Württemberg agree with the building sizes extracted from the cadastral data. It can thus be assumed that the employed assumptions as well as the statistical data used represent the German building stock adequately (see Mainzer et al.7 for more details).

Technical wind potential. The onshore wind cost-potentials were validated with other studies in the literature. With central key assumptions of this method adjusted to reflect those in BWE14, the results in both cases are very similar. Our study estimates a total potential area of about 41,000 km² compared to the BWE's 46,000 km². In terms of the available area for wind energy in Germany, the present work with 41,613 km² also agrees well with the UBA¹⁵ study, which determined 49,000 km². The reason for the lower results in our case lies in the more conservative assumed suitability factors and offset distances from obstacles.

The discrepancy is significantly larger in relation to the installable power and generated energy, however: UBA conclude that about 1,190 GW could be installed to generate about 2,900 TWh/a (i.e. average full load hours of about 2400), compared to 367 GW and 855 TWh/a (full load hours of 2329)⁸. Whilst some of the discrepancy can about 2400), compared to 367 GW and 855 TW h/a (full load notus of 2529). Whilst some of the discrepancy can be explained by the difference in the determined areas, the majority is probably due to the significantly higher tur-bine densities employed by UBA. The high overall turbine density is achieved through an aggregation procedure that clusters nearby polygons together (similarly to Fueyo *et al.*^{16,17}). For Baden-Württemberg, McKenna *et al.*¹⁸ found a technical potential of 29 to 41 GW at costs between 6 and 21 €ct/kWh. This compares well to the technical potential calculated in our study of 34 GW at average costs of 9.5 €ct/kWh. Similarly, McKenna *et al.*¹⁹ found a potential and land area for Germany of 707 TWh and 35,700 km² respectively in a similar study at the European level. The only other recent study that considered the whole of Germany was EEA (www.eea.europa.eu/), which did not differentiate between different land use categories. This explains the very high total technical potential of 4000 TWh/a identified in Germany. The calculated fraction of this potential, which by 2020 should be competitive, was found to be 258 TWh/a, which corresponds to generation costs of about 7 ct/kWh in McKenna et al.⁸. A direct comparison of the generation costs with EEA is not possible, however, as they were not divulged for individual countries or states.

Achievable hydrothermal temperature. As already described above, the data for the achievable hydro-thermal temperature were transferred based on the GeotIS project results¹³. On the one hand, visual inspection of the maps in Fig. 4 shows that the data was transferred correctly. On the other hand, this has also been confirmed by a sampling check.

Cluster data. In the cluster analysis, 26 cluster validation criteria were applied to determine the number of clusters⁶. Further validation methods came to the same results as the 26 cluster criteria.

Code availability

Microsoft Excel version 2010 and ArcGIS version 10.1 were used to determine the technical PV and wind potentials. QGIS version 2.14.10 was used for the transfer of wind and hydrothermal potential to the municipality level. MATLAB version 2017b and Microsoft Excel version 2016 were used to compile the census and mobility data. Microsoft Excel or another program for processing CSV or XLS files is required to process the data set provided with this data descriptor. The codes for generating the data can be made available on request.

Received: 17 April 2019; Accepted: 12 September 2019; Published online: 30 October 2019

- References
 1. McKenna, R. The double-edged sword of decentralized energy autonomy. Energ. Pol. 113, 747–750, https://doi.org/10.1016/j. enpol.2017.11.033 (2018)
- enpol.2017.11.033 (2018).
 McKenna, R., Bger, T. & Fichtner, W. Energieautarkie ausgewählte Ansätze und Praxiserfahrungen im deutschsprachigen Raum. *uwf* 22, 241–247, https://doi.org/10.1007/s00550-014-0339-y (2014).
 McKenna, R., Bertsch, V., Mainzer, K. & Fichtner, W. Combining local preferences with multi-criteria decision analysis and linear optimization to develop feasible energy concepts in small communities. *Eur. J. Oper. Res.* 268, 1092–1110, https://doi.org/10.1016/j. ejor.2018.01.036 (2018).

- ejor. 2018.01.036 (2018).
 4. Alhamwi, A., Medjroubi, W., Vogt, T. & Agert, C. Modelling urban energy requirements using open source data and models. *Appl. Energy*. 231, 1100–1108, https://doi.org/10.1016/j.apenergy.2018.09.164 (2018).
 5. Mainzer, K. Analyse Und Optimierung Urbaner Energiesysteme Entwicklung Und Anwendung Eines Übertragbaren Modellierungswerkzeugs Zur Nachhaltigen Systemgestaltung (KIT Karlsruhe, 2019).
 6. Weinand, J. M., McKenna, R. & Flchtner, W. Developing a municipality typology for modelling decentralised energy systems. Util. Pol. 57, 75–96, https://doi.org/10.1016/j.jup.2019.02.003 (2019).
 7. Mainzer, K. *et al.* A high-resolution determination of the technical potential for residential-roof-mounted photovoltaic systems in Germany. Sol. Energy 105, 715–731, https://doi.org/10.1016/j.solenet.2014.04.015 (2014).
 8. McKenna, R., Hollnaicher, S. & Fichtner, W. Cost-potential curves for onshore wind energy. A high-resolution analysis for Germany. Appl. Energy 115, 103–115, https://doi.org/10.1016/j.apenergy.2013.10.030 (2014).

SCIENTIFIC DATA (2019) 6:243 | https://doi.org/10.1038/s41597-019-0233-0

- Weinand, J. M., Mainzer, K. & McKenna, R. Spatial high-resolution socio-energetic data for municipal energy system analyses. *figshare*, https://doi.org/10.6084/m9.figshare.7964609.v5 (2019).
 Weinand, J. M., Kleinebrahm, M., McKenna, R., Mainzer, K. & Richtner, W. Developing a combinatorial optimisation approach to design district heating networks based on deep geothermal energy. *Appl. Energy* 251, 113367, https://doi.org/10.1016/j.apenergy.2019.113367 (2019)
- (2019).
 (2019).
 11. Kraftfahrt-Bundesamt. Fahrzeugzulassungen (FZ). Bestand an Kraftfahrzeugen und Kraftfahrzeuganhängern nach Gemeinden, https://www.kba.de/SharedDocs/Publikationen/DE/Statistik/Fahrzeug/Z/2017/EZ_2017_gdf.pdf.jsessionid=031266C8B0663 B938AA41410071E0075.ltwe21303?_blob=publicationFile&v=3 (2017).
 12. Mainzer, K., Killinger, S., McKenna, R. & Fichtner, W. Assessment of rooftop photovoltaic potentials at the urban level using publicly available geodata and image recognition techniques. Sol. Energy 155, 561-573, https://doi.org/10.1016/j.solener.2017.06.065 (2017).
 13. Agemar, T. et al. The Geothermal Information System for Germany Geottis. ZDGG, 165,129-144 (2014).
 14. Bundesverband WindEnergie e.V. (BWE). Potenzial der Windenergienutzung an Land, https://www.wind-energie.de/fileadmin/redaktion/dokumente/publikationen-oeffentlich/theme.n/01-mensch-und-unwelU03-naturschutz/bwe_potenzialstudie_kurzfassung_2012-03.pdf (2011).
 15. Umwelbundesamt (UBA). Potenzial der Windenergie an Land-Studie zur Ermittiune des hundesweiten Efficien. und Leitimeentermenden.

- 15. Umweltbundesamt (UBA). Potenzial der Windenergie an Land: Studie zur Ermittlung des bundesweiten Flächen- und Leistungspotenzials der Windenergienutzung an Land, https://www.umweltbundesamt.de/sites/default/files/medien/378/publikationen/potenzial_der_
- windenergie.pdf (2015).
 users in the second se
- Freyo, N., Sattz, F., Kolfigues, M., Johnstes, C. & Zolpazo, C. The use of cosgeneration curves for une analysis of while electricity costs in Spain. Appl. Energy 88, 733–740. https://doi.org/10.1016/j.apenergy.2010.09.008 (2011).
 McKenna, R., Gantenbein, S. & Fichtner, W. Determination of cost-potential-curves for wind energy in the German federal state of Baden-Württemberg. Energ Pol. 57, 194–203. https://doi.org/10.1016/j.enpol.2013.01.043 (2013).
 McKenna, R., Hollnaicher, S., Ostman v. d. Leye, P. & Fichtner, W. Cost-potentials for large onshore wind turbines in Europe. Energy 83, 217–229. https://doi.org/10.1016/j.energy.2015.02.016 (2015).
 Statistisches Bundesamt. Statistik lokal. Dataen für die Kreise, kreisfreien Städte und Gemeinden Deutschlands. Dataset on CD (2011).
 Kollschneit, M. & Sulfie a. Demutymberg. Energiteinge In Date für die Kreise, kreisfreien Städte und Gemeinden Deutschlands. Dataset on CD (2011).
- Salastickies runnes and Salastickies in the second states and states and states and states and candidate countries. Sol. Energy 81, 125–1305, https://doi.org/10.1016/j.solener.2006.12.007 (2007).
 Unger, H. & Skiba, M. Solare Strahlung auf geneigte Flächen. Eine Ergänzung zu den Meteosat-Strahlungskarten. Sonnenenergie 1, docs flackster in the second states and st
- 48-50 (1998)

Acknowledgements

The authors gratefully acknowledge the financial support of the PhD College "Energy and Resource Efficiency" (ENRES), from the Federal State of Baden-Wuerttemberg, for funding the first author's PhD studentship. The second author gratefully acknowledges the support of the Smart City Accelerator project (https:// smartcitiesaccelerator.eu/about-smart-cities-accelerator/), which supported his contribution to this article. Furthermore, we acknowledge support by the KIT-Publication Fund of the Karlsruhe Institute of Technology. The usual disclaimer applies.

Author contributions

J.M.W. collected and synthesized the data, wrote the majority of the manuscript and created/modified the figures. R.M. and K.M. determined the original data on wind and photovoltaic potential, wrote parts of the manuscript and supervised the ENRES project.

Competing interests The authors declare no competing interests.

Additional information

Correspondence and requests for materials should be addressed to J.M.W.

Reprints and permissions information is available at www.nature.com/reprints.

Publisher's note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons license, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons license and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this license, visit http://creativecommons.org/licenses/by/4.0/.

The Creative Commons Public Domain Dedication waiver http://creativecommons.org/publicdomain/zero/1.0/ applies to the metadata files associated with this article.

© The Author(s) 2019

Journal of Cleaner Production 238 (2019) 117756



Contents lists available at ScienceDirect

journal homepage: www.elsevier.com/locate/jclepro



Assessing the potential contribution of excess heat from biogas plants towards decarbonising residential heating



Jann Michael Weinand ^a, Russell McKenna ^{a, b, *}, Katharina Karner ^c, Lorenz Braun ^d, Carsten Herbes ^e

^a Chair of Energy Economics, Karlsruhe Institute of Technology, Germany

^b DTU Management, Technical University of Denmark, Denmark

^c Institute of Energy, Transport and Environmental Management, FH Joanneum, Austria
^d Nuertingen-Geislingen University, Faculty of Agriculture, Economics and Management, Germany

^e Nuertingen-Geislingen University, ISR, Nuertingen, Germany

ABSTRACT

Article history: Received 8 March 2018 Received in revised form 18 May 2019 Accepted 22 July 2019 Available online 30 July 2019

ARTICLE INFO

Handling Editor. Yutao Wang

Keywords: Biogas plant Excess heat Energy autonomy CO₂ abatement costs District heating This paper analyses the technical potential for utilising excess heat from biogas plants, in order to supply local settlements through district heating. Based on a survey of around 600 biogas plant operators, the fractions of excess heat from the cogeneration units in these plants are analysed. A heuristic is developed to match biogas plants (heat sources) with local settlements (heat sinks) in order to determine a least-cost district heating supply for residential buildings. Two criteria are employed, namely the CO_2 abatement costs and the payback period, which represent the macro- and microeconomic perspectives respectively. Based on the survey, the mean fraction of excess heat is 40%, which is negreement with other empirical studies. Extrapolating this fraction to the German biogas plant stock, which is selected as a case study, leads to technically feasible CO_2 savings of around 2.5 M(CO_2/a . Employing the criteria of $200 \in [/CO_2 and below a payback period of 9 years. This represents about 0.25% of the total German <math>CO_2$ emissions in 2016 or around 2.5% of all CO_2 in residential buildings. Alternative threshold values of 80 $\in [/CO_2 and 5 years payback period reduce the carbon reduction potential to about 0.5 MtCO_2 and 0.75 MtCO_2 respectively. These relatively high average costs are related to the typically low population density in rural regions where biogas plants are located. These potentials are concentrated in around 3.500 of 11,400 municipalities, where district heating from biogas plants could reduce <math>CO_2$ emissions per capita by an average of 250 kgCO_2 /a and cover 12% of the total heating demand. Apart from a methodology that the use of excess heat in biogas plants can contribute to global decarbonisation.

© 2019 Elsevier Ltd. All rights reserved.

1. Introduction

The expansion of renewable energy technologies (RETs) is largely being driven by private individuals, farmers and energy cooperatives in the context of *community energy* (Klaus Novy Institut e.V. & trend:research 2011; trend:research 2017). This trend involves these actors investing in and/or operating RETs, including wind, solar and bioenergy plants, in some cases also buying the local energy infrastructure (gas, electricity, heating networks) back from the local utility. They are mainly motivated by

 Corresponding author. DTU Management Engineering, Produktionstorvet, Building 426, Room 130A, 2800 Kgs. Lyngby, Denmark.
 E-mail address: rkennad@ttu.dk (R. McKenna).

https://doi.org/10.1016/j.jclepro.2019.117756 0959-6526/© 2019 Elsevier Ltd. All rights reserved. a desire to 'take control' of their local energy supply (system) and thus become more independent from centralised markets and energy suppliers (for reviews, cf. Müller et al., 2011; Rae and Bradley, 2012). Many community energy projects declare an objective of energy autonomy, with some examples in Germany including the 100% Renewable Energy Regions and the Bioenergy Villages. Despite possibly having potentially negative system wide impacts (McKenna, 2018), municipalities typically define autonomy on an annual basis and for electricity alone.¹ The costs of these RETs were

¹ At the time of writing only one German municipality is known to the authors that is aiming to be completely energy autonomous, namely Bordelum, cf. http:// www.sonnenseite.com/de/energie/norddeutsche-gemeinde-stellt-komplett-aufemeuerbare-energien-um.html.

	Nomencla	ture	HD _{SAtotal} HDC _{SA}	Total heat demand in a settlement area (kWh) Specific heat distribution costs of the excess heat
	af	Annuity factor (–)		(€/GJ)
	a ₁₁₁	Area size of a settlement area with continuous urban fabric (m^2)	I _{con,SA}	Investment in the pipeline for connecting the biogas plant and the settlement area (\mathfrak{S})
	a ₁₁₂	Area size of a settlement area with discontinuous urban fabric (m^2)	I _{DHG,SA}	Investment for the building or the densification of the district heating grid in the settlement area (\in)
	A _{BC}	Share of brown coal in heat supply for settlement area (%)	I _{SA}	Total investment to use the excess heat from the biogas plant to supply the settlement area (\in)
	A _{EL}	Share of electricity in heat supply for settlement area (%)	IR	Investment rate for the district heating pipeline (\in /m)
	A _{Gas}	Share of gas in heat supply for settlement area (%)	LHD _{SA}	Linear heat density in a settlement area (GJ/m)
	A _{HC}	Share of hard coal in heat supply for settlement area (%)	LP _{BGP} LP _{SA}	Excess heat profile of a biogas plant (-) Heat demand load profile of a settlement area (-)
	A _{Oil}	Share of heating oil in heat supply for settlement area (%)	LS ₁ LS ₂	Living space in a settlement area (m^2) Adjusted living space in a settlement area, if LS_{rotal1} is
	BSA _{SA}	Total building space of a settlement area (m^2)	2	not equal to LS_{total2} (m ²)
	CO2 _{costs}	CO_2 abatement costs (\in /tCO ₂)	LS _{total1}	Calculated sum of the living space over all
	CO2 saved SA	Saved CO_2 emissions in a settlement area (tCO_2)	Lottari	settlements in a municipality (m ²)
	C1	Construction cost parameter per length unit for	LS _{total2}	Measured total living space in a municipality (m ²)
		district heating network (€/m)	na	Number of apartments in a settlement area (-)
	C2	Construction cost parameter per area unit for district heating network $(\in m^2)$	n _b	Total number of buildings in a square kilometre of the census data grid $(-)$
	CR	Coverage ratio of the heat supply (%)	n	Number of buildings in a settlement area with
	df	Discount factor (%)		continuous urban fabric $(-)$
	den	Average diameter of the district heating pipelines (m)	116110	Number of buildings in a settlement area with
	DISSA	Distance between a biogas plant and a settlement		discontinuous urban fabric (–)
	5/1	area (m)	NPV _{SA}	Net present value for the supply of a settlement with
	е	Plot ratio, used to categorize typical city districts (–)		district heating (\in)
	EFBC	CO ₂ emission factor of brown coal (kgCO ₂ /kWh)	Р	Total population of a settlement area (Inhabitants)
	EF _{EL}	CO ₂ emission factor of electricity (kgCO ₂ /kWh)	PD _{SA}	Population density of a settlement area
	EF _{Gas}	CO ₂ emission factor of gas (kgCO ₂ /kWh)		(Inhabitants /km²)
	EF _{HC}	CO ₂ emission factor of hard coal (kgCO ₂ /kWh)	PP	Payback period (a)
	EF _{Oil}	CO ₂ emission factor of oil (kgCO ₂ /kWh)	P _{th}	Thermal power of a biogas plant (kW)
	EH _{BGP}	Share of available excess heat in a biogas plant (%)	Qsell	Amount of excess heat that is supplied by the biogas
	Enused	Sum of expenses for the supply of a settlement with	DEL	Sum of revenues for the supply of a settlement with
	E	district heating (\in)	KL V _{SA}	Sum of revenues of the supply of a settlement with district heating (ϵ)
	r_1	Number of residential buildings in a federal state/	SBio	Proportion of heat supplied by district heating (%)
	C .	Number of residential buildings in a federal state i	S _{DH}	Proportion of heat supplied by district fleating (%)
	12	municipality with two apartments (S_{GT}	environmental heat (%)
	F.	Number of residential buildings in a federal state/	San	Proportion of heat supplied by solar energy $(%)$
	13	municipality with three or more anartments $(-)$	Sum	Proportion of heat supplied by solar energy (%)
	Fmean	Mean living space per apartment in a federal state $\binom{m^2}{m^2}$	SLP _{BGP}	Standardised best load profile of a biogas plant $(-)$
	F _{mean,m}	Mean living space per apartment in a municipality	SR HD	Supply ratio of the heat supply (%)
	E	(III^{-}) Total living space in a federal state (m^{2})	w	Effective width, relationship between an area and the
	r _{total}	Iotal living space in a federal state (m [*])		Average number of apartments in a residential
ſ	nu _{mean}	$(12Mb/m^2)$	х	hybriding with more than two anartments (
[HD-/	(KVVII/III ⁻) Total heat demand in a settlement area, reduced by		building with more than two apartments (–)
	nDSA	the share of heat generation technologies that should not be replaced by the excess heat (kWh)		

historically higher than those of conventional technologies, meaning they relied on subsidies to be economic. However, recent rapid reductions in costs for photovoltaics and batteries have provided a renewed incentive for attempts to become energy autonomous at a local scale (Nykvist and Nilsson, 2015).

omous at a local scale (Nykvist and Nilsson, 2015). Many studies have techno-economically analysed the scope to achieve a local energy supply from renewable sources (e.g. Jenssen et al., 2014; Schmidt et al., 2012 and Burgess et al., 2012). Most conclude that a completely autonomous energy supply is only feasible in rural municipalities with large bioenergy resources, and even then, large storage capacities are required which lead to high costs. In the context of decentralised energy supply, a related stream of research is concerned with (industrial) excess heat and the possibilities of utilising this as an energy input, e.g. for space
heating. Several studies in this area have analysed the technical and/or economic potential of excess heat from industry, often based on emissions data from individual industrial plants. Noteworthy are the studies McKenna and Norman (2010), Fang et al. (2013, 2015), Brueckner et al. (2014), Miró et al. (2015, 2016), Bühler et al. (2017), and the ongoing Heat Roadmap Europe project (Persson et al., 2014).

One additional source of excess heat potential are biogas plants, which valorise organic matter into biogas through fermentation (cf. section A.1 in the appendix). This biogas is typically combusted in a cogeneration unit, whereby the electricity is generally fed into the grid. Within Europe, Germany accounts for around 50% (329 PJ in 2015) of the total biogas production (Scarlat et al., 2018). A legal requirement was introduced in Germany with the Renewable Energy Sources Act (EEG) 2012, that new biogas plants must utilise at least 25% of their excess heat in the first year of operation and 60% thereafter (Mergner et al., 2013). Despite this, there are many operating plants, some of which were commissioned before this time, with much lower levels of heat utilisation. Section A.1 in the appendix gives an overview of the current status of the German biogas industry.

Against this background of decentralised energy systems, aspirations for local energy autonomy and excess heat potentials, this study analyses the technical and economic potential for recovering excess heat from biogas plants in Germany to supply local buildings with low-carbon space heating. Germany is selected as the case study due to the high biogas production mentioned above. The objective is to determine the amount of heat that could technically be recovered as well as its associated costs, in order to give indications of promising locations. The paper addresses the following research questions:

- How does the excess heat from biogas plants match the existing heat sinks?
- What contribution can the use of biogas excess heat make to the energy autonomy of municipalities?
- In which locations is the utilisation of biogas excess heat economically interesting for investors?

The employed methodology involves input from a survey of around 600 biogas plant operators for a GIS-based analysis of heat sources and sinks. Thereby, a developed heuristic matches and connects the heat sources and sinks with district heating pipelines based upon a least-cost approach. The results are therefore of relevance to local decision makers, biogas plant operators and researchers in the field of energy system analysis. In particular the quantitative results offer an indication of the costs and related CO_2 saving potentials on a local level, in order to underpin local decision-making.

The remainder of the paper is structured as follows. The following section 2 gives an overview of relevant literature on the subject of designing district heating networks (DHNs). The subsequent section 3 outlines the employed methodology, including the plant operator survey, the approach to spatially locating the analysed plants, the determination of the residential building heat demand at a local level and the heuristic to match heat sources and sinks. Then section 4 presents and discusses the results, first for the survey sample and then for a tentative scale-up for the whole of Germany. The paper closes in section 5 with conclusions and an outlook.

2. Literature review

In the following literature review, Table 1 provides a summary of the methodology and problem sizes used in the evaluated studies. The distinction between bottom-up and top-down studies is made in such a way that bottom-up studies, for example, are based on individual buildings and roads, while the top-down approach uses heat and population densities to calculate potentials. District heating (DH) pipelines with length-dependent costs are required to utilise the excess heat of biogas plants in residential settlements. In section 2.1, studies on small-scale DHNs will be discussed before section 2.2 focuses on large-scale DHNs.

2.1. Small-scale district heating networks

Approaches for the design and dimensioning of small-scale DHNs have already been presented and discussed in numerous studies. In Chinese (2008), DH and cooling systems are designed using a mixed integer optimisation model. Central and decentralised heat production are combined while taking network costs into account. A mixed integer linear program was developed by Casisi et al. (2009) to determine the optimal layout of a DH system and the optimal operating strategy for a distributed cogeneration system while minimising the total cost of ownership and operation. Dobersek and Goricanec (2009) identify the optimal tree branch path of a DHN in an urban area, under consideration of construction costs, pump and electricity costs. The optimal network for a complete supply of all consumers is determined, whereby the locations of the heat source and consumers are defined in advance. Some specific nodes that do not represent consumers are used for branches of district heating pipelines (DHPs). Delangle et al. (2017) investigate the extension of existing DHNs. Besides DH, gas boilers, biomass boilers, heat pumps and heat storages are included in a mixed integer linear optimization. The model can determine the optimal investment plan for a DHN extension and can be applied to other case studies. In the latter case, the existing DHN and the buildings to be connected must be known. Karschin and Geldermann (2015) present an optimisation model which determines the location and capacity of a bioenergy plant and the associated DHN. Possible locations are defined beforehand as the locations of the plant. The study serves to support bioenergy villages in planning their energy systems. The fuzzy optimisation model in Balaman and Selim 2016 designs biomass based renewable energy supply chains and DH systems with heat storages in specific regions. Another study on small scale DHNs is that of Coss et al. (2018). In their article, a multiobjective optimisation model is presented, which determines the energy supply of a biomass plant for DH purposes. However, the model does not optimise the design but only the operation mode of the DHN. Marty et al. (2018) highlight the relevance of a simultaneous optimisation of a DHN and an organic rankine cycle as parts of a geothermal plant. The location of the DH plant is specified in advance for the optimisation model, and one consumer is already connected to the DHN. In the case of the utilisation of excess heat from biogas as in our study, the connection of a customer in advance is not meaningful, since the use of excess heat is not always worthwhile. Karner et al. (2018) investigate heat flexibilities in cities through excess heat from industry. One of the presented options is the use of the heat in DHNs. Nielsen (2014) develops an algorithm for the economic evaluation of possible network expansions based on existing DHNs. Geographical data are used for cost calculations for heat generation, distribution and transmission. In order to reduce model complexity, Fazlollahi et al. (2014) design the DHN on the basis of cluster analyses. The clustering of urban areas is intended to reduce the model complexity of a subsequent optimisation model by aggregating the energy demand and DH distribution costs in the resulting urban districts. In the investigated city there are already 13 clusters needed to adequately represent the demand and costs

In many cases, the studies do not address the applicability of the

J.M. Weinand et al. / Journal of Cleaner Production 238 (2019) 117756

4 Table 1

Studies about designing and dimensioning DHNs (R = residential sector, C = commercial sector, I = industrial sector).

Churches Marchestellerum Danken un eintern Martinen fahrunde beröffelterum ein Carakan Danken

 Study	Methodology	Bottom-up or top-down methodology	Network building or expansion	Sector	Region	Number of customers/ nodes/sinks	Number of suppliers/sources
Chinese (2008)	Optimisation model	Bottom-up	Building	R, C, I	Urban area (Italy)	11	1
Casisi et al. (2009)	Optimisation model	Bottom-up	Building	С	Urban area (Italy)	6	9
Dobersek and Goricanec (2009)	Optimisation model	Bottom-up	Building	n.a.	Urban area	23	1
Fazlollahi et al. (2014)	Cluster analysis and optimisation model	Top-down	Building	n.a.	Urban area	13	n.a.
Nielsen (2014)	Heuristic	Bottom-up	Expansion	n.a.	Arbitrary Danish municipality	n.a.	n.a.
Karschin and Geldermann (2015)	Optimisation model	Bottom-up	Building	R	Rural area	71	1
Bordin et al. (2016)	Optimisation model	Bottom-up	Expansion	R, C	Urban area	1,000	1
Balaman and Selim (2016)	Optimisation model	Bottom-up	n.a.	R	Urban area (Turkey)	10	1
Delangle et al. 2017	Optimisation model	Bottom-up	Expansion	n.a.	Urban area	31	1
Unternährer et al. (2017)	Cluster analysis and heuristic	Top-down	Building	n.a.	Urban area (Switzerland)	410	n.a.
Coss et al. (2018)	Optimisation model	Top-down	Just operation of the DHN	R, C	n.a.	n.a.	n.a.
Guelpa et al. (2018)	Optimisation model	Bottom-up	Expansion	R, C, I	Urban area (Italy)	182	5
Karner et al. (2018)	Heuristic	Bottom-up	Expansion	R, C, I	Urban area (Austria)	n.a.	n.a.
Marquant et al. (2018)	Cluster analysis and optimisation model	Bottom-up	Building	R, C	Urban area (Switzerland)	32	n.a.
Marty et al. (2018)	Optimisation model	Bottom-up	Building	n.a.	n.a.	9	1
Soltero et al. (2018)	Heuristic	Top-down	Just cost calculation	R	Rural areas	499	n.a.
This Study	Heuristic	Top-down	Building	R	Whole Germany	38,414	10,446

methods to large-scale problems with many heat suppliers and it can be assumed that the models are not suitable for this purpose (Casisi et al., 2009; Dobersek and Goricanec, 2009; Karschin and Geldermann, 2015; Marty et al., 2018). In addition, the studies show that the investigation of small-scale problems with the applied methods already leads to computational and time restrictions and is therefore not suitable for large-scale applications (Chinese, 2008; Fazlollahi et al., 2014; Balaman and Selim 2016). Chinese (2008) recommends the use of heuristics instead of optimisation to solve larger problems in a reasonable time. Other studies require extensive data for the application of the models, which are not available for large-scale problems and/or for application in Germany (Nielsen, 2014; Karner et al., 2018).

2.2. Large scale district heating networks

In the literature are only a few studies which design DHNs for large regions. Soltero et al. (2018) follow a similar approach to our study. The authors develop a methodology to design biomass DH systems in rural areas and examine 499 municipalities in Spain that are not connected to the gas network. In the region under consideration, DH from biomass can save 5.4 Mt_{CO2} per year. The study shows several differences to our paper. Firstly, the heuristic estimates the length of the DHN instead of determining the length by geographical methods. Furthermore, no existing plants are considered, but the potentials of new plants are estimated. In addition, the approach is only applicable to rural areas and the investment decision is made for entire municipalities, while in our analysis it is possible to connect parts of municipalities, i.e. settlements, to the DHN. Marquant et al. (2018) develop a combined clustering schema to overcome time-constraints while estimating the potential for DHNs. In this study, however, only 32 buildings are examined in a case study. These buildings are divided into four clusters to reduce the variables for the subsequent DH optimisation model. However, the calculation for the whole model takes even in this case more than 250 h. Unternährer et al. (2017) also use a

clustering approach before determining the optimum design of a geothermal DHN for the resulting clusters. The cluster analysis is performed using an optimisation model. On the computer used in the study with 8 cores and 32 GB RAM, the model could not be executed from a cluster number of 420 clusters or larger due to memory restrictions. The extension of existing DHNs is subject in the study of Bordin et al. (2016). The network can be expanded by potential pipelines and customers in an optimisation model to maximise the net profit. The tree configuration of the network is optimised with one heat supply plant as starting point. However, the model is not suitable for the purposes of our study, as only one network with one heat generation plant can be considered and existing networks are extended. The same applies to the study by Guelpa et al. (2018), in which large DHNs are also expanded with the help of optimisation.

The evaluation of the literature on the design of DHNs shows that a large problem size of the kind described in our article has not yet been investigated in any study. In some methods, even smaller DHNs encounter temporal or computing problems (Unternährer et al., 2017; Marquant et al., 2018). In addition, the methods developed so far are not applicable to the case presented in this paper (Soltero et al., 2018), for example due to the need for extensive building data (Guelpa et al., 2018). Furthermore, no procedure has yet been developed to design many DHNs at the same time. The methodology developed in our study is therefore not only applied to the novel case of utilising excess heat from all German biogas plants, but also represents a further development with regard to dealing with large problems in DH design. Due to the evaluated literature and the time and calculation restrictions mentioned therein, the method developed in our study is not an optimisation but a heuristic. The developed heuristic differs in the following points from the problem formulations in the above mentioned studies:

• It is evaluated whether a connection of a settlement is worthwhile and "to which percentage" a settlement should be

J.M. Weinand et al. / Journal of Cleaner Production 238 (2019) 117756



Fig. 1. Overview of the algorithm developed and used in this study to select biogas plants (BGP) and settlement areas for DH supply

connected. This can result in a biogas plant not being connected to heat consumers, which means that the excess heat remains unused.

- A very large problem has to be solved (with about 50,000 biogas plants/settlements).
- Subtrees/independent DHNs are allowed (up to 10,000, as there are as many sources/biogas plants).
- Possible pipelines go from any biogas plant/settlement to any other nearby settlement.

Besides Germany, the data used to represent settlement areas are available for 38 other European countries (EEA, 2018). The heuristic could be extended to any country with similar available data. As demonstrated in Scarlat et al. (2018), there is also a lot of biogas production in other countries than Germany.

3. Method and approach

This study uses different methods in a multi-stage approach. The summarizing representation in Fig. 1 is intended to provide a better understanding of the algorithm explained in this section. Firstly, the results of an extensive survey of biogas plant operators in Germany are taken into account with regard to their heat utilisation rates (cf. section 3.1, Herbes and Halbherr, 2017). Secondly, 10.446 biogas plants as well as the 38,414 CORINE Land Cover (CLC) settlement areas in Germany are examined with regard to their technical and geographical characteristics (cf. section 3.2). Subsequently, the local demand for heat in buildings in all German settlement areas and the excess heat availability of biogas plants are calculated (cf. section 3.3). In order to provide a least cost solution, options to integrate the excess heat from the biogas plants into DHNs are explored. In this context, in particular the distances between the biogas plants and the settlement areas are determined. Based on published methods for DH systems assessment and dimensioning, the possible CO2

savings and associated costs as well as the payback periods are determined for combinations of biogas plants and their nearest residential areas. By focussing on the connections with the lowest CO₂-abatement costs/payback period, the most environmentally and economically attractive locations for a district heating network development are identified (cf. section 3.4).

5

3.1. Biogas plant survey

In the summer of 2016, an online survey of those members of the German Biogas Association that run a biogas plant was conducted, resulting in a gross sample of 2,724 operators² (Herbes et al., 2018a). After the development of the questionnaire, which was a joint undertaking of Nuertingen-Geislingen University and the German Biogas Association (GBA), extensive cognitive pretesting with external biogas experts, GBA staff and plant opera-tors was carried out before fielding the survey. To decrease nonresponse bias after the first phase, a telephone campaign in those federal states that were underrepresented in the answers was conducted. These efforts resulted in a final data set of n = 602 plant operators, which is equivalent to a response rate of 22% (according to response rate 2 (RR 2), cf. AAPOR (2015)). Regarding the distribution of federal states, the sample shows a small over-representation of Bavaria and Baden-Wuerttemberg (cf. Fig. 5). Regarding size and commissioning year, the sample is statistically representative of the entire German biogas plant stock. Amongst other things, the data set includes location, plant size, percentage of the already used heat, utilisation paths and various data on prices and price models, which are not relevant for the present study (cf. Section A.2 in the appendix and Herbes and Halbherr, 2017 for more information).

² For an overview of the historical development and current status of the German biogas sector, the interested reader is referred to Appendix A.1.



Fig. 2. Exemplary section of the "111" and "112" CORINE areas (left part) and the intersection with the ETRS89-LAEA grid (right part).

3.2. Biogas plant register and CORINE land use data

6

The plants from the survey can be identified and mapped in the Energymap plant register (Engel, 2015) on the basis of their year of commissioning, postcode and nominal power. Further and more detailed information could then be taken from the plant register, such as the full load hours in recent years. The full load hours in this study represent the total operating hours recalculated to full load hour equivalents. The data from the survey were used for heat utilisation (cf. section 4.2).

The Energymap plant register contains the coordinates of the biogas plants with a maximum error of 3 km. Therefore, the locations of the 10,446 plants in Germany could be mapped with the help of the geoinformation system QGIS. The plants from the survey are a subset of these 10,446 plants. CLC data from the European Environment Association (EEA, 2016) were used as a source for the settlement areas. The shapefiles of these areas for Germany are provided by the Federal Office of Cartography and Geodesy (Lenk et al., 2017a). Urban areas are distinguished according to the density of the urban fabric, into "continuous urban fabric" (denoted with the number "111") and "discontinuous urban fabric" areas (denoted with the number "112"). The boundary between the 111 and 112 areas is mainly determined by the presence and quantity of vegetation (EEA, 1995). Overall, the German settlement areas are divided into 38,414 of these areas. The left part of Fig. 2 shows an exemplary section of these areas.

3.3. Heat demand and generation

The method for determining the heat demand of the settlement areas and the excess heat availability of the biogas plants is explained in sections 3.3.1, 3.3.2 and 3.3.3.

3.3.1. Heat demand

Data on the building stock in Germany were taken from the census of the Federal Statistical Office to determine the heat demand in the settlement areas. For more information on determining the census data, see Statistisches Bundesamt (2015b). The census data include data on building age, building type and share of DH, and are assigned with the help of the Lambert-Azimuthal-Equal Area Projection (ETRS89-LAEA) into INSPIRE-compliant 1-km² grid cells (Statistisches Bundesamt, 2016b). Therefore, the CLC settlement areas must also be assigned to this grid. For this purpose, the CLC settlement areas were intersected with the ETRS89-LAEA grid. The result can be seen on the right part of Fig. 2.

The black dots in the CLC areas represent the area centroids. Industrial and commercial areas are excluded from this analysis due to the lack of data for industrial heat demand.

After the allocation of the CLC sub-areas to the square kilometres, the census data were assigned to the CLC areas. The data per square kilometre must be distributed to all CLC sub-areas in the grid. In addition to taking into account the area share, a distinction is also made between 111 and 112 areas (cf. section 3.2). The 111 areas represent settlement areas of which on average 90% are covered with buildings. For the 112 areas, the proportion of building area is on average 65% (EEA, 1995). These values are used as building densities. Since the census data is related to buildings, this information can be used to divide the census data into areas. Section A.3 of the appendix shows the calculation of the heat demand $HD_{SAtotal}$ in the individual settlements using these census data in detail. In many settlement areas there are buildings whose heat demand is covered by DH systems. For these settlements, the heat demand HD_{SA} is deducted from the total heat demand $HD_{SA_{const}}$. Since it is not known which building types have DH systems, the share of DH systems S_{DH} in the settlement is deducted from the total heat demand. In addition, the heat demand is reduced by the share of technologies that should not be replaced by DH with biogas excess heat, such as renewables (cf. Eq. (1)).

$$HD_{SA} = HD_{SA total} \cdot (1 - S_{DH} - S_{WP} - S_{Bio} - S_{SE} - S_{GT})$$
(1)

 S_{WP} stands for the proportion of heat supplied by wood pellets, S_{Blo} for biomass, S_{SE} for solar energy and S_{GT} for geothermal and other environmental heat. These shares are not to be replaced because the emission factors are lower or not significantly higher than those for DH from biogas plants (LfU, 2016). Apart from the share of DH (municipality level), all shares are based on figures at federal state level (Statistisches Bundesamt, 2016a). Eq. (1) is based on the assumption that the heat demand of buildings which already have a building connection for DH is completely covered by the existing DH. With the census of 2011 as a basis, the share of DH in settlement areas in Germany ranges from 0 to 95% (mean value is 2.65%³).

In the next step, a heat load profile LP_{SA} is assigned to each settlement area (cf. Eq. (2)). For this purpose, a standardised profile of a DHN SLP_{HD} with an hourly resolution is used, which is adapted

³ In the calculation of this value, all municipalities are equally weighted. Therefore it must not be confused with the proportion of district heating in the residential heat supply in Germany of 13.8% (Euroheat and Power 2017).

according to the previously determined heat demand HD_{SA} (for the standardised profile, see Karner et al. (2016)).

$$LP_{SA} = HD_{SA} \cdot SLP_{HD} \tag{2}$$

3.3.2. Heat generation

For each biogas plant, the thermal capacity, the full load hours and the share of already used excess heat EHused are known, as explained above. If the thermal power P_{th} is not given, it is determined with the help of the electrical power and a heat-to-power coefficient of 1 (Klein et al., 2014). In case no full load hours are known, these are determined on the basis of the electrical power and the amount of energy generated per year. On the basis of this data, an hourly load profile SLPBGP can now be assigned to each biogas plant. Biogas plants are typically operated as baseload, but recent changes to the energy-political framework (cf. section A.1 in the appendix) have led to more flexible operation. Since the maximum thermal power requirement is in winter, it is assumed that the biogas plant will be primarily operated during these months. On the basis of the full-load hours, the time window of operation during the winter is extended in both directions until the full load hours are reached (cf. Fig. 12). Since excess heat is only generated during operation, the period of electricity production also corresponds to the period during which excess heat is generated. The excess heat profile LP_{BGP} is determined according to Eq.

$$LP_{BGP} = SLP_{BGP} \cdot P_{th} \cdot (1 - EH_{used}) \tag{3}$$

3.3.3. Coverage and supply ratio

Having assigned load profiles to both the biogas plants and the settlement areas, the coverage ratio *CR* and the supply ratio *SR* can be determined. The supply ratio is based on the total amount of available excess heat EH_{BGP} and puts the available excess heat in proportion to the heat demand HD_{SA} of the settlement areas (cf. Eq. (4)).

$$SR = \frac{EH_{BGP}}{HD_{SA}} \tag{4}$$

By contrast, the coverage ratio is based on an hourly coverage and thus takes into account the time characteristics of supply and demand. As a result, the coverage ratio cannot be greater than the supply ratio. The coverage ratio *CR* is calculated in Eq. (5) for every hour. If the heat demand HD_{SA} is greater than the excess heat EH_{BCP} in an hour *t*, the share that can be covered by the excess heat is calculated in Eq. (6). If the share of excess heat is greater, the heat demand can be fully covered (cf. Eq. (7)). The hourly coverage ratios are then integrated in order to obtain the overall coverage ratio. The calculation of this indicator allows a statement about the autonomy of the settlement area.

$$CR = \frac{100}{8760} \cdot \int_{t=0}^{8760} CR_t \cdot dt$$
 (5)

$$HD_{SA,t} \ge EH_{BGP,t} \Rightarrow CR_t = \frac{EH_{BGP,t}}{HD_{SA,t}}$$
(6)

$$HD_{SA,t} < EH_{BGP,t} \Rightarrow CR_t = 1$$
 (7)

3.4. Allocation of the biogas plants to CLC areas

As described in section 3.2, the shape files of the CLC settlement areas and the coordinates of the biogas plants were used in the geoinformation system QGIS. After the calculation of the centroids of the CLC areas, QGIS was used to calculate the distance matrices for the distances between biogas plants and CLC areas as well as for the distances between the CLC centroids. In the first case, the closest 50 CLC areas and their distances to each biogas plant were determined. In the latter case, due to computational restrictions, only the closest 25 CLC areas were determined for each of the 38,414 CLC areas. Our results indicate that the limitation to the next 25 or 50 areas is sufficient. The distances are needed to calculate the costs of the DHPs, as will be explained in the following sections.

Now the loop shown in the algorithm in Fig. 1 after determining the distance matrices is explained. In the first step, CO2 abatement costs/payback periods are calculated for each of the 10,446 biogas plants, which would result from the supply of DH. This is done for every biogas plant for all 50 of the closest CLC areas. In the next step, the connection is selected from the resulting $10,446 \times 50$ connections, which results in the lowest CO2 abatement costs/ payback period. Then the 25 closest CLC areas to the selected CLC area are added to the selected biogas plant because this CLC area can now be considered as a new starting point for the heat supply. The amount of heat provided to the settlement is deducted in the next steps from the heat supply of the biogas plant and from the heat demand of the settlement. The maximum possible amount of heat is provided in each step. If there is no heat supply left after this step, the biogas plant will be removed from consideration, otherwise the CLC area. The abort criterion is then checked, and if the last CO₂ abatement costs/pavback period are above a predefined limit value, the loop is aborted. The limit values are described in more detail in section 4. Otherwise the loop is carried out again, but the recalculation is carried out only for the biogas plants and CLC areas where the heat supply or heat demand has changed since the last iteration in order to reduce simulation time. In addition, all variables that change are recalculated, such as CR and SR from section 3.3.3. All biogas plants that have the selected CLC area among the nearest 50 areas will therefore be included in the new calculation. In each step, for economic reasons it is ensured that the total length of the DHP does not exceed 50 km (Arbeitsgemeinschaft QM Fernwärme, 2017).

The following sections explain the determination of the CO_2 emissions saved (section 3.4.1), the costs for the DHN and the CO_2 abatement costs (section 3.4.2) as well as the payback period (section 3.4.3).

3.4.1. Calculation of the saved CO₂ emissions

The calculation of the saved CO_2 emissions is based on the allocation of used energy to provide heat to the settlement areas. The emission factors (EF) of the energy sources and their average allocation in Germany are shown in Table 2. The allocation of the energy for the determination in the settlement areas is based on the average values in the respective federal state (Statistisches

Table 2 The emission factors Germany (LfU, 2016;	EF of the heating energy Statistisches Bundesam	sources and their average allocation in t, 2016a).
Energy carrier	Allocation A [%]	Emission factor EF [kgCO ₂ /kWh]
Gas	62.50	0.252
Heating oil	21.94	0.215

	62.50	0.252
ng oil	31.84	0.315
icity (EL)	4.90	0.646
n coal (BC)	0.54	0.429
coal (HC)	0.22	0.428
	ng oil ìcity (EL) n coal (BC) coal (HC)	62.50 62.50 icity (EL) 31.84 icity (EL) 4.90 n coal (BC) 0.54 coal (HC) 0.22

Bundesamt, 2016a). Since the shares of DH, wood pellets, biomass, solar energy and geothermal and other environmental heat in the total heat demand are deducted from the total heat demand (cf. section 3.3.1), these types of energy are not listed in Table 2.

The calculation of the saved CO₂ emissions $CO2_{saved,SA}$ in Eq. (8) is based on the calculated coverage ratios per settlement area CR_{SA} . Excess heat replaces part of the fossil energy used to supply the settlement areas. It is assumed that excess heat replaces the existing energy carriers proportionally. The usage of excess heat has an emission factor of 0 kg/kWh (Theissing, 2012).

criterion and is calculated in Eq. (12) using the sum of the discounted revenues REV_{SA} less expenses EXP_{SA} . Discounting over a certain period is taken into account by the discounting factor df of 0.05 (5%), which is intended to represent a compromise between social and commercial discount rates. Finally, the investment I_{SA} is deducted to calculate the NPV_{SA} . Annual costs are caused by maintaining the DHG and driving the DH pumps. As operating power of the pumps, a factor P_p of 10 kWh_{el} /MWh_{th} is used (Good, 2004). This means that 10 kWh_{el} is required for each MWh_{th} of DH

$$CO2_{saved,SA} = \frac{HD_{SA} \cdot CR_{SA} \cdot (A_{Gas} \cdot EF_{Gas} + A_{Oil} \cdot EF_{Oil} + A_{EL} \cdot EF_{EL} + A_{BC} \cdot EF_{BC} + A_{HC} \cdot EF_{HC})}{1000}$$

3.4.2. Determination of the CO₂ abatement costs

The investment for the DH grid construction, or densification in case a DHN already exists, are mainly based on the length of the grid. The grid length cannot be determined without a detailed on site investigation or analysis of the heat demand density/distribution using a geographic information system. Persson and Werner (2011) developed a method to determine the investment for DH grids without the mentioned procedures. In this way, the investment can be estimated based on publicly available data such as population density, specific building space, specific heat demands and some cost parameters.

First, the fraction of the excess heat that is supplied by the biogas plant to the settlement area Q_{sell} is determined using Eq. (9).

$$Q_{sell} = HD_{SA} \cdot CR \tag{9}$$

The investment IDHG, SA for the construction or the densification of the district heating grid (DHG) in a settlement is determined by the specific heat distribution costs HDC_{SA} and the excess supplied heat Q_{sell} . In Section A.4 of the appendix, a detailed description of determining the specific heat distribution costs HDC_{SA} in the settlements is given. Additionally, the investment in the pipeline for connecting the biogas plant and the settlement area Icon, SA has to be calculated. There an investment rate *IR* of 200 €/m for the pipe is multiplied by the distance DISSA between biogas plant and settlement area (Fraunhofer UMSICHT, 1998; C.A.R.M.E.N. e.V, 2012; Pfnür et al., 2016). The relatively low value of 200 €/m is supposed to reflect the fact that biogas plant operators receive a subsidy in the context of the Combined Heat and Power Act (KWKG) amounting to 100 €/m of DHN built (BMJV, 2018). Finally, the investment can be summed up in Eq. (10) and results in the necessary investment I_{SA} to use the excess heat from the biogas plant to supply the settlement area with heat.

$$I_{SA} = I_{DHG,SA} + I_{con,SA} = HDC_{SA} \cdot Q_{sell} \cdot 0.0036 + IR \cdot DIS_{SA}$$
(10)

The multiplication by 0.0036 is performed to convert Q_{self} to G]. In the section 3.4.1, the saved CO₂ emissions were determined. These are now taken into account to calculate the specific CO₂ abatement costs CO2_{costs} (cf. Eq. (11)).

$$CO2_{costs} = \frac{I_{SA}}{CO2_{saved,SA}}$$
(11)

3.4.3. Determination of the net present value and payback period The net present value (NPV) is used as a further economic to the customers. The used input data is shown in Table 3, which gives the mean heat prices from a random sample of ten district heat providers in Germany.

(8)

$$NPV_{SA} = \sum \frac{REV_{SA} - EXP_{SA}}{(1 + df)^{t}} - I_{SA}$$
(12)

The NPV is not a key result for this analysis, but is used to determine the payback period *PP* of the excess heat utilisation. In the algorithm, a loop is used to identify the time *t* at which the NPV becomes positive.

4. Results and discussion

In this section, the algorithm is validated before the results are discussed in more detail (cf. section 4.1). Initially, the calculations are only carried out with 241 biogas plants included in the survey. Section 4.2 demonstrates how these 241 plants were selected. Thereafter, it will be shown in section 4.3 how the results change when all 10,446 biogas plants in Germany are included in the analysis. In order to provide the algorithm with appropriate abort criteria, the limit value of the CO2 abatement costs is set in all calculations to $1,000 \in /tCO_2$, the limit value of the payback period to 20 years. The reason for these high values is that we wish to economically assess the technical potential, rather than to directly determine an economic potential based on some predefined criteria. Finally, the procedure is critically appraised in section 4.4. The algorithm was implemented in MATLAB and solved using a standard desktop PC (2x Intel Xeon 5430 Processor and 24GB RAM). The solution time for the 241 plants is around 10 min and that for the whole German biogas plant stock about two days. The code can be made available by the authors upon request.

4.1. Validation of the algorithm

Fig. 3 gives an overview of the possible types of DH connections

Table 3

nput data for economic evaluation, (Statista, 2018a; BDEW, 2018; SWB, 2014).				
Asset	Costs	Units		
Maintenance cost rate of the DHG	0.5	% of investment		
Electricity purchase price EP	0.2324	€/kWh		
Thermal connection power Pa	5	kW		
Heat selling prices HS:				
Energy price	0.0664	€/kWh		
Demand charge	30.99	€/kW		
Base price	10.81	€/month		



Fig. 3. Exemplary illustration of resulting DHPs for the use of excess heat from biogas plants in several municipalities in Baden-Württemberg. The background map is from OpenStreetMap contributors (2018).

that can be created by the algorithm in Fig. 1. The red DHPs lead from a biogas plant to a settlement area and the blue ones lead from one settlement area to another. The black lines represent municipal borders. The results shown in this figure were derived from the calculation with all biogas plants in Germany and the CO2 abatement costs as a selection criterion (cf. scenario A.1 in section 4.3). The first case shows biogas plant 7 (number in box), which has no connection to a settlement area. This can have several reasons. On the one hand, 100% of the excess heat from the biogas plant could already be used. In this case, this biogas plant would be one of the 241 plants from the survey, since only in these plants over 60% heat is used (cf. section 4.3). On the other hand, the settlement area in the vicinity could have no heat demand, e.g. because all buildings are already supplied with heat from alternative technologies. In addition, a DH supply might not be worthwhile in this case, since the limit values of the target criteria are exceeded in the algorithm. The second case is the one in which a settlement area is supplied with DH by only one biogas plant, and these biogas plants also supply only one settlement area (biogas plants 3 and 8). If a settlement area is supplied by several biogas plants, this is the third case (biogas plants 1, 2, 5, 6 and 9). The fourth case is the supply of several settlement areas by a biogas plant, as in the case of the biogas plants 4, 5 and 6. Finally, the connections of biogas plant 5 show that the DHP can also lead from one settlement area to the next, as shown by the blue-coloured DHP.

The heat demand of the settlement areas from section 3.3.1 can also be validated. The sum for all settlement areas $HD_{SA_{total}}$ results in a heat demand of 576 TWh (cf. Eq. (18) in the appendix). The heat demand for space heating and hot water in private households (mainly gas demand, cf. DIW Berlin and EEFA, 2017) is subject to large annual fluctuations (Umweltbundesamt, 2017a) and ranged from 544 to 578 TWh between 2010 and 2015 (Statista, 2018b). In 2010, from when the census data, heat consumption data and renovation data originate, the heat requirement was 578 TWh (Deviation: 0.35%). Thus, the procedure presented in section 3.3.1 can be deemed to be very accurate.

4.2. Results from the survey

In the survey, respondents provided information on the percentage share of heat utilisation for different categories such as in schools, hospitals and fish farming. In 262 of the 602 plants, the total shares were above 100% (cf. Fig. 4). The plants with a heat utilisation of more than 100% are excluded from the analysis in this study, as it seems the survey participants misunderstood the questions and/or made mistakes in stipulating these shares. In total, this results in an average heat utilisation of 60% and an unused heat fraction of 40%.

Of the 602 biogas plants from the survey, 241 plants (40%) could be identified and mapped in the Energymap plant register (cf. section 3.2). 111 of the 241 plants, however, are among the 262 plants in which a heat utilisation value of above 100% was reported. As the left part of Fig. 5 shows, the majority of the 241 biogas plants (green circles) identified are located in the federal state of Bavaria (143 plants \triangleq 59%). This also corresponds to the distribution of the responses from the survey (cf. section 3.1).

The heat map on the left side of Fig. 5 shows the population density. White areas mark settlements with a low population density and the population density then rises to dark red. The heat map shows that the calculation with only 241 plants does not take into account the settlements with the highest population densities. A comparison of the 241 plants with all plants in Germany (cf. right part of Fig. 5) shows that the latter are clearly more evenly distributed across Germany. Of the 241 biogas plants allocated, only



Fig. 4. Number of biogas plants surveyed as a function of the specified share of heat that is already being used.



Fig. 5. Location of the 241 biogas plants identified among the plants surveyed in Germany and a heat map showing the population density in the settlements (left part). Location of all biogas plants in Germany (right part) (Engel, 2015; Statistisches Bundesamt, 2017c).

121 plants still have unused heat. The remaining 120 plants already use 100% of their excess heat.

First of all, the results of the calculations with CO₂ abatement costs as a decision criterion are presented in section 4.2.1, followed by the results of the calculations with payback period as a decision criterion in section 4.2.2. A sensitivity analysis is performed for both cases to show how some key parameters affect the results. The values of the parameters were changed in 10% steps from -50% to +50%. The only exception is the distance of the biogas plant from a specified location. As described in section 3.2, the location of the biogas plant can have a maximum uncertainty of 3 km. The smallest distance between a biogas plant and the centroid of a settlement

area is 10 m, which lies far below this maximum uncertainty. However, it can also be a so-called satellite CHP unit located within the settlement area (Rutz and Güntert, 2012). After all, 17% of the biogas plant operators in Germany have a satellite CHP unit (Liebetrau et al., 2017). To cover this uncertainty, the distance of the plants to the settlement areas was varied between -3 km and +3 km. Since this parameter is shown together with the other parameters in Fig. 6 below, -3 km corresponds to -50% and +3 km to +50% and the change of 10% corresponds to 600 m steps. A further examination of over 700 plants in the most recent EEG biogas plant register from Bundesnetzagentur (2018) enabled them to be matched to their records in Energymap (Engel, 2015). The

J.M. Weinand et al. / Journal of Cleaner Production 238 (2019) 117756



Fig. 6. Change in the mean CO₂ abatement costs of the reference scenario when changing specific parameters.

average distance between the matched plants was 1.5 km, which corresponds to a range in Fig. 6 of \pm 25%, but it is unclear which of these two sources is more accurate.

4.2.1. CO₂ abatement costs as decision criterion

In the reference scenario, defined by the parameter values from section 3, the CO₂ abatement costs range from 55 \in /tCO₂ to 987 \in /tCO₂ (mean: 120 \in /tCO₂) and 41,500 tCO₂/a is saved. 129 DHPs are built in this case.

Fig. 6 shows the mean CO2 abatement costs resulting from the sensitivity analyses. The strongest deviation occurs when the coordinates of the biogas plants are changed (distance of biogas plant from specified location). The mean CO2 abatement costs change by -41% if the biogas plants are 3 km closer to the settlement areas and by +97% if the biogas plants are 3 km further away from the settlement areas. The deviation for the 3 km closer plants is smaller, since some plants are already less than 3 km away from the settlement areas. The gradient of the curve becomes more constant the less the distance is reduced. Nevertheless, the curves do not have constant gradients. This is related to the abort criteria, as a result of which more and more connections between plants and settlements are excluded from the analysis or included in the analysis when parameters are changed. This will be further explained in section 4.2.2. If the biogas plants deviate 1.5 km from their location as described above, the costs could change between -31% and +52%.

The pipe specific investment outside a settlement IR and the discount factor df have approximately the same effect on the mean CO_2 abatement costs and reduce them by a maximum of 19%

at -50% and increase them by a maximum of 19% at +50%. It is interesting to note that the pipe specific investment inside a settlement HDC_{SA} has a stronger effect on costs than *IR*. This means that the DHPs within a settlement have a greater influence on costs than the DHPs leading to the settlement, i.e. the population density of a settlement is of crucial importance. The same conclusion can be drawn when looking at the curve of the plot ratio *e*. This is the only parameter that leads to a reduction of the average cost when it is increased. An increase of *e* is equivalent to an increase of the building density in the settlements. In this case, more heat can be delivered per settlement area that would otherwise remain unused.

As illustrated in Fig. 7, the variation of the parameters does not have a large influence on the total CO₂ abatement. The change in total CO₂ abatement of the reference scenario (41,483 tCO₂/a) varies only between -1.5% and +0.5%. The greatest influence has once again the variable "distance of biogas plant from specified location".

Some of the curves in Fig. 7 are not linear, which would not be directly suspected. For example, it has been concluded that an increase of the plot ratio *e* continuously reduces the CO₂ abatement costs. It is also likely that an increase of *e* would increase the total CO₂ abatement. However, the curve in Fig. 7 shows that, for example, with an increase of *e* from +40% to +50%, the total CO₂ abatement decreases again (by 28 tCO₂ /a). This example is now explained on behalf of the other non-linearities in Fig. 7. In the case of +40%, a biggas plant in a municipality in Bavaria will initially supply the CORINE area 8940 with 485 tCO₂ /a for 108 \in /tCO₂ and with 65 tCO₂ /a for 345 \in /tCO₂ (cf. left part of Fig. 8). In the case of +50%,



Fig. 7. Change in the total CO₂ abatement of the reference scenario when changing specific parameters.



Fig. 8. DHN for the utilisation of the excess heat from a biogas plant in a municipality in Bavaria. The left part of the figure shows the connections that result in the case where plot ratio e is increased by 40%, in the right part e is increased by 50%. The background map is from Microsoft (2018).

the CORINE area 8876 will be selected directly with 522 tCO₂ /a for 105 (cf. right part of Fig. 8).

In the first case, CORINE area 8876 is also supplied, since approx. 14% of the excess heat in the biogas plant remains after supplying CORINE area 8940. In the second case, however, only 7.4% excess heat is left after supplying CORINE area 8876, so that the supply of area 8940 is no longer worthwhile (CO₂ abatement costs of 1,180 €/tCO₂ \geq 1,000 €/tCO₂). The reason for the fact that settlement area 8940 is supplied before area 8876 (up to case +50%), although in settlement area 8876 there is more heat demand, is, among other things, the almost twice as high population density in settlement area 8940 (3,400 persons per km² compared to 1900 persons per km²). In order to avoid fluctuations in the curves of the total CO₂ abatement costs in Fig. 7, the total CO₂ abatement costs would have to be used as a decision criterion in the algorithm. In this case, however, the CO₂ abatement costs involved could be excessively high in some cases.

4.2.2. Payback period as decision criterion

If the payback period is selected as the decision criterion, the reference scenario results in a mean payback period of 7.2 years (payback periods of between 2.5 and 20 years). In this case, 122 DHPs are built and 40,600 tCO₂ /a is saved.

The mean payback period increases the most when HS decreases (+69%) and is the lowest when the distance of biogas plant from specified location decreases (-44%, cf. Fig. 9). If HS decreases, only a lower profit can be achieved by selling the heat. Since the variation of DH pump power P_p has exactly the same effect on the payback period, only the electricity price EP is shown in the figure. If EP increases, the costs for driving the DH pumps and thus the payback period increase. P_a has an opposite effect on the payback costs, since the rate at which the heat is delivered, and hence the amount of heat sold, depends strongly on this parameter. The distance, e, df, HDC_{SA} and IR have almost the same effect on the man payback period as on the mean CO_2 abatement costs (cf. section 4.2.1).



Fig. 9. Change in the mean payback period of the reference scenario when changing specific parameters.

However, the declining mean payback period with a change of IR from +40% (8.3 years) to +50% (7.8 years) is remarkable. The other non-linearities in the curves in Fig. 9 can also be explained using the following example: Fig. 10 shows the change in the number of DHPs when increasing IR from -50% to +50%. The number of connections is reduced from 132 at -50% to 107 at +50%. In each step, the changes are within a range of one to three no longer existing connections. However, in the last step from +40% (115 connections) to +50% (107 connections), eight connections no longer exist due to a payback period of more than 20 years. As a result, many of the connections with a long payback period will be eliminated in this step. Therefore, the mean payback period decreases in this case. This can be proven by repeating the calculation in the case of +50%, and this time not using the payback period as a criterion for aborting the algorithm, but by aborting at a number of 115 DHPs as in case +40%. This would result in a mean payback period of 8.8 years, which is higher than in the case of +40%.

4.3. Results for all German biogas plants

The analysis with all 10,446 biogas plants assumes an average value of 40% unused heat in the biogas plants that were not part of the survey. Whilst there will obviously be deviations in individual cases, the assumption of 40% excess heat seems reasonable and is verified by other studies (e.g. DBFZ, 2015). Table 4 shows the various scenarios developed with all biogas plants.

When using the CO₂ abatement costs as a decision criterion in scenario **A.1**, the excess heat from 9,790 different biogas plants is used. A total of 10,989 DHPs are built in this case and 2.55 MtCO₂ /a can be saved (cf. yellow line Fig. 11). This corresponds to 0.3% of the amount of CO₂ emitted in Germany in 2015, equivalent to 792 MtCO₂ /a (Umweltbundesamt, 2017c). 85% of the 2.55 MtCO₂ /a can be saved with CO₂ abatement costs below 200 €/tCO₂. The first 78 tCO₂ are saved at a minimum cost of 13.5 €/tCO₂, which is 41.5 €/tCO₂ less than the best connection in the case of the 241 plants. The reason for this is that this biogas plant is located only 70 m from the settlement area centroid. In total, around 8 TWh of heat demand in German households are



Fig. 10. Change in the number of DHPs when changing IR.

Overview of the scenarios carried out with all biogas plants.

Table 4

covered by the biogas plants.

Scenario A.2 assumes that the biogas plant can replace all forms of existing heat supply (i.e. neglecting Eq. (1), so $HD_{SA} = HD_{SAtotal}$). In this case, the CO₂ abatement is only 5 ktCO₂ /a higher at almost the same cost. This can be explained as follows. At the beginning of the calculation with all plants, approx, 10 TWh of excess heat are available from all biogas plants. In the end of calculation A.1 above, only 0.22 TWh (2%) remain at the end. This shows that there is scarcely any potential for an increase and that there is a bottleneck on the supply side of the biogas plants, rather than on the demand side in the existing heat supply systems. For the second scenario (without Eq. (1)) only 0.03 TWh more heat is used, due to the small amount of non-fossil-fuel-based existing heating that is replaced in this case. Since the CO2 abatement cost curve does not change significantly, the curve for A.2 is not shown in Fig. 11. The bottleneck on the supply side of the biogas plants explains the lower $\ensuremath{\text{CO}}_2$ abatement here than in Soltero et al. (2018) (cf. section 2.2). In their study, 5.4 MtCO₂ were avoided, however, the entire heat of biomass plants was used instead of unused excess heat as in our study.

As explained in section 3.3.2, a constant load profile for the excess heat from the biogas plants was assumed. However, this profile could look different in reality, since the digester or residential buildings that are already supplied with heat require more heat in the colder months (Rutz et al., 2015). Based on the values from Rutz et al. (2015), the load profile of a biogas plant that supplies residential buildings and the digester with excess heat is changed in scenario **A.3** according to Fig. 12. The short period in summer, in which the load is 0 kW, is due to the fact that the biogas plant shown as an example has full load hours of less than 8760.

Actually, curve A.3 would be expected to differ significantly from curve A.1, as more heat is now available in summer, i.e. at a time when less heat is required. However, this is not the case due to the bottleneck in the heat supply of the biogas plants already mentioned in the description of scenario A.2. This means that in most cases, however, the settlement areas require more heat in summer than the biogas plants can provide. In total, 5 ktCO₂ / a or 0.2% less is saved than in scenario A.1, with slightly higher mean CO₂ costs.

Compared to scenario A.1 with CO_2 abatement costs as a decision criterion, if the payback period is used as a decision criterion in **B.1**, the number of biogas plants whose heat is used decreases by 835 to 8,955 plants and 344 DHPs fewer are built (10,645 pipelines). Overall, 0.08 MtCO₂ /a less than in scenario A.1 is saved (2.47 MtCO₂ /a). The first 85% CO₂ can be saved with a payback period of fewer than 10 years (cf. Fig. 11). Only 0.53 TWh (5%) of excess heat remains unused in this calculation.

The scenarios discussed above do not include the cogeneration bonus from the Renewable Energy Sources Act 2009, so that **B.2** is carried out with the cogeneration bonus in order to estimate its effect on the result. It is assumed that the cogeneration bonus for the heat supply of residential buildings in the amount of 3 ct/kWh (Clearingstelle EEG, 2009) will be paid to all plant operators whose plants were commissioned before 31.12.2011 (Clearingstelle EEG, 2012) and have a nominal power of less than 5 MW (Clearingstelle EEG, 2009).

Number	Decision criterion	Adjustment
A.1	CO ₂ abatement costs <1,000 €/tCO ₂	
A.2	CO ₂ abatement costs <1,000 €/tCO ₂	Not considering Eq. 1
A.3	CO ₂ abatement costs <1,000 €/tCO ₂	New excess heat profile, cf. Fig. 12
B.1	Payback period <20 years	
B.2	Payback period <20 years	Considering the cogeneration bonus of 3 ct/kWh for district heat from biogas plants



Fig. 11. Cumulative CO₂ abatement in the scenarios with CO₂ abatement costs or payback period as the decision criterion. The curves refer to the y-axes with the corresponding decision criterion of the scenario.



Fig. 12. Change in excess heat load profile, without (left) and taking into account monthly differences in heat utilisation (right).

Taking into account the cogeneration bonus, the amount of CO_2 saved increases by 70 ktCO₂/a (3%). This means that the amount cannot be significantly increased due to the above mentioned heat supply bottleneck. However, the mean payback period is considerably shorter than for calculation B.1. Thus, 2 MtCO₂/a are already saved within a payback period of 6 years (instead of 9 years in B.1). This has to be considered as an upper limit, however, as it is not certain whether all biogas plants will receive the cogeneration bonus to the extent assumed here. This is especially the case for plants with a nominal power of more than 500 kW, as they only receive the cogeneration bonus of 3 ct/kWh for 500 kW (Clearingstelle EEG, 2009). However, the plants commissioned be regardless of their nominal power (Clearingstelle EEG, 2006).

Using the criteria of CO₂ abatement costs and payback period, a yield of 2 MtCO₂/a is achieved for values below 200 $€/tCO_2$ (A.1) and 9 years (B.1). This corresponds to around 0.25% of the total German CO₂ emissions in 2015 (Umweltbundesamt, 2017c) or around 2.5% of all CO₂ in residential buildings (Umweltbundesamt)

2017b). The CO₂ reduction potential is approximately 0.5 MtCO₂ and 0.75 MtCO₂, if thresholds of 80 ϵ /tCO₂ and 5 years are set to reflect the proposed external cost of carbon and an expected payback period from an investor's point of view respectively (Schwermer, 2012). However, if the current price of CO₂ in the EU Emissions Trading System of about 7 ϵ /tCO₂ is taken as a benchmark, the economic fraction of this technical potential saving diminishes to 0 tCO₂.

Fig. 13 shows the share of residential heat demand in German municipalities that can be covered by the excess heat from biogas plants with the help of DH. The calculation A.1 serves as a basis for the figure, and DH connections where the payback period is longer than 20 years were not taken into account. The figure shows that especially in the southern (Baden-Wuerttemberg and Bavaria) and northern federal states (Schleswig-Holstein, Mecklenburg-Western Pomerania, Lower Saxony, Saxony-Anhalt and Brandenburg), the most heat demand per municipality can be covered. In total, there is a potential in 3,591 (32%) of the 11,400 German municipalities, in the other municipalities the value is 0%. The mean value in the

J.M. Weinand et al. / Journal of Cleaner Production 238 (2019) 117756



Fig. 13. Share of heat demand that can be covered by DH from biogas in German municipalities. The numbers in brackets represent the number of municipalities to which the shares can be allocated. The background map is from OpenStreetMap contributors (2018).

3,591 municipalities is 12% and the CO₂ emissions per capita are reduced by an average of 250 kgCO₂/a. Some of the municipalities can cover almost all of their heat demand with DH from the biogas plant(s). More than 85% of the heat demand is met in 21 municipalities, while a maximum value of 98% is reached in the municipality of "Bresegard bei Eldena" (200 inhabitants) in Mecklenburg-Western Pomerania. In some municipalities, excess heat from biogas plants can therefore contribute to achieving energy autonomy. These results at the municipal level can be found in the online supplementary material.

4.4. Critical appraisal of the methodology

The sensitivity analysis considered the uncertainties in the parameters used, such as costs. However, the algorithm applied here also has some weaknesses that would not occur in the detailed planning of each individual biogas plant. This section first explains the weaknesses, which may lead to overestimating the calculated technical potential of DH by excess heat from biogas plants. Subsequently, the weaknesses that reduce the real technical potential are identified. Finally, the uncertainties relating to the method are discussed.

First of all, some assumptions are made in the algorithm used here, which can lead to a moderate to severe underestimation of the costs. For example, the shortest route from the biogas plant to the settlements is always used for the DHPs. Here, however, the topology and other obstacles should be considered as in Nielsen (2014). Whilst data on surface topology is available in the form of Digital Elevation Models, it was not possible in this study to consider other obstacles. Given the objective of the study, to determine the overall technical potential for excess heat utilisation, this would also have been beyond the scope. In practice, however, a detailed DHN planning process would have to be carried out for individual municipalities.

In addition, the profiles for heat supply and demand were assumed. On the one hand, it is not certain that the biogas plants



Fig. 14. DHN for the utilisation of the excess heat from a biogas plant in the municipality of Cavertitz in Saxony. The background map is from OpenStreetMap contributors (2018).

will actually be operated in the periods in which the heat demand in the settlements is highest (winter). On the other hand, the standard load profile SLP_{HD} from Eq. (2) can only be accurately be applied in municipalities with several hundred households. In the smallest settlement area, there are only three households with a total of five inhabitants. However, the influence of this uncertainty is low, since in the calculations only 7% of the connected settlement areas have less than 100 households. Furthermore, the standard load profile was assumed to be equal for all settlements in Germany. In reality, differences between settlements from different regions could be found. In general, however, for this study the standard load profiles in the settlements.

In some cases the costs are overestimated, however, as the following example illustrates. Fig. 14 shows the DHN for using the excess heat from a biogas plant in the municipality of Cavertitz in Saxony. The problem lies in the fact that the algorithm iteratively selects the optimal connection in each step. The result is that the blue-coloured DHP furthest to the left is selected before the red coloured DHP furthest to the left. If the red DHP had been "built" first, there would have been a shorter distance to the settlement area in the bottom left, which is supplied by the blue DHP. This uncertainty has a low impact on costs, due to the fact that this phenomenon only occurs very rarely. In scenario A.1, for example, only 6% of DH connections start in settlement area. Of this 6%, only a few connections will show this fault, in Fig. 14 it is only one connection of five.

Fig. 14 also shows that the selection of the area centroid for distance measurement overestimates the length of the DHPs from the biogas plants to the settlement areas in this case. Actually, only the distance to the border of the settlements should be measured here. However, this determination was not possible since the distance calculation is only possible with point coordinates. This could lead to a moderate increase in costs. For example, the settlement area at the bottom left of Fig. 14 has a width of 4.2 km. In extreme cases, the length of the DHPs could therefore be overestimated by around 2 km.

Additionally, an existing DH supply can be deducted from the heating demand of the settlements, but no existing DHG can be taken into account due to a lack of data. If these data were available, the length of the DHPs to be built could be reduced, as they would no longer have to lead to the settlement, but only to the nearest connection point of the existing DHG. The impact on costs is estimated to be negligible, due to the limited number of existing district heating networks in Germany. In other countries, such as leeland or Denmark, the costs for the use of waste heat from biogas plants could be considerably lower due to the significantly higher DHG share (Euroheat and Power 2016).

A further overestimation of costs results from the fact that only residential areas are considered and not industrial or commercial areas. Some promising progress in this area has been made by the Pan-European Thermal Atlas (PETA) in the context of the Heat Roadmap Europe project.⁴ Within the PETA urban areas are considered as "coherent urban areas", including industry and commercial sector, and existing DHNs are considered on a detailed level. However, at the time of carrying out this study, this data was not publicly available.

Fig. 15 shows the municipality of Leutenbach in Baden-Württemberg. Blue surfaces are used here to represent the CLC areas. Red areas are individual buildings from Geofabrik (2018). The figure shows that the CLC data do not cover all existing settlement areas (cf. red circles in the Fig. 15), as there is a minimum threshold for the size of an urban area to be differentiated in the CLC data (0.05 km²) and the data are from 2012 (so do not consider newer buildings). In the example shown in Fig. 15, the excluded settlement areas are relatively small, so the implications for the results are likely to be only marginal.

A further uncertainty arises from the age of biogas plants. The age profile of the existing plants means that some of these plants will cease to benefit from the feed-in tariffs (FIT) of the Renewable Energy Sources Act upon which they rely to be economic. After this time, their continued operation is uncertain. The plants should already have broken even after twenty years, as studies have shown payback periods in the region of 7-12 years depending on the substrate and the output (Balussou et al., 2011). Hence the plant may continue to operate, but the negative impact on the economics of losing the FITs could force the operators to seek other business models such as biomethane uprading and feed-in (cf. section A.1 in the appendix). In the absence of new business models, the plants might be forced to close, which would mean the excess heat considered here is no longer available. Instead, this study assumes constant cash flows over the 20 year lifetime of the DH project, which obviously neglects this issue. However, this study aimed to

⁴ http://www.heatroadmap.eu/maps.php.



Fig. 15. Settlement areas from the CLC data set and buildings in the municipality of Leutenbach in Baden-Württemberg (Geofabrik, 2018). The background map is from OpenStreetMap contributors (2018).

assess the current technical potential for excess heat use from biogas plants, and there are always future uncertainties associated with such analyses. Potential future business models for biogas plant operators in a post-EEG context could be the subject of a future contribution.

Finally and partly independently of the future development of the biogas sector, the dynamic of the surrounding energy system was not analysed in any detail in this study. In particular, the decarbonisation of the energy system means a shift away from fossil fuels towards renewable heat and electricity supply. The German government has ambitious climate targets in the context of the energy transition, including 50% renewable electricity consumption by 2030 and an 80% reduction in building primary energy demand by 2050. Assuming these targets are met within this timeframe implies a deep insulation of existing buildings as well as shift towards gas-fired cogeneration (in the short term) and heat pumps (in the long term, cf. Merkel et al., 2016). Hence the CO₂ intensity of existing heat supply would significantly reduce and similarly tend to decrease the potential impact of the results for CO₂ abatement through district heating from biogas plants presented in this paper.

5. Summary and conclusions

This paper has analysed the current technical potential for utilising excess heat from biogas plants, in order to supply local settlements through district heating. Due to the high production of biogas, Germany was selected as the case study. Based on a survey of around 600 biogas plant operators, the fractions of excess heat from the cogeneration unit in these plants have been analysed. The analysis was carried out for the surveyed population as well as scaled up to the whole German biogas plant stock. A heuristic was developed to connect biogas plants (heat sources) with local settlements (sinks) in order to determine a least-cost district heating supply for residential buildings. Thereby two criteria were employed, namely the CO_2 abatement costs and the payback period, which represent the macro- and microeconomic perspectives respectively.

Based on the survey, a mean fraction of 40% excess heat was determined, which is in agreement with other empirical studies.

Extrapolating this fraction to the German biogas plant stock leads to technically feasible CO2 savings of around 2.5 MtCO2/a. Employing the criteria of CO2 abatement costs and payback period yields about 2 MtCO₂/a below CO₂ abatement costs of 200 €/tCO₂ and 9 years respectively. These relatively high average costs are related to the typically low population density in rural regions where biogas plants are located. The potential CO2 savings represent about 0.25% of the total German CO2 emissions in 2016 or around 2.5% of all CO₂ in residential buildings. If a threshold value of 80 €/tCO2, to reflect the German government's suggested external cost of carbon, is employed, the carbon reduction potential is about 0.5 MtCO₂. Similarly, a threshold for the expected payback period of 5 years, to reflect an investor's point of view yields potential savings of 0.75 MtCO2. These potentials are concentrated in around 3,500 municipalities, where district heating from biogas plants could reduce CO2 emissions per capita by an average of 250 kgCO₂ /a and cover 12% of the total residential heating demand. In some of these municipalities, large proportions of their heating demand could be economically met (according to the criteria employed here) by this excess heat, hence assisting in the transition to more decentralised autonomous energy systems. On the other hand, if the current price of CO2 in the EU Emissions Trading System of about 7 €/tCO₂ is taken as a benchmark, the economic fraction of this technical potential saving reduces to 0 tCO2. Although these results are relatively modest in the overall context of decarbonising the energy system, this study does provide a quantitative basis for decision makers, researchers and energy planners. The detailed results provided as supplementary material should offer useful insights for local planners and authorities when considering the sustainable energy options at their disposal.

The developed method was applied to a German case study, but could be equally relevant for other countries with high biogas production. Some of these countries (e.g. Denmark) have a significantly more widespread district heating network than Germany. Thus, the use of biogas excess heat could lead to similar or even better results for these countries than in the case study for Germany. Apart from a methodology that can be transferred to any country with similar data availability, the present study therefore demonstrates that the use of excess heat in biogas plants can be one contribution towards a global energy system decarbonisation.

The employed methodology, whilst adequate for a national estimation of the technical potential and associated costs, has several uncertainties. Most importantly, the shortest birds-eye route from the biogas plant to the centroid of the settlement is used as the required distance for the pipeline. Whilst a good estimate for the order of magnitude, this obviously leads to uncertainties relating to the required district heating pipeline length, and therefore also to the costs. In addition, the developed heuristic does not (necessarily) determine the optimum allocation of heat sources to heat sinks, and may also therefore overestimate the costs. Finally, the focus on residential buildings and the rough consideration of existing district heating supply (but not infrastructure) add additional uncertainties. All of these aspects remain areas for future work.

Acknowledgements

The authors gratefully acknowledge the financial support of the PhD College "Energy and Resource Efficiency" (ENRES), from the Federal State of Baden-Wuerttemberg, for funding the first author's PhD studentship. The authors also thank five anonymous reviewers, who provided valuable comments on earlier versions of this manuscript.

Appendix A. Supplementary data

Supplementary data to this article can be found online at https://doi.org/10.1016/j.jclepro.2019.117756.

Appendix A.1 Current status of the German biogas industry

At the end of 2017, the number of biogas plants in Germany stood at around 10,500 plants. This makes Germany the country with the largest biogas plant population in Europe by far (European Biogas Association, 2016), clearly over-fulfilling the National Renewable Energy Action Plan (NREAP) targets for 2015 (Pablo-Romero et al., 2017). The 10,500 plants relate to 4.3 GW_{el} and 25.7 TWh_{el} of capacity and generation respectively in 2015 (Engel, 2015).

A period of fast growth between 2009 and 2011 saw the

installation of more than 1,300 new plants per year on average. The expansion of the sector subsequently came to a halt in 2012 and the average number of newly built plants dropped to an average of below 150 per year for the period from 2014 to 2016 (German Biogas Association, 2017). The main growth driver until 2012 was the German Renewable Energy Sources Act, which guaranteed generous feed-in-tariffs (FIT) for electricity produced in biogas plants. It especially promoted the use of energy crops, which is why 51% of all material input of German biogas plants consists of energy crops, of which 73% is maize silage (FNR, 2017). But the public as well as the political discourse have turned against the use of energy crops (Herbes et al., 2014). Subsequently, policy makers first introduced a cap on the amount of maize silage that can be used in new plants and finally redesigned the subsidy system in a way that only allows very few new plant projects to be financially viable (Markard et al., 2016; Herbes et al., 2014). Recently, the German government introduced a tender system for electricity from biomass, but the first tender in September 2017 was perceived as unsuccessful, since the bids totalled to only about a third of the tender volume (Bundesnetzagentur, 2017a). While not many new biogas plants are built at the moment, the outlook for the existing sites beyond the end of their 20 year FIT period is unclear, and without a new remuneration period, a large part of the plant population in Germany will be shut down (Lauer and Thrän. 2017).

The majority of German plants use the gas onsite to fuel combined heat and power (CHP) units, thus cogenerating heat and electricity. The electricity is fed into the electricity network to receive a feed-in tariff, and heat is used locally if at all feasible. A further around 200 plants (in 2016, around 9 TWh of gas, cf. dena 2016) upgrade the biogas into biomethane or Bio-SNG (Synthetic Natural Gas) and inject it into the public gas grid (cf. Fig. 16). Particularly where a local heat sink is lacking, despite the overall utilisation efficiency of biogas valorisation. For example, a typical CHP unit can achieve 40% electrical efficiency, reaching an overall efficiency of up to 80% if all the generated heat can be used locally (Pöschl et al., 2010). In the absence of a local heat sink, biogas upgrading and feed-in can result in overall efficiencies of around 75–80%, based on the utilisation of Bio-SNG as a fuel for transport, for heating and/or power generation (Niesner et al., 2013; Köppel et al., 2014).



Fig. 16. Schematic of biogas plant, showing key valorisation steps and pathways (Figure from Bidart (2013)).

What are potential future perspectives for biogas in Germany, given the reduced financial support for electricity from biogas and the widespread reservations against biogas from energy crops? In order to compensate the reduced income from FIT or auction prices for electricity from biogas, plant operators need to open up new income streams in the near and middle future. At the moment, three such streams look promising.

The first one is offering system services to network operators, e.g. balancing energy (Bundesnetzagentur, 2017b). In order to supply these services, plant operators have to invest into flexibility, i.e. gas storage and additional CHP units (Hahn et al., 2014; Hochloff and Braun, 2014; Ertem et al., 2016; Lauer et al., 2017). The German Renewable Energy Sources Act incentivizes flexibilization through a premium but the prices for these services have decreased significantly reducing financial attractiveness (Bundesnetzagentur, 2017b; Purkus et al., 2018). Plants with a cumulative installed capacity of 2.8 GWel which equals about 60% of installed electrical capacity of biogas plants overall in Germany are already pursueing this strategy and benefiting from the flexibility premium under the REA (Fraunhofer IEE, 2018).

A second income stream is revenues from selling heat. Many plants have unused heat and/or are selling their heat into applications with a low value added that only pay low prices, such as grain or wood drying (Herbes et al., 2018a). The application that we found most often in the survey dataset was heating of the digester (91% of all plants under review, multiple answers possible), followed by heating of residential buildings (81%), wood drying (47%), heating commercial buildings (45%) and grain drying (37%). Innovative applications like using the heat for pre-treating the input material (e.g. Maroušek et al., 2018) or producing bio coal (e.g. Maroušek, 2014) did not play any role for the biogas plants under review. This is in line with the literature as reviewed in Herbes et al. (2018a).

A third income stream is the upgrading and monetarization of digestates. The majority of digestates today is sold to agricultural businesses in the vicinity of the biogas plant without upgrading. Farmers pay very low prices and in some regions that are rich in nutrients, they even ask fees for taking on digestate. On the other hand, hobby gardeners, landscapers and other potential target groups outside the agricultural sector are looking for organic fertilizers and soil products that do not deplete finite resources e.g. in phosphorous. By upgrading their digestate and turning it into a marketable product, biogas plant operators can increase their value added considerably (Dahlin et al., 2015, 2016, 2017, 2018).

A fourth income stream that would replace rather than supplant income from FIT, is revenues from selling biomethane. In order to realize this stream, biogas plant operators have to invest into biogas upgrading facilities (Herbes, 2015).

Another important issue is which feedstock German biogas plants will use in the future, especially given the widespread resistance against using energy crops. This will influence the longterm future of biogas in Germany. Researchers and practitioners alike are constantly looking into new types of feedstock, such as aquatic macrophytes or algae that do not compete with food pro-duction (Ertem et al., 2017; Herbes et al., 2018b). However, when looking at the total technical biomass potential in Germany in 2050, it becomes clear that the vast majority comes from energy crops (41%) that are cultivated on arable land and wood (38%), where the latter is not suitable for biogas plants. Household waste as a noncompeting biomass on the other hand is rather negligible with only 7% (FNR, 2017). As of today, two thirds (69%) of the technical potential of residual and waste materials, both from agriculture and from other sectors in Germany, are already used, which leaves an unused technical potential of only 31 Mt of dry matter that is either unused or where the use is unclear. If we look at material that is suitable for biogas plants, the unused potential is even much smaller with around 18 Mt of dry matter and a total energetic value of 213 PJ. 66% of that potential are straw and the rest consists mostly of various types of manure (FNR, 2015). To summarize, if the biogas sector is to grow significantly, there is no way around energy crops. In other words: if biogas will be largely restricted to waste and residues in the future, a growth of this sector is impossible.

A.2. Information about biogas plant survey

able 5						
ocations of	biogas	plants	in	the	samp	le

Federal state	Number of biogas plants in the sample (total: 602)
Baden-Wuerttemberg	93
Bavaria	280
Berlin	0
Brandenburg	21
Bremen	0
Hamburg	0
Hesse	19
Mecklenburg-Western Pomerania	1
Lower Saxony	86
North Rhine-Westphalia	30
Rhineland-Palatinate	16
Saarland	1
Saxony	4
Saxony-Anhalt	4
Schleswig-Holstein	21
Thuringia	9
No location available	17

Table 6 Size of biogas plants in the sample

	orogas	Plane	 line	Sum	pic		

Size	Number of biogas plants in the sample (total: 602)
Below 70 kW _{el}	11
71 to 150 kW _{et}	44
151 to 500 kWet	238
500 to 1,000 kWet	214
Above 1,000 kWet	83
Missing value	12

A.3. Heat demand calculation

The number of buildings in the 111 areas n_{b111} and the number of buildings in the 112 areas n_{b112} is calculated in equation Eqs. (13) and (14) using the total number of buildings in the square kilometre n_b as well as the area sizes of the 111 areas a_{111} and 112 areas a_{112} .

$$n_{b111} = \frac{\sum 0.9 \cdot a_{111}}{\sum 0.9 \cdot a_{111} + \sum 0.65 \cdot a_{112}} \cdot n_b \tag{13}$$

$$n_{b112} = \frac{\sum 0.65 \cdot a_{112}}{\sum 0.9 \cdot a_{111} + \sum 0.65 \cdot a_{112}} \cdot n_b \tag{14}$$

The values 0.9 and 0.65 are the mentioned building densities. With the help of this procedure, the data from the census is assigned to the settlement areas. Furthermore, for comparison with data at the municipal level, it is necessary to assign settlement areas to municipalities. In QGIS, the settlement areas were also intersected with administrative boundaries of the German municipalities from Lenk et al. (2017b) for this purpose.

In Landesamt für Statistik Niedersachsen (2014), the average living space per apartment can be found for all federal states. For federal states, districts and municipalities, Statistisches Bundesamt (2015a) indicates the number of residential buildings with one apartment (F_1), two apartments (F_2) and three or more apartments (F_3). In addition, the total living space is given in m² (F_{total}). The mean living space (F_{mean}) could be calculated for each federal state using Eq. (15).

$$F_{mean} = \frac{F_{total}}{F_1 + F_2 \cdot 2 + F_3 \cdot x}$$
(15)

Variable x represents the average number of apartments in a residential building with more than two apartments. The variable is adjusted iteratively for each federal state until the mean living space corresponds to the specified value of Landesamt für Statistik Niedersachsen (2014) at the federal state level. Then the value of x was adopted for all municipalities in the state. In this way, the average living space $F_{mean,m}$ was determined for each German municipality *m*. For 141 of the 38,414 CLC areas, the average living space of the federal state was adopted, since there were no values for the municipalities in the housing data.

In order to calculate the living space LS_1 on CLC settlement area level, the number of apartments n_a in the CLC area is required. The census data contains the categories "number of buildings with living space" with: 1 apartment, 2 apartments, 3–6 apartments, 7–12 apartments and more than 13 apartments. In the last 3 categories, the average value was estimated so that the sum of the living space in the CLC areas is equal to the total living space in of the apartment number (40,411,000) calculated in this way is taken as the number of households in the settlement areas for later calculations. Compared with the actual number of apartments on 31.12.2010 of 40,479,000, the deviation is only -0.15% (Statistisches Bundesamt, 2017a). Then the living space per CLC area *LS*₁ was calculated using the number of apartments *n*_a and the mean living space *F*_{mean.m} in the municipality (cf. Eq. (16)).

$$LS_1 = F_{mean,m} \cdot n_a \tag{16}$$

Now the sum of the CLC's living space LS_{total1} for each municipality is compared with the measured total living space of this municipality LS_{total2} (for which data are available). If the sum of the living space LS_{total1} deviates from the total living space of the municipality LS_{total2} , the new living space of each CLC area LS_2 will be adjusted accordingly to Eq. (17).

$$LS_2 = LS_1 - \left(\frac{LS_1}{LS_{total1}} \cdot (LS_{total1} - LS_{total2})\right)$$
(17)

As described above, percentages for building age are also given in the census. With the help of these parameters and specific heat demand per year, square meter and building type, the total specific heat demand per settlement area can be determined. For this purpose, the consumption values for single-family houses, twofamily houses and multi-family houses are taken from Walberg et al. (2011). The calculation of consumption values in Walberg et al. (2011) also takes into account the modernisation rate for each age group. The building age classes from the census and in Walberg et al. (2011) are not completely identical, as Table 7 shows. The data has therefore been assigned in such a way that a minimal error occurs. The assignment was made according to the colours in Table 7.⁵

Table 7

Comparison of the building age classes from Statistisches Bundesamt (2015b) and from Walberg et al. (2011) as well as the allocation of the classes by colour.

Statistisches Bundesamt (2015b)	Walberg et al. (2011)
Before 1919	Before 1918
1919-1948	1918-1948
	1949-1957
1949-1978	1958-1968
	1969-1978
1979-1986	1979-1987
1987-1990	1988-1993
1991-1995	
1996-2000	1994-2001
2001-2004	
2005-2008	2002-2008
2009 and later	

residential buildings in Germany (i.e. $3,670,870,000 \text{ m}^2$). For this purpose, an average value of 3-6 apartments was assumed to be 5, for 7-12 apartments 9.5 and for 13 or more apartments 16. The sum

⁵ Please refer to the online version of the article for the colours in this table and all figures.

The census data does not reveal the age profiles of different building types in a settlement. Therefore, a mean heat demand must be calculated for the different building types. The building stock model from McKenna et al. (2013) was used to determine how the single-family houses/two-family houses (SFHs) and multifamily houses (MFHs) are distributed among the building age classes. This was differentiated according to new and old federal states. With the help of the specific heat demand per age group and building type from Walberg et al. (2011), the total heat demand for old and new federal states can be determined for SFHs and MFHs. These total heat demands divided by the number of SFHs/MFHs gives the mean specific heat demand per SFHs or MFHs in a settlement area.

In order to calculate the total heat demand in a settlement area (SA), the share of SFHs or MFHs in the settlement must be known. For this purpose, the mean living space (differentiated by age group, building type and federal state from the building stock model) was multiplied by the number of apartments in SFHs or MFHs in the settlement. In the case of MFHs, the number of apartments x from Eq. (15) was used to determine the number of apartments. The shares of SFHs in the living space were then multiplied by the specific heat demand of the SFHs in the settlement area (MFHs analogously). The two mean specific heat demands are then added together to form the mean specific heat demand hdmean per settlement area. Now the total heat demand per settlement area HD_{SAtotal} can be calculated according to Eq. (18).

$$HD_{SAtotal} = hd_{mean} \cdot LS_2 \tag{18}$$

A.4. Determination of specific heat distribution costs

An important parameter to determine the specific heat distribution cost is the plot ratio e. The plot ratio is a city planning parameter that captures the building density within an area. Plot ratio values are used to categorize typical city districts: (A) inner city areas (e = 0.5 - 2.0), (B) outer city areas (e = 0.3 - 0.5) and (C) sparse areas (e = 0-0.3). Those parameter ranges are based on Statens Planverk (1985). The plot ratio is calculated with the population density PDSA and the total building space BSASA of the settlement area divided by the total population P of the settlement. By dividing the calculated residential area per settlement area by the residential area per person of the municipality from Statistische Ämter des Bundes und der Länder (2011), the number of inhabitants per settlement area can be estimated. All in all, Germany will then have a population of 81.711,000, close to the actual population of 2010 (81.750,000, Statistisches Bundesamt, 2017b). By dividing the number of inhabitants by the settlement area, the population density can be determined and in the following step the plot ratio e (Eq. (19), cf. Persson and Werner, 2011).

$$e = PD_{SA} \cdot BSA_{SA}/P \tag{19}$$

The effective width w is a parameter that describes the relationship between a land area and the length of district heat pipelines within this land area. It can also be seen as a correction factor to avoid the overestimation of distribution costs and is based on the plot ratio e (Persson and Werner, 2011). The effective width is calculated by using Eq. (20).

$$w = 61.8 \cdot e^{-0.15} \tag{20}$$

In order to calculate the investment for heat distribution, the linear heat density LHDSA is necessary. The linear heat density is calculated in Eq. (21) based on the effective width, the plot ratio and the specific heat demand.

$$LHD_{SA} = e \cdot w \cdot \frac{HD_{SA}}{BSA_{CA}}$$
(21)

Subsequently the average diameter of the DH grid d_{SA} is calculated as this is one of the major cost influencing factors (cf. Eq. (22)). The determination of the diameter depends on the linear heat density.

$$d_{SA} = 0.0486 \cdot \ln(LHD_{SA}) + 0.007 \tag{22}$$

Finally, the determined values are used to calculate the specific heat distribution costs HDC_{SA} using Eq. (23). These costs represent the investment for distributing 1 GJ of heat inside a settlement area. The construction cost parameters C1 and C2 vary depending on the plot ratio e. C1 is a base cost that solely depends on the length of the heat pipe, whereas C2 depends on the pipe diameter d_{SA} . Three district types are considered in this study and for each type there are different cost parameters applied. The higher the plot ratio e, the higher are C1 and C2. In Table 8 the values for the construction cost parameters C1 and C2 can be found for the according district plot ratio and the district type. Furthermore, an annuity factor af of 0.08 is integrated in the calculation.

$$HDC_{SA} = \frac{af \cdot (C1 + C2 \cdot d_{SA})}{LHD_{SA}}$$
(23)

Table 8

Cost parameter values and district type according to plot ratio range (Persson and Verner, 2011)

Plot ratio (e)	C1 [€/m]	C2 [€/m ²]	District Type
0.5 < e	286 214	2022	Inner city area (A) Outer city area (B)
0 < e < 0.3	151	1378	Park area (C)

References

- American Association for Public Opinion Research (AAPOR), 2015. Standard definitions. Final dispositions of case codes and outcome rates for surveys. Available online at. https://www.aapor.org/AAPOR_Main/media/publications/Standard-Definitions2015_8theditionwithchanges_April2015_logo.pdf. checked on 2/22/ 2018.
- Arbeitsgemeinschaft QM Fernwärme, 2017. Planungshandbuch Fernwärme. Balaman, Yilmaz, Sebnem, Selim, Hasan, 2016. Sustainable design of renewable energy supply chains integrated with district heating systems. A fuzzy opti-mization approach. J. Clean. Prod. 133, 863–885. https://doi.org/10.1016/ j.jclepro.2016.06.001. Issou, D., Kleyböcker, A., McKenna, R., Möst, D., Fichtner, W., 2011. An econom
- Balu Balissott, D., Reporter A., MCRenna, R., Most, D., Fichner, W. 2011. All economic analysis of three operational co-digestion biogas plants in Germany. Waste and Biomass Val 3 (1), 23–41.
 BDEW Bundesverband der Energie- und Wasserwirtschaft e.V. 2018. BDEW-strompreisanalyse Mai 2017. Berlin.
- Bidart, C. 2013. A Techno-Economic Assessment of the Generation and Usage of Biogenic Gases in Chile as a Substitute of Natural Gas. Dissertation at the Karlsruhe Institute for Technology (KIT), Karlsruhe. https://publikationen. bibliothek.kit.edu/10000336622.
- BMJV, 2018. Gesetz für die Erhaltung, die Modernisierung und den Ausbau der Kraft-Wärme-Kopplung (Kraft-Wärme-Kopplungsgesetz - KWKG). Edited by Bundesministeriums der Justiz und für Verbraucherschutz. Available online at. ://www.destatis.de/DE/Methoden/Rechtsgrundlagen/Statistikbe eiche
- https://www.destatis.de/DE/Methoden/Rechtsgrundlagen/Statistikbereiche/ Inhalte/1058_KWKG.pdf?__blob=publicationFile. checked on 3/2/2018.
 Bordin, Chiara, Gordini, Angelo, Vigo, Daniele, 2016. An optimization approach for district heating strategic network design. Eur. J. Oper. Res. 252 (1), 296–307. https://doiorg/10.1016/j.ejor.2015.12.049.
 Brueckner, S., Miró, L., Cabeza, L.F., Pehnt, M., Laevemann, E., 2014. Methods to estimate the industrial waste heat potential of regions a categorization and literature review. Renew. Sustain. Energy Rev. 2014 (38), 164–171.
 Bühler, F., Petrović, S., Karlsson, K., Elmegard, B., 2017. Industrial excess heat for district heating in Denmark. Appl. Energy 205, 991–1001, 2017.

- Bundesnetzagentur, 2017a. Hintergrundpapier Ergebnisse der Ausschreibung für Biomasse vom 1. September 2017. Available online at. https://www. bundesnetzagentur.de/sharedDocs/Downloads/DE/Sackgebiete/Energie/ Unternehmen_Institutionen/Ausschreibungen_2017/Biomasse17/
 - Hintergrundpapier_0109_2017,pdf; jsessionid=1378DD2EAA88519EF6C80AB26B404C85? blob=nublicationFile&v=6
- Bundesnetzagentur, 2017b. Monitoring report. Available online at https://www bundesnetzagentur.de/SharedDocs/Downloads/EN/Areas/ElectricityGas/ CollectionCompanySpecificData/Monitoring/MonitoringReport2017.pdf;
- Collection.companySpecificultal/Monitoring/MonitoringReport2017.pdf; jsessionid=71D2FECI53FC2E747ABA881ED01A05F8? __blob=publicationFile&v=2. checked on 10/29/2018. Bundesnetzagentur, 2018. EEG-Anlagenstammdaten (außer PV-Gebäudeanlagen). Available online at https://www.bundesnetzagentur.de/DE/Sachgebiete/ ElektrizitaetundGas/Unternehmen_Institutionen/ErneuerbareEnergien/ ZahlenDatenInformationen/EEG Registerdaten/EEG Registerdaten node.html. checked on 3/4/2018.
- checked on 3/4/2018.
 Burgess, P.J., Casado, M.R., Gavu, J., Mead, A., Cockerill, T., Lord, R., Van der Horst, D., Howard, D.C., 2012. A framework for reviewing the trade-offs between, renewable energy, food, feed and wood production at a local level. Renew. Sustain. Energy Rev. 16 (1), 129–142.
 CA.R.M.E.N. e.V. 2012. Nahwarmenetze und Bioenergieanlagen. Ein Beitrag zur
- effizienten Wärmenutzung und zum Klimaschutz, Available online at. https:// ennstoffe/merkblatt_Nahwae
- www.carmen-ev.de/files/festbrennstoffe/merkblatt_Nahwaermenetz_carmen_ ev.pdf. checked on 2/16/2018.
 Casisi, M., Pinamonti, P., Reini, M., 2009. Optimal lay-out and operation of combined heat & power (CHP) distributed generation systems. Energy 34 (12), 2175–2183. https://doi.org/10.1016/j.energy.2008.10.019.
 Chinese, D., 2008. Optimal size and layout planning for district heating and cooling
- Chinese, D., 2008. Optimal size and layout planning for district heating and cooling networks with distributed generation options. In. J. Energy Sect. Manag. 2 (3), 385–419. https://doi.org/10.1108/17506220810892946.
 Clearingstelle EEG, 2006. EEG 2004. Available online at. https://www.clearingstelle-eeg-kwkg.de/eg2004. checked on 3/2/2018.
 Clearingstelle EEG, 2009. EEG 2009. Available online at. https://www.clearingstelle-eeg-kwkg.de/eg2004. checked on 3/2/2018.
- e/eeg2009, checked on 2/26/2018. Clearingstelle EEG, 2012. EEG 2012. Available online at. https://www.clearingstelle
- Ceaningstele Liv, 2012. Liv 2012. Vanalos comine at https://www.ceaningstele-eeg-kwkg.de/eeg2012. checked on 2/26/2018. Coss, Stefano, Verda, Vittorio, Le-Corre, Oliver, 2018. Multi-objective optimization of district heating network model and assessment of demand side measures using the load deviation index. J. Clean. Prod. 182, 338–351. https://doi.org/10.1016/
- iclepro.2018.02.083.
- Dahin, Johannes, Herbes, Carsten, Nelles, Michael, 2015. Biogas digestate market-ing. Qualitative insights into the supply side. Resour. Conserv. Recycl. 104, 152–161. https://doi.org/10.1016/j.resconrec.2015.08.013.

- Dahlin, Johannes, Halbherr, Verena, Kurz, Peter, Nelles, Michael, Herbes, Carsten, 2016. Marketing green fertilizers. Insights into consumer preferences. Sustainability 8 (11), 1168. https://doi.org/10.3390/su8111169.
 Dahlin, Johannes, Nelles, Michael, Herbes, Carsten, 2017. Biogas digestate management. Evaluating the attitudes and perceptions of German gardeners towards digestate-based soil amendments. Resour. Conserv. Recycl. 118, 27–38. https://doi.org/10.1016/j.resconrec.2016.11.020.
 Dahlin, Johannes, Beuthner, Christoph, Halbherr, Verena, Kurz, Peter, Nelles, Michael, Herbes, Carsten, 2018. Sustainable compost and potting soil marketing. Private gardener preferences. J. Clean. Prod. https://doi.org/10.1016/j.jclepro.2018.10.068.
- DBFZ, 2015. Stromerzeugung aus Biomasse. Available online at. https://www.dbfz de/fileadmin/eeg_monitoring/berichte/01_Monitoring_ZB_Mai_2015.pdf. checked on 2/19/2018.
- CHECKEU 011 // 197/2018.
 Delangle, Axelle, Lambert, Romain S.C., Shah, Nilay, Acha, Salvador, Markides, Christos N., 2017. Modelling and optimising the marginal expansion of an existing district heating network. Energy 140, 209–223. https://doi.org/10.1016/j.energy.2017.08.066.
 dena, 2016. Branchenbarometer Biomethan: Daten, Fakten und Trends zu Biomethan. dena, Berlin.
 Dobergek D. Gorizanec D. 2000. Optimisation of the staff size.
- Dobersek, D., Goricanec, D., 2009. Optimisation of tree path pipe network with
- Dobriek, D., donane, D., 2009. Optimisation of the path pipe network with nonlinear optimisation method. Appl. Therm. Eng. 29 (8–9), 1584–1591. https://doi.org/10.1016/j.applthermaleng.2008.07.017.
 DIW Berlin, D.I.W., EEFA, 2017. Atswertungstabellen zur Energiebilanz Deutschland. 1990 bis 2016. Edited by AGEB AG Energiebilanzen e.V. Available online at https://www.ag-energiebilanzen.de/. checked on 1/4/2018.
 EEA, 1995. CORINE Land Cover Part 2: Nomenclature. European Environment
- Agency. Available online at. https://www.eea.europa.eu
- Agency. Available of thinks at https://www.cea.europa.eu/downloads/ 5c537b70226944b8bfe7bdc9470100ff/1488296244/clc-2012-vector.pdf. checked on 11/28/2017.
- EEA. 2018. Land Cover Country Fact Sheets 2012. European Environment Agency.
- EEA, 2018. Land Cover Country Fact Sheets 2012. European Environment Agency. https://www.eea.europa.eu/themes/landuse/land-cover-country-fact-sheets? b_start:int=0, zuletzt geprüft am 18.10.2018. Online verfügbar unter. Engel, T., 2015. Die daten der Energymap zum download. Deutsche Gesellschaft für Sonnenergie e.V. (DGS). Available online at. http://www.energymap.info/ download.html. checked on 11/28/2017.
- Ertem, Funda Cansu, Martínez-Blanco, Julia, Finkbeiner, Matthias, Neubauer, Peter, Junne, Stefan, 2016. Life cycle assessment of flexibly fed biogas processes for an

- improved demand-oriented biogas supply. Bioresour. Technol. 219, 536-544. https://doi.org/10.1016/j.biortech.2016.07.123. Ertem, Funda Cansu, Neubauer, Peter, Junne, Stefan, 2017. Environmental life cycle
- em, runa cansu, recubater, reter, june, stelan, 2017. Environmental nie cycle assessment of biogas production from marine macroalgal feedstock for the substitution of energy crops. J. Clean. Prod. 140, 977–985. https://doi.org/ 10.1016/j.jclepro.2016.08.041. opean Biogas Association, 2016. 17,358 biogas plants in Europe (31/12/2015). Available online at. http://european-biogas.eu/wp-content/uploads/2017/01/ Graph-1-Number-of-biogas-plants.png. checked on 10/29/2018. Nature 2017. Environment of the substant of the subs
- Europear
- Graphe 1-Munder-of-Diogas-phants, Julg. Checked on 10(29)/2016.
 Euroheat and Power, 2016. Top district heating countries Euroheat & power 2015 survey analysis. Available online at. https://www.euroheat.org/news/district-neargy-in-the-news/top-district-heating-countries-euroheat-power-2015-survey-analysis/. checked on 5/14/2019.
 Euroheat and Power, 2017. District energy in Germany. Available online at. https://www.euroheat.org/knowledge-centre/district-energy-germany/. checked on 9/22/016.
- 27/2018.
- Fang, H., Xia, J., Zhu, K., Su, Y., Jiang, Y., 2013. Industrial waste heat utilisation for low
- Fang, H., Xia, J., Zhu, K., Su, Y., Jiang, Y., 2013. Industrial waste heat utilisation for low temperature district heating. Energy Policy 52, 236–246, 2013.
 Fang, H., Xia, J., Jiang, Y., 2015. Key issues and solutions in a district heating system using low grade industrial waste heat. Energy 86, 589–602, 2015.
 Fazlollahi, Samira, Girardin, Luc, Maréchal, François, 2014. Clustering urban areas for optimizing the design and the operation of district energy systems. In: 24th
- For optimizing the design and the operation of use first energy systems. In: 24th European Symposium on Computer Aided Process Engineering, vol. 33. Elsevier (Computer Aided Chemical Engineering), pp. 1291–1296. hagentur Nachwachsende Rohstoffe (FNR), 2015. Biomassepotenziale von Rest-und Abfallstoffen status Quo in Deutschland. Available online at. https:// mediathek.fnr.de/band-36-biomassepotenziale-von-rest-und-abfallstoffen. Fachagentu
- mediathek.fnr.de/ omassepotenziale-von-rest-und html. checked on 10/29/2018.
- html. checked on 10/29/2018. Echagentur Nachwachsende Rohstoffe (FNR), 2017. Basisdaten bioenergie deutschland 2017. Available online at. http://www.fnr.de/fileadmin/allgemein/ pdf/broschueren/Basisdaten_Bioenergie_2018.pdf. checked on 10/29/2018. Fraunhofer IEE, 2018. Vorbereitung und Begleitung bei der Erstellung eines Erfah-rungsberichts gemäß §97 Erneuerbare-Energien-Gesetz. Teilvorhaben II at: biomasse. Zwischenbericht. Available online at. https://www.erneuerbare-energien.de/EE/Redaktion/DE/Downloads/bmwi_de/bericht-eeg-2-biomasse. odf/_bibm=aublestichtestiebienese_bergien_de/bergienestiebienese.pdf _blob=publicationFile&v=8. checked on 10/29/2018.
- Fraunhofer UMSICHT, 1998. Leitfaden Nahwärme. Fraunhofer IRB Verlag, Stuttgart. https://doi.org/10.24406/UMSICHT-PX-52182.
 Geofabrik, 2018. Regierungsbezirk Stuttgart. Available online at. http://download.
- geofabrik.de/europe/germany/baden-wuerttemberg/stuttgart-regbez.html. checked on 2/15/2018.
- German Biogas Association, 2017. Biogas market data in Germany 2016/20176/2017. Available online at https://www.biogas.org/edom/webfi/d.pc Gernam biogas Association, 2017. biogas inarket tada in centrality 2016/2017/ Available online at. https://www.biogas.org/edcom/web/vb.nsf/id/DE_ Branchenzahlen/Sfile/17-10-13_Biogasindustryfigures-2016-2017.pdf. Good, Jügern, 2004. Planungshandbuch. Straubing: CA.R.M.E.N. e.V. vol. 4. Schrif-
- tenreihe QM Holzheizwerke.
- Cuelpa, Elisa, Mutani, Cuglielimina, Todeschi, Valeria, Verda, Vittorio, 2018. Reduction of CO2 emissions in urban areas through optimal expansion of existing district heating networks. J. Clean. Prod. 204, 117–129. https://doi.org/ 10.1016/j.jclepro.2018.08.272. 10.1016/j.jclepro.2018.08.272. Hahn, Henning, Ganagin, Waldemar, Hartmann, Kilian, Wachendorf, Michael, 2014.
- Cost analysis of concepts for a demand oriented biogas supply for flexible po-wer generation. Bioresour. Technol. 170, 211-220. https://doi.org/10.1016/ i.biortech.2014.07.085.
- J. biorrech. 2014.07.085. Herbes, Carsten, 2015. Marketing für Biomethan. In: Friege, Christian, Herbes, Carsten (Eds.), Marketing Erneuerbarer Energien. Grundlagen, Geschäftsmodelle, Fallbeispiele. Springer Gabler, Wiesbaden, pp. 183–201. Herbes, C., Halbhert, V., 2017. Stärkere Wärmenutzung in Biogasanlagen kann sich lohnen. Biogas J. 1, 68–71, 2017.
- Ionnen. Biogas J., 168–71, 2017.
 Herbes, Carsten, Jirka, Eva, Braun, Jan Philipp, Pukall, Klaus, 2014. Der gesell-schaftliche Diskurs um den "Maisdeckel" vor und nach der Novelle des Erneuerbare-Energien-Gesetzes (EEG) 2012The Social Discourse on the "Maize Cap" before and after the 2012 Amendment of the German Renewable Energies Act (EEG). In: GAIA Ecological Perspectives for Science and Society, vol. 23, pp. 100–108. https://doi.org/10.14512/gaia.23.2.7, 2.
- bes, Carsten, Halbherr, Verna, Braun, Lorenz, 2018a. Factors influencing prices for heat from biogas plants. Appl. Energy 221, 308–318. https://doi.org/10.1016/ j.anenergy.2018.03.188. Herb
- J.apenergy.2018.03.188. bes, Carsten, Brummer, Vasco, Roth, Sandra, Röhl, Markus, 2018b. Using aquatic plant biomass from de-weeding in biogas processes—an economically viable option? Energ Sustain Soc 8 (1), 175. https://doi.org/10.1186/s13705-018-0163-
- Hochloff, Patrick, Braun, Martin, 2014, Optimizing biogas plants with excess power unit and storage capacity in electricity and control reserve markets. Biomenergy 65, 125–135. https://doi.org/10.0106/j.biombioe.2013.12.012. assen, T., Kõnig, A., Eltrop. L., 2014 Bioenergy villages in Germany: bringing a low carbon energy supply for rural areas into practice. Renew. Energy 61, 74–80.
- Karner, Katharina, Theissing, Matthias, Kienberger, Thomas, 2016, Energy efficiency
- Karner, Katharina, Iheissing, Matthias, Kienberger, Thomas, 2016. Energy efficiency for industries through synergies with urban areas. J. Clean. Prod. 119, 167–177. https://doi.org/10.1016/j.jclepro.2016.02.010.
 Karner, Katharina, McKenna, Russell, Klobasa, Marian, Kienberger, Thomas, 2018. Industrial excess hear tercovery in industry-city networks. A technical, envi-ronmental and economic assessment of heat flexibility. J. Clean. Prod. 193, 771, 729. https://doi.org/10.1016/j.jclepro.2016.05.045. oi.org/10.1016/j.jcl 771–783 http enro 2018 05 04³
- Karschin, Ingo, Geldermann, Jutta, 2015. Efficient cogeneration and district heating

- is in bioenergy villages. An optimization approach. J. Clean. Prod. 104, 305-314. https://doi.org/10.1016/j.jclepro.2015.03.086. Klaus Novy Institut e.V. trend:research, 2011. Marktakteure: Erneuerbare-Energien-Anlagen in der Stromerzeugung", [Market actors: renewable energy plants in
- Anlagen in der Strömerzeugung", [Market actors: renewable energy plants in electricity generation], Köln.
 Klein, C., Rozanski, K., Gailfuß, M., Kukuk, J., Beck, T., 2014. BHKW-kenndaten 2014/ 15. ASUE Arbeitsgemeinschaft für sparsamen und umweltfreundlichen Ener-gieverbrauch e-V. Berlin. Awailable online at. http://www.asue.de/ blockheizkraftwerke/broschueren/05_10_14_bhkw-kenndaten_2014-15. checked on 2/13/2018.
- checked on 2/13/2018.
 Köppel, W., Schreck, H., Lubenau, U., Erler, R., 2014. Abschlussbericht. Monitoring Biogas II. DVGW Deutscher Verein des Gas- und Wasserfaches e. V. Available online at. https://www.dvgw.de/index.php?elD=dumpFile&t=I6K=2167& toleen=7ce5a7b7c61b4d21er9f3a6fd694/67/d0dece9b5. checked on 3/2/2018.
 Landesamt für Statistik Niedersachsen, 2014. Gebäude- und Wohnungsbestand in
- Juesania für Statistik viedersächsen, 2014. Gebaude- und wonnungsbestand in Deutschland. Erste Ergebnisse der Gebäude- und Wohnungszbählung 2011. Statistische Ämter des Bundes und der Länder, Hannover. Available online at. https://www.statistik.rlp.de/fileadmin/dokumente/gemeinschaftsveroeff/zen/ Zensus_GWZ_2014.pdf. checked on 11/28/2017. Jer, Markus, Thrän, Daniela, 2017. Biogas plants and surplus generation: cost driver or reducer in the future German electricity system? Energy Policy 109, 324–336. https://doi.org/10.1016/ji.appd.101702016
- aiver of reducer in the function certain electricity system/ Energy Pointy 109, 324–336. https://doi.org/10.1016/j.enpol.2017.07016.
 Lauer, Markus, Dotzauer, Martin, Hennig, Christiane, Lehmann, Monique, Nebel, Eva, Postel, Jan, et al., 2017. Flexible power generation scenarios for biogas plants operated in Germany. Impacts on economic viability and GHG emissions. Int. J. Energy Res. 41 (1), 63–80. https://doi.org/10.1002/er.3592.
 Lenk, M., Boumann, J., Sippel, T., 2017a. CORINE land cover 10 ha. Available online at http://www.good.becomtem.com/paced.ten/ards.ten/paced.ten/paced
- http://www.geodatenzentrum.de/geodaten/gdz_rahmen.gdz_div?gdz_ spr=deu&gdz_akt_zeile=5&gdz_anz_zeile=1&gdz_unt_zeile=22&gdz_user_ id=0. checked on 11/28/2017.
- Intervet dir 11/26/2017. k, M., Bourmann, J., Sippel, T., 2017b. Verwaltungsgebiete 1:250 000 stand 010.12017. Available online at. http://www.geodatenzentrum.de/geodaten/gdz_ rahmen.gdz_div?gdz_spr=deu&gdz_akt_zeile=5&gdz_anz_zeile=1&gdz_unt_ mile_1/vermin_werside_0_babeder_ar_210/010217. =14&gdz_user_id=0. checked on 12/18/2017.
- LfU, 2016. Berechnung der CO2-Emissionen. Bayrisches Landesamt für Umwelt. Available online at. https://www.umweltpakt.bayern.de/energie_klima/ fachwissen/217/berechnung-co2-emissionen. updated on 9/21/2016, checked on 1/0108.
- IachWitsell/21/Jerechnung Gereinstein on 1/9/2018.
 Liebetrau, Jan, Sträuber, Heike, Kretzschmar, Jörg, Denysenko, Velina, Nelles, Michael, 2017. Anaerobic digestion. Advances in Biochemical Engineer-international Conference on Co
- ing/biotechnology. https://doi.org/10.1007/10_2016_67. Markard, Jochen, Wirth, Steffen, Truffer, Bernhard, 2016. Institutional dynamics and Markau, Jochen, With, Schein, Huner, Bernarda, 2016. Institutuolar dynamics and technology legitimacy – a framework and a case study on biogas technology. Res. Policy 45 (1), 330–344. https://doi.org/10.1016/j.respol.2015.10.009.
 Maroušek, Josef, 2014. Significant breakthrough in biochar cost reduction. Clean
- Technol. Environ. Policy 16 (8), 1821-1825. https://doi.org/10.1007/s10098-014
- 0/30-y. Maroušek, Josef, Stehel, Vojtěch, Vochozka, Marek, Maroušková, Anna, Kolář, Ladislav, 2018. Postponing of the intracellular disintegration step im-proves efficiency of phytomass processing. J. Clean. Prod. 199, 173–176. https:// doi.org/10.1016/j.jclepro.2018.07.183.
 Marquant, Julien F., Bollinger, L. Andrew, Evins, Ralph, Carmeliet, Jan, 2018. A new
- combined clustering method to Analyse the potential of district heating net-works at large-scale. Energy 156, 73–83. https://doi.org/10.1016/ jenergy.2018.05.027.
- Jennegy.2018.05.027. Marty, Fabien, Serra, Sylvain, Sochard, Sabine, Reneaume, Jean-Michel, 2018. Simultaneous optimization of the district heating network topology and the organic rankine cycle sizing of a geothermal plant. Energy. https://doi.org/ 10.1016/j.energy.2018.05.110.
- McKenna, R., 2018. The double-edged sword of decentralized energy autonomy Energy Policy 113, 747-750. February 2018. https://doi.org/10.1016/j.enp
- McKenna, R.C., Norman, J.B., 2010. Spatial modelling of industrial heat loads and recovery potentials in the UK. Energy Policy 38, 5878–5891. https://doi.org/ 10.1016/j.enpol.2010.05.042.
- ICLUDIOJ. enpol. 2010.05.042.
 McKenna, Russell, Merkel, Erik, Fehrenbach, Daniel, Mehne, Stephanie, Fichtner, Wolf, 2013. Energy efficiency in the German residential sector. A bottom-up building-stock-model-based analysis in the context of energy-political targets. Build. Environ. 62, 77–88. https://doi.org/10.1016/j.jbuildenv.2013.01.002.
- j.buildemv.2013.01.002. Mergner, et al., 2013. National Policy Enforcement for Heat Use from Biogas in Austria, Coratia, Czech Republic, Denmark, Germany, Italy, Latvia, Poland and Romania, Biogas Heat Project Deliverable 2.4, Project IEE/11/025. Merkel, E., McKenna, R., Fehrenbach, D., Fichtner, W., 2016. A model-based assessment of climate and energy targets for the German residential heat system. J. Clean. Prod. 142 (4), 3151–3173. https://doi.org/10.1016/j.jclepro.2016. 10.153
- Microsoft, 2018, Bing maps, Available online at, https://www.bing.com/maps/. Min
- checked on 2/22/2018 maps. Perinder of mile at: https://www.bing.com/maps/. checked on 2/22/2018.
 6, L, Brückner, S., Cabeza, L.F., 2015. Mapping and discussing Industrial Waste Heat (IWH) potentials for different countries. Renew. Sustain. Energy Rev. 2015 (51), 847–855. Miró, L., Brueckner, S., McKenna, R., Cabeza, LF., 2016. Methodologies to estimate
- industrial waste heat potential by transferring key figures: a case study for Spain. Appl. Energy 2016 (169), 866–873.

- Müller, M.O., Stämpfli, A., Dold, U., Hammer, T., 2011. Energy Autarky: a conceptual framework for sustainable regional development. Energy Policy 39, 5800–5810. Nielsen, Steffen, 2014. A geographic method for high resolution spatial heat plan-
- Indiata, Scherk Verschlier, Holding (Joseph and Karley Karley
- Nykvist, B., Nilsson, M., 2015. Rapidly falling costs of battery packs for electric ve hicles. Nature Clim. Change Lett. 5, 329–332. https://doi.org/10.1038 NCLIMATE2564.
- NCLIMATE2564.
 OpenStreetMap contributors, 2018. OpenStreetMap. Available online at. https:// www.openstreetmap.org/#map=5/51.330/10.453. checked on 1/29/2018.
 Pablo-Romero, Maria del, P., Sánchez-Braza, Antonio, Salvador-Ponce, Jesús, Sánchez-Labrador, Natalia, 2017. An overview of feed-in tariffs, premiums and tenders to promote electricity from biogas in the EU-28. In: Renewable and Sustainable Energy Reviews, vol. 73, pp. 1366–1379. https://doi.org/10.1016/ i.exec701201132 j.rser.2017.01.132
- Person, Urban, Werner, Sven, 2011. Heat distribution and the future competitive-ness of district heating. Appl. Energy 88 (3), 568–576. https://doi.org/10.1016/ j.apenergy.2010.09.020. Person, U., Möller, B., Werner, S., 2014. Heat Roadmap Europe: identifying strategic
- heat synergy regions. Energy Policy 2014, 74-81. Pfnür, A., Winiewska, B., Mailch, B., Qschatz, B., 2016. Dezentrale vs. zentrale
- Print, A., Wintewska, B., Maitch, B., QSchatz, B., 2016. Dezentrale vs. zentrale Wärmeversorgung im deutschen Wärmemarkt. Vergleichende Studie aus energetischer und ökonomischer Sicht. Darmstadt. Available online at. http:// www.bdh-koein.de/fileadmin/user_upload/pressemitteilungen_pdf/studie_ dezentrale.vs.zentrale.waerneversorgung.pdf. checked on 2/16/2018. Pöschl, Martina, Ward, Shane, Owende, Philip, 2010. Evaluation of energy effciency.
- of various biogas production and utilization pathways. Appl. Energy 87 (11), 3305-3321. https://doi.org/10.1016/j.apenergy.2010.05.011.
- Dirkus, Alexandra, Gawel, Erik, Szarka, Nora, Lauer, Markus, Lenz, Volker, Ortwein, Andreas, et al., 2018. Contributions of flexible power generation from biomass to a secure and cost-effective electricity supply-a review of potentials, incentives and obstacles in Germany. Energ Sustain Soc 8 (1), 151. https:// doi.org/10.1186/s13705-018-0157-0.
- Rae, Callum, Bradley, Fiona, 2012. Energy autonomy in sustainable communities a

- Rae, Callum, Bradley, Fiona, 2012. Energy autonomy in sustainable communities a review of key issues. In: Renewable and Sustainable Energy Reviews, vol. 16, pp. 6497–6506.
 Rutz, Dominik, Güntert, David, 2012. Biogas checklist for administrative bodies. In: WIP–Renewable Energies.
 Rutz, D., Mergner, R., Janssen, R., 2015. Sustainable Heat Use of Biogas Plants. A Handbook, second ed. WIP Renewable Energies, Munich, Germany.
 Scarlat, Nicolae, Dallemand, Jean-François, Fahl, Fernando, 2018. Biogas. Developments and perspectives in Europe. Renew. Energy 129, 457–472. https://doi.org/10.1016/j.renene.2018.03.006.
 Schmidt, J., Schönhart, M., Biberacher, M., Guggenberger, T., Hausl, S., Kalt, G., Leduc, s., Schardinger, I., Schmid, E., 2012. Regional energy autarky: potentials, costs and consequences for an Austrian region. Energy Policy 47, 211–221.
- Schwermer. S., 2012. Ökonomische Bewertung von Umweltschäden. METHOD-Schwermer, S., 2012. Ökonomische Bewertung von Umweltschäden. METHOD-ENKONVENTION 2.0 ZUR SCHÄTZUNG VON UMWELITKOSTEN. Umweltbunde-samt. Available online at. https://www.umweltbundesamt.de/sites/default/ files/medien/378/publikationen/uba_methodenkonvention_2.0__2012_ gesamt.pdf. checked on 2/19/2018.Sollero, V.M., Chacartegui, R., Ortiz, C., Velazquez, R., 2018. Potential of biomass district heating systems in rural areas. Energy 156, 132–143. https://doi.org/ 10.1016/j.energy.2018.05.051.Statista, 2018a. Gewerbe- und Industriestrompreise in Deutschland bis 2017. Available online at. https://de.statista.com/statistk/daten/studie/154902/ umfrage/strompreise- fuer-industrie-und-gewerbe-seit-2006/, checked on 02/
- umfrage/strompreise-fuer-industrie-und-gewerbe-seit-2006/. checked on 02/ 05/2017
- 05/2017. Statista, 2018b. Endenergieverbrauch der privaten Haushalte für Wohnen in Deutschland nach Anwendungsbereich im Jahresvergleich 2005 und 2015 (in Terawattstunden). Available online at. https://destatista.com/statistik/daten/ studie/165364/umfrage/energieverbrauch-der-privaten-haushalte-fuer-wohnen-2000-und-2009/. checked on 1/4/2018.
- Wohnen-2000-und-2009, checked on 1/4/2018.
 Statistische Ännter des Bundes und der Länder, 2011. Gebäude mit Wohnraum und Wohnungen nach Gebäudearten stichtag 09.05.2011. Available online at. https://www.regionalstatistik.de/genesis/online/data; jsessionid=5D5DC080C0G6EDCC42E56152EFF2DA884.reg27 operation=abruftabelleAbrufen&selectionname=31211-01-01-5&levelindex=1&levelid=1513613620454&index=2. checked on 12/18/2017.
- Szerveindex=1 retevende=1515/35/35/35/35/36/acindex=2. checked on 12/18/2017. Statistisches Bundesamt, 2015a. Fortscherbiung auf Basis GWZ 2011 Wohngebäude-und Wohnungsbestand stichtag 31.12. regionale Tiefe: gemeinden, Samt-/ Verbandsgemeinden. Available online at. https://www.regionalstatistik.de/ genesis/online/datajsessionid=E0E1257886684086722411D184342C44.reg27 operation=abrufabelleAbrufen&selectionname=31231-02-01-5& levelindex=1 & levelid=151186272847kindex=5. checked on 11/28/2017. Statistisches Bundesamt, 2015b. Zensus 2011. Methoden und Verfahren. Statistische den Ruden zur den Zufen Wichenden, Auslieble anglina et. https://
- Änter des Bundes und der Länder, Wiesbaden, Available online at. https:// www.zensus2011.de/SharedDocs/Downloads/DE/Publikationen/Aufsaetze_ Archiv/2015_06_MethodenUndVerfahren.pdf?__blob=publicationFile&v=6. checked on 11/28/2017
- Statistisches Bundesamt, 2016a. Bauen und Wohnen. Mikrozensus zusatzerhe-bung 2014 Bestand und Struktur der Wohneinheiten Wohnsituation der Haushalte. Wiesbaden. Available online at. https://www.destatis.de/DE/

J.M. Weinand et al. / Journal of Cleaner Production 238 (2019) 117756

- Publikationen/Thematisch/EinkommenKonsumLebensbedingungen/Wohnen/ WohnsituationHaushalte2055001149004.pdf; jsessionid=1970E8F571F3A560CA79DE823C7889CF.InternetLive17 __blob=publicationFile. checked on 1/9/2018. Statistisches Bundesamt, 2016b. Datenangebot zum Zensusatlas Klassifizierte Ergebnisse. Ergebnisse des Zensus am 9. Mai 2011 pro km². Statistische Änter des Bundes und der Länder, Wiesbaden. Available online at. https://www. opengeodata.nrw.de/produkte/jbevoelkerung/zensus2011/grgebnisse_Ikm-gitter/Datensatzbeschreibung_klassierte_Werte_Ikm-Gitter.pdf. checked on 11/ 28/2012 28/2017.
- Statistisches Bundesamt. 2017a. Bestand an Wohnungen und Wohngebäuden Justiches bundesami, 2017a, bestand an wohnungen und wohngebauden -bauabgang von Wohnungen und Wohngebäuden - Jange Reihen ab 1969 – 2016. Available online at. https://www.destatis.de/DE/Publikationen/Thematisch/ Bauen/Wohnsistuation/FortschreibungWohnungsbestandPDF_5312301.pdf? __blob=publicationFile.checked on 12/18/2017.
- __blob=publicationFile. checked on 12/18/2017. Statistisches Bundesamt. 2017b. Bevölkerungsstand. Available online at. https:// www.destatis.de/DE/Zahlenfakten/GesellschaftStaat/Bevoelkerung/ Bevoelkerungsstand/Tabellen_lirbevO3.html.checked on 12/18/2017. Statistisches Bundesamt, 2017c. Gemeindeverzeichnis Gebietsstand: 30.06.2017 (2. Quartal). Alle politisch selbständigen Gemeinden (mit Gemeindeverband) in Deutschland nach Fläche, Bevölkerung (auf Grundlage des Zensus 2011), Bev-ölkerungsdichte und der Postleitzahl des Verwaltungssitzes der Gemeinde. Statistische Amster des Bundes und der Lönder Wischbuden. Statistische Ämter des Bundes und der Tänder, Wiesbaden. Available online at https://www.destatis.de/DE/ZahlenFakten/LaenderRegionen/Regionales/ Gemeindeverzeichnis/Administrativ/Archiv/GVAuszugQ/AuszugGV2QAktuell.
- Gemeindeverzeichnis/Administrativ/Archiv/GVAuszugQ/AuszugGV2QAktuell. html. checked on 9/7/2017. SWB, 2014. Technische Anschlussbestimmungen für den Anschluss an das Fernwärmenetz der Stadtwerke Bönnigheim. Stadtwerke Bönnigheim. Available

online at. http://www.stadtwerke-boennigheim.de/resources/ecics_42.pdf. checked on 3/1/2018. Theissing, M., 2012. CO2 - emissionen und Primärenergiefaktor der Fernwärme in Österreich. In: Fernwärmetage 2012. Wels.

- Osterietch in: reinwalmetage 2012. wells. trend: research, 2017. Eigentümerstruktur: Erneuerbare Energien. Entwicklung der Akteursvielfalt, Rolle der Energieversorger, Ausblick bis 2020 (20-01174), Available online at. http://www.trendresearch.de/studie.php?s=672, checked on 2/28/2018.
- Umweltbundesamt, 2017a. Energieeffizienz in Zahlen. Edited by Bundesministe-Omweinbundesami, 20174. Einergeeinzeitz im Zahleit. Earderd by Bundeshiniste-rium für Wirtschaft und Energie (BMWi). Available online at. http://www. bmwi.de/Redaktion/DE/Publikationen/Energie/energieeffizienz-in-zahlen.pdf? __blob=publicationFile&v=10. checked on 1/4/2018. Umweltbundesamt. 2017b. Energieverbrauch privater haushalte. Available online at. https://www.umweltbundesamt.de/daten/private-haushalte.konsum/ wohnen/energieverbrauch-privater-haushalte. checked on 2/19/2018.

- wonnenjenergieverbrauch-privater-hausnatte, checked on 2/19/2018.
 Umweltbundesamt, 2017. Treibhausgas-emissionen in deutschland. Available on-line at. https://www.unweltbundesamt.de/daten/klima/treibhausgas-emissionen-in-deutschland#textpart-1. checked on 1/29/2018.
 Unternährer, Jrérémy, Moret, Stefano, Jost, Stéphane, Maréchal, François, 2017.
 Spatial clustering for district heating integration in urban energy systems. Application to geothermal energy. Appl. Energy 190, 749–763. https://doi.org/ 10.1016/j.anergery.2015.12.136. 10.1016/j.apenergy.2016.12.136.
- IU.UU6/j.apenergy.2016.12.136.
 Walberg, Dietmar, Holz, Astrid, Gniechwitz, Timo, Schulze, Thorsten, 2011. In: Wohnungsbau in Deutschland 2011 Modernisierung oder Bestandsersatz. Studie zum Zustand und der Zukunftsfähigkeit des deutschen "Kleinen Woh-nungsbaus". Arbeitsgemeinschaft für Zeitgemäßes Bauen, Kiel (Baufor-schungsbericht, 59).

Applied Energy 251 (2019) 113367



Contents lists available at ScienceDirect

Applied Energy



journal homepage: www.elsevier.com/locate/apenergy

Developing a combinatorial optimisation approach to design district heating networks based on deep geothermal energy



Jann Michael Weinand^{a,*}, Max Kleinebrahm^a, Russell McKenna^b, Kai Mainzer^a, Wolf Fichtner^a

^a Chair of Energy Economics, Institute for Industrial Production (IIP), Karlsruhe Institute for Technology, Germany
^b Energy System Analysis, Division of Sustainability, Department of Technology, Management and Economics, Technical University of Denmark (DTU), Denmark

HIGHLIGHTS

• Development of combinatorial optimisation model to design district heating networks.

- Optimisation of the location of the district heating plant in municipalities.
- Deep geothermal energy is used as heat source.
- Heuristic has high accuracy and is significantly faster than the optimisation.
 Applicable for any German municipality and extendable to other locations.
- Applicable for any derman municipality and extendable to other locations.

ARTICLE INFO

Keywords: District heating network design Deep geothermal energy Geothermal plant Location planning Combinatorial optimisation Graph theory

ABSTRACT

Plants increasingly exploit high geothermal energy potentials in German district heating networks. Municipal planners need instruments to design the district heating network for geothermal heat. This paper presents a combinatorial mixed-integer linear optimisation model and a three-stage heuristic to determine the minimum-cost district heating systems in municipalities. The central innovations are the ability to optimise both the structure of the heating network and the location of the heating plant, the consideration of partial heat supply from district heating and the scalability to larger municipalities. A comparison of optimisation and heuristic for three exemplary municipalities demonstrates the efficiency of the latter: the optimisation takes between 500% and $1 \times 10^7\%$ more time than the heuristic. The deviations of the heuristic's calculated total investments for the district heating system compared to the optimisation are in all cases below 5%, and in 80% of cases below 0.3%. The efficiency of the heuristic is further demonstrated by comparison with the Nearest-Neighbour-Heuristic; which is less efficient and substantially overestimates the total costs by up to 80%. The heuristic can also be used to design district heating networks in holistic energy system optimisations due to the novel possibility of conscription with the scale of heat losses, as well as taking additional geological and topographical conditions into account.

1. Introduction

The German energy sector is currently undergoing radical structural change due to ambitious national climate targets and supportive energy policy. This change is dominated by the expansion of renewable energy generation technologies, which are mainly decentrally exploited due to their characteristics [1]. This development has led to the generation of 33% of the electricity in Germany by renewable energies in 2017 [2]. In contrast, the proportion of renewable heat supply is around 13% [3]. The installed capacity of renewable heat supply is around 155 GW of wind energy (on- and offshore), 42 GW photovoltaic systems, about 7

GW of bioenergy [4] and 39 MW of geothermal energy [3]. In comparison to the other renewable energies, relatively little electricity is generated by geothermal plants (GTPs, for abbrevations see Table 1) in Germany. Despite the potential of GTPs to make a major contribution to reducing greenhouse gas emissions, the installed capacity in Germany is still very low compared to the available resources [5].

The currently (2017) installed 30 GTPs in Germany generate about 155 GWh of electricity [3] and 1.3 TWh of heat [5] annually, which is generally used for district heating (DH) applications [6]. However, the German district heating networks (DHNs) are currently mainly supplied with heat by conventional gas and coal-fired plants [7]. Most of the

* Corresponding author.

E-mail address: jann.weinand@kit.edu (J.M. Weinand).

https://doi.org/10.1016/j.apenergy.2019.113367

Received 5 March 2019; Received in revised form 3 May 2019; Accepted 17 May 2019 0306-2619/@2019 Elsevier Ltd. All rights reserved.

Table 1

Abbreviation	Full description
CLC	CORINE Land Cover
DCMST	Degree-constrained minimum spanning tree
DH	District heating
DHN	District heating network
GTP	Geothermal plant
MST	Minimum spanning tree
NGB	North German Basin
ORC	Organic Rankine cycle
URG	Upper Rhine Graben

heat is consumed in the residential sector [7], in which the market share of DH is 13.8% [8]. Between 2000 and 2017, DHNs were constantly expanded [9] and the consumption of DH in the residential sector increased by about 40% [7].

The low emission of pollutants during the operation of GTPs means that they could provide low-carbon heat for future DH systems. They could thereby contribute to higher levels of renewable heat supply, as already achieved in some DHNs based on GTPs at the municipal level [10]. As the literature review in Section 2 shows, transferable planning instruments are needed to support decision-making in the dimensioning of DHNs. The focus of this study is on the development of a novel method for designing a DH system based on geothermal heat in an arbitrary municipality. In comparison to existing work, this paper presents a generic approach for the simultaneous optimisation of DHN and DH plant locations under local geographic conditions. The novel possibility of connecting an arbitrary number of buildings to the DHN is thereby also given. Due to the high complexity of this combinatorial optimisation problem, an optimisation is not possible for a municipality with a large number of settlements. Therefore, a heuristic is developed for the planning of the DHNs, whose efficiency is evaluated with the aid of the optimisation model. The location of the GTP, as well as the DHN, are specified depending on the location, heat demand and heat density of the individual settlements within a municipality. In the first stage of the three-stage heuristic, the integrated selection of the GTP location and the initial design of the DHN is performed. In the second stage, an algorithm is employed to optimise the design of the DHN. Based on the results, the optimum heat allocation to the settlements connected to the DHN is determined in the third stage.

The paper is structured as follows. Section 2 provides an overview of the literature and clarifies the context of this paper. Section 3 then explains the methodology, before Section 4 presents the results. A sensitivity analysis with subsequent critical discussion follows in Section 5. The paper concludes with a summary and conclusions in Section 6.

2. Literature review

This section highlights the novelty of the methodology developed in this study. In the literature many studies analyse existing DHNs and do not design new networks. For example, based on the results of a demonstration system for decentralized cooling in DHN, a simulation is developed in Udomsri et al. [11] to improve the performance of the existing network. Furthermore, Fang & Lahdelma [12] design a new methodology to minimise production and distribution costs in existing DHNs during simultaneous heat production in multiple heating plants. In Pini Prato et al. [13], an existing DHN is considered as a test case for optimising the matching between the heat generation of a combined heat and power plant and the heat demand of the thermal consumers of the network. The development of biomass-based industrial DHNs in Italy in Chinese & Meneghetti [14] involves the sizing of new DHNs, however, these networks are not designed from a topographical perspective. Finally, the study by Mørck et al. [15] analyses the EU CONCERTO project class 1, in which settlements with low energy buildings were constructed which are supplied with heat by biomass and solar based DHNs.

In fewer studies, which are discussed in the following, the design of new DHNs is optimised. First of all, Section 2.1 shows that previous energy system analyses with geothermal DH systems do not dimension the DHNs. Afterwards, approaches for dimensioning DHNs are analysed and the differences to the developed methods are identified (cf. Section 2.2). Section 2.3 concludes by clarifying the differences between the developed methods and typical combinatorial optimisation problems.

2.1. Energy system analyses with geothermal district heating

A few studies have already investigated energy systems in municipalities in which geothermal DH is used. Østergaard et al. [16] investigate the feasibility of supplying the municipality of Aalborg in Denmark with renewable energy through a combination of low-temperature geothermal heat, wind power plants and biomass. The results of the simulations show that these technologies cannot cover the demand of Aalborg in every hour and that energy has to be imported instead. The study of Østergaard & Lund [17] also takes deep geothermal energy into account as a heat generation plant with DHN. It analyses a 100% renewable energy supply for Frederikshavn in Denmark. The simulations demonstrate that the operation of the GTP and the DHN reduces the energy imports of Frederikshavn. In the study of Sveinbjörnsson et al. [18], the municipality of Sønderberg in Denmark is considered, which is aiming for zero net CO₂ emissions by 2029. As in Østergaard & Lund [17], the GTP is combined with an absorption heat pump. The optimisations indicate that by supplementing combustion with modern energy conversion technologies, the climate targets can be achieved in a cost- and energy-efficient manner.

In the studies described above, the DHN already exists and is not further analysed. Besides, the location of the DH plant is not optimised but is specified manually beforehand. Some studies on DH in energy system analyses have been conducted in which the DH is not based on geothermal energy such as Möller & Lund [19]. The authors examine the expansion of DH in a region in Denmark which is currently supplied with natural gas. The energy system of the region, however, is not optimised, but only analysed with the help of various scenarios. One conclusion is to replace natural gas with DH, especially in the vicinity of cities, and to increase the share of DH in the heat supply from 46% to 50–70%. However, this paper indicates that the use of geothermal energy should be included in future analyses when considering DH. According to this literature review, so far, energy system optimisations of entire municipalities in which the DHNs are designed simultaneously with other technologies have not been implemented.

2.2. Algorithms for district heating design

Studies in which the design of DHNs is determined without the involvement of other energy technologies, by contrast, have been conducted with several different approaches. In general, however, these studies are based on a basic methodology, in which different nodes/ buildings/settlements are connected by edges/pipelines. This methodology is also the basis of the analysis in the present study. Table 2 summarises important characteristics of the studies discussed in this section. This paper distinguishes between bottom-up methods, in which the calculations are based on data of individual buildings and roads, and top-down approaches, which use e.g. heat densities and population densities.

The majority of studies dealing with the dimensioning and placement of DHNs are limited to DH pipelines within one urban area and do not consider the connection of several urban areas. Casisi et al. [20], for example, determine the optimum layout of a DHN in conjunction with combined heat & power units in the city centre of Pordenone, Italy. The distances between the six considered buildings are specified

Table 2

Overview of relevant studies of algorithms for DH design (R = residential sector, C = commercial sector, I = industrial sector, BU = bottom-up method, T	D = top
down method).	

Study	Method	Network building or expansion?	Determination of plant location?	Sector	Region	Easily <i>trans</i> - ferable?
[21]	Optimisation model (BU)	Building	No	R, C, I	Urban area (Italy)	No
[20]	Optimisation model (BU)	Building	No	С	Urban area (Italy)	No
[23]	Optimisation model (BU)	Building	No	n.a.	Urban area	No
[31]	Cluster analysis and optimisation model (TD)	Building	No	n.a.	Urban area	No
[27]	Heuristic (BU)	Expansion	No	n.a.	Arbitrary Danish municipality	Yes
[29]	Optimisation model (BU)	Expansion	No	R, C	Urban area	No
[25]	Heuristic (BU)	Building	Yes	R	German town	No
[22]	Heuristic (BU)	Building	No	R, C	Urban area	No
[28]	Optimisation model (BU)	Expansion	No	n.a.	Urban area	No
[33]	Optimisation model (BU)	Building	No	R, C	Urban area	No
[32]	Cluster analysis and graph theory (TD)	Building	No	n.a.	Urban areas (Switzerland)	No
[24]	Optimisation model (BU)	Building	No	n.a.	n.a.	No
Present Study	Optimisation model and heuristic (TD)	Building	Yes	R	Arbitrary German municipality	Yes

beforehand. The location of the DH plant is not optimised here, but the authors note that this could lead to cost reductions. Chinese [21] designs two district heating and cooling networks in the municipality of Udine in Italy with the help of an optimisation model. In designing the size and layout of the network, central and distributed generation of heating and cooling are combined and compared in consideration of the network costs. In the simulation model of Bratoev et al. [22], the DHN with connections to the buildings is first generated on the basis of a road network. Afterwards, the location of the DH plant can be selected, which is then connected to the DHN via the shortest distance. The algorithm requires an extensive building database, which must be collected in advance, for example through surveys. Dobersek & Goricanec [23] determine the optimal tree branch path of a DHN in an urban area taking economy and functionality into account. The influence of construction cost, pump and electric energy cost on the type of optimal network is investigated. The locations of heat source and consumers are defined in advance and the optimal network for a complete supply of all consumers has to be determined. The edges are also fixed in advance and cannot lead from each node to any other node. Marty et al. [24] highlight the relevance of a simultaneous optimisation of the DHN and the Organic Rankine Cycle (ORC) as parts of a GTP. Although the algorithm is similar to the methodology of the present study, some differences are apparent. The location of the DH plant is specified in the study and is not optimised. Furthermore, there is always a consumer who is connected, i.e. the construction of the DH plant is not decided but the DHN is built in any case. The heat demand of the consumers has to be completely covered and no statement is made on the transferability of the model. Overall, however, it can be stated that the model in the study cannot be used for the purposes intended by the present study

Falke et al. [25] is the only other study, compared to the present study, which determines the location of the district heating plant. However, the entire planning of the DHN is based on a heuristic whose performance is not demonstrated in comparison to an optimal solution. First, the heuristic determines a few random DHNs that connect all buildings. The minimum network length is then calculated using the Kruskal [26] algorithm. This algorithm, which is based on the Greedy heuristic, would not be suitable for the problem presented in the present study, since it connects all nodes with each other. In the present study, however, it must be decided simultaneously which nodes should be part of the network at all.

In Nielsen [27], Delangle et al. [28] and Bordin et al. [29], existing DHNs are expanded. Nielsen [27] developed an algorithm to economically evaluate possible network expansions based on existing DHNs.

This is done on the basis of cost calculations for heat production, distribution and transmission, based on geographical data. Although the model presented is transferable to Danish regions, its application in German municipalities is hindered by the unavailability of data. To the authors' knowledge, shapefiles with building locations are not available for the whole of Germany. An extension of existing DHNs is also investigated by Delangle et al. [28]. In addition to DH, gas boilers, biomass boilers, heat pumps and heat storages are considered in a mixedinteger linear optimisation. The model can identify the optimal investment schedule for a DHN extension by a couple of buildings. The methodology can be applied to other DHNs, but in this case, the existing DHN and the potential buildings to be connected have to be known. In the optimisation approach of DHNs in Bordin et al. [29] the tree configuration of a network is determined by starting from one heat supply plant. An existing network is assumed, which can be extended by potential pipelines and customers to maximise the net profit. For more complex analyses in future studies, the authors propose the use of heuristics to reduce computing times. Like the studies described above. the model from Bordin et al. [29] is suitable for the individual planning of a DHN, but not as part of a larger energy system optimisation model of a municipality or region. Although there have been some studies on the expansion of existing DHNs, a planning instrument should focus more on the construction of new DHNs, since the share of DH in residential heating systems is less than 5% in over 85% of German municipalities [30].

In Fazlollahi et al. [31], Unternährer et al. [32] and Marquant et al. [33], the DHN is dimensioned based on cluster analyses, in order to reduce the model complexity. The clustering of urban areas in Fazlollahi et al. [31] results in combined urban districts whose energy demand and distribution costs of DH can be aggregated in order to subsequently optimise the DHN design. In Unternährer et al. [32], the optimum design of the DHN is determined for each cluster. As in the present analysis, geothermal energy is used for DH. The authors highlight the economic value of geothermal energy for DH supply. However, memory restrictions occur during this analysis. Since only one settlement and not an entire municipality is analysed, this approach is not suitable to meet the objectives of the present study. Marquant et al. [33] use a density-based cluster analysis to divide large city-scale problems into sub-problems. The subsequent optimisation of the DHNs in the individual clusters requires demand profiles of the individual buildings and is therefore not automatically transferable.

In summary, the studies analysed use optimisation models and heuristics respectively (cf. Table 2). In articles about optimisation models, the choice of heuristics is suggested for more complex studies,

but these two approaches are not compared in any of the studies. In addition, none of the studies in which the networks are newly built is suitable for application to entire municipalities, but only for planning DHNs in urban areas. In addition, the novel possibility to connect an arbitrary number of buildings to the DHN is given in this study. This results in various optimal solutions for the connected buildings, depending on the arbitrary specification of heat coverage by the DHN. This also enables the use of discrete DHN solutions in holistic energy system optimisations (cf. Sections 2.1 and 5.4). Furthermore, the methodologies of the studies cannot be applied to arbitrary municipalities in Germany. All of these points are addressed with the approaches in the present article.

2.3. Combinatorial optimisation of networks

Most, if not all of the studies in the previous section use the methodology of combinatorial optimisation. As particularly evident in the studies of Dobersek & Goricanec [23] and Bordin et al. [29], the optimisation of network design in DH systems falls within the scope of minimum spanning tree (MST) problems. Therefore, articles on MST problems are analysed in the following in order to draw conclusions for DHN optimisations.

Smith & Walters [34] describe an evolutionary approach based on genetic algorithms to find the optimal trees in undirected water, gas or material networks. Based on the construction and operating costs, a network tree is identified that is close to the optimum. In order to be able to execute the genetic algorithm, however, starting solutions must first be identified. Blanco et al. [35] extend the MST problem so that the nodes are not points but belong to regions with a certain geometry to reflect the uncertainty of the location of a node. Efficient algorithms and formulations of the problem have been implemented in Blanco et al. [35] to solve the problem in a reasonable time. The methodology with the uncertain locations of the heat consumers could also be relevant for DH designing models (cf. Section 5.4). Fernández et al. [36] focus on multiobjective spanning trees and aggregate the vector of the objective values. Formulations of the problem are developed which reduce the number of decision variables and thus the time needed to solve the problem.

Gao & Jia [37], Salgueiro et al. [38] and Gouveia et al. [39] investigate another form of the MST, the degree-constrained minimum spanning tree (DCMST). The problem formulation of the DCMST contains additional restrictions regarding the degree of the nodes, i.e. how many edges are connected to a node. This restriction is not meaningful for the DHNs considered in the present study, as a large number of pipelines should be allowed to enter and leave the nodes. In addition, there is no need to limit the number of end nodes/leaves as done in Gouveia & Simonetti [40], whereby they develop a competitive model for the max-leaves problem. Cerrone et al. [41] show in their paper that a spanning tree with a minimum number of leaves is better suited to minimising the number of light-splitting devices required in optical networks than two other MST problems. In Marín [42], the MST problem is adjusted to minimise the number of branches, i.e. nodes with a degree greater than two. As in the present study, a multi-stage heuristic is developed for this problem. However, the limitation of the number of branches should not apply to DH systems either.

The approach for DHNs in the present study differs from the MST problems described above in the following points:

- Not every node has to be connected (cf. Eq. (8) in section 3.2).
- Not only the edges but also the nodes have weights (cf. Eq. (3) in section 3.2).
- Edges can connect any node to any other node (cf. Eq. (2) in section 3.2).
- A decision must be made "to which percentage" a node is connected (cf. Eq. (3), (5) and (9) in section 3.2).
- Instead of one starting point/source, there is an arbitrary number of

sources, of which one must be selected (cf. section 3.1 and Eq. (6) in section 3.2).

The aim of this paper is not only to provide a unique optimisation problem/heuristic for the planning of DHNs in municipalities but also to solve the problem in a reasonable time. Furthermore, the method presented in the following section is applicable to every municipality in Germany.

3. Methodology

The distance to the heat consumers is crucial for the costs of DHNs. Therefore, an optimisation model, as well as a heuristic, to design DHNs and position the DH plant is developed in this study. The location of the GTP and the connection of the settlements by a DHN are determined for an arbitrary municipality considering the minimisation of the investment for the DHN. Possible locations for the GTP are specified in advance, as the following section 3.1 shows, which also describes the determination of input data and the most important assumptions (see also Table 3). The optimisation model for determining the optimal DH system is presented in section 3.2 and the heuristic in 3.3.

3.1. Input data generation and assumptions

The optimisation model and the heuristic are demonstrated using the German municipality "Groß Kreutz", which is shown in Fig. 1. The figure illustrates the possible locations for the GTP as purple circles and the pink areas are CORINE Land Cover (CLC) settlement areas from EEA [43]. The number of possible locations of the GTP can be specified manually via the distances between the purple circles. The locations were set so that 35 locations are between the minimum and maximum longitude coordinates and/or latitude coordinates. In Groß Kreutz this results in 497 possible locations i.e. 5 locations per km². Section 5.2 shows that this number is sufficient, as a further increase does not significantly affect the results. The shapefiles of these CLC areas are provided by the Federal Office of Cartography and Geodesy [44]. Some of the purple circles are close to the settlement areas since GTPs do not have to keep a minimum distance to settlements as long as measurements are taken to assess vibrations according to DIN 4150-3 [45]. Using OpenStreetMap, inadmissible areas such as settlements, water areas and forests were excluded as locations for the GTP.¹ The settlements marked in pink differ in terms of their heat demand and building density. The most important assumptions include the specific costs c_p for pipelines outside the settlements, which are estimated to be 200 ϵ/m [46,47]. In addition to pipelines, these costs also include network pumps and planning costs [46]. The relatively low value of 200 €/m is supposed to reflect the receipt of a subsidy in the context of the Combined Heat and Power Act (KWKG) amounting to 100 €/m [48]. However, the 200 €/m are above the 160 €/m in the geothermal project Grünwald [49]. The pipeline costs within the settlements, which depend largely on building density, are estimated using the methodology of Persson & Werner [50]. By applying this methodology, exact roads within the settlement do not have to be known in order to estimate the costs. The DH substations are assumed to have a reference size of 250 kW [51,52]. If the heat demand in a settlement exceeds 250 kW, more than one substation has to be built. According to Connolly et al.

¹The red encirclement at the right of Fig. 1 represents a possible error in the placement of the purple circles. The error occurs due to the fact that Open-StreetMap uses slighdy different municipal boundaries, which differ from the municipal boundaries of the shapefiles from the Federal Office of Cartography and Geodesy used here to illustrate the municipal boundaries. Due to the large distance to the settlements, they are rather unlikely to be chosen as the location for the GTP.

Applied Energy 251 (2019) 113367

Table 3 Data and methods used in the present study.			
Description	Value	Unit	Sources of data / method
DH substation size	250	kW	Connolly et al. [51], Le Truong et al. [52]
DH substation cost c_s	21,500	e	Connolly et al. [51], Le Truong et al. [52]
Inadmissible areas	-		OpenStreetMap contributors [60]
Pipeline costs within the settlements		1	Persson & Werner [50]
Settlement areas (CORINE Land Cover data)	<u></u>		EEA [43], Lenk et al. [44]
Settlement data (i.e. heat demand)			Weinand et al. [53]
Specific costs c _p for pipelines outside the settlements	200	€/m	C.A.R.M.E.N. e.V [46], Pfnür et al. [47]



Fig. 1. Possible locations (purple circles) for building the GTP in the municipality Groß Kreutz (Havel). The background maps in this and the following figures are from OpenStreetMap contributors [60].

[51] and Le Truong et al. [52], the costs for one substation c_s are assumed to be 4000 ε and 70 $\varepsilon/kW,$ i.e. 21,500 ε for 250 kW.

Parameters for all German settlements required for this study are taken from Weinand et al. [53], including the heat demands, living spaces, population densities and coordinates of the centroids. This also includes the shares of existing DH supplies in the total heat supply, which are subtracted from the heat demand of the settlements in the optimisation as well as in the heuristic. This is intended to prevent DHNs from being built where networks already exist. These data for all German settlements and the heuristic are made available as online supplementary material associated with this article. In addition, the necessary input data for the municipality Groß Kreutz are provided as an example. This input is needed to execute the optimisation, which is explained in the following section.

3.2. District heating system optimisation

In this section, variables are marked in bold letters. A nomenclature of variables and parameters can be found in Table 4.

The combinatorial optimisation model for the cost-optimised placement of the GTP as well as the determination of the optimal DHN minimises the total costs C_{total} (cf. Eqs. (1)–(13)). These costs consist of the costs for the DH pipelines in and outside the settlements, $C_{pipe,in}$ and $C_{\rm pipe,out}$ as well as the DH substations $C_{\rm sub}$ according to Eq. (1).

 $minC_{total} = C_{pipe,out} + C_{pipe,bn} + C_{sub}$ (1) subject to

 $C_{pipe,out} = \left(\sum_{l=1}^{m} \sum_{i=1}^{n} \mathbf{x}_{G,l,i} \cdot l_{G,i,l} + \sum_{i=1}^{n} \sum_{j=1}^{n} \mathbf{x}_{S,i,j} \cdot l_{S,i,j}\right) \cdot c_{p}$ (2)

$$C_{pipe,in} = \sum_{i=1}^{n} (HDC_i \cdot \sum_{t=1}^{T=3760} \dot{Q}_{i,t} \cdot \boldsymbol{p}_i)$$

$$C_{sub} = \sum_{i=1}^{n} (N_{S,i} \cdot \mathbf{p}_i) \cdot c_s$$
(4)

(3)

 $\dot{Q}_{G,t} - \sum_{l=1}^{n} \dot{Q}_{l,t} \cdot \mathbf{p}_{l} = 0 \qquad \forall t = 1, 2, \cdots, 8760$ (5)

$$\sum_{l=1}^{m} \boldsymbol{b}_{GL,l} = 1 \tag{6}$$

$$\sum_{l=1}^{n} x_{G,l,l} \le \mathbf{b}_{GL,l} \cdot n \qquad \forall l = 1, \cdots, m$$
(7)

Table 4

Nomenclature.		
Variable / Parameter	Description	Unit
b _{GL}	Binary variable for deciding whether a geothermal plant should be built at a specific location	-
b _{GS}	Binary variable for deciding whether a settlement can be reached by a certain path	-
C _p	Specific costs for pipelines outside the settlements	€/m
Ĉ _{pipe in}	Investment for district heating pipelines inside a settlement	€
Cpine.out	Investment for district heating pipelines outside a settlement	€
C _s	Investment for one district heating substation	€
Csub	Investment for all district heating substations	€
Ctotal	Investment for the entire district heating network	€
Dheat	Heat demand of the municipality	kWh
HDC	Specific heat distribution costs	€/kWh
l _G	Distance between a district heating plant and a settlement	m
ls	Distance between two settlements	m
М	A big number	-
Ns	Number of district heating substations	-
р	Proportion of the heat demand in a settlement covered by district heating	-
Q.	Heat supply to a settlement	kWh
Q _G	Heat generation of a geothermal plant	kWh
Si	Matrix in which every path from every possible source to the sink i is contained	-
x _G	Binary variable for deciding whether a district heating pipeline between a district heating plant and a settlement should be built	-
Xs	Binary variable for deciding whether a district heating pipeline between two settlements should be built	-

$\begin{pmatrix} \mathbf{b}_{GS,i,1} * 1 \\ \mathbf{b}_{GS,i,2} * 2 \\ \vdots \\ \mathbf{b}_{GS,i,\left[\sum_{a=1}^{n} (n-1) / (n-a) \right] - 1} \\ n \\ \leq S_{i} \qquad \forall i = 1,, n$	
$b_{GS,i,\sum_{a=1}^{n}(n-1)!/(n-a)!}$	(8)
$p_i \le b_{GS,i,1} + b_{GS,i,2} + \dots + b_{GS,i,\sum_{a=1}^n (n-1)!/(n-a)!} \forall i$	$= 1, \dots, n$ (9)
$\mathbf{x}_{G,l,i} \in \{0, 1\}$ $\forall l = 1, \cdots, m$ $\forall i = 1, \cdots, n$	(10)
$\mathbf{x}_{\mathbf{S},ij} \in \{0, 1\}$ $\forall i, j = 1, \cdots, n$	(11)
$\boldsymbol{b}_{GL,l} \in \{0, 1\} \qquad \forall \ l = 1, \cdots, m$	(12)

 $b_{GS,i,k} \in \{0, 1\}$ $\forall i = 1, \dots, n$ $\forall k = 1, \dots, \sum_{a=1}^{n} (n-1)!$ /(n - a)!

(13)

The costs for the pipelines outside the settlements $C_{pipe,out}$ are determined using the vectorised geodetic distances $I_{G,i,1}$ between the GTP at location I and a settlement area i as well as the distances $I_{S,i,j}$ between two settlement areas i and j (cf. Eq. (2)). The binary variables x_G and x_S are used to decide which pipelines should be constructed. Thereby m possible locations of the GTP and n different settlements are given. One of the m different geothermal locations has to be selected via the binary variable b_{GL} (cf. Eq. (6)). The variable b_{GL} then also restricts the possible pipelines that can be built using Eq. (7). This ensures that only connections starting from the chosen geothermal location can be selected.

The costs for the DH pipelines inside the settlements $C_{pipe,in}$ are determined with the help of the specific heat distribution costs HDC_i and the heat supply Q per hour t in a settlement i (cf. Eq. (3)). The variable p_i gives an indication of which proportion of the settlement is supplied with heat and therefore takes values between 0 and 1. The heat distribution cost is calculated on the basis of Persson & Werner [50], depending on the population and building density as well as the specific heat demand. The settlements have to consume the entire heat Q_G of the GTP (cf. Eq. (5)). This amount of heat should be adjusted in advance according to the specific use case or the geothermal potential available in the municipality. Fig. 2 qualitatively illustrates the influence of p on the proportion of heat demand is covered, if p = 1 is

selected. The GTP, if operated during the whole year, would generate surplus energy in summer. Although the use of the surplus energy is not relevant in this study, electricity could be generated in a GTP in this time period.

For each settlement at least one DH substation is required to connect the heating system of the buildings with the DHN. The costs for one substation c_s are multiplied by the number of substations N_s and the proportion of the settlement supplied with heat to determine the costs for DH substations C_{sub} in Eq. (4). A subtour is the term used when two or more coherent graphs are

created as a solution instead of one coherent graph. In order to prevent subtours, it must be ensured that each sink/settlement is connected with the source/GTP via a coherent path. For this purpose, equations are set up for each possible path from the source to the sink (cf. Eq. (8)). S_i represents a matrix in which every path from every possible source to the sink i is contained. The binary variable b_{GS} is used to decide by which path a certain settlement can be reached. This path must begin at the GTP. For a certain settlement, several paths could be selected, but this would lead to higher costs according to Eq. (2). For example, in the first row in Eq. (8) the pipeline would lead directly from the GTP to a settlement ($b_{GS,i,1} = 1$), while in the last row in Eq. (8) the pipeline would lead across all other settlements $(b_{GS,i,\sum_{k=1}^{n}(n-1)!/(n-k)!} = 1)$. In the penultimate row of Eq. (8) b_{GS} is also multiplied by n, since the path leads over as many settlements as in the last row, but the order of the settlements on this path is different. The number of restrictions in Eq. (8) depends on the number of settlements and can be calculated via $\Sigma^{\rm B}$ n!/(n - a)!. Therefore, the number of equations increases exponentially with the number of settlement areas in a municipality. Eq. (9) has to be included so that no settlement can be supplied with heat that is not connected to the GTP via DH pipelines. The optimisation problem described above can only be solved for municipalities with less than eight settlements due to calculation time limitations (cf. section 4.1). Even the building of equations is not possible for municipalities with more than ten settlements due to computational constraints (cf. section 4.1). In order to solve the problem for larger municipalities, a heuristic has been developed, which is described in section 3.3.

3.3. Heuristic for designing district heating systems

The heuristic is similar to algorithms for solving the combinatorial optimisation problem of the minimum spanning tree. In contrast to the general minimum spanning tree problem, the nodes to be connected are not previously fixed and the weights of the edges can change in each



Fig. 2. Influence of the variable p on the heat demand covered by the DH plant in a settlement.

iteration. The heuristic is executed for each possible GTP location, which is determined in advance (cf. Fig. 1). Before starting the heuristic, the percentage of the municipality's heat demand to be covered by the GTP must be specified (cf. Fig. 4). After determining the distance matrices $l_{\rm S}$ and $l_{\rm G},$ a loop is performed for all GTP locations. Due to these iterations the rectangular boxes in Fig. 4 are shown nested. The heuristic then selects a connection from the geothermal system to one of the settlements in the municipality. In the example of the municipality Groß Kreutz, a choice must be made between twelve settlements. The decision is made on the basis of the investment for the DH system Court per supplied amount of heat. If only costs for the pipelines would be considered, the nearest and most densely built-up areas would be selected. This would lead to high costs, as in most cases more settlements would have to be connected with DH pipelines since the first connected settlements might not occur in the optimal solution. Subsequently, the hourly heat demand Q of the settlement is subtracted from the heat supply Q_G of the GTP. The next DH connection is selected if the GTP still has heat available and there is still unmet demand in the settlements. All settlements to which a connection already exists are now included as possible heat sources. This means that in each iteration there is an additional heat source that can be connected to the other settlements.

The first stage of the heuristic is completed as soon as the heat supply of the geothermal plant is exhausted and is a modified form of the algorithm from Weinand et al. [53]. The upper part of Fig. 3 shows the result after the first stage for the case that 100% of the heat demand is covered in Groß Kreutz. The red lines represent the DH pipelines originating from the GTP, the blue ones those originating from a settlement. The GTP is illustrated as a purple circle. The upper part of the figure shows, above all in the part encircled in red, that the shortest connections are not always used. This is due to the iterative approach of the heuristic, whereby one connection is chosen in each step. Previous iterations could have selected connections that would not have been chosen if all connected settlements had been known beforehand. Therefore, the heuristic is extended by a second stage with another target criterion: the costs for the connections outside the settlements. However, the selection of settlement areas in this second stage is limited to those settlements that were connected in the previous stage. This leads to a more economical connection, as shown in the lower part of Fig. 3.

Another problem can be caused by the iterative approach. In case the heat demand is not met to 100%, the last selected settlement will only be partly supplied with heat (p < 1). However, this last connected settlement could have a higher heat density than another settlement connected to the DHN. Then it would be reasonable to supply the last selected settlement with more heat. In this case, the settlement with the lowest heat density would be supplied with less heat than in the solution of the second stage. Therefore, the heat supply is reassigned to all selected settlements on the basis of the HDC in the third stage of the heuristic.

Finally, the DH connections and the associated costs and CO_2 abatements per year are given as output. For calculating the CO_2 abatement, the actual heating technologies in the various settlements were determined in Weinand et al. [53].

4. Results

In Weinand et al. [54], the 11,100 German municipalities were clustered into ten groups with regard to their suitability for decentralised energy systems. Two of the 34 socio-energetic indicators in the cluster analysis were used to measure the potential for geothermal energy: the attainable geothermal temperature and the required drilling depth. The municipalities in Germany are suitable for low-temperature GTPs with achievable temperatures up to 190 °C. As case studies for the analysis in this paper, four municipalities are selected from the clusters with the highest potential for geothermal energy (cf. Table 5). Municipalities are selected which differ particularly in the number of settlements, heat demand and the population density. Three of the municipalities are located in the North German Basin (NGB) and Bensheim lies in the Upper Rhine Graben (URG). The parameters of the municipalities for the following analyses are listed in Table 5.

Firstly, the results and the solving time of the optimisation and the heuristic are compared in section 4.1. The results for the four municipalities from Table 5 are then presented in section 4.2. The results are determined for different levels of heat supply, from 10% in 10% steps up to 100%. The proportion of existing DH systems in the municipalities is subtracted from the heat demand in the settlements.

4.1. Comparison of optimisation and heuristic approaches

The heuristic was executed in Matlab and the optimisation was performed in GAMS using the solver CPLEX. The selected computer has the following performance properties: Intel Xeon E-1650 v2, 12 threads and 128 GB RAM. The results of the optimisation model and the heuristic are compared for the municipalities of Billerbeck, Dümmer and Bensheim, which contain three, five and seven settlements respectively. The optimisation for Groß Kreutz with more than seven settlements was not conducted because time constraints already arose during the optimisation for Bensheim with seven settlements. Thus the optimisation for Bensheim had to be terminated after 7 days in the case of 100% heat



Fig. 3. Location of the GTP (purple circle) and DH pipelines to the settlement areas for the target criteria investment per kW (upper part) and costs for the DH pipelines outside the settlements (lower part).

supply with a MIP gap of 4.8%. Building the equations in Matlab for a municipality with 8 settlements alone would take two hours and the optimisation problem would contain over 1×10^5 binary variables. In the municipalities of Billerbeck, Dümmer and Bensheim, the equations take between 35 s and 12 min to build. The time required to build the equations is not included in the following analysis.

Fig. 5 shows the percentage by which the calculated total costs are lower in the optimisation compared to the heuristic. In the case of 100% heat supply in Bensheim, the result of the lower bound was adopted due to the above-mentioned termination of the optimisation. However, it should be noted that the deviation could actually be much smaller in this case. Therefore, the following analysis will focus on how the deviations in Billerbeck are caused in the 30–70% cases and the deviation in Dümmer is caused in the 50% case.

In Billerbeck up to the 20% case, heat is supplied to settlement 2 in the heuristic as well as in the optimisation cases, since this settlement has the highest building density (for settlement 2 see Fig. 6). In the 30% case, however, there would still be heat remaining, so that another settlement has to be supplied. The error is caused by the fact that with the heuristic, after selecting a settlement, the maximum possible

Applied Energy 251 (2019) 113367



Fig. 4. Heuristic for determining the cost-optimal DH system for an arbitrary municipality considering the heat demand and heat densities of the settlements. The nested illustration of some steps in Stage 1 shows that these steps are executed several times, in this case one iteration per GTP location.

 Table 5

 Characteristics of the German municipalities Billerbeck, Groß Kreutz, Bensheim and Dümmer.

Municipality	Biller-beck	Groß Kreutz	Bens-heim	Dümmer
Cluster [54] Basin [61] Area [km ²] Population Population density in	3 NGB 91 11,593 127	8 NGB 99 8,133 82	3 URG 58 40,051 693	8 NGB 32 1,430 46
municipality [1/km ²] Number of Corine settlement areas	3	12	7	5
Average Population density in settlements [1/km ²]	3,000	1,000	3,300	1,300
Yearly heat demand [GWh] Average share of DH in settlements [%]	135 2	55 4	320 2	9 8

amount of heat is supplied to this settlement. Therefore, in the cases 30-70%, only settlement 3 with a lower building density than settlement 2 is supplied in the case with the heuristic (cf. left part of Fig. 6). By contrast, in the optimisation in these cases, settlement 2 is always supplied entirely with heat and settlement 3 proportionately (cf. right part of Fig. 6). Although this means that the costs for the pipelines outside the settlements are higher, the overall costs are lower. Above the 30% case, the deviation between optimisation and heuristic decreases further, as the proportion of the heat quantity delivered to settlement 2 becomes smaller and smaller in relation to the total heat supply.

The difference in the municipality of Dümmer in the 50% case is due

to the same reason as in Billerbeck. The right part of Fig. 7 shows the DHN resulting from the optimisation. Here settlement 4 represents an intermediate station and only 15% of the heat demand is covered. This would not be possible in the heuristic since settlement 4 would have been supplied with the complete remaining heat supply. Therefore, the more favourable solution in the heuristic is to supply settlement 3 and 5 (cf. left part of Fig. 7).

The two discussed examples show that with regard to the amount of heat that is delivered to the settlements, there is still potential for improvement in the heuristic. However, the largest deviation from the optimisation is only 5% and in most cases less than 0.1% of the total investment. If the 100% case in Bensheim is neglected in the calculation due to the termination of the optimisation, a mean absolute percentage error of 0.7% results. As these errors are deemed acceptable for this application, the heuristic can be used for estimating DH costs and for planning the DHN including location planning of the plant.

Nevertheless, the use of the heuristic would not be reasonable, unless the application contains a significant reduction of the solving time. As Fig. 8 shows, the solving time of the heuristic is between 2 and 35 s, depending on the case. Leaving aside the 100% case in Billerbeck, the optimisations take between 5 and 100,000 times longer compared to the heuristic. The more settlements that can be connected in a municipality, the longer the optimisation takes (cf. 80% case and 90% case in Bensheim). Whilst the time increases linearly in the heuristic, an exponential increase can be observed in the optimisation. Billerbeck has the fewest settlement areas and yet in some cases, the calculations/ optimisations take longer than in the other municipalities. This is due to the possible locations of the GTP, of which there are about 200 more in Billerbeck compared to the other two municipalities.

Interpreting the fluctuations of the solution time when optimising the different cases is not possible since CPLEX is provided as a black



Fig. 5. Deviation of the total investment of the optimisation from the heuristic for the municipalities Billerbeck, Dümmer and Bensheim.

box. However, it is certain that the solution space will become smaller or larger, depending on the specified amount of heat. With a duration of up to 7 days, the optimisation would be quite time-consuming. For municipalities with more than seven settlements, optimisation should be avoided at all due to the computing constraints as stated at the beginning of this section.

In order to show that the heuristic also runs in an acceptable time for larger municipalities, the heuristic was executed for the municipality Gardelegen. This municipality has the eleventh largest number of settlements of all municipalities in Germany with 42 settlements. In total, the calculation of all 10 cases from 10% to 100% took 4h. The heuristic was not executed for larger municipalities than Gardelegen since in this case, the limitation to only one heating plant would not be appropriate due to district heating pipelines with a length of more than 100 km. 4.2. Resulting costs for district heating networks

Since the employment of the heuristic was evaluated as appropriate in the previous section, it is applied to the four municipalities from Table 5 in this section. Thereby the assumptions from section 3.1 are used. The comparison of the results reveals the following main interdependencies:

- The higher the household density in the settlements, the lower the investment.
- The more settlements in a municipality and the further apart they are, the higher the costs.
- The smaller the largest settlement, the faster the costs increase with the proportion of heat supplied.

As these findings mainly confirm expectations, no detailed



Fig. 6. DHN in the 30% case in Billerbeck with the heuristic (left part) and with the optimisation (right part). The settlements are numbered from 1 to 3.



Fig. 7. DHN in the 50% case in Dümmer with the heuristic (left part) and with the optimisation (right part). The settlements are numbered from 1 to 5.

explanation is given in this section, but instead, only a few aspects are discussed. The dependency of costs on household density is clarified in Fig. 9. The figure shows the investment in relation to the connected heat capacity. The specific connection costs are between 500 ℓ /kW and 1,900 ℓ /kW when using the assumptions from section 3.1. Bensheim and Billerbeck with the highest household densities lead to the lowest costs per kW. The specific costs in Bensheim exceed those in Billerbeck in the 100% case in which all settlements are supplied for the first time and thus a long pipeline is built. In Dümmer and Groß Kreutz, the long distances between the settlements play an important role in addition to the low household density, which leads to high specific costs. If the costs increase more rapidly, as in the 50% case in Dümmer, this indicates that more or different settlements are connected to the heating network than in the previous case.

The calculated costs in Fig. 9 are close to the average costs for the DHN of the geothermal projects Grünwald, Unterföhring, Unterhaching and Neustadt-Glewe of 615 ¢/kW and can therefore be interpreted as plausible [55,56,49,57,58]. Since the percentage of the heat demand satisfied by the geothermal projects in the municipalities is not known, the 615 ¢/kW is illustrated as a horizontal line. The costs for the

municipalities Dümmer and Groß Kreutz are above this average, as these municipalities have a low building and population density. The costs in the municipalities of Billerbeck and Bensheim with higher building and population density are lower, as the maximum length of the DHN outside the settlements in these municipalities is 3 km and 15 km respectively and therefore below the average network length of 38 km in the above mentioned geothermal projects.

The CO₂ abatement cost curves are similar to those in Fig. 9 and lie between 0.7 €/t_{CO2} and 2.4 €/t_{CO2} in the 100% case in Billerbeck and Dümmer respectively. The similar curves result from the fact that the energy mix in the examined municipalities is almost the same, and therefore the amount of CO₂ abatement per kW of geothermal district heating is nearly the same. Therefore, the following sensitivity analysis is performed only with regard to the total costs, since CO₂ abatement would remain constant.

5. Sensitivity analysis and critical discussion



A sensitivity analysis is conducted on the basis of the municipality of Groß Kreutz, which was already used as an example in section 3.

Fig. 8. Solving time of the optimisation (dashed lines) and the heuristic (continuous lines) for the municipalities Billerbeck, Dümmer and Bensheim.



Fig. 9. Specific investment for DH pipelines in the four municipalities depending on the proportion of heat supply compared to the mean value from German geothermal DH projects. The investments are related to the connected nominal capacity of the heating network.



Fig. 10. Settlement centroids in the municipality of Groß Kreutz as well as heat demand, heat density and heat demand proportion of the settlements.

Groß Kreutz is shown again in Fig. 10 in reduced form (left side), together with important information in the table on the right side. The black squares represent the centroids of the settlements and above the squares, the numbering of the settlements is shown. The red circles indicate the size of the heat demand in the settlements.

The example presented in section 3 with 497 possible locations for the DH plant as well as the parameters and locations of the settlements shown in Fig. 10 represents the reference case for this section. The costs for the reference scenario are determined using the heuristic developed in this study. Table 6 shows which conditions have been changed as part of the sensitivity analyses in this section. In section 5.1, scenario 1 shows the influence of a modification of the heat density on the solution. The variation regarding the number of possible locations for the DH plant in scenarios 2 to 7 is the subject of section 5.2. Subsequently, section 5.3 explores the impact of the stages of the heuristic using scenarios 8 and 9. Afterwards, the heuristic developed for this study is compared with the Nearest-Neighbour-Heuristic in this subsection. Finally, the methodology developed in this study is subjected to a critical discussion in section 5.4.

Fig. 11 shows the deviation of the total costs in scenarios 1-10 compared to the reference case for 10% heat supply steps. For reasons of clarity, continuous deviations of less than 1.5% have been removed

from Fig. 11. Therefore, the curves for scenarios 6 and 9 are not shown at all. In the following sections, the figure is explained in more detail.

5.1. Heat demand and heat density

On the left side of Fig. 12, the reference case with 70% heat coverage is shown. The five settlements 5, 6, 9, 11 and 12 with the highest heat demand are connected to the DH plant in this case. To examine the sensitivity in scenario 1, the heat density is now reduced from 5.1 to 0.5 GWh/km² in settlement 5 with the highest heat demand. Settlement 5 must be partially supplied in order to cover 70% of the heat demand in the municipality, as it accounts for more than 30% of the total heat demand (cf. table in Fig. 10). All other settlements, except settlement 4, are now connected to the DHN to ensure that as few households as possible are connected in settlement 5 (cf. right part of Fig. 12). Settlement 4 also has a very low heat density and heat demand, therefore the pipeline to this settlement is not worthwhile.

In the 60% case, settlement 5 is not connected to the DHN at all. In comparison to the reference scenario, the costs increase strongly in scenario 1 if settlement 5 is included in the solution. Thus the total costs increase in the 70% case only by 17%, while in the 100% case the costs increase by over 80% since settlement 5 is completely supplied with

Scenario	Heat density	Number of possible locations for DH plant	Heuristic
Reference	cf. Fig. 10	497	Stages 1, 2 and 3
1	Reduced to 0.5 GWh/km ² for settlement 5, for the other settlements cf. Fig. 10	497	Stages 1, 2 and 3
2	cf. Fig. 10	1	Stages 1, 2 and 3
3	cf. Fig. 10	9	Stages 1, 2 and 3
4	cf. Fig. 10	46	Stages 1, 2 and 3
5	cf. Fig. 10	171	Stages 1, 2 and 3
6	cf. Fig. 10	1299	Stages 1, 2 and 3
7	cf. Fig. 10	2749	Stages 1, 2 and 3
8	cf. Fig. 10	497	Stage 1
9	cf. Fig. 10	497	Stages 1 and 2
10	cf. Fig. 10	497	Nearest-Neighbour-Heuristia

Table 6 Overview of the ten scenarios examined as part of the sensitivity analysis. Changed conditions are shown in ita

heat (cf. Fig. 11).

5.2. Number of locations

As already described in section 3, when determining the possible locations of the DH plant, the number of locations between the minimum and maximum coordinates of the municipality is specified. 1, 5, 10, 20, 55 and 80 locations are defined for the scenarios 2 to 7 respectively. The potential locations for the scenarios are illustrated in Fig. 13, with the resulting number of locations shown in parentheses under the scenario names. In cases with a lower percentage of heat supply, the number of locations has the greatest impact on the total costs, as the costs for the pipelines outside the settlements account for a larger share. Thus the costs in the 10% case in scenario 2 are 32% higher than in the reference case with 497 locations (cf. Fig. 11). With a rising number of potential locations from scenario 2 to scenario 5, however, the costs increasingly match with the costs in the reference scenario. In scenario 6 there are only very small deviations, so that the curve is not shown in Fig. 11. A further increase in the number of locations in scenario 7 also reduces the costs only marginally.

As shown, the costs in the scenarios change due to the varying distance of the DH plant to the settlements. Therefore the costs are calculated again for the reference scenario and scenarios 2 to 7, however, this time with costs for the pipelines outside the settlements of 500 C/m instead of 200 C/m. As a result of this adjustment, deviations also occur in the cases with higher heat supply proportions and the other deviations increase strongly. For example, the deviations in scenario 2 increase from 32% to 68% in the 10% case. The cost of pipelines outside settlements now accounts for up to 35% of total costs in

scenarios 2 to 7 instead of up to 20% in the case with pipeline costs of 200 C/m. Even with 500 C/m, the costs decrease only slightly in scenarios 6 and 7 compared to the reference scenario. With an increasing number of locations, the time for solving the heuristic increases linearly. Thus the selection of the 497 locations for the municipality Groß Kreutz can be evaluated as reasonable. Therefore, a number of 5 locations per km² can be recommended for the study of other municipalities considering the size of Groß Kreutz.

5.3. Number of stages

Section 3 and Fig. 3 have already indicated that an appropriate DHN is not achieved by performing only stage 1 of the heuristic. Since the settlements are iteratively connected to the DHN, the optimal connections rarely arise. This is reflected in the costs of scenario 8, which are at least 5% and on average 12% above the costs of the reference scenario for each case (cf. Fig. 11). Considering the reference scenario 8 with pipeline costs of 500 €/m instead of 200 €/m, the costs in scenario 8 increase on average by 30%. The omission of step 3 in scenario 9 results in a slight cost increase of no more than 1.3%, so that the curve is not illustrated in Fig. 11. However, the third stage leads to slight improvements and should be executed as it has negligible influence on the solution time.

In scenario 10, a standard algorithm is used to demonstrate the performance of the heuristic developed in this study. Thereby, the heuristic of this study is applied with a modified target criterion: the Nearest-Neighbour-Heuristic connects the nearest settlement to the DHN in each iteration. The DHN in the reference case on the left side in Fig. 12 could therefore never result with this heuristic. Instead of



Fig. 11. Deviations in the total costs for scenarios 1-10 in relation to the reference scenario.



Fig. 12. Location of the DH plant and DHN with a heat supply of 70% of the heat demand in Groß Kreutz in the reference case (left part) and for scenario 1 (right part).

settlement 12, which is the furthest away from all other settlements, the Nearest-Neighbour-Heuristic connects settlements 3 and 10 in the 70% case (cf. Fig. 14). In the next step and if more heat were available, settlement 8 would be connected by the grey dotted line, which is 1.76 km away from settlement 6. The Nearest-Neighbour-Heuristic developed in this study would be executed with the target criterion of stage 2.

As shown in Fig. 11, the Nearest-Neighbour-Heuristic results in significantly higher costs than the reference scenario, with deviations reaching values of up to 78%. Obviously, the more heat demand covered in the municipality, the lower the deviations will be until the costs match in the 100% case. The calculation of the nine cases up to the 90% case requires between 50% and 250% more time compared to the reference scenario, as more settlements are connected. The results of this section indicate again that the application of the three-stage heuristic is reasonable.



5.4. Critical appraisal

The sensitivity analyses in the previous section revealed that the heuristic developed in this study yields better results than simpler heuristics, especially in the case of low heat coverage in the

Fig. 14. Location of the DH plant and DHN in the 70% case resulting from the Nearest-Neighbour-Heuristic.



Fig. 13. Possible locations for the DH plant in scenarios 2–7. The resulting number of locations is shown in parentheses under the scenario names.
municipalities. In German municipalities that already have DH systems, the average share of DH in all heating technologies is only 3.5% [30]. Therefore, the municipalities with low heat coverages are of particular importance. This share is lower than the share of district heating in the total German heat supply of 13.8% (cf. section 1), as all municipalities are equally weighted in the calculation. For high heat coverages, e.g. the 100% case, simpler heuristics such as the Nearest-Neighbour-Heuristic are also suitable. However, since this takes more time than the developed heuristic, it is not recommended.

The heuristic can be used in this form or slightly modified for DH plants different to GTPs. This would only require a change in the possible locations of the DH plant. For other plant types, the location could even be within a settlement.

In addition, with the help of the presented approach, DHNs can be designed in holistic energy system analyses. This would be a novelty, as demonstrated in section 2.1, and will be presented by the authors in a forthcoming study. As shown in section 4, the DHNs and the associated costs could be determined for different levels of heat coverage. The results could then be used in the energy system analysis as discrete options, which would enable the optimisation of the DHN. In this study, the developed heuristic was applied to individual municipalities. However, the applicability of the heuristic is not limited to this administrative level. The heuristic can also be used for other regions with little effort. Fig. 15, for example, shows the simultaneous application to four municipalities.

This is an important aspect, since energy system analyses should not be limited to individual municipalities, but should determine the optimal aggregation level for the energy system. In small municipalities, the construction of a GTP would probably not be worthwhile as the fixed costs of these plants are very high. However, if several municipalities are involved in the analysis, the construction of such a plant could become economically viable.

In addition, the heuristic can be applied to other countries. In this case, the heat demand, as well as the population and building density for each settlement, have to be determined as in Weinand et al. [53]. The Corine land use areas are available for 38 other European countries besides Germany and can be used as a basis [59].

For the study presented here, improvements can be made in a couple of areas. First of all, the optimisations in this study do not always yield the realistic optimum. This is due to the fact that, in reality, the pipelines cannot always run straight ahead and no local topographical conditions are taken into account. It is further neglected that a pipeline could be divided into two or more pipelines to connect several settlements with a pipeline starting from one settlement. In order to overcome this problem, many points could be introduced as possible

branches in further work, such as the points of the GTPs in Fig. 3, at which pipelines can divide. However, the investment would probably not be significantly reduced. In addition, since the centroids of the settlements are used as connection points and not the border of a settlement, the costs for pipelines outside the settlements are slightly overestimated.

Furthermore, the feasibility of GTPs depends strongly on the local geological conditions. This means that the GTP location determined by the heuristic may not be technically feasible at all. In this case, the location could be excluded and the heuristic could be executed again. On the other hand, a location could also be specified beforehand and the heuristic could be used to only determine the connections of the pipelines. In addition, some locations for the GTP could be manually excluded in advance, as it is clear that these are not optimal due to a high distance to the settlements (cf. Fig. 1). However, due to the short time required by the heuristic, this is not necessary.

The use of a constant heat loss via an efficiency and thus neglecting heat losses per kilometre also means that the optimal location of the GTP and the optimal DHN are not necessarily determined. If the specific heat losses were taken into account, the GTP in Fig. 3 would probably be closer to the largest settlement on the left-hand side of the figure. This is due to the fact that a large amount of heat could then be supplied without a large heat loss occurring before. However, a different location of the GTP than in the optimal case would not increase the costs significantly. Nevertheless, the heuristic should be improved in further work with regard to heat losses. Related to this, pressure losses should also be taken into account in future work. These make further DHN pumps necessary at certain points in the DHN. In the case of an optimisation model, non-linear equations would have to be integrated into the model, to take the properties mentioned in this paragraph into account [29].

It has been described above that the DH share in German municipalities is on average 3.5%, but there are also municipalities with a high DH share of over 90%. In future studies, it is therefore necessary to extend the heuristic in such a way that existing networks can be recognised as well as used to integrate the heat and thus reduce the investment for pipelines. As already explained, the share of existing DH is only deducted from the heat demand in this study. One possibility would be to apply the methodology of Blanco et al. [35] which has been mentioned in section 2.3. Uncertainty about the location of heat consumers could take into account the fact that it is not known where district heating connections already exist and where the greatest heat demand within a settlement exists. However, these approaches would require a significantly higher spatial resolution and the consideration of individual buildings and roads.



Fig. 15. Resulting district heating network, if the heuristic is applied to several small municipalities.

6. Conclusions

Against the background of a trend towards decentralised and community-owned energy systems, this paper develops a method to set up a minimum-cost geothermal-based municipal district heating system. To this end, two approaches based on combinatorial optimisation were presented, in order to support local planners in the design of geothermal district heating systems. The first approach involves a combinatorial optimisation of the district heating network layout, including geothermal plant location and network topology, which is applicable to municipalities with less than eight discrete settlement areas. The second approach is a three-stage heuristic, which serves the same purpose but can be applied to a much larger number of municipalities with many more settlement areas.

One of the innovations of the developed optimisation model and the three-stage heuristic compared to previous work is the fact that not only the district heating network but also the location of the district heating plant is optimised. Furthermore, the nodes/settlements to be connected are not fixed in advance and do not have to be supplied completely with heat. The two approaches presented in this work can be applied to every municipality in Germany and the methodology could be extended to an arbitrary country with equivalent data. The 38 other European countries are particularly suitable for this extension due to the availability of Corine Land Cover data.

A comparison of optimisation and heuristic for three exemplary municipalities demonstrates the efficiency of the developed heuristic. For municipalities with three, five and seven settlements respectively, the optimisation takes between 500% and $1\times 10^7\%$ more time than the heuristic. The resulting deviations in the calculated total investment for the district heating from the results of the optimisation are in all cases below 5%, and in 80% of cases below 0.3%. The efficiency of the heuristic is demonstrated by comparison with the Nearest-Neighbour-Heuristic. The latter is not only less efficient, it substantially overestimates the total costs by up to 80% in all cases with less than 100% heat coverage. In addition, the calculated investments in the investigated municipalities ranged from 500 €/kW to 1,900 €/kW, values which could be validated with investments for existing geothermal district heating networks in Germany.

The developed heuristic consistently yields results within acceptable margins of error of its equivalent combinatorial optimisation problem, is efficient and scales well to other regions or contexts. The developed methodology would benefit from some further improvements, for example some of the technical aspects such as heat and pressure losses within the district heating network could be modelled more precisely in the heuristic. This would be particularly important in order to plan the district heating network within the municipality, which is considered beyond the scope of this contribution. Furthermore, additional geological and topographical conditions in the municipalities should be taken into account in order to better identify the optimal location of the geothermal plant and the type of network. Finally, the heuristic should be extended in such a way that the district heating pipelines can also branch off in order to reach several endpoints from one starting point. All of these aspects remain areas for future work and will be the subject of a forthcoming contribution, alongside a more holistic energy system analysis.

Notwithstanding these shortcomings, the developed method, comprising combinatorial optimisation and heuristic, provides a sound basis for decision support for municipal-scale geothermal district heating systems. The heuristic for cost-optimal placement of the geothermal plant (provided as supplementary material) can be extended and should offer useful insights for local planners and authorities when considering the heat source options at their disposal. In addition to supporting the planning of municipal district heating networks, the heuristic can also be used to design district heating networks in holistic energy system optimisations due to the novel possibility of connecting an arbitrary number of buildings to the district heating network.

Acknowledgement

The authors gratefully acknowledge the financial support of the PhD College "Energy and Resource Efficiency" (ENRES), from the Federal State of Baden-Wuerttemberg, for funding the first author's PhD studentship. The contribution of the second author was supported by the Helmholtz Association under the Joint initiative "Energy Systems Integration" (funding reference: ZT-0002). The authors also thank three anonymous reviewers and the editor, who provided valuable comments on earlier versions of this manuscript. The usual disclaimer applies.

Appendix A. Supplementary material

Supplementary data to this article can be found online at https:// doi.org/10.1016/j.apenergy.2019.113367.

References

- [1] Wirth H. Aktuelle Fakten zur Photovoltaik in Deutschland (Recent Facts about Null in Andrei in Germany.); 2018. Edited by Fraunhofer ISE Available and at < https://www.ise.fraunhofer.de/content/dam/ise/de/documents/ publications/studies/aktuelle-fakten-zur-photovoltaik-in-deutschland.
- > checked on 10/26/2018. Statistisches Bundesamt. Bruttostromerzeugung in Deutschland für 2015 bis 2017 [2] (Gross electricity generation in Germany for 2015 to 2017.); 2017. Available online at < https://www.destatis.de/DE/ZahlenFakten/Wirtschaftsbereiche/Energie/ ung/Tabellen/Brutt erzeugung.htm > checked on 5/9/2018
- Brzeugung/Tabellen/Bruttostromerzeugung.htm > cnecked on 5/9/2018.
 UBA. Erneuerbare Energien in Deutschland. Daten zur Entwicklung im Jahr 2017 (Renewable energies in Germany. Data on the development in 2017.); 2018. Edited by Umweltbundesamt. Available online at < https://www.umweltbundesamt.de/ es/default/files/medien/376/publikationen/180315_uba_hg_eeinzahlen_2018_bf. df > checked on 6/28/2018.
- pdf > checked on 6/28/2018.
 BMWL Emeuerbare Energien in Zahlen. Nationale und internationale Entwicklung im Jahr 2016 (Renewable energies in figures. National and international develop-ment in 2016.); 2017. Available online at < https://www.bmwl.de/Redaktion/DE/ Publikatione/Energie/enruerbare-energien-in-zahlen.2016.pdf?_blob = publicationFile&w = 8 > checked on 5/9/2018.
- [5] Agemar T, Weber J, Moeck I. Assessment and public reporting of geothermal resources in Germany. Review and Outlook. In Energies 2018;11(2):332. http: prg/10.3390/en11020332.
- Org/10.3590/CH1102032.
 Weber J, Ganz B, Sanner B, Moeck I. Geothermal Energy Use, Country Update for Germany. Strasbourg, France (European Geothermal Congress 2016); 2016.
 Available online at < https://www.geothermal-energy.org/pdf/IGAstandard/EGC/ 2016/GERMANY%20country%20update%20EGC%202016.pdf > checked on 6/ 28/2018.
- 20/2010. AGFW; AGEB; BMWI. Verwendung von Fernwärme in Deutschland (Use of district heating in Germany); 2018. Available online at < https://www.bmwi.de/ Redaktion/DE/Infografiken/Energie/Energiedaten/Energietraeger/energiedaten-energietraeger32.html > checked on 9/27/2018.
- [8] Euroheat & Power. District Energy in Germany; 2017. Available online at < https:// www.euroheat.org/knowledge-centre/district-energy-germany/ > checked on 9/ ww.euroheat.org/knowledge-ce 27/2018
- [9] AGFW. Hauptbericht 2016 (Annual Report 2016); 2017. Available online at d/Zahlen_und_Statistik 2016.pdf > checked on 9/27/2018.
- [10] Hechler H, Bredel-Schürmann S. Tiefe Geothermie (Deep geothermal energy). [10] HECHEF H, Bread-Schurmann S. Here Geonemic Usep geotermal energy).
 Edited by ASUE Ar-beitsgemeinschaft für sparsamen und unweltfreundlichen Energieverbrauch e.V. Berlin; 2011. Available online at < http://asue.de/sites/ default/files/asue/themen/unwelt/dimaschutz/2011/broschurern/07.05_11_ asue-tiefe-geothermic.pdf > checked on 10/2/2018.
 [11] Udomsri Seksan, Bales Chris, Martin Andrew R, Martin Viktoria. Decentralized
- Udomst Sessan, Jakes Chris, Martin Andrew K, Martin Viktoria. Determinized cooling in district heating network. System simulation and parametric study. Appl Energy 2012;92:175–84. https://doi.org/10.1016/j.apenergy.2011.10.009. Fang Tingting, Lahdelma Risto. Genetic optimization of multi-plant heat production in district heating networks. Appl Energy 2015;159:610–9. https://doi.org/10. 1016/j.apenergy.2015.09.027. Alessandro Phil Prato, Strobino Fabrizio, Broccardo Marco, Parodi Giusino Luigi. Interconted measurement of commention leafer and district heating networks. [12]
- [13]
- Aussainto Fin Frao, Subolico foi cogeneration plants and district heating networks. Appl Energy 2012;97:590–600. https://doi.org/10.1016/j.apenergy.2012.02.038.
 Chinese Damiana, Meneghetti Antonella. Optimisation models for decision support in the development of biomass-based industrial district-heating networks in Italy. Appl Energy 2005;82(3):228–54. https://doi.org/10.1016/j.apenergy.2004.10.
- [15] Ove Mørck, Engelund Thomsen Kirsten, Jørgen Rose. The EU CONCERTO project Class 1 – demonstrating cost-effective low-energy buildings – recent results with special focus on comparison of calculated and measured energy performance of Danish buildings. Appl Energy 2012;97:319–26. https://doi.org/10.1016/j. agenergy.2012.01.057.
- [16] Alberg Østergaard Poul, Vad Mathiesen Brian, Bernd Möller, Henrik Lund. A re newable energy scenario for Aalborg Municipality based on low-temperatu

geothermal heat, wind power and biomass. Energy 2010;35(12):4892-901. https:// .2010.08.041. .org/10.1016/i.ene

- [17] Alberg Poul Østergaard, Lund Henrik. A renewable energy system in Frederikshavn using low-temperature geothermal energy for district heating. Appl Energy 2011;88(2):479–87. https://doi.org/10.1016/j.apenergy.2010.03.018.
 [18] Sveinbjörnsson Dadi, Sara Ben Amer-Allam, Hansen Anders Bavnhøj, Algren Loui,
- [16] Svenufjorisson Pau, Sai zel Panel Ander Anani, Harsen Anders Bavinog, Aggen Dud, Pedersen Allan Schröder. Energy supply modelling of a low-CO 2 emitting energy system. Case study of a Danish municipality. Appl Energy 2017;195:922–41. https://doi.org/10.1016/j.apenergy.2017.03.086.
 [19] Möller Bernd, Lund Henrik. Conversion of individual natural gas to district heating: geographical studies of supply costs and consequences for the Danish energy system. Appl Energy 2010;87(6):1846–57. https://doi.org/10.1016/j.apenergy. 2000.12.011 009.12.001.
- 2009.12.001.
 [20] Casisi M, Pinamonti P, Reini M. Optimal lay-out and operation of combined heat & power (CHP) distributed generation systems. Energy 2009;34(12):2175–83. https://doi.org/10.1016/j.energy.2008.10.019.
 [21] Chinese D. Optimal size and layout planning for district heating and cooling networks with distributed generation options. Int J of Energy Sector Man 2008;2(3):385–419. https://doi.org/10.1108/17506220810892946.
 [20] Interest P. Benergt C. Childrahden Schuhm C. Durald E. Autur T. Depining and Colling and Coling and Colling and Colling and Co
- [22] Bratoev I, Bonnet C, Chokhachian A, Schubert G, Petzold F, Auer T. Designing and Bratoev I, Bonnet C, Cnoknachian A, Schubert G, Petzold P, Auer T. Designing and evaluating district heating networks with simulation based urban planning. In: URBANCEQ-2017. International conference on urban comfort and environmental quality. Genova, Italy, 28-29 September 2017. Dobersek D, Goricance D. Optimisation of tree path pipe network with nonlinear optimisation method. Appl Therm Eng 2009;29(8–9):1584–91. https://doi.org/10. 0016/dom/biomeduce.pdoi/00.00101
- [23] 1016/j.applthermaleng.2008.07.017
- [24] Marty Fabien, Serra Sylvain, Sochard Sabine, Reneaume Jean-Michel. Simultaneous optimization of the District Heating Network topology and the Organic Rankine Cycle sizing of a geothermal plant. Energy 2018. https://doi.org/10.1016/j.energy
- [25] Falke Tobias, Krengel Stefan, Meinerzhagen Ann-Kathrin, Schnettler Armin. Multiobjective optimization and simulation model for the design of distributed energy systems. Appl Energy 2016;184:1508–16. https://doi.org/10.1016/j.apenergy.

- 2016.03.044.
 [26] Kruskal Joseph B. On the shortest spanning subtree of a graph and the traveling salesman problem. Proc Am Math Soc 1956;7(1):48. https://doi.org/10.1090/S0002-9939-1956-0078686-7.
 [27] Nielsen Steffen. A geographic method for high resolution spatial heat planning. Energy 2014;67:351-62. https://doi.org/10.1016/j.energy.2013.12.011.
 [28] Delangie Axelle, Lambert Romain SC, Shah Nilay, Acha Salvador, Markides Christos N. Modelling and optimising the marginal expansion of an existing district heating network. Energy 2017;140:209-23. https://doi.org/10.1016/j.energy.2017.08.066.
- [29] Bordin Chiara, Gordini Angelo, Vigo Daniele. An optimization approach for district heating strategic network design. Eur J Oper Res 2016;252(1):296–307. https:// doi.org/10.1016/j.ejor.2015.12.049.
- [30] Statistisches Bundesamt. Gebäude und Wohnungen (Buildings and apartments). Edited by Statistische Ämter des Bundes und der Länder. Wiesbaden; 2014. Available online at < https://www.regionalstatistik.de/genesis/online/ data;jsessionid=62C84E17497FA378AD49A9004D9A6DF6.reg2?operation statistikenVerzeichnisNextStep&levelindex = 0&levelid = 1504786087209&index = ructurelevel = 3 > checked on 9/7/2017.
- 3&structurelevel = 3 > checked on 9/7/2017.
 [31] Faziollahi Samira, Girardin Luc, Marchal François. Clustering urban areas for optimizing the design and the operation of district energy systems. In: 24th European symposium on computer aided process engineering, vol. 33: Elsevier (Computer Aided Chemical Engineering); 2014, p. 1291-6.
 [32] Unternährer Jerémy, Moret Stefano, Joost Stéphane, Marchal Prançois, Spatial
- Charling & Jordany, Mort Sociation, Jose Suphian, Jaccular François, Spaca clustering for district heating integration in urban energy systems. Application to geothermal energy. Appl Energy 2017;190:749–63. https://doi.org/10.1016/j. appenergy.2016.12.136.
 Marquant Julien F, Ralph Evins, Andrew Bollinger L, Jan Carmeliet. A holarchic Application of the systems of the system of the systems of t
- approach for multi-scale distributed energy system optimisation. Appl Energy 2017;208:935-53. https://doi.org/10.1016/j.apenergy.2017.09.057. Smith David K, Walters Godfrey A. An evolutionary approach for finding optimal
- [34] trees in undirected networks. Eur J Oper Res 2000;120(3):593-602. ht
- Blanco Víctor, Fernández Elena, Puerto Justo. Minimum Spanning Trees with neighborhoods: mathematical programming formulations and solution methods. Eur J Oper Res 2017;262(3):863-78. https://doi.org/10.1016/j.ejor.2017.04.02
- [36] Fernández Elena, Pozo Miguel A, Puerto Justo, Scozzari Andrea, Ordered weighted
- remainez lefta, rozo ingleta A, retero Jaso, dozani Almara, Outereu wegneti average optimization in multiobjective spanning tree problem. Eur J Oper Res 2017;260(3):886–903. https://doi.org/10.1016/j.ejor.2016.10016. Gao Xin, Jia Lifen. Degree constrained multinium spanning tree problem with un-certain edge weights. Appl Soft Comput 2017;56:580–8. https://doi.org/10.1016/j. oc.2016.07.054
- [38] Salgueiro Rui, de Almeida Ana, Oliveira Orlando. New genetic algorithm approach
- Sugter Viri, et al. (1997) and a set of the original of the generating approximation of the mini-degree constrained minimum spanning tree. Eur J Oper Res 2017;258(3):877–86. https://doi.org/10.1016/j.ejor.2016.11.007. Gouvela L, Moura P, Ruthmair M, Sousa A. Spanning trees with variable degree bounds. Eur J Oper Res 2014;239(3):830–41. https://doi.org/10.1016/j.ejor.2014.
- [40] Gouveia Luis, Simonetti Luidi. Spanning trees with a constraint on the number of leaves. A new formulation. Comput Oper Res 2017;81:257-68. http

Applied Energy 251 (2019) 113367

1016/j.cor.2016.12.005

- [41] Cerrone C, Cerulli R, Raiconi A. Relations, models and a memetic approach for three
- degree-dependent spanning tree problems. Eur J Oper Res 2014;232(3):442-53. https://doi.org/10.1016/j.ejor.2013.07.029.
 Marín Alfredo. Exact and heuristic solutions for the minimum number of branch vertices spanning tree problem. Eur J Oper Res 2015;245(3):680–9. https://doi. org/10.1016/j.ejor.2015.04.011. [43] EEA. Corine Land Cover 2012 seamless vector data. Edited by European
- Environment Agency; 2016. Available online at < https://www.eea.europa. downloads/5c537b70226944b8bfe7bdc9470100ff/1488296244/clc-2012-ve checked on 11/28/2017
- [44] Lenk M, Boumann J, Sippel T. CORINE Land Cover 10 ha; 2017. Available online at < http://www.geodatenzentrum.de/geodaten/gdz_rahmen.gdz_div?gdz_sp anz zeile = 1&gdz unt zeile = 22&gdz user id = deu&odz akt zeile = 58 odz > checked on 11/28/2017.
- [45] HLUG Leitfaden Tiefengothermile. Planung und Durchführung von Projekten (Guide to deep geothermal energy. Planning and implementation of projects.). Edited by Regierungspräsidium Darmstadt, Hessisches Landesant für Unwelt und Geologie; 2011. Available online at < https://www.energieland.hessen.de/pdf/</p> en Tiefengeothermie - Bergrechtlicher Teil.pdf > checked on 3/15/2018.
- [46] C.A.R.M.E.N. e.V. Nahwärmenetze und Bioenergieanlagen. Ein Beitrag zur effi [46] C.A.R.M.E.N. e.V. Nahwärmenetze und Bioenergicanlagen. Ein Beitrag zur effi-zienten Wärmenutzung und zum Klimaschutz (Local heating networks and bioe-nergy plants. A contribution to efficient heat utilisation and elimate protection.); 2012. Available online at < https://www.carmen.ev.de/files/festbrennstoffe/ merkbalkt.Nahwaementz_carmen_ev.pdf > checked on 2/16/2018.
 [47] Pfnür A, Winiewska B, Mallch B, Oschatz B. Dezentrale vs. zentrale
- Wärmeversorgung im deutschen Wärmemarkt. Vergleichende Studie aus en-ergetischer und ökonomischer Sicht (Decentralised vs. central heat supply in the German heat market. Comparative study from an energy and economic point of view.); 2016. Darmstadt. Available online at < http://www.bdh-koeln.de/ udie dezentrale vs zentrale ser_upload/pressemitteilungen_pdf/stu orgung.pdf > checked on 2/16/2018.
- WaermeVersorgung, put > cnecked on 2/16/2018.
 [48] BMJV. Gestez für die Erhaltung, die Modernisierung und den Ausbau der Kraft-Wärme-Kopplung (Kraft-Wärme-Kopplungsgesetz KWKG) (Jaw for the maintenance, modernisaterina and expansion of cogeneration). Edited by Bundesministerium der Justiz und für Verbraucherschutz; 2018. Available online at < https://www.destatis.de/DE/Methoden/Rechtsgrundlagen/Statistikbereiche/Inhalter/1058 KWKG/af2⁺. Dio by PublicationFile Checked on 3/2/2018.
 [49] Erdwärme Grünwald GmbH. Erdwärme Grünwald: "Hinter dem Mehr an Kosten atomotica in Statistica Behren an Kosten eine Mehr an Kosten.
- [49] Erdwärme Grinnwald GmbH. Erdwärme Grinnwald, Hinter dem Mehr an Kosten steckt ein Mehr an Leistung. "(Geothermal energy Grinwald." Behind the increase in costs is an increase in performance".); 2013. Available online at < http://www.erdwaeme-gruenwald.de/_URL/E1086.htm >, checked on 3/20/2018.
 [50] Persson Urban, Werner Sven. Heat distribution and the future competitiveness of district heating. Appl Energy 2011;88(3):568–76. https://doi.org/10.1016/j.
- penergy.2010.09
- [51] Connolly D, Lund H, Mathiesen BV, Werner S, Möller B, Persson U, Heat Roadmap Europe. Combining district heating with heat savings to decarbonise the EU energy system. Energy Policy 2014;65:475–89. https://doi.org/10.1016/j.enpol.2013.10.
- [52] Le Truong Nguyen, Dodoo Ambrose, Gustavsson Leif. Renewable-based heat supply
- Le Irtong Nguyen, Dodoo Ambrose, Gustavsson Leir, Kenewanie-based neat supply of multi-apartment buildings with varied heat demands. Energy 2015;93:1053–62. 10.1016/j.energy.2015.09.087. Weinand Jann, McKenna Russell, Karner Katharina, Braun Lorenz, Herbes Carsten. Assessing the potential contribution of excess heat from biogas plants towards decarbonising German residential heating. In: Working paper series in production and energy 2018;31.10.5445/IR/1000086374. [53]
- [54] Weinand Jann, McKenna Russell, Fichtner Wolf, Developing a municipality pology for modelling decentralised energy systems. Utilities Policy 2019;57:75–96.
- [55] Enerchange. Sauberes Jubiläum: 15 Jahre Geothermie in Neustadt-Glewe (Clean anniversary: 15 years of geothermal energy in Neustadt-Glewe.); 2009. Available online at < http://www.tiefegeothermie.de/top-themen/sauberes-jubilaeum-15 ahre-geothermie-in-neustadt-glewe > checked on 3/20/2018.
- [56] Enerchange. Projekte Tiefe Geothermie (Deep Geothermal Projects.); 2018.
 Available online at < http://www.tiefegeothermie.de/projekte > checked a 20/2018. > checked on 3/
- Geothermie Unterhaching.); 2017. Available online at < https://www.geothe aching.de/cms/geothermie/web.nsf/gfx/
- 0189A5952BAEB5E8C1258183003FA71A/\$file/Datenblatt%20Geothermie [58]
- 0189A5952BAB5BSC1258183003FA71A/Shie/Datenbiatt%20Gothermie %20Unterhaching%202017.pdf > checked on 3/20.2018. GTN. Geothermisches Heizwerk Neustadt-Glewe (Geothermal heating plant Neustadt-Glewe). Edited by Geothermie Neubrandenburg GmbH; 2012. Available online at < http://www.ginonline.de/media/PDF/GTN Tlefe, Geothermie. Geothermisches Heizwerk Neustadt, Glewe.052012.pdf > checked on 3/20/2018. [59] EEA. Land cover country fact sheets 2012. Edited by European Environment
- [59] EEA Land cover country ract sheets 2012. Edited by European EnVronment Agency; 2018. Available online at < https://www.eea.europa.eu/themes/landuce.land-cover-country-fact-sheets?b_start:int=0 > checked on 10/18/2018.
 [60] OpenStreetMap contributors. OpenStreetMap; 2018. Available online at < https://www.openstreetmap.org/#map =>5/51.330/10.453 > checked on 1/29/2018.
 [61] Agemar Thorsten, Weber Josef, Schulz Rüdiger. Deep geothermal energy production in Germany. Energies 2014;7(7):4397–416. https://doi.org/10.3390/en7/4307
- en7074397.

Applied Energy 255 (2019) 113824



Contents lists available at ScienceDirect

Applied Energy



journal homepage: www.elsevier.com/locate/apenergy

Assessing the contribution of simultaneous heat and power generation from geothermal plants in off-grid municipalities



Jann Michael Weinand^{a,*}, Russell McKenna^b, Max Kleinebrahm^a, Kai Mainzer^a

^a Chair of Energy Economics, Karlsruhe Institute of Technology, Germany
^b Energy Systems Analysis, DTU Management, Technical University of Denmark, Denmark

HIGHLIGHTS

- Analysis of hydrothermal potential in German municipalities.
- · Optimisation of simultaneous geothermal heat and electricity generation.
- Drilling depth and hydrothermal temperature are implemented endogenously.
- Integration of geothermal plants in a holistic energy system optimisation.
 Geothermal plants reveal a potential for cost reduction in off-grid municipalities.

ARTICLE INFO

Keywords: Mixed-integer linear optimisation Geothermal plant Hydrothermal potential Variable drilling depth Energy autonomy

ABSTRACT

A growing number of German municipalities are striving for energy autonomy. Geothermal plants are in-creasingly constructed in municipalities in order to exploit the high hydrothermal potential. This paper analyses the potential contribution of simultaneous geothermal power and heat generation in German municipalities to achieving energy autonomy. A linear regression estimates the achievable hydrothermal temperatures and the required drilling depths. Technical restrictions and cost estimations for geothermal plants are implemented within an existing linear optimisation model for municipal energy systems. Novel modelling approaches, such as optimisation with variable drilling depths, are developed. The new approach is validated with data from existing geothermal plants in Germany, demonstrating a Root Mean Squared Error of about 15%. Eleven scenarios show that achieving energy autonomy is associated with at least 4% additional costs, compared to scenarios without it. The crucial role of geothermal plants in providing base load heat and power to achieve energy autonomy is demonstrated. The importance of simultaneous modelling of electricity and heat generation in geothermal plants is also evident, as district heating plants reduce the costs, especially in municipalities with high hydrothermal potential. Further work should focus on the optimal spatial scale of the system boundaries and the impact of the temporal resolution of the analysis on the costs for achieving energy autonomy.

1. Introduction

The radical change in the energy sector due to ambitious national targets in energy policy is characterised in particular by the expansion of renewable energies. Renewable energies are mainly utilised decentrally due to their characteristics. Therefore, municipalities are often referred to as the driving force behind the energy transition. This particularly applies to Germany, where the decentralised structure also applies to the owners and operators of energy plants: private individuals increasingly invest in renewable energy systems or form socalled citizen-energy cooperatives [1]. The majority of regenerative plants in Germany are actually owned and operated by private individuals, farmers and communities [2]. This development is due to various motivations: among other things, citizens intend to play an active role in energy supply and to be less dependent on central markets and structures (e.g. [3]). For homeowners, energy autonomy is one of the main factors that drives them to install renewable energies [4].

Besides energy autonomy for homeowners, the concept of municipal energy autonomy [5] has become established, which is employed here to also include energy autarky [6], self-sufficiency [7] and integrated community energy systems [8]. Alone the number of terms for this concept illustrates the diversity within the literature, which also

* Corresponding author.

E-mail address: jann.weinand@kit.edu (J.M. Weinand).

https://doi.org/10.1016/j.apenergy.2019.113824

Received 11 February 2019; Received in revised form 28 August 2019; Accepted 31 August 2019 0306-2619/ © 2019 Elsevier Ltd. All rights reserved.

Nomenc	lature		[m]
	a	D _{el}	electricity demand in the municipality [kW]
A_D	area for the well site [m ²]	D _{el total}	sum of electricity demand in the municipality and elec-
A _{el}	area for the ORC plant [m ²]		tricity consumption of the feed pump [kW]
A_S	area for the 3D seismology [m ²]	D _{heat, total}	total heat demand of the municipality in a time step [kW]
Ath	area for the district heating plant [m [*]]	D_p	percentage of electricity consumption of the feed pump in
D _{D,1900}	whether the drilling depth is between 0 m and 1900 m	D	geothermal power plants [%]
,	(binary) [-]	D _{pump}	electricity consumption of the feed pump [kW]
D _{D,3250}	whether the drilling depth is between 1900 m and 3250 m	M	large number [-]
1	(Dinary) [-]	P_{el}	power generation of the geothermal ORC plant in a time
D _{D,3850}	whether the drilling depth is larger than 3250 m (binary)	n	step [KW]
1	[-]	P _{el max}	nominal power of the God summer
D _{D,1}	(hinem) []	P_{pump}	nominal power of the geothermal district heating plant in
11000 m	(Dinary) [-]	Qth	a time step [[W]]
D _{DHP}	whether the district heating plant is built (binary) [-]	Ô.,	nominal nower of the district heating plant [kW]
DDHP,op	(binow) []	≺m, max r_	range of a section for the drilling depth [m]
h	(Dillary) [-]	ΔT	temperature gradient [°C/m]
boat der	and of the municipality (hipary) []	T.	annual mean ambient temperature [°C]
heat den	whether the geothermal plant is built (hipary) [-]	Tour meur	temperature of the geothermal water after the heat
b _{GP}	whether the ORC plant is built (binary) [_]	19111, FOCUS.	transfer to the district heating system [°C]
C ₁	investment for feasibility study and preliminary planning	$T_{DH return}$	min minimum return temperature of the district heating
01	[€]	211,700074	system [°C]
C.	investment for properties and infrastructure [€]	T_{DHmin}	minimum forward temperature of the district heating
C2	investment for exploration of the reservoir $[\mathbf{\xi}]$		system [°C]
C_4	investment for production well pump [€]	$T_{DH,pinch}$	pinch temperature of the heat exchange to the district
C_5	investment for thermal water system (above-ground) [€]	-	heating system [°C]
C_6	investment for power generation plant [€]	$T_{ORG,min}$	minimum temperature of the geothermal water after the
C7	investment for district heating system [€]		heat transfer to the ORC plant [°C]
C_8	investment for project management, control and finance	$T_{ORC,out}$	temperature of the geothermal water after the heat
	planning as well as other investments [€]		transfer to the ORC Plant and before the heat transfer to
C_D	costs for the drilling work and the construction and re-		the district heating system [°C]
	cultivation of the well site $[\in]$	T_{PW}	hydrothermal temperature in the production well in a time
C_{DR}	annual costs for working fluid and other demand related-	T.	step ['C]
_	resources [€/a]	I pW, max	maximum nydrotnermal temperature in the production
C_{DW}	costs for the drilling work [€]	ц,	wen (wennead temperature) [0] mean volumetric flow rate in the production well [1/s]
C_{FS}	costs for the feasibility study $[\in]$	* B % c	drilling denth [m]
C_{GI}	costs for geochemical investigations $[C]$	20D	drilling depth if drilling is up to 1000 m at maximum [m]
CHT	costs for hydraulic tests [6]	≈D,1900 %D,0055	drilling depth, if drilling is up to 3250 m at maximum [m]
CL	annual labour costs [€/a]	×D,3250	drilling depth, if drilling is above 3250 m [m]
COP	annual other operating costs, e.g. costs for operating of	<i>∾D,3850</i> %70.1	drilling depth, if drilling is up to 1000 m at maximum [m]
C	seismic monitoring [€/a]	20,1 20.1	drilling depth, if drilling is up to 2000 m at maximum [m]
COV	annual variable costs for insurance and legal assistance $\left[\frac{\rho}{2}\right]$		drilling depth, if drilling is up to 3000 m at maximum [m]
C-	[∇a]	20,5 Хл л	drilling depth, if drilling is up to 4000 m at maximum [m]
	costs for the preliminary planning of the power generation	Z _{D.5}	drilling depth, if drilling is up to 5000 m at maximum [m]
GPP	plant and the above-ground plant components [4]	Pw	mean density of the geothermal water [kg/l]
Cn w	mean heat capacity of the geothermal water $\begin{bmatrix} k \\ -1 \end{bmatrix}$	η_{el}	efficiency of the ORC plant [-]
Car	costs for stimulation $[\mathcal{F}]$	η_{pump}	efficiency of the feed pump [-]
Cm	costs for the underground planning $[\ell]$	η_{th}	efficiency of the geothermal district heating plant [–]
d _p	distance between the production and the injection well		
u _D	assume serveen the production and the injection wen		

extends to its definition. Three rough distinctions are made between complete energy autonomy (i.e. off-grid), net or balanced energy autonomy, whereby local supply equals or exceeds demand on an annual basis, and a tendency towards higher energy autonomy through decentralised renewables [9]. The extensive survey of Engelken et al. [10] shows that the majority of municipalities with energy autonomy aspirations strive for balanced energy autonomy and the focus is usually on electrical energy. This can also be observed by analysing the energy project "100%-Renewable-Energy-Communities", in which the participating municipalities strive for energy autonomy. The exponential development of those projects since 1995 indicates that an increasing number of German municipalities is striving for energy autonomy.¹ The 1300 municipalities of the project correspond to 12% of all municipalities in Germany and account for 15% of the population.

According to the authors' knowledge, only one entire municipality in Germany strives for complete energy autonomy, namely Bordelum in Schleswig Holstein [11]. Another example for complete energy autonomy is the village Feldheim, which is a part of the municipality

¹Own determination with data from the websites of the municipalities as well as the following websites, among others: http://www.kommunal-erneuerbar. de/startseite.html and http://www.100-ee.de/.

Treuenbrietzen. A study of the impacts of autonomy in this village showed that, after the energy system transition, every inhabitant was in an "economically more favourable situation" than before [12]. However, balanced energy autonomy could make network expansion even more essential and also make new allocation systems for grid fees necessary [1]. This could result in economic inefficiencies compared to the established system of centralised generation, transmission and distribution [13]. The present study therefore considers complete energy autonomy, as this state could be advantageous for the energy system. In other words, no imports are possible and the electricity and heat demand must be covered by local renewable energies.

These renewable energies already account for 33% of electricity generation in Germany in 2017 [14]. In contrast, the proportion of renewable heat supply is much lower at around 13% [15]. The renewable energies include around 55 GW of wind energy (on- and offshore), 42 GW photovoltaic (PV) systems, about 7 GW of bioenergy $\left[16\right]$ and 39 MW of deep geothermal energy $\left[15\right] .$ In comparison to the other renewable energies, relatively little electricity is generated by geothermal plants (GTPs) despite the fact that deep geothermal energy could make a major contribution to reducing greenhouse gas emissions. However, the relative growth from 7 MW_{el} to 39 MW_{el} (+450%) in geothermal power generation in Germany between 2010 and 2017 is one of the highest in the world [17]. The currently (2017) installed 30 GTPs in Germany generate about 155 GWh of electricity [15] and 1.3 TWh of heat [18] annually, which is generally used for district heating (DH) applications [19]. The locations of the 30 GTPs are shown in The electricity generation is still very low compared to the available resources with a technical potential of 4155 TWh per year [20]. This potential is significantly higher than the 654 TWh of gross electricity generated in 2017 in Germany [21].

The use of GTPs could be relevant for municipal energy autonomy efforts due to the following advantages: on the one hand, in contrast to many alternative renewable energy plants, the plants are able to provide heat and electricity as base load. As the Literature Review in Section 2 demonstrates, the costs of energy autonomy are very high due to the volatile generation of electricity by PV and wind and the resulting need for high storage capacities. In addition, resources for the use of biomass or gas networks are not available in every municipality. In this case, the use of the DH generated by the GTPs would be particularly important. Furthermore, GTPs emit the lowest amount of pollutants during the life cycle of the plant after hydropower [22].

The objective of this study is to answer the following research questions, developed from an analysis of existing literature (cf. Section 2):

- Could the high costs for off-grid municipal energy systems be reduced through the use of geothermal plants?
- Is it sufficient to consider only the electricity generation of the geothermal plant or would the use of the geothermal heat in district heating networks create an additional benefit?

To answer these research questions, a generic approach for the modelling of GTPs in municipalities is developed, which is included in a holistic² energy system optimisation. In order to ensure the transferability of the method, a novel modelling approach for the simultaneous generation of electricity and heat as well as the calculation of costs for GTPs is demonstrated. Furthermore, a basis for the assessment of the hydrothermal potential in each German municipality is provided, which also serves as input for the optimisation model. The

Applied Energy 255 (2019) 113824

hydrothermal potential is determined by the achievable temperature of the water in aquifers at depths of up to 5 km. The hydrothermal potential therefore also determines the geothermal potential (i.e. electricity or heat generation potential), since the temperature of the hydrothermal fluid serves as a heat source for the geothermal plants in the present study. The approach is not intended to replace the detailed assessment of the hydrothermal potential on site, but is suitable to demonstrate a possible utilisation of this potential. In addition to the transferability of the approach, the effect of GTPs on the costs and emissions of energy autonomous municipal energy systems is investigated in several municipalities by energy system optimisations. Thereby the focus is on the residential sector. On the one hand, the focus of the energy projects mentioned above is on this sector. On the other hand, this sector can be implemented most precisely due to the comprehensive data availability.

The paper is structured as follows. Section 2 provides an overview of the literature and clarifies the context of this paper. Section 3 then explains the methodology, before Sections 4 and 5 present and discuss the results. The paper concludes with a summary and conclusions in Section 6.

2. Literature review

In this section, the novelty of the study is demonstrated by discussing peer-reviewed literature on energy autonomy in municipalities (Section 2.1) and on the optimisation of GTPs (Section 2.2).

2.1. Municipal energy autonomy

Several case studies have already examined the feasibility of municipal energy autonomy (cf. Table 1). In this context, only studies in which at least balanced autonomy is investigated are referred to. Balanced autonomy of a region is examined in Scheffer [23], Schmidt et al. [24], Peura et al. [25] and Burgess et al. [26]. Scheffer [23] focuses on a rural model region with 10,000 inhabitants and agriculture as well as trade and commerce, but without large-scale industry. High investments are necessary for the balanced energy autonomy and storable biomass is highlighted as the most important energy source for achieving this status. Schmidt et al. [24] investigate the advantages and disadvantages of energy autonomy compared to conventional energy supply in Sauwald, a rural region with 21,000 inhabitants in Austria. The study comes to the conclusion that attaining energy autonomy implies a decline in local production of food and feed as well as high costs for consumers. Peura et al. [25] investigate the potential economic effects of balanced energy autonomy in the municipalities of Perho and Jepua in Finland and take into account the consumption sectors of households, industry, transport and commerce. The study finds that energy autonomy can be technically feasible and economically advantageous in rural areas. Finally, the study by Burgess et al. [26] explores the Marston Vale region in the UK, which in particular would have to import heat energy and fuels, while a large proportion of its electricity needs could be covered by energy from the region itself.

Complete municipal energy autonomy, on the other hand, is investigated in Peter [27], Jenssen et al. [28] and Woyke & Forero [29]. The suitability of a rural settlement structure for complete energy autonomy is analysed in Peter [27], who demonstrates that renewable energies could cover the electricity demand of an "example village" with 3850 inhabitants but with tremendous storage costs. Jenssen et al. [28] conclude that complete energy autonomy is technically achievable in an "average" German municipality through the "Bioenergy Village" approach, albeit at high costs. Woyke & Forero [29] assess complete energy autonomy in Pellworm, an island municipality in Germany with 1100 inhabitants, which has been regarded as a model location for the construction of renewable energy autonomy is not possible with the current energy system in Pellworm due to grid restrictions.

² We mean by "holistic" that not only the geothermal plant itself is optimised. Instead, the geothermal plant is only a small part of the overall optimisation, in which the costs for energy supply in municipalities are minimised and other technologies and measures can be applied alongside / instead of the geothermal plant.

Table 1

Municipal energy autonomy studies and the considered energy supply technologies (BE = Bioenergy, WE = Wind energy, PV = Photovoltaics, ST = Solar thermal energy, GE = Geothermal energy, AHP = Air-source heat pump, GHP = Ground-source heat pump).

Study	Conside	red energy sup	oply technolo	gies		Type of energy autonomy	Context
	BE	WE	PV	ST	GE		
Burgess et al. [26]	+	+	1	1	GHP	Balanced	Region in UK
Jenssen et al. [28]	1	×	×	×	×	Complete	German "Bioenergy village"
Peter [27]	×	*	1	1	GHP	Balanced and complete	German village
Peura et al. [25]	1	1	×	×	×	Balanced	Finnish municipalities
Scheffer [23]	1	1	1	1	×	Balanced	Rural German region
Schmidt et al. [24]	1	×	1	1	GHP	Balanced	Rural Austrian region
Woyke & Forero [29]	1	1	1	×	×	Balanced and complete	German Island
This study	1	1	1	×	AHP/GHP and GTPs	Complete	German municipalities

Overall, it appears that energy autonomy can only be achieved at high cost. In the case of complete energy autonomy, this is related to high storage costs. The present study considers all energy supply technologies applied in the presented literature except for solar thermal (cf. Table 1). So far, none of the studies on complete municipal energy autonomy includes a possible contribution from GTPs, which could reduce the high storage costs through base load capacity.

2.2. Geothermal plant optimisation

In most studies on GTPs, process parameters are optimised, such as power output or various efficiencies of the plant (cf. 43 summarised studies in Table A1³ in the Appendix A). Usually only electricity generation is considered. Van Erdeweghe et al. [30] and Marty et al. [31] alone consider a combined power and heat production, whereby in Van Erdeweghe et al. [30] the focus is on the optimisation of the power generation as well. Furthermore, in most cases low brine temperatures are investigated, which are also present in Germany. Only in very few cases the investigated energy system includes not only the GTP but a more extensive system with other energy supply technologies. These cases of interest are examined in more detail in the following.

Østergaard et al. [32] investigate the feasibility of supplying the municipality of Aalborg in Denmark with renewable energy through a combination of deep geothermal heat, wind power plants and biomass. GTPs are used for heat generation only. The results of the simulations show that these technologies cannot cover the demand of Aalborg in every hour and that energy has to be imported instead. The study of Østergaard & Lund [33] on Frederikshavn in Denmark, in which the city's energy demand should be 100% renewable, also takes deep geothermal energy into account as a heat generation plant with a district heating network (DHN). The simulation includes wind turbines to cover the electricity demand, therefore a balanced electricity autonomy is considered. The study shows that the operation of the GTP and the DHN reduces the energy imports of Frederikshavn. In the study of Sveinbjörnsson et al. [34], the municipality of Sønderberg in Denmark is considered, which is aiming for zero net CO₂ emissions by 2029. Exactly as in Østergaard & Lund [33], the GTP is combined with an absorption heat pump. The mixed-integer linear optimisation shows that by supplementing combustion with modern energy conversion technologies, the climate targets can be achieved in a cost- and energyefficient manner. The most relevant paper for the analysis carried out in the present study is Marty et al. [31]. They highlight the relevance of a simultaneous optimisation of the DHN and the organic rankine cycle (ORC) of a GTP. However, the model in their study is not automatically applicable to any location, as hydrothermal temperatures as well as distances to heat consumers have to be specified manually and geographical conditions are not considered. Furthermore, only parts of the GTP are optimised and not the entire plant. For example, fixed drilling costs are specified in the optimisation and it is not possible to optimise the depth of the boreholes. Beyond that, the greatest need for improvement lies in the fact that the GTP is optimised without being in an energy system context with competition from other technologies such as PV, wind turbines and biomass plants.

Table 2 summarises the main characteristics of the discussed studies. In the 39 other studies on GTP energy systems from Table A1, only those properties are given for which a clear statement was possible. More detailed information can be found in Table A1. As a summary, the evaluation of the studies demonstrates that GTPs are usually optimised with regard to power generation. This is justified as most studies are technical analyses to optimise parameters of the ORC plant. On the other hand, the ORC system is neglected and only the DH supply is considered when optimising municipal energy systems. Furthermore, only specific municipalities were examined and no attempt was made to generate transferable methods. The following gaps were identified in the review of the literature, which the authors fill with the present study:

- (1) Development of a generic optimisation model for the optimisation of a GTP, which simultaneously generates electricity and heat. The model also optimises the drilling depth and the related hydrothermal temperature (cf. Section 3).
- (2) Integration of the GTP model in a holistic and transferable⁴ energy system optimisation model applicable to every German municipality (cf. Section 3.5).
- (3) Investigation of the economic feasibility of complete municipal energy autonomy including GTPs (cf. Section 4).

3. Methodology

Energy system analyses in municipalities usually require a lot of data. In addition, the possible measures have to be adapted to the individual municipality. The methodology in this study aims to be transferable and applicable to each municipality without further effort⁴. This means that no data needs to be collected. Instead, only the name of the municipality needs to be provided and all required input data are determined automatically. This approach enables the investigation of several municipalities and the identification of the impacts of different prerequisites in the municipalities on the optimal energy systems. First of all, this methodological section gives an overview of the general approach (Section 3.1). In Section 3.2 the achievable hydrothermal temperatures at certain locations in Germany as well as the varying depths at which these temperatures can be found, depending on the location, are demonstrated. Afterwards, the most important model

 $^{^{\}rm 3}$ Tables, figures and sections in the Appendix are marked with an A before the numbering.

⁴ This refers to the developed methods. Some assumptions, especially about certain parameter values, are not necessarily transferable to every municipality (cf. Section 5).

Table 2

Applied Energy 255 (2019) 113824

Main characteristics of the	geothermal plant op	timisation studie				
Study	Methodology	Focus on DH and/or power	Variable temperature and drilling depth	Optimisation of the district heating network	Easily transfer- able? ⁴	Investigated energy system
39 other studies (Section A.1)	Mainly optimisation	Power	×	×	-	GTP
Østergaard et al. [32]	Simulation	DH	×	×	×	Municipality
Østergaard & Lund [33]	Simulation	DH	×	×	×	Municipality
Sveinbjörnsson et al. [34]	Optimisation	DH	×	×	×	Municipality
Marty et al. [31]	Optimisation	Both	×	1	×	GTP and few consumers
This study	Optimisation	Both	1	1	*	Municipality

equations for representing a GTP are presented in Section 3.3 before the economic assessment of a GTP is considered in Section 3.4. Finally, Section 3.5 explains the integration of the developed GTP model into the holistic energy system model. An overview of the existing methodology and the extensions presented in this study are given in Fig. 1.

3.1. General approach

For the determination of the optimal municipal energy system design in this study, the "Renewable Energies and Energy Efficiency Analysis and System OptimisatioN" (RE^3ASON) model is used and further extended ([35], cf. Fig. 1). In the first step of the model ("Input data determination") the required input data are calculated with the use of a Java model (Eclipse). The input data are applied in the second step, the actual optimisation model, which is implemented by using the General Algebraic Modeling System (GAMS). The RE³ASON model consists of several parts, which provide transferable methods for determining the existing technologies, infrastructure, the heat and electricity demand of residential buildings as well as the potential and associated costs for energy supply from photovoltaic (PV), wind and biomass in an arbitrary location. Due to the transferability, this model is applied in the present study, as various municipalities in different locations are investigated. RE^3ASON further provides a deterministic model of optimal investment and dispatch for new energy conversion

technologies at the community level. In the mixed integer linear program (MILP), the optimal technology investment and unit commitment of all technologies as well as energy flows between districts is identified. The model serves to cope with the complexity resulting from the number and combinations of the individual measures and their dependencies that would otherwise not be feasible. Included in the model are the above-mentioned energy supply technologies as well as measures such as insulation, heating technologies or appliances. The mu-nicipality under consideration is divided into districts, in which buildings are grouped into building types according to the TABULA building typology [36]. The spatial resolution consists of these districts as nodes to which the input, like heat and power demand, is assigned. In addition, the existing infrastructure such as gas and electricity grids are identified. However, no DHNs could be built in the old version of the model. Therefore, it is extended for the utilisation of district heat from the GTP (cf. Section 3.5). In the present study, the model is used to perform a long-term energy system optimisation (from 2015 to 2030), whereby each 5th year is modelled explicitly and divided into 72 time slices (4 seasons, 2 day types, 9 time slices within each day). An energy system must ensure security of supply even under extreme conditions of energy demand and climatic conditions [37]. Therefore, one extreme day is designed for every season. In the extreme days, the energy demands in the residential buildings reach the maximum values of the originally determined demand load profiles for the respective season

	Input data determination (Java, Eclipse)	Energy system optimisation (GAMS)
REªASON	 Simulation of energy service demand Appliances Lightning Room comfort Identification of existing technologies Energy generation plants Heating technologies Identification of existing infrastructure (gas, electricity grid, etc.) Calculation of technical and economic renewable energy potential for Biogas, PV and Wind 	 Optimisation of cost for energy system (CO₂ emissions and other target oriterias also possible) 4 model years, 72 time silces Time resolution: hourly Spatial resolution: tistricts Energy imports (electricity, coal, gas, oil), which are prohibited in case of complete energy autonomy Electricity demand has to be covered in every time step Heat demand has to be covered per day, as buildings are considered as natural heat storage units
RE ^s ASON _{GTP} Extensions	 Identification of geothermal potential (borehole temperature and drilling depth, section 3.2) in 8.5 km² resolution DH designing (Weinand et al. 2019a): Determination of a cost-effective district heating network design for different heat coverages and the associated costs 	 GTP model equations Technical (section 3.3) Economic (section 3.4) Integration in RE³ASON (section 3.5)
L		

Fig. 1. Overview of RE³ASON and the extensions developed in the present study.

and no solar radiation or wind is present.

The model can be used to minimise total discounted system costs, CO2 emissions or energy imports of the municipal energy system. The calculation of the potential of PV, wind and biomass is briefly explained in the following, for further information about the model including the mathematical model formulation the reader is referred to McKenna et al. [38] and Mainzer [35]. In order to determine the PV potential, OpenStreetMap is used to identify building data and roof areas [39]. Satellite data is used to detect roof type and orientation. In addition, a methodology based on neural networks detects already installed PV modules. More information about the PV potential calculation can be found in Mainzer et al. [40]. The potential calculation of wind turbines uses minimum distance specifications to, for example, settlements or airports to determine the number of wind turbines that can be placed in the municipality. Based on the minimum levelized cost of electricity (LCOE), the wind turbine type is selected. For wind and PV potential determination, different local climates are also included in the calculation. In the case of biomass plants, the maximum amount of substrates that can be produced on the agricultural and forest areas of the municipality is determined for different types of plants. Implemented plants are waste-to-energy plants, woody biomass combustion plants and biogas plants. The LCOE calculation includes investments as well as transport costs. The amount of energy that can be supplied using the substrates can be divided arbitrarily over the year. The biomass plants, therefore, can be used flexible and can serve peak loads as well as base loads. Results of the input determination for photovoltaics, wind power and biomass as well as for the district heating network for an exemplary municipality can be found in the electronic Appendix. The RE³ASON model extensions shown in Fig. 1 are explained hereafter. The extensions are implemented within the program environment of the existing RE³ASON model.

3.2. Achievable hydrothermal temperature and required drilling depth

The achievable hydrothermal temperature $T_{FW,\max}$ (wellhead temperature) is based on the values of the Geothermal Information System GeotIS [41]. For the present study, the data from the geographical map of Germany in GeotIS were transferred to a CSV file in 8.5 km² resolution and the vertices of the grid now serve as input for the optimisation model. Fig. 2 shows the geographical map of the 11,100 German municipalities with achievable temperatures at a depth of up to 5 km resulting from the data transfer. The achievable temperatures reach up to 190 °C and the prevailing pressure conditions in the brine are assumed to be high enough, so that the water does not boil below this temperature. If more than one point of the 8.5 km² grid is located in the municipality, the mean value of the achievable temperatures is employed. Since the transfer of the data to create the grid with the hydrothermal temperatures required a substantial amount of work, the CSV is provided as supplementary material.

As soon as the hydrothermal temperature is known, the required drilling depth z_D can be determined. For this purpose, according to Bauer et al. [42], the mean German temperature gradient of 32 °C/km is assumed for the South German Molasse Basin (MB) and 35 °C/km for the North German Basin (NGB). In the Upper Rhine Graben (URG), on the other hand, the temperature gradients are much higher with 43 °C/km or average (locally up to 110 °C/km, Bauer et al. [42], especially in the area up to 3 km [43]). MB, NGB and URG are shown in Fig. 2. In order to be able to determine the temperature gradient more precisely, the temperature gradient resulting from the drillings for the GTP in Landau are taken over for the entire URG [44], but adjusted in such a way that the mean value is 43 °C/km. This results in the three temperature scetions in Fig. 3. The temperature gradient ΔT decreases with depth and adapts to the German average. Above 3250 m, a constant temperature gradient is assumed.

Applied Energy 255 (2019) 113824

3.3. GTP model for simultaneous heat and electricity generation

This section describes the equations used to implement the GTP model. In the equations, variables are shown in bold, the other terms are parameters. Thereby the most important model equations will be described. The authors could provide the complete modeling equations upon request. The schematic illustration of the GTP in Fig. 4 is intended to help identify the most important variables and parameters of the model, which are introduced in this section. Since existing DHNs in Germany cannot be identified without further research, it is assumed that no DHN already exists in the municipalities (areen field approach).

that no DHN already exists in the municipalities (green field approach). The power generation P_{el} of the ORC plant and heat generation \dot{Q}_{eh} of the district heating plant (DHP) per time step t can be determined using Eqs. (1)–(2).

$$\dot{V}_{B} \cdot \boldsymbol{\rho}_{w} \cdot \boldsymbol{c}_{p,w} \cdot (T_{PW}(t) - T_{ORC,out}(t)) = \boldsymbol{P}_{el}(t) / \boldsymbol{\eta}_{el} \quad \forall t$$
(1)

$$\dot{V}_{B'}\boldsymbol{\rho}_{m'}\boldsymbol{c}_{D'm'}(T_{OBC,out}(t) - T_{DHP,return}(t)) = \dot{\boldsymbol{Q}}_{fb}(t)/\boldsymbol{\eta}_{fb} \quad \forall t$$
(2)

The maximum possible volumetric flow rate depends on the local geological conditions. Furthermore, in most GTPs the economically and energetically optimal volumetric flow rate is significantly below the maximum possible flow rate, since the installation depth and design of the feed pump depend on the flow rate [45]. Therefore, the mean flow rate of the existing deep geothermal projects in Germany of 751/s is used as the volumetric flow rate $V_{\rm B}$ in the model (cf. Table A2). The heat capacity $c_{p,w}$ of geothermal water increases from approximately 4.18 kJkg⁻¹.K⁻¹ at 20 °C to about 4.45 kJkg⁻¹.K⁻¹ at 100 °C. In the same temperature interval the water density p_w decreases from approximately 1 kg/1 to about 0.9 kg/1 [46]. In this temperature interval, the product of these two coefficients differ by 4%. Therefore, a constant mean heat capacity of 4.31 kJ'kg⁻¹.K⁻¹ and a mean density of 0.95 kg/l are assumed in the following. The GTP is assumed to operate in base load, therefore P_{el} and \dot{Q}_{vh} are limited in such a way that these take



Fig. 2. Achievable average hydrothermal temperature (°C) at a depth of up to 5 km in German municipalities according to Agemar el al. [41] and locations of GTPs in Germany (blue circles). The background map in this Fig. and in all of the following figures with background maps is from OpenStreetMap contributors [39]. For a better analysis of the colours in this figure, please refer to the online version of the paper.



Fig. 3. Assumed temperature sections as a function of the drilling depth in the Upper Rhine Graben.

constant values in a season, i.e. four different values per year.

The injection temperatures are known for the power plants Insheim, Dürnhaar and Kirchstockach. No heat is generated in these plants (cf. Table A2). Thus, the efficiency of the ORC plants could be determined with the help of Eq. (1). The resulting mean value 13% is assumed for the electrical efficiency $\pmb{\eta}_{el}$ in the following, which is a relatively high value for binary ORC plants and almost corresponds to the typical efficiency of ORC plants of 16% [47]. In accordance with Ozgener and Ozgener [48], 65% is assumed for the efficiency η_{th} of the geothermal DH system, including heat exchange and heat losses in the DHN. Thus, the power and heat generation depends only on the temperature in the production well T_{PW} and the temperature after the heat transfer to the ORC process $T_{ORC,outs}$ as well as the temperature after the heat transfer to the DH system $T_{DHP,return}$, which are further constrained in the following. Based on the values from Zarrouk & Moon [49], a minimum output temperature TORC, min of 50 °C is assumed after the heat transfer to the binary power plant. To avoid that the ORC plant is necessarily built, this condition only applies if the binary variable b_{PP} equals 1 (cf. Eq. (3)).

$$b_{PP} \cdot T_{ORC,min} \leq T_{ORC,out}(t) \quad \forall t$$
(3)

The same applies to heat extraction for DH purposes. A conventional DHN is assumed here with a minimum forward temperature $T_{DH,\min}$ of 65 °C [50], a pinch temperature $T_{DH,pinch}$ of 5 °C at the heat exchanger and a return temperature $T_{DHP,reaur}$ of $T_{DH,reaurn,\min} = 55$ °C [51]. This condition also only applies if the DHP is built ($D_{DHP} = 1$). In order to

ensure that $T_{ORG,out}$ is not always at least $T_{DH,min} + T_{DH,pmch} = 70$ °C in this case, the time-dependent variable $b_{DHP,op}$ is introduced, which indicates whether heat is generated in a time step or not (cf. Eqs. (4) and (5)).

$(T_{DH,return,min} +$	- $T_{DH,pinch}$) $\boldsymbol{b}_{DHP,op}(t)$	$\leq T_{DHP,return}(t)$	$\forall t$	(4)

 $b_{DHP,op}(t) \cdot (T_{DH,min} + T_{DH,pinch}) \le T_{ORC,out}(t) \qquad \forall t$ (5)

$$b_{DHP,op}(t) \cdot M \ge T_{DHP,return}(t) \quad \forall t$$
(6)

Eq. (6) is introduced to ensure that the temperature $T_{DH,return,min}$ is maintained. *M* is a large number which should be as small as possible and at least as large as the maximum value of the restricted variable [52], in this case 190 °C. The injection temperature is at least 50 °C if no energy is used for DH purposes and 60 °C otherwise. As described in more detail in Section 3.4, the drilling cost function is divided into five sections, to obtain a linear function. Only one of the sections may be selected with the binary variable $b_{D,1}$ (cf. Eq. (7)):

$$\sum_{i=1}^{s} b_{D,i} \leq 1 \tag{7}$$

The selected range then determines the possible minimum and maximum value of the drilling depth $z_{D,*}$. The range of a section r_s in Eqs. (8)–(10) is set to 1000 m.

$$\boldsymbol{b}_{D,i} \cdot (i-1) \cdot \boldsymbol{r}_{z} \leq \boldsymbol{z}_{D,i} \quad \forall \ i \tag{8}$$

 $b_{D,i} \cdot i \cdot r_z \ge z_{D,i} \quad \forall i \tag{9}$

$$z_D \le 5 * r_z \tag{10}$$

For example, if $b_{D,2}$ is set to 1, then the drilling depth is in the range of 1000 m to 2000 m. However, since there are now five cost functions for the different sections of the drilling depth z_D , only the cost function that has been selected may be used. Therefore, the drilling depth $z_{D,1}$ is determined for each section, which is zero if the section is not selected (cf. Eq. (11)).

$$z_{D,i} = z_D \cdot b_{D,i} \quad \forall i \tag{11}$$

In the following, the costs can be assigned to the variables $z_{D,i}$ and $b_{D,i}$ (see Section 3.4). The achievable temperature is in turn determined via z_{D} . To avoid that all $b_{D,i}$ and $z_{D,i}$ are set to zero, the following equation must be added:

$$z_D = \sum_{i=1}^{5} z_{D,i}$$
 (12)



Fig. 4. Schematic illustration of the GTP considered in this study. The heat flow in the condenser of the ORC process could be water or air. For a better analysis of the colours in this figure, please refer to the online version of the paper.

7

The equation and the defined temperature sections from Section 3.2 are used to determine the temperature $T_{\ensuremath{\text{PW}},\ensuremath{\text{max}}}$ of the water in the production well if the investigated municipality is located in the Upper Rhine Graben (cf. Eq. (13)).

$$\begin{array}{l} \left(\frac{\mathbf{z}D_{1200}}{1_{900}}\right) \cdot (100^{\circ}C - T_{a}) + T_{a'}\mathbf{b}_{D,1900} + \left(\frac{\mathbf{z}D_{1250} - 1_{900m}\mathbf{b}_{D,1250}}{1_{350m}}\right) \cdot 55^{\circ}C + 100^{\circ}C \cdot \mathbf{b}_{D,3250} \\ + \left(\frac{\mathbf{z}D_{1350m}\mathbf{c}}{600m}\right) \cdot 20^{\circ}C + 155^{\circ}C \cdot \mathbf{b}_{D,3850} \ge T_{PW,max} \end{array}$$

(13)

The sections of the drilling depth $z_{D,1900}$, $z_{D,3250}$ and $z_{D,3850}$ to determine the temperature do not correspond to the sections of the drilling depths $z_{D,i}$ to determine the drilling costs. Therefore, the Eqs. (7)-(12) from above have to be replicated for these drilling depth sections as well. To limit the temperature $T_{PW, \max}$ to the maximum values of the ranges from Eq. (13), Eq. (14) must be implemented:

$$T_{PW,max} \le 100^{\circ}C \cdot b_{D,1900} + 155^{\circ}C \cdot b_{D,3250} + 190^{\circ}C \cdot b_{D,3850}$$
 (14)

For the North German Basin and the South German Molasse Basin, the temperature linear depends on the drilling depth (cf. Section 3.2). In addition, equations are needed to ensure that the GTP has to be built In addition, equations are needed to characterize that the of r has been been defined by $(b_{GP} = 1)$ for DHP is built and to limit the nominal power of ORC and DH plant, $P_{el_{eff}max}$ and $\dot{Q}_{th,max}$ to zero if not. The operating life of the plant is assumed to be 30 years based on Budisulistyo et al. [53].

3.4. Economic assessment

In GTP projects, four main phases occur to which costs can be allocated. The four phases are resource identification, resource exploration, drilling and energy production [54]. The costs for GTPs are subject to uncertainties. As an extreme example, investments in the Sauerlach project have increased from the original estimate of 25 M€ to 90 M€ (cf. Table A2, [55]). The cost-functions in this section are mainly based on Schlagermann [56]. Three of the eight components of the investment in Table 3 that are particularly relevant for this study are explained in this section. An explanation of the remaining cost functions can be found in Section A.4.

The costs C_1 are divided into the costs for the feasibility study C_{FS} , the preliminary underground planning C_{UP} as well as the preliminary planning of the power generation plant and the above-ground plant components C_{pp} [56]:

$$C_1 = C_{FS} + C_{UP} + C_{PP} = (180, 000 \pounds \cdot b_{GP})$$

with $A_{S} = (z_{D} + 4 \text{ km} + d_{D}) \cdot (z_{D} + 4 \text{ km})$

+ (100, 000
$$\boldsymbol{\epsilon} \cdot \boldsymbol{b}_{GP}$$
 + 25, 000 $\boldsymbol{\epsilon} / \mathrm{km}^2 \cdot \boldsymbol{A}_S$) + (150, 000 $\boldsymbol{\epsilon} \cdot \boldsymbol{b}_{GP}$) (15)

$$= z_D^2 + 9, 5 \ km \cdot z_D + 22 \ km^2$$
(16)

The preliminary underground planning includes a 3D seismology for which the area A_s must be known. The vertical depth of the drillings z_D and the distance between the two drillings d_D is required to calculate A_S . For d_D , deflections of the drillings are also taken into account. According to Stober & Bucher [57], the distance in the underground, in Applied Energy 255 (2019) 113824

the area of the working horizon, is usually 1000–2000 m. Therefore d_D is assumed to be 1500 m in the following. In order to use Eq. (15) in the linear optimisation problem, the function must be linearised. This was achieved by dividing the function into five sections in 1000 m steps. Now the costs can be calculated using the linearised cost functions (cf. Equations in red boxes in Fig. 5).

The costs for the exploration of the reservoir C_3 are divided into costs for drilling C_D , stimulation C_{ST} , hydraulic tests C_{HT} and geochemical investigations C_{GI} [56]. With a share of up to 70% of the capital costs, drilling costs account for the largest share of the investment for a GTP [58]. The drilling costs are estimated according to the following equation [59]:

$$C_D = 610, \ 000 \boldsymbol{\ell} * \boldsymbol{b}_{GP} + 1.015 \cdot C_{DW} = 610, \ 000 \boldsymbol{\ell} * \boldsymbol{b}_{GP} + 1.015 * [1.198 \cdot e^{0.00047894} \sqrt{z_D^2 + d_D^2} \cdot 10^6 \boldsymbol{\ell}]$$
(17)

The term in the square brackets covers the costs of the actual drilling work C_{DW} , the rest is for the construction and recultivation of the well site, among other things. This study assumes that all drilling is done at one well site. Therefore, the costs of $610,000 \in$ are only charged for the first well.

The drilling costs are assumed to be subject to economies of scale and the costs of the second well is only 90% of the costs of the first well. The same applies to the hydraulic tests with 67% as well as the geochemical investigations of the wells with 70% for the second well. The costs for the exploration of the reservoir can therefore be calculated using the following equations [56]:

 $C_3 = C_D + C_{ST} + C_{HT} + C_{GI}$ (18)

with $C_D = 610, 000 \notin b_{GP} + 1.015 \cdot C_{DW} \cdot (1 + 0.9)$ (19)

$$C_{ST} = 450,\ 000 \boldsymbol{\epsilon} \cdot (1+0.67) \cdot \boldsymbol{b}_{GP} \tag{20}$$

$$C_{HT} = 500, \ 000 \boldsymbol{\epsilon} \cdot \boldsymbol{b}_{GP} \tag{21}$$

$$C_{GI} = 65 \frac{\pi}{m} \cdot \sqrt{z_D^2} + d_D^2 \cdot (1 + 0.7)$$
(22)

The investment for exploration of the reservoir C_3 thus depend on the vertical drilling depth according to the function in Fig. 5. The deviations of the linearised curves from the actual costs C_1 and C_3 are between -2% and +1%.

3.5. Integration in RE³ASON

.

.

The entire GTP model presented above is integrated in the optimisation model of the RE3ASON model, as explained in the following using the three model extensions.

(1) Determination of the hydrothermal potential in a municipality (cf. ection 3.2)

The achievable hydrothermal temperature serves as input for the optimisation model and is determined within the scope of the input calculation. The basin is also transferred to the model in order to

fable	3			

Components of the investment for the GTP and their dependence on variables of the GTP model.

Variable of the component	Description of component	Depending on the following variables
$egin{array}{cccc} G_1 & & & \\ G_2 & & & \\ G_3 & & & \\ G_4 & & & \\ G_5 & & & \\ G_6 & & & \\ \end{array}$	Feasibility study and preliminary planning Properties and infrastructure Exploration of the reservoir Production well pump Thermal water system (above-ground) Power generation plant	b _{GP} , z _D bGP, bPP, b _{DHP} , P _{elmax} , Q _{thmax} b _{GP} , b _{W,3} , b _{W,2} , z _D b _{W,3} , P _{el} , Q _{th} b _{GP} P _{elmax} , b _{PP}
C_7 C_8	District heating system Project management, control and finance planning as well as other investments	$b_{DH,S,i}$ All variables from C_1 - C_7



Applied Energy 255 (2019) 113824

Fig. 5. The costs for the feasibility study and the preliminary planning C_1 (red) and the costs for exploration of the reservoir C_3 as a function of vertical drilling depth as well as linearization of the functions. The costs C_3 are shown for the first (blue) and second (yellow) drilling. For a better analysis of the colours in this figure, please refer to the online version of the paper.

determine the required drilling depth.

(2) GTP model equations for simultaneous heat and electricity generation (cf. Section 3.3) and

(3) GTP cost determination (cf. Section 3.4)

The technical and economic model equations are integrated into the RE³ASON optimisation model. The total costs for the GTP are determined as the sum of all individual costs C_1 to C_8 . These total costs and the variable costs of the GTP are part of the objective function to calculate the total discounted system costs for the entire energy system (cf. Eqs. (A1)–(A2) in the Appendix A of McKenna et al. [38]).

At this point it is necessary to clarify the consideration of the construction of a new DHN for the utilisation of the geothermal heat. The spatial resolution of the model is that of the district as a node (cf. 3.1). The costs for the DHN are calculated in several percentage steps of heat supply based on the heuristic developed in Weinand et al. [60]. In the following a segmentation into ten steps is chosen, i.e. costs for the supply of 10%, 20%, 30% etc. up to 100% of the municipal heat demand. This also includes determining which buildings should optimally be connected to the DHN in the various steps. Therefore, the investigated municipality has to be divided into districts to also take account of energy flows in the optimisation process. For this purpose, a Voronoi clustering is performed on the basis of the settlements in the municipality. The resulting Voronoi cluster for an example municipality can be found in the electronic Appendix. For a given number of points in a plane, a Voronoi diagram divides the plane according to the rule of the nearest neighbour: each point is associated to the region closest to it [61]. The resulting districts are used to determine which settlements will be connected to the geothermal-based DHN. The investment for the DHN in the different steps serves as input for the optimisation model in RE³ASON. In the optimisation model, the DH connections and costs then depend on the choice of the binary variable $b_{\text{DH},\text{S},i}$ (cf. Eq. (23)).

$\sum_{i=1}^{10} b_{DH,S,i} \leq 1$

For example, if all buildings/settlements in a municipality should be supplied with DH, then $b_{DH,5,10}$ has to be set to 1. If only 70% of the heat demand should be covered by DH, then $b_{DH,5,7} = 1$ applies. The DHN that results in this latter case can be found in the electronic Appendix for an example municipality. By selecting the binary variables, the output \dot{Q}_{th} of the DH plant is limited by a proportion of the heat demand

of the municipality $\mathrm{D}_{heat,total}$ (cf. Eq. (24)).

$\dot{Q}_{th}(t) \leq \sum_{i=1}^{9} (\mathbf{b}_{DH,S,i} \cdot 10\% \cdot i \cdot D_{heat,total}(t)) + (\mathbf{b}_{DH,S,10} \cdot M)$ (24)

The case with 100% is not included in the sum so that more DH can be generated than required. Therefore, the binary variable is multiplied by a large number *M*. In addition, the investment for the district heating system C_7 for the DHN can be specified with the binary variable $b_{DH,S,F}$.

4. Results

First, the GTP model is validated in Section 4.1. Subsequently, in Section 4.2, general information on the energy system optimisations are given and the results are presented in detail.

4.1. Validation

Validations are performed for the achievable hydrothermal temperatures and associated drilling depths as well as for the cost calculations (investment and LCOE). Table A2 shows a list of 31 deep geothermal projects in Germany, with corresponding data. The findings of this section can be analysed in more detail by examining the columns Temperature deviation, Temperature gradient and Depth deviation of Table A2 as well as the Figs. A1-A3. Since the drilling depths as well as the achieved temperatures are indicated in the projects, the model can be validated with the data from Table A2. For this purpose, the temperatures in the model are calculated on the basis of the actual drilling depths of the projects. The locations show deviations from the real achievable temperatures in the existing geothermal projects. The deviation could occur due to the fact that locations with above-average temperature gradients have been chosen for the geothermal projects realised so far. Furthermore, the hydrothermal temperatures could only be transferred as point coordinates of an $8.5 \,\mathrm{km^2}$ grid. The root-meansquared-error (RMSE) between the real and calculated temperature values is 27 °C. Actually, the RMSE calculated with the model equations would be only 16 $^\circ\text{C},$ however, the calculated temperatures are limited by the maximum achievable temperatures from Fig. 2. At 17 of the 31 locations, the deviations are less than 15%, all deviations are on average 19%. Apart from a few municipalities in the North German Basin, the accuracy of the model is considered sufficient for generic application and identification of municipalities with potential geothermal exploitation. For a detailed planning of a GTP plant, specific

(23)

measurements have to be performed. Since hydrothermal temperature and drilling depth are directly interdependent in the model, the deviations are the same for both parameters. The validation of the investment for the GTP can be found in Section A.4.1 of the Appendix A. All in all, the validation shows satisfactory results for the model. Due to the lack of data, however, it is not possible to predict the exact hydrothermal temperature and temperature gradient at a specific location. As already mentioned in the objectives in Section 1, the applied methods are sufficient to estimate the geothermal potential for the energy system optimisations, since this study is intended to demonstrate possible utilisation of this potential. Fig. 6 shows the calculated LCOE for geothermal power plants using

Fig. 6 shows the calculated LCOE for geothermal power plants using the model equations developed in this study (upper part). The temperature $T_{ORC,out}$ is set to 50 °C for the calculation of the reference cases. Based on the achievable temperature, the LCOE are given for all three large basins in Germany. Values between $0.05 \notin$ /kWh and $0.5 \notin$ /kWh are reached. In 2017 the LCOE of selected international geothermal projects ranged between $0.03 \notin$ /kWh and $0.14 \notin$ /kWh (cf. area between black dotted lines in Fig. 6, [62]). The range is given in general, not depending on the hydrothermal temperature. Since only projects with Applied Energy 255 (2019) 113824

Characteristics of the municipalities Westheim, Prinzenmoor and Groß Kreutz examined in this study.

Municipality	Westheim (Pfalz)	Prinzenmoor	Groß Kreutz (Havel)
Cluster [63]	3	4	8
Basin [41]	URG	NGB	NGB
Area [km ²]	7	6	99
Population	1731	179	8133
Population density in municipality [1/km ²]	243	32	82
Number of districts	1	1	12
Maximum hydrothermal temperature [°C]	190	175	80

achievable temperatures of over 80 °C are usually realised, the model can be assessed as plausible in this respect. For the NGB, four scenarios with changed parameter values compared to the reference case are also shown as sensitivities in the lower part of Fig. 6. If the ORC efficiency η_{el} is reduced or T_{ORGout} is increased, the LCOE increase. Thereby the



Table 4

Fig. 6. LCOE of geothermal power plants calculated by the model presented in this study using different achievable temperatures for the three large basins in Germany (upper part). The dotted black horizontal lines mark the area in which the LCOE of real deep geothermal projects fall according to IRENA [62]. As a sensitivity analysis, the change in LCOE for the NGB are shown with parameter values deviating from the reference case (lower part). For a better analysis of the colours in this figure, please refer to the online version of the paper.

deviation of the LCOE in comparison to the reference case decreases the higher the achievable hydrothermal temperature is. Furthermore, the deviation for parameter changes that reduce electricity production ($\eta_{el} = 5\%$ and $T_{ORC,out} = 80$ °C) is higher due to the high fixed costs of the GTP. Since the LCOE deviation is largest in the scenario with an efficiency of 5% (in the relevant temperature range above 100 °C), this sensitivity is also investigated in the optimisations in Section 4.2.2. A validation of the RE³ASON model can be found in Mainzer [35].

4.2. Results of the energy system optimisation case studies

Three German municipalities are selected as case studies. These municipalities are taken from three different municipality clusters [63], so that they differ considerably in terms of relevant indicators. The indicators of the municipalities important for the analyses are listed in Table 4. Westheim is selected to examine a densely populated municipality in the URG in which the maximum hydrothermal temperature of 190 °C can be reached. The case of Prinzenmoor, on the other hand, should show how the energy system changes, if a sparsely populated municipality is investigated. Furthermore, Groß Kreutz is chosen to investigate whether geothermal plants are also installed in municipalities with a large population and widely dispersed settlements in various districts as well as low hydrothermal potential.

The objective value of the optimisations are the total discounted system costs (TDSC). The energy systems are optimised for the period between 2015 and 2030. In addition, the majority of the optimisations consider complete municipal energy autonomy (CMEA), i.e. the demand must be met by local renewable energy and no energy imports are allowed. In all optimisations, the MIP-Gap, the relative tolerance on the gap between the best integer objective and the objective of the best node remaining, is set to 5%. The maximum capacity for the electricity storage per district is given as 20 MWh. Optimisation times depend largely on the number of districts in a municipality, as more energy flows need to be optimised in cases with many districts. In addition, the optimisations for Westheim lasted longer than for Prinzenmoor since Westheim is located in the Upper Rhine Graben and therefore Eqs. (13)–(14) become relevant. The calculation times ranged from a few hours to seven days. More information about the optimisation problem and the computation is given in Table 5. The number of variables and equations depends on the number of districts, therefore the optimisation problem for Groß-Kreutz is considerably larger. In addition, more equations and variables are contained for Westheim than for Prinzenmoor, due to the different modelling of the drilling depth (cf. Eq. (13)).

Further input data, including energy carrier price developments and techno-economic assumptions, can be found in the electronic Appendix.

Whilst scenario specifications and the optimisation results concerning the geothermal plant can be found in Table 6, the resulting TDSC and their development are shown in Fig. 7. Before presenting specific results in detail in Sections 4.2.1 and 4.2.2, the following list gives an overview of the most important results, as well as in which section and by which scenarios these results can be explained:

- Achieving CMEA is associated with a significant increase in TDSC and is never achieved unless it is enforced (all scenarios).
- The construction of a GTP is not worthwhile, if CMEA does not have to be achieved (W1, P1, GK1 in Sections 4.2.1 and 4.2.2).
- The construction and operation of a GTP can significantly reduce the TDSC, if CMEA has to be achieved (W2 and GK2 in comparison with W5 and GK3 in Section 4.2.1).
- The previous statement also applies to the case with significantly lower ORC efficiency (W3 in Section 4.2.2).
- Neglecting the district heating option of a GTP by only considering the ORC could lead to higher TDSC (W2 in comparison with W4 in Section 4.2.2).
- In municipalities with low population, the installation of a GTP is not worthwhile due to high fixed costs (P2 in comparison with P3 in

Section 4.2.2).

• In contrast, the TDSC can be significantly reduced, even at low hydrothermal potential, if the fixed costs of the GTP are distributed over a large population (GK2 in comparison with GK3 in Section 4.2.2).

In the description of the results, the focus is on the GTP and electricity.

4.2.1. Impact of geothermal plants on complete municipal energy autonomy Scenario W2 demonstrates on the one hand the importance of a GTP

in achieving CMEA and on the other hand the importance of considering the heat side and the electricity side of the GTP. In comparison to scenario W5, in which the installation of a GTP is prohibited (cf. Table 6), the use of the GTP saves $1.6\,\mathrm{k}\mathbb{C}$ per inhabitant for the energy supply system until 2030 (cf. Fig. 7). Thus, the use of a GTP can significantly reduce the cost of achieving CMEA. In scenario W2, only one biomass plant with a capacity of 225 kWel is built in addition to the GTP to cover the demand. Due to the base-load operation of the $1.1 \, \text{MW}_{el}\text{ORC}$ plant, the electricity demand is exceeded in some time steps (cf. Fig. 8). In these cases, the excess energy is temporarily stored in a 1.3 MWh battery storage and reused as required. Scenario W5, on the other hand, requires an 18.2 MWh storage capacity in order to achieve CMEA using biomass and the volatile energy supply technologies PV and wind. This clarifies the high increase of the costs for CMEA through the use of large storage capacities which was already found in the literature in Section 2.1. By the use of the GTP in scenario W2, these costs can be lowered, however, the TDSC increase by 70% in comparison to the reference scenario W1 without CMEA.

Since almost all heat demand is covered by the GTP via DH in scenario W2, only a few power-operated heat generators (e.g. electric storage heaters) are operated. The electric storage heaters are only needed for peak loads, which cannot be covered by the DH base load. A large part of the municipal power consumption is caused by the pump in the production well of the GTP. For further information on the distribution of the consumption patterns, please refer to the electronic Appendix.

The preference for biomass plants and the GTP in scenario W2 over the volatile energy from wind and PV can be explained by the base load capability of the plants. However, it is not so obvious why the GTP is preferred instead of biomass plants, which can be operated flexibly in addition to the base load. On the one hand this is due to the low technical biomass potential in Westheim. On the other hand, the average LCOE of the potential biomass plants of 0.25 ϵ /kWh is significantly higher than that of the GTP amounting to (at least) 0.07 ϵ /kWh with a hydrothermal temperature of 98 °C. Even with assumptions that would reduce the performance of the GTP, the LCOE would still be below 0.25 ϵ /kWh (cf. Section 4.1).

4.2.2. Sensitivity analysis

A particularly critical assumption in the modelling of the GTP in Section 3.3 is the assumption of a constant ORC efficiency of 13%. For

Table 5

Structure of the optimisation problem as well as properties and hardware for solving the model.

Case study	Westheim (Pfalz)	Prinzenmoor	Groß Kreutz (Havel)
Number of variables	602,619	573,177	5,945,481
Number of binary variables	3235	3180	6724
Number of equations	1,074,006	1,031,469	8,484,842
Programming language Solver	General Algebrai CPLEX (branch 8	c Modeling System & cut)	(GAMS)
Hardware properties	Intel Xeon E-165	0 v2; 3.5 GHz; 12	Threads; 128 GB RAM

Table 6

Scenarios including s	pecification	ns as well as th	e results for the	geothermal plant	in the individual s	cenarios.			
Scenario	Scenario sp	pecifications			GTP results				
	CMEA forced?	ORC allowed?	DH allowed?	Special characteristics	Drilling depth [m]	Tempera-ture [°C]	ORC capac-ity [kW]	DH capac-ity [kW]	GTP invest- ment [M€]
Westheim 1 (W1) Westheim 2 (W2) Westheim 3 (W3) Westheim 4 (W4) Westheim 5 (W5)	× • • •	* * * *	* * * *	$\boldsymbol{\eta}_{el} = 5\%$ ORC forced	- 1850 2130 1580 -	- 98 109 85 -	- 1100 610 1390 -	- 1970 1920 - -	- 22 22 18 -
Prinzenmoor 1 (P1) Prinzenmoor 2 (P2) Prinzenmoor 3 (P3)	× • •	\$ \$	* * *	ORC forced	- - 1590	- - 56	- - 230	- -	- - 15
Groß Kreutz 1 (GK1) Groß Kreutz 2 (GK2) Groß Kreutz 3 (GK3)	× ≮ ≮	≮ ≮ ×	≮ ≮ ×		- 2290 -	- 80 -	- 1200 -	-	- 20 -



Fig. 7. TDSC per inhabitant for the energy supply between 2015 and 2030 in various scenarios for the municipalities Westheim, Prinzenmoor and Groß Kreutz.

this reason, scenario W2 is changed in scenario W3 in such a way that the efficiency is only 5%. As shown in Table 6, even in this case, an ORC plant and a DH system are built. The $1.5 \,\mathrm{Ke}$ higher TDSC per inhabitant result primarily from the larger biomass plant and battery storage. An ORC plant is therefore constructed even with very low efficiency. The TDSC with the real efficiency would most likely be between the TDSC of scenarios W2 and W3.

The literature review in Section 2.2 shows that GTPs are usually investigated with regard to electricity generation and not with regard to simultaneous electricity and heat generation. The results of scenario W4 demonstrate that simultaneous electricity and heat optimisation could create added value. The TDSC in scenario W4, in which no DH system may be installed, are 2.1 k€ higher per inhabitant than in scenario W2 (cf. Fig. 7). The ORC system in scenario W4 has a higher capacity than in scenario W2, since more of the generated electricity has to be used for heat supply with heat pumps and electric storage heaters. Furthermore, the different drilling depths and the resulting hydrothermal temperatures in scenarios W2, W3 and W4 show the added value which results from the novel modelling of the drilling depth as a variable.

For the municipality Prinzenmoor, the costs in the scenarios with CMEA show a significant increase compared to the reference scenario P1 without CMEA (cf. Fig. 7). This indicates that the small rural municipalities which are mainly examined in the literature are not necessarily more suitable for achieving CMEA than larger municipalities (cf. Section 2.1), as will be clarified further below using scenario P3.

As Table 6 shows, no GTP is built to achieve CMEA in scenario P2.

The energy demand is covered by PV rooftop modules, biomass plants and a battery storage. The largest peak loads, which are mainly caused by heat pumps, are covered by biomass and the battery (cf. Fig. 8). If the construction of a GTP is forced (scenario P3), the TDSC increase by 472% compared to the reference scenario P1. This is due to the high fixed costs of the GTP, whose investment is not much lower than in the other scenarios despite significantly lower capacity (cf. Table 6). The two drillings alone account for 10 M€, corresponding to 66% of the total investment, which is distributed among significantly fewer inhabitants than in Westheim. This demonstrates that in smaller municipalities the fixed costs of the energy plants in particular lead to high TDSC per inhabitant. Therefore, economies of scale lead to lower TDSC in larger municipalities. Furthermore, the disadvantage of the lower temperature gradient in the NGB is evident, since the depth of the drillings in the URG would have to be only 970 m in order to reach the °C, instead of 1590 m in the NGB. 56

The dependence of the TDSC per inhabitant on the number of inhabitants is further confirmed in Groß Kreutz. Since the costs of the GTP are distributed among significantly more inhabitants than in the other two municipalities, CMEA only leads to a 4% increase in costs in scenario GK2 compared to reference scenario GK1. The increase in TDSC per inhabitant is significantly lower than in Prinzenmoor and Westheim (cf. Fig. 7). Thus, in the case of CMEA, the construction of a geothermal plant is worthwhile even with a low hydrothermal potential. However, the construction of a DH plant is probably prevented by the low achievable temperature. A comparison of scenario GK2 with scenario GK3, in which no GTP may be installed, shows this again. The cost to achieve CMEA without GTP almost doubles in GK3 (cf. Fig. 7). In GK3, wind turbines, PV modules, biomass plants and battery storages are installed.

5. Critical appraisal

This study is not without weaknesses which should be addressed in future work. While the developed model provides transferability to any German municipality⁴, the geological conditions on site must always be considered when planning a GTP. As shown in the previous sections, the model cannot guarantee 100% accuracy in the calculation of achievable hydrothermal temperatures, drilling depths and associated costs. This probably results from the average temperature gradients assumed for the different basins. The geothermal projects with which the model was compared were probably realised at sites with exceptionally high temperature gradients.

In addition, GTPs are sensitive to various types of scaling, including carbonate minerals, amorphous silicates, metal oxides and sulphides. Silica (SiO2), and calcite (CaCO3) are the most common. One result of the scaling is the degradation of plant components, which requires their





Fig. 8. Optimised energy supply patterns (upper part) - divided into energy supply technologies - on a typical weekday for all seasons in 2015 for the municipalities of Westheim (upper part) and Prinzenmoor (lower part). For a better analysis of the colours in this figure, please refer to the online version of the paper.

premature replacement [64]. In addition, the scaling reduces the diameter of the pipes and thus increases the total pressure drop and the friction factor in the pipe [65]. In the water of low-temperature GTPs (T < 150 °C at 1 km depth) there is no silica saturation even when cooling to approximately 20 °C in DH systems [64]. This would therefore be applicable to all locations in Germany. Problems caused by the high salt content in the water of the North German Basin are avoided by the relatively high injection temperature of at least 50 °C, which is applied in this study [66].

Furthermore, some assumptions about parameters like constant efficiencies, temperatures and flow rates were necessary to represent the GTP in a linear optimisation model. The temperature dependent ORC efficiency is one of the most critical assumptions and has been investigated in the sensitivity analysis (cf. Section 4.2.2). The ORC efficiency did not have a significant impact: although the TDSC have increased while the efficiency was considerably lower (+6%), the GTP has been built nevertheless to achieve CMEA. For the other parameters, the assumptions are rather conservative, but the impacts should also be further examined in the future. Due to the employed assumptions, the statement that a simultaneous optimisation of heat and electricity of the geothermal plant is recommended has to be critically evaluated. The fact that the installation and operation of the district heating system in scenarios W2 and W3 leads to lower TDSC could be explained by the uncertainties of the model. On the other hand, the efficiency of the competing ORC plant is overestimated. Since the district heating option is applied nevertheless, district heating should also be considered in future modelling of geothermal plants.

In addition, the GTP is assumed to operate without outages. This leads to a moderate underestimation of the LCOE of the GTP. A possible improvement of this weakness could be to create redundancy, which still allows the operation of the plant in the case of a pump outage. To this end, the model could be extended by the option to build a triplet instead of a doublet [67], which consists of an injection well and two production wells [68]. The heat from the reservoir could be used for up to ten years longer with a triplet well layout in comparison to a doublet well layout due to improved system performance [69].

Another important issue for future studies is the investigation of the influence of the time series structure on the results. In the $\rm RE^3ASON$ model an hourly resolution is used. A higher resolution would involve higher peak loads, which could make CMEA no longer achievable in municipalities, or with greater effort. Furthermore, only extreme days were considered in this study. In future analyses longer extreme periods could be considered to take periods of very low solar and wind production into account. However, even with only one extreme day per season, geothermal plants were built in the cases considered in this study.

Furthermore, the fact already mentioned in the literature was confirmed: the total costs to achieve CMEA increase very much, if large battery storage capacities are required (cf. Section 4.2.2). Besides battery storages for short-term purposes, it would also be sensible to include seasonal storages, such as hydrogen storages, in upcoming analyses.

Finally, the choice of the system boundaries must also be considered. The analysis of the municipality of Prinzenmoor showed that CMEA in small municipalities is associated with high additional costs due to the high fixed costs of the energy supply technologies. These municipalities could therefore be analysed together with smaller neighbouring municipalities. Since the municipality Groß Kreutz with the least TDSC in a CMEA scenario is the largest of the investigated municipalities with regard to the population, the conclusion is obvious that there could be an optimal municipality scale for CMEA. All of the aspects mentioned in this section will be the focus of forthcoming contributions.

6. Conclusions

The objective of this paper is the investigation of the economic feasibility of complete municipal energy autonomy in an energy system including geothermal plants. Therefore, a generic optimisation model of a geothermal plant, which simultaneously generates electricity and heat, is developed and integrated into an existing holistic energy system optimisation model. Variable drilling depths and thus hydrothermal temperatures represent one of the novel modelling approaches. As input for the optimisation, a linear regression estimates the achievable hydrothermal temperatures and the required drilling depths in the municipalities. Some cost estimations for the geothermal plant, such as drilling costs, had to be linearised for this purpose. A validation of the cost and the input determination with data from actual plants shows that the model presented in this work can reasonably be applied to any municipality in Germany without additional efforts⁴.

Related to the above objective, the specific research questions addressed in this paper are as follows. Could the high costs for off-grid municipal energy systems be reduced through the use of geothermal plants? Is it sufficient to consider only the electricity generation of the geothermal plant or would the use of the geothermal heat in district heating networks create an additional benefit? In order to answer these questions, the developed optimisation model was applied to three different municipalities from different municipal clusters. Eleven scenarios demonstrated that achieving energy autonomy is associated with high additional costs. Compared to the scenarios without energy autonomy, total discounted system costs for the period between 2015 and 2030 have increased by at least 4%. Thereby, the utilisation of geothermal plants can significantly reduce the costs for achieving energy

Appendix A

A. 1. Literature review table

See Table A1.

autonomy, which answers the first research question above. The electricity generation is preferred to heat generation in geothermal plants, which is related to the high costs for the district heating network. However, the importance of simultaneous modelling of electricity and heat generation in geothermal plants is evident, as district heating plants reduce the costs, especially in municipalities with high hydrothermal potential. This provides an answer to the second research question, i.e. that in the context of municipal energy system planning it is not sufficient to only consider the electricity side of the plant. Therefore, the installation of geothermal plants could help to decarbonise the energy system through energy autonomy.

This paper has developed a generally-applicable method for the optimal setup of a GTP within or near a residential area and considering both heat and electricity generation. Together with the related contribution for optimally locating the GTP plant within an existing or new district heating network [60], the consideration of the heat side represents a significant step forward. Compared to previous studies that focussed on a detailed GTP system setup, typically optimised for power generation, the present paper adopts a more holistic approach. The current and the above cited paper together provide a methodological framework for the economically effective and energetically efficient integration of GTPs into local energy systems. For renewable energy system planning this therefore represents an invaluable tool in the context of the energy transition. Furthermore, the employed method is intended to be highly transferable, both within Germany and, by employing additional data sources, beyond. It can provide decision support to local energy planners and other relevant stakeholders when considering the renewable energy options at their disposal.

Due to the fact that the employed methodology is intended as an early-stage planning tool, it has several uncertainties, however. Hence the authors emphasize the need for a more detailed energy system planning, especially but not only relating to the GTP, before entering the implementation phase. Most importantly, the costs of geothermal plants are very uncertain and depend on local geological conditions. Whilst the model presented provides a good estimate of the hydrothermal temperatures, the investment can rise due to uncertain incidents. Further work should also include the investigation of the impact of a higher temporal resolution as well as the identification of the optimal spatial scale for energy autonomy.

Acknowledgements

The authors gratefully acknowledge funding by the German Federal Ministry of Education and Research (BMBF) within the Kopernikus Project ENSURE 'New ENergy grid StructURes for the German Energiewende' (funding reference: FKZ 03SFK1N0) as well as the financial support of the PhD College "Energy and Resource Efficiency" (ENRES), from the Federal State of Baden-Wuerttemberg, for funding the first author's PhD studentship. The contribution of the second author (RM) was supported by the Heat 4.0 project from the Danish Innovation Fund (Reference 8090-00046B). The contribution of the third author (MK) was supported by the Helmholtz Association under the Joint initiative "Energy Systems Integration" (funding reference: ZT-0002). The authors also thank five anonymous reviewers and the editor, who provided valuable comments on earlier versions of this manuscript. The usual disclaimer applies.

J.M. Weinand, et al.	J.M.	Weinand,	et al.
----------------------	------	----------	--------

Table A1 Literature review on env	ergy system analyses with geothermal plants	,					
Study	Methodology	Assessment area/ region or plant site	Objective function	Energy system	Plant type	Energy Generation	Brine temperature
Aali et al. [70]	Exergoeconomic analysis and multi-objective	Site: Sablan seothermal field (Iran)	Specific cost of output power and	Geothermal plant	Combined flash-binary ORC	Power	165-183
Astolfi et al. [71] Budiadiotro et al. [62]	optimisation Dptimisation	Scourstand note (man) 	Thermodynamic efficiency	Geothermal plant	Binary ORC	Power	120-180
loci us to otisuusuud	LIETIDO-ECOLOLIUC ALERIYSIS	raupo vorcanic zone (New Zealand)		ocomennar prant		POWEL	101
Chagnon-Lessard et al.	Thermodynamic numerical simulation and ontimisation	I	Specific power output	Geothermal plant	ORC	Power	80-180
Clarke et al. [73]	Comparison of genetic algorithm with particle swarm optimisation for the constrained, non-	I	Plant performance (power output)	Geothermal plant	Double flash	Power	I
Clarke & McLeskey [74]	linear, simulation-based optimisation Multi-objective particle swarm optimisation (MOPSO)	I	Pareto-optimal set of designs depending on brine temperature and	Geothermal plant	Binary ORC	Power	80-180
Ghaebi et al. [75]	Exergoeconomic analysis & single- and multi- objective optimisations (using genetic aloration (GA))	I	ury-outo temperature Total levelized cost of energy system products; thermal efficiency along with the exercy efficiency	Geothermal plant	Cascade Kalina cycle (CKC)	Heat	170
Ghasemi et al. [76]	Parametric optimisation	1	Net power output	Geothermal plant	Binary ORC	Power	135
Guzović et al. [77]	Optimisation of the main parameters of ORC and Kalina cycle.	Lunjkovec-Kutnajk, Croatia	Power output, by optimising the main cycle parameters: ORC: upper cycle pressure Valina: concentration of ammonia	Geothermal plant	ORC and Kalina cycle	Power	140
Huster et al. [78]	Deterministic global optimisation of component size and operating conditions (Non Linear Prostam)	I	Net power, LCOE	Geothermal plant	ORC	Power	135.85
Karimi & Mansouri [79]	Exergeeconomic analysts and optimisation of different ORC configurations	20 countries (e.g. Australia, Brazil, Jordan)	Exergy efficiency, specific investment cost (SIC) and a combination of exergy and SIC	Geothermal plant	Organic Rankine Cycle (ORC), Regenerative Rankine Cycle (RORC) and Two-Stage Evaporation Organic Rankine Cycle (TSFORC)	Power	61.85-211.85
Kolahi et al. [80]	Particle swarn optimisation	Sabalan geothermal Dower plant (Iran)	Maximum total output power	Geothermal plant	Flash-binary ORC	Power	31-188
Lazzarin et al. [81]	Cycle and turbine re-optimisation on cothermal resources		Net electrical output	Geothermal plant	1	Power	145
Liu et al. [82]	Parametric optimisation and performance analysis	I	Net power output	Geothermal plant	ORC	Power	110-150
Liu et al. [83]	Analysis of the thermodynamic performance of a hybrid geothermal- fossil power generation system for various	I	I	Plant	Hybrid geothermal-fossil power generation system	Power	100-160
Liu et al. [84]	geothermal resource temperatures Thermo-economic analysis (multi-objective	I	Thermal efficiency, exergy efficiency,	Geothermal plant	Binary ORC	Power	80-95
Lu et al. [85]	optimisation	I	power output, capital cost Net power output	Geothermal plant	Single-flash system, double-flash system, flash-ORC system; and double-flash-	Power	170
Makhanlall et al. [86]	Optimisation by an exergy-topological methodoloov (coonitive thermodynamic tool)		Optimal operating conditions	Geothermal plant	ORC system Binary medium grade plant	Power	25-150
	Multi-objective optimisation	I		Geothernal plant	ORC	Power (co	167.75 ntinued on next page)
							•

Applied Energy 255 (2019) 113824

J.M. Weinand, et al.

Study	Methodology	Assessment area/ region or plant site	Objective function	Energy system	Plant type	Energy Generation	Brine temperature
Martínez-Gomez et al. [87] Marty et al. [31]	Simultaneous optimisation	1	Maximise profit from energy sales, minimise risk and minimise environmental impact (CO ₂ emissions) Net ammal profit	ORC plant, district	ORC	Heat and	185
	-		-	heating plant and a few consumers		Power	
Mohammadzadeh Bina et al. [88]	Thermo-economic analysis (EES optimisation (multi-criteria))	Sabalan geothermal power plant (Iran)	Energy efficiency, exergy efficiency, net power output, production cost, total cost rate	Geothermal plant	Basic ORC (B-ORC), Dual fluid ORC, Regenerative ORC (R-ORC), ORC with Internal Heat Exchanger (IHE-ORC)	Power	165
Mohammadzadeh Bina et al. [89]	Exergoeconomic analysis and optimisation	Sabalan geothermal power plant (Iran)	Net power output	Geothermal plant	Single and Double flash cycles	Power	149-242.96
Mosaffa et al. [90]	Thermodynamic and economic analysis (optimisation)	. 1	Maximise energy and exergy efficiency levels, minimise total product unit cost	Geothermal plant	ORC with Internal Heat Exchanger (IHX), Dual fluid ORC, Regenerative ORC	Power	160-180
Nakomcic Smaragdakis et al. [91]	Determination of geomophological, geological, geophysical, hydrogeological and hydrothernal daracteristics of the area, 2, gaothysis of physicodemical daracteristics of geothermal and mixeral waters, as well as drill's capacity; 3. Possble applications of the geothermal that are considered and recommendations for there are are even	Indjija (Serbiz)	T	Geothermal plant		Heat	1.
Østergaard et al. [32]	Analysing scenario for supplying Aalborg through renewable energies on the basis of the EnergyPLAN model	Aalborg (Denmark)	1	Energy system of the municipality Aalborg	1	Heat	I
Østergaard & Lund [33]	Analysing scenario for supplying Frederikshavn through renewable energies on the basis of the EnergyPLAN model	Frederikshavn (Denmark)	1	Energy system of the municipality Frederikshavn	1	Heat	I
Pambudi et al. [92]	Optimisation	Dieng Geothermal Power Plant (Indonesia)	Energy exergy flow and efficiency	Geothermal plant	Single flash binary	Power	18–179.9
Pambudi et al. [93]	Thermodynamic and silica scaling analysis	Dieng Geothermal Power Plant (Indonesia)	Increase capacity, reduce the impact of silica scaling	Geothermal plant	Single flash binary, double flash binary	Power	116.4
Peña-Lamas et al. [94]	Optimisation	I	Structure of the cycle, operating conditions and best fluid	Geothermal plant	Binary ORC	Power	164.34–170
Pollet et al. [95] Shengjun et al. [96]	Optimisation Parametric optimisation and performance analysis	1.1	Total energy output Thermal efficiency, exergy efficiency, recovery efficiency, heat exchanger area per unit power output and the levelized energy cost	Geothermal plant Geothermal plant	ORC Subcritical ORC and transcritical power cycle	Power Power	120-150 80-100
Sigurdardottir et al. [97]	Lumped parameter modelling (LPM) combined with a mixed integer linear programming (MILP)	Laugames geothermal system (South-West Iceland)	Profit	Geothermal reservoir	I	Heat	127.55
Sun et al. [98]	Thermodynamic optimisation (parametric analysis by means of genetic algorithm)	I	Turbine high-level inlet pressure and temperature, turbine low-level inlet pressure	Geothermal plant	Double-pressure ORC	Power	120
						103)	timed on next page)

Applied Energy 255 (2019) 113824

154

Table A1 (continued)							
Study	Methodology	Assessment area/ region or plant site	Objective function	Energy system	Plant type	Energy Generation	Brine temperature
Sun et al. [99]	Thermodynamic optimisation		Evaporation temperature and net nower output	Geothermal ORC	ORC	Power	100-150
Sveinbjörnsson et al. [34]	Energy supply modelling using the Sifre tool	Sonderborg Municipality (Denmark)	power output Operating expenses of the specified energy demand during all time steps	Energy system of the municipality Sonderborg	1	I	I
van Erdeweghe et al. [30]	Thermodynamic optimisation		Electrical power output	Combined heat-and- power (CHP)	ORC	Heat and Power	110-150
van Erdeweghe et al. [100]	Two-step optimisation	Balmatt site, Belgium	Design case: Net present value Off-Design case: Net electrical power outbut	Geothermal plant	ORC	Power	110-150
Walraven et al. [101]	Optimisation with different types of cooling vorems	Belgium	Levelized cost of electricity	Geothernal plant	ORC	Power	100-150
Wu et al. [102]	optimization (parametric optimization via optimization (parametric optimization via parteen search alcorithm (DSA))	I	Turbine inlet temperature, turbine inlet pressure, power output	Geothermal plant	Transcritical Power Cycle (TPC)	Power	100-150
Yang & Yeh [103] Yilmaz [104]	Economic performance optimisation Economic performance optimisation Thermo-economic optimisation using genetic	1 1	Net power output Cost optimal exergetic efficiency	Geothermal plant Geothermal plant	Transcritical Rankine Cycle (TRC) Combined flash-binary	Power Power	44.13–66.02 150–159
Zare [105]	agoutunt neutoor Thermodynamic and exergoeconomic optimisation with own models & profitability evaluation	I	Optimisation: total capital investment and payback period	Geothermal plant	ORC-based binary power plant	Power	160-170
Zhao & Wang [106]	Exergeconomic analysis and optimisation	ī	Exergoeconomic: Minimise levelized cost per unit of exergy; Thermodynamic: maximise exergy efficiency.	Geothernal plant	Flash-binary ORC cycle	Power	200

Applied Energy 255 (2019) 113824

projects
geothermal
deep
ио
tables
Data
A.2.

See Table A2.

Data on German de indicated in the tab	ep geothern de column "	nal projects [4 'sources". If th	he maximum we	ellhead tempera	iture of the mod	iei is less man t				
Location	Latitude	Longitude	Wellhead temperature reality [°C]	Wellhead temperature model [°C]	Max. Wellhead temperature model [°C]	Temperature deviation	Depth reality [m]	Depth model [m]	Depth Deviation	Temperature gradient ["C/km]
Groß Schönbeck	52.9099	13.5544	150	151	8	47%	4309	4286	-1%	34.8
Hannover	52.4872	9:9095	169	137	80	-53%	3901	4829	+24%	43.3
Neubrandenburg	53.5547	13.2357	71	44	115	-37%	1268	2029	+60%	56.0
Neuruppin	52.9244	12.8150	64	57	115	-11%	1620	1829	+13%	39.5
Neustadt-Glewe	53.3714	11.5930	67	86	115	-11%	2450	2771	+13%	39.6
Waren	53.5253	12.6548	63	55	115	-13%	1565	1800	+15%	40.3
Average NGB			102	88	103	29%	2519	2924	21%	42.2
Bruchsal	49.1261	8.5669	123	126	8	-22%	2542	2465	-3%	48.4
Brühl	49.3924	8.5352	160	157	150	% 9-	3318	3400	+2%	48.2
Insheim	49.1539	8.1534	165	173	121	+3%	3800	3550	%2-	43.4
Landau (Palatinate)	49.1864	8.1228	159	156	121	-24%	3291	3370	+2%	48.3
Average URG			152	153	134	14%	3238	2926	4%	47.1
Aschheim	48.1701	11.7352	85	84	8	~9~	2630	2656	+1%	32.3
Dümhaar	47.9960	11.7260	135	132	115	-15%	4114	4219	+3%	32.8
Erding	48.3087	11.8828	63	75	22	-10%	2359	1969	-17%	26.7
Garching	48.2571	11.6672	74	71	80	-4%	2226	2313	+4%	33.2
Geretsried	47.8432	11.4903	165	193	145	-12%	6036	5156	-15%	27.3
Grünwald	48.0349	11.5223	130	131	107	-17%	4083	4063	-1%	31.8
Ismaning	48.2310	11.6986	75	88	80	%2+	2738	2344	-14%	27.4
Kirchstockach	48.0274	11.6877	139	124	115	-17%	3881	4344	+12%	35.8
Kirchweidach	48.0991	12.6454	130	121	115	-12%	3800	4063	%/+	34.2
München (Riem)	48.1349	11.7153	95	88	80	-16%	2747	2969	+8%	34.6
Poing	48.1673	11.7915	76	88	8	+5%	3049	2375	-22%	24.9
Pullach	48.0567	11.5140	104	112	83	-11%	3505	3250	-7%	29.7
Sauerlach	47.9723	11.6686	140	143	127	% 6 -	4480	4375	-2%	31.3
Simbach- Braunau	48.2571	13.0105	81	62	69	-23%	1941	2531.25	+30%	41.7
Straubing	48.9751	12.5553	37	26	0	-100%	824	1156.25	+40%	44.9
Taufkirchen	48.0237	11.6208	136	136	115	-15%	4258	4250	%0	31.9
Traunreut	47.9732	12.5792	108	149	115	%9+	4645	3375	-27%	23.3
Unterföhring	48.1933	11.6593	86	64	80	-26%	1986	2687.5	+35%	43.3
Unterhaching	48.0660	11.6131	123	115	115	%2-	3590	3843.75	+7%	34.3
Unterschielßheim	48.2725	11.5806	78	63	8	-20%	1960	2437.5	+24%	39.8
Waldkraiburg	48.2024	12.4206	109	87	8	-27%	2718	3406.25	+25%	40.1
Average MB			104	103	92	17%	3242	3262	14%	36.6

J.M. Weinand, et al.

(continued on next page)

J.M. Weinand, et al.

Location	Injection Temperature [°C]	Forward/return temperature [°C]	Volumetric flow rate [L/s]	Nominal power (electrical) [MW]	Nominal Power (thermal) [MW]	Calculated electrical efficiency [%]	Sources
Groß Schönbeck			21	0.07	7 6		
Neubrandenburg		65/35	11-28	0	3.8		Schallenberg
Neuruppin			14	0	2.1		[801]
Neustadt-Glewe		09/06	11-33	0.2	5/6.5		Schallenberg
Waren		70/50	17	0	3.6		[oot]
Average NGB Bruchsal	60	90/65	14-20 24	0.55	5.5		Eyerer et al. [59]
Brühl Insheim	70		86	4.8	-	¥	
Landau	50	70/40	70	3.8	6	21	Eyerer et al. [59]
(Palatinate)			Ę			;	
Average URG Aschheim			00 X	-	76	15	
Dümhaar	45		130	5.5/7	0	15	
Erding			48	0	34/10.2		
Garching			100	0	15/8		
Geretsried		1000		Ļ	c L		1000 I
Lerunwaid Lemoning		C0/071	15U 8E	6.4 0	01		bauer et al. [42]
Kirchstockach	45		145	5.5	0	10	
Kirchweidach			50	0.7	12		
München (Riem)			75	0	10		
Poing			80	0	6		
Pullach			71	0	15		
Sauerlach	45	90-105/60	110	4	م		
Simbacn-Braunau Strenbing			80 45	0.2	8		
Taufkirchen		115/70	120	4.3	35		
Traunreut	55	100/65	150	5	12		
Unterföhring			75	0	10		
Unterhaching	60	80-110/50-60	150	3.4	38		Eyerer et al. [59]
Unterschleißheim			06	0	28/8		
Waldkraiburg			65	4.8	14		
Average MB			95			12.5	
							continued on next page)

Applied Energy 255 (2019) 113824

Applied Energy 255 (2019) 113824

Location	Total investment reality [m. 6]	Total investment model [m. €]	Total investment deviation	Investment for drilling [m. €]	Investment distric heating network [m. €]	Investment for plant process [m. €]	Plant Process	Commissioning year	Number of wells	Sources
Groß Schönbeck	12.1	38	+ 212%				ORC	Testing phase	2	GTN [109]
Hannower	1	17	+ 15%					develonment		Rindesverhand Geothermie [110]
Manhandarhuna	3	4 6	A/01 -					1007	- 0	Bunderrethand Contraints [110
Neubrandenburg		17						1907	^ c	
Neuruppill		01						70.07	7	DUIDESVERVATION OCOULENTING
Neustadt-Glewe	10.8	24	+122%	1.7	1.8	3.2	ORC	2003	c1 •	GTN [1111],Enerchange [112]
Waren		17.						1984	4	
Average NGB	13	24	116%	:					5.3	
Bruchsal	17	.24	+ 43%	8.1			Kalina	1002	7	
Brühl								development		
Insheim	50	40	- 20%				ORC	2012	2	Eyerer et al. [59]
Landau	20	39	+ 93%				ORC	2007	2	
(Palatinate)										
Average URG	19	34	52%						2	
Aschheim	60	45	- 24%					2009	2	Bundesverband Geothermie [11
Dürnhaar	60	50	-17%					2012	2	Bundesverband Geothermie
										[110], Eyerer et al. [59]
Erding	18.7	51	+172%					1998	2	Bundesverband Geothermie [11
Garching		32		13				2011	2	Bundesverband Geothermie [11
Geretsried								development		
Grünwald	150.4	92	- 39%		53.8		ORC	2011	2	Erdwärme Grünwald GmbH
										[113]
Ismaning	71.6	36	-50%	18	41			2012	en en	
Kirchstockach	62	44	- 30%	23			ORC	2013	2	Eyerer et al. [59]
Kirchweidach		40						2012	2	Bundesverband Geothermie [11]
München (Riem)		31						2004	2	Bundesverband Geothermie [11]
Poing		30						2013	2	
Pullach		47		6				2005	en	
Sauerlach	90 (initially 25)	62	- 31%				ORC	2014	en 1	Bundesverband Geothermie
										[110] Zeitungsverlag tz [55]
Simbach-Braunau	21	24	+14%				ORC	2001	2	Bundesverband Geothermie [11]
Straubing	12	16	+ 33%					1999	2	Bundesverband Geothermie [11]
Taufkirchen	65	78	+ 20%				Kalina	Testing phase	2	Bundesverband Geothermie
								(2014)		[110]. Everer et al. [59]
Traumrent	80	62.	- 23%				ORC	2014	2	
Unterföhring	37.5	36	-5%	12.5	14.5			2009	4	
Unterhaching	80	73	- 9%		48	16	Kalina	2009	2	Geothermie Unterhaching [114]
Unterschleißheim	22	44	+ 99%					2013	2	Bundesverband Geothermie [11]
Waldkrathure	12	46	+ 281%					2012	6	OVB24 GmbH
>										[115],Bundesverband
	:	ļ							0	Geothermie [116]
AVAYOR NIK	5	47	E A UK							



Fig. A2. Drilling depth of geothermal projects in reality compared to the model. (cf. columns "Depth reality" and "Depth model" in Table A2 for precise values). For a better analysis of the colours in this figure, please refer to the online version of the paper.



Fig. A3. Total investment of geothermal projects in reality compared to the model. (cf. columns "Total investment reality" and "Total investment model" in Table A2 for precise values). For a better analysis of the colours in this figure, please refer to the online version of the paper. 21

(A1)

(A4)

(A5)

A.4. Further economic assessment

The purchase values for properties C_P in German municipalities are given in Statistisches Bundesamt [117] divided into municipal size classes, i. e. depending on the population in the municipality. A distinction is made here between *developed building land, undeveloped building land and other building land.* Other building land can be developed building land and undeveloped building land, but differs from both in its fixed use to date. It includes industrial land, land for transport and open spaces [118]. As commercial and industrial sites are included in the other building land, it is assumed that the geothermal plants will be built on this kind of land. Fig. A4 shows the cost structure of other building land for different municipality sizes [117].



Fig. A4. Average purchase value for other building land for different municipality sizes [117].

The figure shows that the land prices can differ significantly from the $60 \notin m^2$ assumed in Schlagermann [56]. With the exception of the outlier in municipalities with between 50,000 and 100,000 inhabitants, property costs increase with the number of inhabitants, probably due to the scarcity of free space in municipalities with many inhabitants. However, as Schmalwasser & Brede [119] show, land prices depend more on population density. Therefore, the average population densities are determined for the municipal size classes in Table A3 by using the municipality data from Statistisches Bundesamt [120] and the prices are set for the resulting population density classes.

Table A3

Costs for other building land depending on the number of inhabitants of the municipality.

Population (in thsd.)	< 2	2–5	5-10	10-20	20-50	50-100	100-200	200-500	> 500
Mean population density in municipality [Inhabitants/km²]	80	170	245	380	565	885	1385	1695	2700
Purchase value [€/m²]	21.5	31.0	42.0	50.5	64.0	37.5	103.0	117.5	449.0

Finally, the cost of land and infrastructure C_2 can be estimated using the following equation:

 $C_2 = C_P \cdot (A_D + A_{el} + A_{th})$

$$= C_{F'} \left(2400 \mathrm{m}^{2} \cdot \boldsymbol{b}_{GF} + \left(2000 \mathrm{m}^{2} \cdot \boldsymbol{b}_{PF} + 0.25 \frac{\mathrm{m}^{2}}{\mathrm{kW}_{\mathrm{el}}} \cdot \boldsymbol{P}_{el,max} \right) + \left(100 \mathrm{m}^{2} \cdot \boldsymbol{b}_{DHF} + 0.01 \frac{\mathrm{m}^{2}}{\mathrm{kW}_{\mathrm{th}}} \cdot \boldsymbol{Q}_{gh,max} \right) \right)$$

 A_D is the area required for the well site, A_{el} for the power generation plant and A_{th} for the district heating plant.

The pumps in a low-temperature geothermal plant consume approximately $D_p = 30\%$ of the gross output [47,121]. Therefore, the demand D_{pump} has to be added to the electricity demand D_{el} if a geothermal plant is built during the optimisation in order to determine the total electricity demand $D_{el,total}$

$$(A2)$$

with

$$D_{pump}(t) = D_{p'} \left(P_{el}(t) + \dot{Q}_{gl_1}(t) \cdot \frac{\eta_{el}}{\eta_{el}} \right)$$
(A3)

As a conservative estimate, the assumption is made that the required amount D_{pump} is not yet included in the calculation to determine η_{th} in Ozgener & Ozgener [48]. The nominal power P_{pump} of the pump is determined via Eq. A(4) with a pump efficiency η_{pump} of 0.95.

$P_{pump} \ge D_{pump}(t) \cdot \eta_{pump}$

Eyerer et al. [59] specify the installation depth of the pumps in the production wells for eight German power plants. With the help of the average value of 700 m, the costs for the production well pump C_4 can then be calculated according to Eq. (A5) [56].

$$C_4 = 15.7 \frac{1}{kM_d} P_{pump} + 98, 500 \mathbf{\hat{c}} \cdot b_{GP}$$

The costs for the thermal water circuit C_5 are determined as a function of the volumetric flow rate rate \dot{V}_B , assuming a constant length of the

circuit [122]:

Applied Energy 255 (2019) 113824

(A10)

 $C_5 = 5,000 \frac{\boldsymbol{\epsilon} \cdot \boldsymbol{s}}{1} \cdot \dot{V}_B + 600,000 \boldsymbol{\epsilon} \cdot \boldsymbol{b}_{GP}$ (A6) For the ORC plant and the grid connection, a linear cost function C_{δ} is assumed according to the values from Jancik & Kupfermann [122], Campos Rodríguez et al. [123] and Rubio-Maya et al. [124]: $C_6 = 2, 470 \frac{\epsilon}{kW_{el}} \cdot P_{el,max} + 14, 100 \epsilon \cdot b_{PP}$ (A7) The costs for project management, control and finance planning as well as other capital-related costs such as insurance, seismic monitoring and public relations C_8 are determined according to the following equation [56]: $C_8 = 0.08 \cdot (C_1 + C_2 + C_3 + C_4 + C_5 + C_6 + C_7) + 0.035 \cdot C_3 + 0.005 \cdot (C_4 + C_5 + C_6 + C_7) + 650, 000 \pounds \cdot b_{GP}$ (A8) A.4.1. Total investment in model and reality The total investment is determined for comparison with the real investments for the projects in Table A2. Since the costs for district heating depend on the population density and thus on the individual case of consideration, an equation is derived from [122] for the calculation of the total investment in Table A2: $f(\dot{Q}_{ih,max}) = 815 \epsilon/kW_{th} \cdot \dot{Q}_{ih,max} - 130,000 \epsilon$ (A9) The more accurate investment calculation for the district heating plant from Section 3.5 cannot be used in this comparison but is used in the optimisations. The total investment was determined only for those geothermal plants for which sufficient information was available. By comparing the total investment calculated in the model and in reality, it is noticeable that the total investment is usually overestimated in the model (40% on average). However, a RMSE of 22 M€ is achieved (cf. Fig. A3). This conservative assessment is taken over for the optimisation.

A.4.2. Demand-related costs, operating costs and other variable costs

The yearly costs for working fluid and other demand-related resources C_{DR} are applied in accordance with Schlagermann [56]:

$C_{\text{DP}}(a) = 0.01 \cdot (C_4 + C_5 + C_4 + C_7 + 150, 000 \textbf{C} \cdot \textbf{b}_{\text{CP}})$

The operating costs include yearly labour costs C_L, which are calculated as a function of the thermal output of the thermal water mass flow [59]: $C_L(\boldsymbol{a}) = 220, \ 000 \cdot e^{5 \cdot 10^{-6} \cdot \dot{\boldsymbol{v}}_B \cdot \boldsymbol{\rho}_W \cdot \boldsymbol{c}_{\boldsymbol{p}, W} \cdot (T \boldsymbol{p}_W, max - T \boldsymbol{p}_H \boldsymbol{p}, return)}$ (A11) This equation has to be linearised. However, since it depends on the temperature of the water in the production well T_{PW,max} and the injection temperature T_{IIV} , a value must be set for T_{IIV} . Since the influence of the injection temperature on the costs is not high, 65 °C is set as an estimate for

the safe side. This results in the following equation for labour costs:	
$C_L(a) = 380 \mathbf{\epsilon}^{\circ} C \cdot T_{PW,max} + 205, 500 \mathbf{\epsilon} \cdot \mathbf{b}_{GP}$	(A12)
Other operating costs C _{OP} include costs for remote monitoring and operation of seismic monitoring as well as mainten	ance and repair costs and
other variable costs C_{OV} include costs for insurance and legal assistance:	

$C_{OP}(a) = 61, \ 000 \notin b_{GP} + 1.25 \cdot 0.005 \cdot C_D + 0.03 \cdot (C_4 + C_5 + C_6 + C_7)$	(A13)
$C_{OV}(a) = 0.006 \cdot (C_4 + C_5 + C_6 + C_7) + 115, \ 900 \textbf{\ell} \cdot \textbf{b}_{GP}$	(A14)

Appendix B. Supplementary material

Supplementary data to this article can be found online at https://doi.org/10.1016/j.apenergy.2019.113824.

References

- McKenna R. The double-edged sword of decentralized energy autonomy. Energy Policy 2018;113:747-50. https://doi.org/10.1016/j.enpol.2017.11.033.
 trend:research. Eigentümerstruktur: Emeuerbare Energien. Entwicklung der Atteuryvieliah, Rolle der Energieversorger, Ausblick bis 2020 (Ownership struc-ture: Renewable energies. Diversity of actors, role of energy suppliers, outlook until 2020), 2017. Available online at https://www.trendresearch.de/studie.php?
- s=672, checked on 5/9/2018.[3] Pieter Boon Frank, Carel Dieperink. Local civil society based renewable energy organisations in the Netherlands. Exploring the factors that stimulate their emergence and development. Energy Policy 2014;69:297–307. https://doi.org/10. 1016/j.enpol.2014.01.046.
- [4] Engelken Maximilian, Römer Benedikt, Drescher Marcus, Welpe Isabell. Why homeowners strive for energy self-supply and how policy makers can influence them. Energy Policy 2018;117:423-33. https://doi.org/10.1016/j.enpol.2018.02.
- U26.
 [5] McKenna Russell, Merkel Erik, Fichtner Wolf. Energy autonomy in residential buildings. A techno-economic model-based analysis of the scale effects. Appl Energy 2017;189:800–15. https://doi.org/10.1016/j.apenergy.2016.03.062.
 [6] Müller Mathias Otto, Stämpfi Adrian, Dold Ursula, Hammer Thomas. Energy autarky. A conceptual framework for sustainable regional development. Energy

- Policy 2011;39(10):5800–10. https://doi.org/10.1016/j.enpol.2011.04.019.
 [7] Balcombe Paul, Rigby Dan, Azapagic Adisa. Energy self-sufficiency, grid demand variability and consumer costs. Integrating solar PV, Stirling engine CHP and battery storage. Appl Energy 2015;155:393–408. https://doi.org/10.1016/j. apenergy.2015.06.017
- apenergy. 2015.06.017.
 [8] Prasad Koirala Binod, Elta Koliou, Friege Jonas, Hakvoort Rudi A, Herder Paulien M. Energetic communities for community energy. A review of key issues and trends shaping integrated community energy systems. Renew Sustain Energy Rev 2016;56:722–44. https://doi.org/10.1016/j.nser.2015.11.080.
 [9] Russell McKenna, Carsten Herbes, Wolf Fichtner. Energieautatkie Vorschlag einer
- [9] Russell McKenna, Carsten Herbes, Wolf Flehtner, Energieautarkie Vorschlag einen Arbeitstekefinition als Grundlage für die Bewertung konkreter Projekte und Szenarien (Energy autonomy. Proposal of a working definition as a basis for the evaluation of concrete projects and scenarios). Z Energiewirtsch 2015;39(4):235-52. https://doi.org/10.1007/s12398-015-0164-1.
 [10] Engelken Maximilian, Römer Benedikt, Drescher Marcus, Welpe Isabell.
- Transforming the energy system. Why municipalities strive for energy self-sufficiency. Energy Policy 2016;98:365-77. https://doi.org/10.1016/j.enpol.2016.07.
- 049. [11] Alt B. Norddeutsche Gemeinde stellt komplett auf erneuerbare Energien um (Northern German community switches completely to renewable energies); 2017. Available online at http://www.sonnenseite.com/de/energie/norddeutsche-gemeinde-stellt-komplett-auf-erneuerbare-energien-um.html, checked on 5/9/ 2018.

23

J.M. Weinand, et al

- [12] Mundaca Luis, Busch Henner, Schwer Sophie. 'Successful' low-carbon energy transitions at the community level? An energy justice perspective. Appl Energy /doi.org/10.1016/i.apener 2018 02 14
- 2018;218:292–303. https://doi.org/10.1016/j.apenergy.2018.02.146. Jågenann (c, Hagspiel S, Lindenberger D. The economic inefficiency of grid parity: the case of German photovoltaics. EWI Working Paper No 13/19, December 2013; [13] 2013
- 2013. Statistisches Bundesamt. Bruttostromerzeugung in Deutschland für 2015 bis 2017 (Gross electricity generation in Germany for 2015 to 2017); 2017a. Available online at https://www.destatis.de/DE/ZahlenFakten/Wirtschaftsbereiche/ Energie/Enzeugung/Tabellen/Bruttostromerzeugung.htm [checked on S/9/2018]. [14]
- [15] UBA. Enception of the state of the s le/sites/default/files/medien/376/publikationen/180315_uba_hg_eeinzahlen_ 2018_bf.pdf, checked on 6/28/2018.
- [16] BMWL Enreuerbare Energien in Zahlen. Nationale und internationale Entwicklung im Jahr 2016 (Renewable energies in figures. National and International devel-opment in 2016); 2017. Available online at https://www.bmwLeR.edaktion/DE/ Publikationen/Energie/eneuerbare-energien-in-zahlen-2016.pdf?_blob= publicationFile&v=8, checked on 5/9/2018.
- [17] Bertani Ruggero. Geothermal power generation in the world 2010–2014 update report. Geothermics 2016;60:31–43. https://doi.org/10.1016/j.geothermics.2015
- Agemar T, Weber J, Moeck I. Assessment and public reporting of geothermal re-sources in Germany. Review and outlook. In Energies 2018;11(2):332. https://doi. org/10.3390/en11020332. [18]
- [19] Weber J. Ganz B, Samer B, Moeck I. Geothermal energy use, country update for Germany. Strasbourg, France (European geothermal congress 2016); 2016. Available online at https://www.geothermal-energy.org/pdf/GAstandard/EGC/ 2016/GERMANY%20country%20update%20EGC%202016.pdf, checked on 6/28/
- [20] Jain Charitra, Vogt Christian, Clauser Christoph. Maximum potential for geo
- [21]
- https://www.iniweiroundesanit.de/dateir/energie/stronerzeuguig-endedraf-konventionell#textpart-1, checked on 4/6/2018.
 [22] Purkus Alexandra, Barth Volker. Geothermal power production in future elec-tricity markets—a scenario analysis for Germany. Energy Policy 2011;39(1):349–57. https://doi.org/10.1016/j.enpol.2010.10.003.
 [23] Scheffer K. Vom Bioenergiedorf zur autonomen Solarenergieregion (From a bioe-nergy village to an autonomous solar energy region). Solarzeitalter 2008;20(4):23–30. checked on 11/2/2017. 20(4):23-30. checked on 11/2/2017
- [24] Schmidt J, Schönhart M, Biberacher M, Guggenberger T, Hausl S, Kalt G, et al. Regional energy autarky. Potentials, costs and consequences for an Austrian re gion. Energy Policy 2012;47:211–21. https://doi.org/10.1016/j.enpol.2012.04.
- [25] Peura Pekka, Haapanen Ari, Reini Kaarina, Törmä Hannu. Regional impacts of sustainable energy in western Finland. J Cleaner Prod 2018;187:85–97. https:// loi.org/10.1016/j.jclepro.2018.03.194
- Burgess Paul J, Rivas Casado Monica, Gavu Jerry, Mead Andrew, Cockerill Tim, Lord Richard, van der Horst Dan, Howard David C. A framework for reviewing the trade-offs between, renewable energy, food, feed and wood production at a local level. Renew Sustain Energy Rev 2012;16(1):129–42. https://doi.org/10.1016/j. [26] er.2011.07.1
- [27] Peter Stefan. Modellierung einer vollständig auf erneuerbaren Energien basier recei avocali avocanicatano chia voltavanta dai cancorrataria Strukturen enden Stromerzeugang im Jahr 2050 in uatarken, dezentralen Strukturen (Modelling of electricity generation based entirely on renewable energies in 2050 in self-sufficient, decentralised structures). Edited by Unweltbundessamt; 2013. Jenssen Till, König Andreas, Eltrop Ludger. Bioenergy villages in Germany.
- [28]
- F291
- Jenssen Titl, König Andreas, Eltrop Ludger. Bioenergy villages in Germany. Bringing a low carbon energy supply for rural areas into practice. Renew Energy 2014;61:74-80. https://doi.org/10.1016/j.renene.2012.08.014. Woyke Wolfgang, Forero Mario. Methoden zur Bewertung regionaler Energieautarkie (Methods for assessing regional energy self-sufficiency). Beitrag auf dem 13. Symposium Energiefinnovation. Graz, Österreich; 2014. van Erdeweghe Sarah, van Bael Johan, Laenen Ben, Dhaeseleer William. Optimal combined heat-and-power plant for a low-temperature geothermal source. Energy 2018;150:396-409. https://doi.org/10.1016/j.energy.2018.01.136. [30]
- F311 Marty Fabien, Serra Sylvain, Sochard Sabine, Reneaume Jean-Michel, Simultaneous optimization of the district heating network topology and the Organic Rankine Cycle sizing of a geothermal plant. Energy 2018. https://do
- 10.1016/j.energy.2018.05.110.
 Østergaard Poul Alberg, Mathiesen Brian Vad, Möller Bernd, Lund Henrik. A re-
- newable energy scenario for Aalborg Municipality based on low-temperature geothermal heat, wind power and biomass. Energy 2010;35(12):4892–901. ro/10.1016/i.en 2010/08/041 [33]
- https://doi.org/10.1016/j.eiergy.2010.08.091. Østergaard Poul Alberg, Lund Henrik. A renewable energy system in Frederikshavn using low-temperature geothermal energy for district heating. Appl Energy 2011;88(2):470–87. https://doi.org/10.1016/j.apenergy.2010.03.018. Sveinbjörnsson Dadi, Ben Amer-Allam Sara, Hansen Anders Bawnhøj, Agren Loui, [34]
- Svemiojomsson badi, per Anter-Aulam Sara, nansen Anders Bavning), Aigren Lou, Pedersen Allan Schröder. Energy supply modelling of a low-CO 2 emitting energy system. Case study of a Danish municipality. Appl Energy 2017;195:922–41. https://doi.org/10.1016/j.apenergy.2017.03.086 Mainzer K. Analyse und Optimierung urbaner Energiesysteme: Entwicklung und Anwendung eines übertragbaren Modellierungswerkzeugs zur nachhaltigen
- [35]

Systemgestaltung, (Analysis and optimisation of urban energy systems: development and application of a transferable modelling tool for sustainable system de-sign). Dissertation. Karlsruhe Institute of Technology. Institute for Industrial

- sign). Dissertation. Karlsruhe institute of Technology. Institute for Industrial Production (IIP); 2019 [in press].
 IWU (Ed.). Deutsche Wohngebäudetypologie. Beispielhafte Maßnahmen zur Verbesserung der Energieeffizienz von typischen Wohngebäuden (zweite erweivertices and user the get muscle is volicity building typology. Exemplary measures to improve the energy efficiency of typical domestic buildings). With assistance of Tobias Loga, Britta Stein, Nikolaus Diefenbach, Rolf Born, IWU, Darmstadt; 2015 [accessed 21 January 2016].
 [37] Petrakopoulou Fontina, Robinson Alexander, Loizidou Maria. Simulation and evaluation of a hybrid concentrating-solar and wind power plant for energy autonomy on islands. Renew Energy 2016;96:863–71. https://doi.org/10.1016/j.renewer.2016.05.030
- 2016/05/030
- [38] McKenna R, Bertsch V, Mainzer K, Fichtner W. Combining local preferences with eterence feasible energy multi-criteria decision analysis and linear optimization to develop feasible er concepts in small communities. Eur J Oper Res 2018;268(3):1092–110. http doi.ore/10.1016/i.eior.2018.01.036.
- [39] OpenStreetMap contributors. OpenStreetMap; 2018. Available online at https:// map.org/#n 5/51.330/1 453, checked on 1/29/2018
- [40] Mainzer Kai, Killinger Sven, McKenna Russell, Fichtner Wolf, Assessment of manizer kan, knimger sven, wickenna kussen, Fichther wohr Assessmen rooftop photovoltaic potentials at the urban level using publicly available and image recognition techniques. Sol Energy 2017;155:561–73. https:// 10.1016/i.solener.2017.06.065 geodata
- Agemar Thorsten, Weber Josef, Schulz Rüdiger. Deep geothermal energy pro-duction in Germany. Energies 2014;7(7):4397–416. https://doi.org/10.3390/
- (42) Bauer Mathias, Freeden Willi, JacobiHans, Neu Thomas, 2014. Handbuch Tiefe Geothermie. Prospektion, Exploration, Realisierung, Nutzung (Handbook deep geothermal energy: Prospecting, exploration, realization, use). Berlin, Heidelberg: Springer Spektrum (SpringerLink Bücher). http://doi.org/10.1007/978-3-642-
- [43] Agemar Thorsten, Schellschmidt Rüdiger, Schulz Rüdiger. Subsurface temperature . 1.1016/i distribution in Germany. Geothermics 2012;44:65-77. https://doi.org/10 geothermics.2012.07.002.
- geouenines.2012.07.002.
 [44] Frey M, Milles U. Geothermische Stromerzeugung in Landau (Geothermal power generation in Landau). Edited by FIZ Karlsruhe; 2007. Available online at http:// www.bine.info/fileadmin/content/Publikationen/Projekt-Infos/2007/Projekt-Info_14-2007/projekt_1407Internet-x.pdf, checked on 1/16/2018.
- Info 14-2007/projekt 1407/Internet x.pdf, checked on 1/16/2018.
 [45] Rohloff K, Kather A. Geothermische Stromerzeugung. Kraftwerkstechnologien und Technologien zur gekoppelten Erzeugung von Strom und Wärne (Geothermal power generation. Power plant technologies and technologies for combined gen-eration of electricity and heat). Edited by Bundesministerium für Umwelt, Naturschutz und Reaktorsicherheit; 2011. Available online at https://www.bgr. bund.de/MAGS/DE/Downloads/BMU Strom.pdf? blob=publicationFile&v=1 checked on 4/10/2018.
- Generation (1997) (2010) (2 015/15022.pdf [checked on 4/3/2018].
- [48] Ozgener Leyla, Ozgener Onder. Monitoring of energy exergy efficiencies and ex-Orgener Leyia, Ozgener Onder, Monitoring of energy exergy emicencies and ex-ergeoconomic parameters of geothermal district heating systems (GDHSs). Appl Energy 2009;86(9):1704–11. https://doi.org/10.1016/j.apenergy.2008.11.017. Zarrouk Sadiq J, Moon Hyungsul. Efficiency of geothermal power plants. A worldwide review. Geothermics 2014;51:142–53. https://doi.org/10.1016/j. geothermics.2013.11.001. Enderborence D. Baudenes T, Greanth H, M. Kluppel D, Ontincington et calm. [49] Zarr
- [50] Lindenberger D, Bruckner T, Groscurth H-M, Kümmel R. Optimization of solar
- [30] Endenberger D, Bickkier J, Hoschurt F-W, Knimer K Optimization of solar district heating systems. Seasonal storage, heat pumps, and cogeneration. Energy 2000;25(7):501–608. https://doi.org/10.1016/S0360-5442(99)00082-1.
 [51] Elmegaard Brian, Ommen Torben Schmidt, Markussen Michael, Iversen Johnny. Integration of space heating and hot water supply in low temperature district heating. Energy Build 2016;124:255–64. https://doi.org/10.1016/j.enbuild.2015. 00.002
- [52] Griva Igor, Nash Stephen G, Sofer Ariela. Linear and nonlinear optimization. 2nd
- Grivangor, Nash Sepheri V, Sorer Artezi. Linear and nonimizer optimization. 2nd ed. Philadelphia, PA: Society for Industrial and Applied Mathematics (OT/SIAM, Society of Industrial and Applied Mathematics, 108); 2009. Budisulistyo Denny, Wong Choon Seng, Krumdleck Susan. Lifetime design strategy for binary geothermal plants considering degradation of geothermal resource productivity. Energy Convers Manage 2017;132:1–13. https://doi.org/10.1016/j. Energyment 2016.10.007 [53]
- [54] Clauser Christoph, Ewert Markus. The renewables cost challenge. Levelized cost of geothermal electric energy compared to other sources of primary energy – Review and case study. Renew Sustain Energy Rev 2018;82:3683–93. https://doi.org/10
- 1016/j.rser.2017.10.095.
 [55] Zeitungsverlag tz. Erkwärme-Projekt: Viel zu teuer gebohrt (Geothermal energy project: drilled far too expensively). Edited by Zeitungsverlag tz München GmbH & Co. KG; 2014. Available online at https://www.tz.de/muenchen/stadt/ erdwaerme-projekt-viel-teuer-gebohrt-tz-3701680.html [checked on 3/26/2018].
- [56] Schlagermann Pascal. Exergo@konomische Analyse geothernischer Strombereitstellung am Beispiel des Oberrheingrabens (Exergoeconomic analysis of geothermal power supply using the Upper Rhine Graben as an example).

München: Dr. Hut.; 2014.

- [57] Stober Ingrid, Bucher Kurt, Geothermische Nutzungsmöglichkeiten (Geothermal
- Stober Ingrid, Bucher Kurt, Geöthermische Nutzungsmöglichkeiten (Geöthermal applications). In: Ingrid Stober, Kurt Emil Bucher, editiors, Geöthermie. 2., überarb. u. aktualisierte Aufl. 2014; 2014. Berlin: Springer Spektrum, pp. 37–64. Leipziger Institut für Energie, 2007. Tiefe Geöthermie in Deutschland (Deep geo-thermal energy in Germany). Edited by Bundesministerium für Umwelt, Naturschutz und Reaktorsicherheit. Available online at https://www.mags-projekt.de/MAGS/DE/Downloads/BMU_TiefeGeothermie.pdf?_blob = [58]
- publicationFile&v=1 [checked on 1/18/2018].
 [59] Eyerer Sebastian, Schifflechner Christopher, Hofbauer Sebastian, Wieland Christoph, Zosseder Kai, Bauer Wolfgang et al. Potential der hydrothermalen Geothermie zur Stromerzeugung in Deutschland (Potential of hydrothermalen thermal energy for power generation in Germany). Bayerisches Staatsministerium für Bildung und Kultus, Wissenschaft und Kunst; 2017, Available online at https:// www.es.mw.tum.de/illeadmin/w00bhq/www/pdf/Potential_der_ hydrothermalen_Geothermie_zur_Stromerzeugung_in_Deutschland.pdf, checked on ñ /2 /2017
- [60] Weinand Jann Michael, Kleinebrahm Max, McKenna Russell, Mainzer Kai, Fichtner Wolf. Developing a combinatorial optimisation approach to design district heating networks based on deep geothermal energy. Appl Energy 2019;251. https://do rg/10.1016/j.apenergy.2019.11 rticle/pii/S0306261919310414 019.113367. https
- Aurenhammer Franz. Voronoi diagrams–a survey of a fundamental geometri data structure. ACM Comput Surv 1991;23(3):345–405. https://doi.org/10.11
- IRENA. Renewable power generation costs in 2017. Edited by International [62] Renewable Energy Agency. Abu Dhabi; 2018. Available online at htt irena.org/-/media/Files/IRENA/Agency/Publication/2018/Jan/IREN n/2018/Jan/IRENA 2017
- Power Loss 2015, D15, Checked on 9/57/2015. Weinand Jam Michael, McKenna Russell, Fichtner Wolf. Developing a munici-pality typology for modelling decentralised energy systems. Utilities Policy 2019;57:75-96. https://doi.org/10.1016/j.jup.2019.02.003. Gunnlaugsson E, Armanusson H, Thorhallsson S, Steingrimsson B. Problems In [63]
- [64] [64] Gunnlagsson E, Armannsson H, Informatisson S, Steingrimsson B. Proteims in Geothermal operation - scaling and corrosion. Santa Tecla, El Salvador (Short Course VI on Utilization of Low- and Medium-Enthalpy Geothermal Resources and Financial), 2014. Available online at https://orkustoinun.is/gogn/unu-gtp-sc/ UNU-GTP-Sc-18-10-pdf [checked on 3/9/2018].
 [65] Zarrouk Sadiq J, Woodhurst Blair C, Morris Chris. Silica scaling in geothermal heat
- exchangers and its impact on pressure drop and performance. Wairakei binary plant, New Zealand. Geothermics 2014;51:445-59. https://doi.org/10.1016/j.
- Magri Fabien, Bayer Ulf, Pekdeger Asaf, Otto Roland, Thomsen Claudia, Maiwald Ulrike. Salty groundwater flow in the shallow and deep aquifer systems of the Schleswig-Holstein area (North German Basin). Tectonophysics [66]
- 2009;470(1-2):183-94. https://doi.org/10.1016/j.tecto.2008.04.019. Baasch R. Pullach schlägt neues Kapitel in der Geothermiegeschichte auf (Pullach [67] ens new chapter in geothermal history). Geothermische Energie 11;1(70):10-
- Majorowicz Jacek, Grasby Stephen E. High potential regions for enhanced geo-thermal systems in Canada. Nat Resour Res 2010;19(3):177–88. https://doi.org [68] .1007/s11053-010-9119-8.
- Jong Fagning, Chen Jilang, Huang Wenbo, Luo Liang. A three-dimensional transient model for EGS subsurface thermo-hydraulic process. Energy 2014;7:300-10. https://doi.org/10.1016/j.energy.2014.05.038.
 Aali A, Pourmahmoud N, Zare V. Exergoeconomic analysis and multi-objective
- polimization of a novel combined flash-binary cycle for Sabalan geothermal power plant in Iran. Energy Convers Manage 2017;143:377–90. https://doi.org/10. 1016/j.enconman.2017.04.025.
- [71] Astolff Marco, Romano Matteo C, Bombarda Paola, Macchi Ennio, Binary ORC [71] Astoin Warco, komano Watteo C, boimbarca Paola, Maccin Emilio, Binary Orc, (organic Rankine cycles) power plants for the exploitation of medium-low tem-perature geothermal sources – Part A. Thermodynamic optimization. Energy 2014;66:423–34. https://doi.org/10.1016/j.energy.2013.11.056.
 [72] Chagnon-Lessard Noémie, Mathieu-Povin François, Gosselin Louis. Geothermal
- bower plants with maximized specific power output. Optimal working fluid and operating conditions of subcritical and transcritical Organic Rankine Cycles. Geothermics 2016;64:111–24. https://doi.org/10.1016/j.geothermics.2016.04.
- [73] Clarke Joshua, McLay Laura, McLeskey James T. Comparison of genetic algorithm to particle swarm for constrained simulation-based optimization of a geothermal wer plant. Adv Eng Inf 2014;28(1):81-90. https://doi.org/10.1016/j.aei.2013
- [74] Clarke Joshua, McLeskey James T. Multi-objective particle swarm optimization o binary geothermal power plants. Appl Energy 2015;138:302-14. https://doi.org
- 10.1016/j.apenergy.2014.10.072.[75] Ghaebi Hadi, Namin Amin Shekari, Rostamzadeh Hadi. Exergoeconomic optimication of a novel cascade Kalina/Kalina cycle using geothermal heat source and LNG cold energy recovery. J Cleaner Prod 2018. https://doi.org/10.1016/j. telence.2019.04.040
- [76] Ghasem Hadi, Paci Marco, Tizzanini Alessio, Mitsos Alexander. Modeling and optimization of a binary geothermal power plant. Energy 2013;50:412–28. https://doi.org/10.1016/j.energy.2012.10.039.
 [77] Guzović Z, Majcen B, Cvetković S. Possibilities of electricity generation in the
- Republic of Croatia from methods in respective science of exercicly generation in une Republic of Croatia from methods une generature genethemal sources. Appl Energy 2012;98:404–14. https://doi.org/10.1016/j.apenergy.2012.03.064. Huster Wolfgang R, Bongartz Dominik, Mirisos Alexander. Deterministic global optimization of the design of a geothermal organic Rankine cycle. Energy Procedia
- [78]

Applied Energy 255 (2019) 113824

2017;129:50-7. https://doi.org/10.1016/j.egypro.2017.09.181.

- Zu17;129:50-7. https://doi.org/10.1016/j.egypr0.2017.09.181.
 Karimi Shahram, Mansouri Sima. A comparative profitability study of geothermal electricity production in developed and developing countries. Exergoeconomic analysis and optimization of different ORC configurations. Renew Energy 2018;115:600-19. https://doi.org/10.1016/j.renee.2017.08.098.
 Kolahi Mohammad-Reza, Nemati Arash, Yari Mortaza. Performance optimization
- and improvement of a flash-binary geothermal power plant using zeotropic mix-tures with PSO algorithm. Geothermics 2018;74:45-56. https://doi.org/10.1016/ 2018 02 004
- [31] Lazzarin Nicolò, Zanellato Luca, Frassinetti Marco. Cycle and turbine re-optimi-zation on geothermal resources significantly deviating from the expected condi-tions. Energy Procedia 2017;129:615–20. https://doi.org/10.1016/j.egypro.2017
- [82] Liu O, Shen A, Duan Y, Parametric optimization and performance analyses of geothermal organic Rankine cycles using R600a/R601a mixtures as working fluids. Appl Energy 2015;148:410–20. https://doi.org/10.1016/j.apenergy.2015.
- al-fossil power [83] Liu Q, Shang L, Duan Y. Performance analyses of a hybrid geothern generation system using low-enthalpy geothermal resources. Appl Energy 2016;162:149-62. https://doi.org/10.1016/j.apenergy.2015.10.078.
- 2016;16:2:149-62. https://doi.org/10.1016/j.apenergy.2015.10.078.
 [84] Liu Xiaonin, Wei Ming, Yang Luona, Wang Xing, Thermo-economic analysis and optimization selection of ORC system configurations for low temperature binary-cycle geothermal plant. Appl Therm Eng 2017;125:155-64. https://doi.org/10.1016/j.applthermaleng.2017.07.016.
 [85] Lu X, Zhao Y, Zhu J, Zhang W. Optimization and applicability of compound power
- cycles for enhanced geothermal systems. Appl Energy 2018;229:128-41. htt
- doi.org/10.1016/j.apenergy.2018.07.033.
 Makhanlall Deodat, Zhang Fuzzken, Xu Ruina, Jiang Pekue. Exergy-topological analysis and optimization of a binary power plant utilizing medium grade geo-thermal energy. Appl Them Eng 2015;88:459–63. https://doi.org/10.1016/j. applthermaleng.2014.09.017. [87] Martínez-Gomez Juan, Peña-Lamas Javier, Martín Mariano, Ponce-Ortega José
- Maria. A multi-objective optimization approach for the selection of working fluids of geothermal facilities. Economic, environmental and social aspects. J Environ Manage 2017;203(Pt 3):962–72. https://doi.org/10.1016/j.jenvman.2017.07.
- Mohammadzadeh Bina Sacid, Jalilinasrabady Sacid, Fujii Hikari. Thermo-eco-nomic evaluation of various bottoming ORCs for geothermal power plant, de-termination of optimum cycle for Sabalan power plant exhaust. Geothermics 2017;70:181–91. https://doi.org/10.1016/j.geothermics.2017.06.007.
 Mohammadzadeh Bina Sacid, Jalilinasrabady Sacid, Fujii Hikari. Exergoeconomic analysis and optimization of single and double flash cycles for Sabalan geothermal power plant. Geothermics 2018;72:74–82. https://doi.org/10.1016/j.geothermics. 2017 10.013
- [90] Mosaffa AH, Mokarram N Hasani, Farshi L Garousi. Thermo-economic analysis of combined different ORCs geothermal power plants and LNG cold energy. Geothermics 2017;65:113–25. https://doi.org/10.1016/j.geothermics.2016.09
- [91] Nakomcic-Smaragdakis Branka, Dvornic Tijana, Cepic Zoran, Dragutinovic Natasa. Analysis and possible geothermal energy utilization in a municipality of Panonian Basin of Serbia. Renew Sustain Energy Rev 2016;59:940–51. https://doi.org/10. 1016/j.rser.2015.12.337.
- [92] Pambudi Nugroho Agung, Itoi Ryuichi, Jalilinasrabady Saeid, Jaelani Khasani. Exergy analysis and optimization of Dieng single-flash geothermal power plant. Energy Convers Manage 2014;78:405–11. https://doi.org/10.1016/j.enconman.
- [93] Pambudi Nugroho Agung, Itoi Ryuichi, Jalilinasrabady Saeid, Gürtürk Mert Sustainability of geothermal power plant combined with thermodynamic and silica scaling model. Geothermics 2018;71:108–17. https://doi.org/10.1016/j. nics.2017.09.003.
- [94] Peña-Lamas Javier, Martinez-Gomez Juan, Martín Mariano, María Ponce-Ortega José. Optimal production of power from mid-temperature geothermal sources. Scale and safety issues. Energy Convers Manage 2018;165:172–82. https://doi. 2018.03.048
- [95] Pollet Mathieu, Goselin Louis, Dallaire Jonathan, Mathieu-Potvin François.
 Optimization of geothermal power plant design for evolving operating conditions. Appl Therm Eng 2018;134:118–29. https://doi.org/10.1016/j.applthermaleng.
- [96] Shengjun Z, Huaixin W, Tao G. Performance comparison and parametric optimi-[96] Shengjini Z, muaxini W, Tao G. Performance comparison and parametric optimi-zation of subcritical Organic Rankine Cycle (ORC) and transcritical power cycle system for low-temperature geothermal power generation. Appl Energy 2011;88(8):2740–54. https://doi.org/10.1016/j.genergy.2011.02.034.
 [97] Sigurdardottir Silja R, Valfells Agust, Palsson Halldor, Stefansson Hlynur. Mixed integer optimization model for utilizing a geothermal reservoir. Geothermics 2015;55:171–81. https://doi.org/10.1016/j.geothermics.2015.01.006.
 [98] Sun Qingxuan, Wang Yaxiong, Cheng Ziyang, Wang Jiangfeng, Zhao Pan, Dai Vining. Dearmodynaptic entruingating of double preservo. Geomic Dendito Colle
- Virjing, Thermodynamic optimization of a double-pressure Organic Rankine Cycle Driven by geothermal heat source. Energy Procedia 2017;129:591-8. https://doi. org/10.1016/tecvpro.2017.09.214.
- org/10.1016/j.egypro.2017.09.214. [99] Sun Jie, Liu Qiang, Duan Yuanyuan. Effects of evaporator pinch point temperature difference on thermo-economic performance of geothermal organic Rankine cycle systems. Geothermics 2018;75:249–58. https://doi.org/10.1016/j.geothermics. 018.06.001
- [100] van Edeweghe S, van Bael J, Laenen B, Dhaeseleer W. Design and off-design optimization procedure for low-temperature geothermal organic Rankine cycles. Appl Energy 2019;242:716–31. https://doi.org/10.1016/j.apenergy.2019.03.142.

- [101] Walraven D, Laenen B, D'haeseleer W. Minimizing the levelized cost of electricity production from low-temperature geothermal heat sources with ORCs: Water or air cooled? Appl Energy 2015;142:144–53. https://doi.org/10.1016/j.apenergy.
- [102] Wu Chuang, Wang Shun-sen, Jiang Xihang, Li Jun. Thermodynamic analysis and [102] Wu Chuang, Wang Shun-sen, Jiang Athang, Li Jun. Thermodynamic analysis and performance optimization of transcritical power cycles using CO 2 - based binary zeotropic mixtures as working fluids for geothermal power plants. Appl Therm Eng 2017;115:292-304. https://doi.org/10.1016/j.applthermaleng.2016.12.077.
 [103] Yang Min-Hsiang, Yeh Rong-Hua. Economic performances optimization of the transcritical Rankine cycle systems in geothermal application. Energy Convers Manage 2015;95:20-31. https://doi.org/10.1016/j.enconman.2015.02.021.
 [104] Yilmaz Ceyhun. Thermoeconomic modeling and optimization of a hydrogen pro-duction system using seathermal energy. Geothermics 2017;65:32-43. https://doi.org/10.1016/j.
- duction system using geothermal energy. Geothermics 2017;65:32-43. https:// doi.org/10.1016/j.geothermics.2016.08.008.
- doi.org/10.1016/J.gcothermics.2016.08.008.
 [105] Zare V. A comparative exergoeconomic analysis of different ORC configurations for binary geothermal power plants. Energy Convers Manage 2015;105:127–38. https://doi.org/10.1016/j.enconman.2015.07.073.
 [106] Zhao Yajing, Wang Jiangéng, Exergoeconomic analysis and optimization of a flash-binary geothermal power system. Appl Energy 2016;179:159–70. https://
- doi.org/10.1016/j.apenergy.2016.06.108. [107] Enerchange. Projekte Tiefe Geothermie (Projects Deep Geothermal Energy); 2018. vailable online at http:// ww.tiefege
- 2018. [108] Schallenberg K. Vergleich des energiekonzeptionellen Aufbaus von drei geother-mischen Heizzentralen (Comparison of the energy conceptual structure of three geothermal heating plants). Edited by Geoforschungszentrum Potsdam (Geothermie-Report 96-1), 1996. Available online at http://gitpublic.gfz-potsdam.de/pubman/item/escidoc8432:7/component/escidoc10033/9608.pdf
- [Internet and Acceleration of the second of t communicationsecter interest and animology der Ork-Anlager ("downenna-energy Groß Schönebeck. Preparation of the communication experiment and in-tegration of the ORC systems). Edited by Geothermie Neubrandenburg GmbH; 2012a. Available online at http://www.gtn-online.de/media/PDF/GTN_Tiefe_ Geothermie_Kommunikationsexperiment_Gro%C3%9F_Schonebeck_052012.pdf, checked on 3/20/2018.
 [110] Bundesverband Geothermie. Tiefe Geothermieprojekte in Deutschland (Deep exectbermel paraiett in Germany); 2018. Auriliable apiling at http://unite.
- geothermal projects in Germany); 2018. Available online at http://w geothermal.dc/fileadmin/useruploads/wissenswelt/Projektliste_Tiefe_ Geothermie_2018.pdf, checked on 3/20/2018.
- [11] GTN. Geothermisches Heizwerk Neustadt-Glewe (Geothermal heating plant Neustadt-Glewe). Edited by Geothermie Neubrandenburg GmbH; 2012b. Available online at http://www.gtn-online.de/media/PDF/GTN_Tlefe_ Geothermic_Geothermisches_Heizwerk, Neustadt, Glewe, 052012.pdf, checked on /2018
- [112] Enerchange. Sauberes Jubiläum: 15 Jahre Geothermie in Neustadt-Glewe (Clean aniversary: 15 years of geothermal energy in Neustadt-Gleweb; 2009. Available online at http://www.tiefegeothermie.de/top-themen/suberes-jubilaeum-15-jahre-geothermie-in-neustadt-gleweb, checked on 3/20/2018.
 [113] Erdwärme Grünwald GmbH. Erdwärme Grünwald: Hinter dem Mehr an Kosten
- Extention of the structure of the struct
- [114] Goothermie Unterhaching. Datenblatt: Geothermie Unterhaching (Data sheet: Geothermal Unterhaching); 2017. Available online at https://www.geotherm. unterhaching.de/cms/geothermie/web.nsf/gfx/ 0189A5952BAEB5E8C1258183003FA71A/\$file/Datenblatt%20Geothermie
 - %20Unterhaching%202017.pdf [checked on 3/20/2018].

Applied Energy 255 (2019) 113824

- [115] OVB24 GmbH. Geothermie: Startklar für Anfang September (Geothermal energy: Ready for the beginning of September); 2009. Available online at https: nsalzach24.de/innsalzach/region-waldkraiburg/geothermie-projekt-aldkraiburg-startklar-anfang-september-is24-388190.html [checked on 3/20/ innsalzach24 de/inn 2018]
- [116] Bundesverband Geothermie. Neue Projektliste: Tiefe Geothermie schreitet voran (New project list: Deep geothermal energy is progressing); 2013. Available online at http://www.geothermie.de/news-anzeigen/2013/07/02/neue-projektliste-. eothermi schreitet-voran.html. checked on 3/20/2018.
- [117] Statistisches Bundesamt, Kauffälle, Veräußerte Fläche, Durötschnittlicher Kaufwert für Bauland: Deutschland, Jahre, Baulandarten, Gemeindegrößenklassen (Purchase cases, area sold, average purchase value for building land: Germany, years, building land types, municipal size classes); 2016a. Available online at https https://www-genesis.destatis.de/genesis/online/logon?language = de&sequenz tabelleErgebnis&selectionname = 61511-0005 [checked on 3/15/2018].
- [118] Statistisches Bundesamt. Perise. Kaufwerte für Bauland (Prices. Purchase values for building land); 2016b. Available online at https://www.destatis.de/DE/ Publikationen/Thematisch/Preise/Baupreise/ KaufwerteBaulandVj2170500163234.pdf?_blob = publicationFile [checked on 3/
- 15/2018]. [119] Schmalwasser O. Brede S. Grund und Boden als Bestandteil der volkswirtschaf-Sumanwaser O, Breue's, offind und boder als bestatuter der volkswirtschart füchen Vermögensbilanzen (Land as a component of the economic balance sheets of assets). Edited by Statistisches Bundesamt (WISTA); 2015. Available online at https://www.lestatis.de/DE/Publikationen/WirtschaftStatistik/2015/O6/GnundBodenBestandteilVermoegensbilanzen_062015.pdf?_blob=publicationFile [checked on 4/16/2018].
- [120] Satistisches Bundesamt. Gemeindeverzeichnis Gebietsstand: 30.06.2017 (2. Quartal). Alle politisch selbständigen Gemeinden (mit Gemeindeverband) in Deutschland nach Fläche, Bevölkerung (auf Grundlage des Zensus 2011), Bevölkerungsdichte und der Postleitzahl des Verwaltungssitzes der Gemeinde (Municipality directory Area status: 30.06.2017 (2nd quarter). All politically independent municipalities (with municipal association) in Gemany by area, penultation (an the besie of the 2011, aprave). Desvultation dessitier und abs portee population (on the basis of the 2011 census), population density and the postcode of the administrative seat of the municipality). Edited by Statistische Ameted Bundes und der Länder. Wiesbaden; 2017b. Available online at https://www. destatis.de/DE/ZahlenFakten/LaenderRegionen/Regionales/ Gemeindeverzeichnis/Administrativ/Archiv/GVAuszugQ/AuszugGV2QAktuell. html [checked on 9/7/2017].
- html [checked on 9/7/2017].
 [121] Kraml M, Walzer M. Geothermie. Wertschöfpfung und Wirtschaftlichkeit von Geothermieprojekten in Deutschland (Geothermal energy. Value creation and profitability of geothermal projects in Germany). Edited by forse of mbH; 2008. Available online at http://www.endurat-kommunal.de/uploads/media/ Geothermie in Deutschland 18.pdf, checked on 4/10/2018.
 [122] Janck S, Kupfermann GA. Vorhaben Ib. Stromerzeugung aus Geothermie (Project Ib. Deutschland). Stromerzeugung aus Geothermie (Project).
- Ib. Power generation from geothermal energy); 2014. Available online at https:// www.clearingstelle-eeg-kwkg.de/files/zwischenbericht-vorhaben-2b.pdf, checked n 3/21/2018
- on 3/21/2018.
 [123] Campos Rodríguez Carlos Eymel, Escobar Palacio José Carlos, Venturini Osvaldo J, Silva Lora Electo E, Cobas Vladimir Melián, Marques dos Santos Daniel, et al. Exergetic and economic comparison of ORC and Kalina cycle for low temperature enhanced geothermal system in Brazil. Appl. Them. Eng. 2013;52(1):109–19. https://doi.org/10.1016/j.applthermaleng.2012.11.012.
 [124] Rubio-Maya Carlos, Pastor Martínez Edgar, Romero Carlos E, Díaz Ambriz Victor M, Pacheco-Ibarra J Jesús. Techno-economic assessment for the integration into a multi-product plant based on cascade utilization of geothermal energy. Appl Therm Eng 2016;108:84–92. https://doi.org/10.1016/j.applthermaleng.2016.07. 108. 108

1

Identification of Potential Off-Grid Municipalities with 100% Renewable Energy Supply

Jann M. Weinand, Sabrina Ried, Max Kleinebrahm, Russell McKenna, Wolf Fichtner

Abstract—An increasing number of municipalities are striving for energy autonomy. This study determines in which municipalities and at what additional cost energy autonomy is feasible for a case study of Germany. An existing municipal energy system optimization model is extended to include the personal transport, industrial and commercial sectors. A machine learning approach identifies a regression model among 19 methods, which is best suited for the transfer of individual optimization results to all municipalities.

The resulting levelized cost of energy (LCOE) from the optimization of 15 case studies are transferred using a stepwise linear regression model. The regression model shows a mean absolute percentage error of 12.5%. The study demonstrates that energy autonomy is technically feasible in 6,314 (56%) municipalities. Thereby, the LCOEs increase in the autonomous case on average by 0.41 *e/kWh* compared to the minimum cost scenario. Apart from energy demand, base-load-capable bioenergy and deep geothermal energy appear to have the greatest influence on the LCOEs.

This study represents a starting point for defining possible scenarios in studies of future national energy system or transmission grid expansion planning, which for the first time consider completely energy autonomous municipalities.

Index Terms—Energy autonomy, renewable energy, geothermal power generation, electric vehicles, vehicle-to-grid, mixed integer linear programming, regression analysis.

I. INTRODUCTION

THE share of renewable energies (RE) in electricity generation has increased steadily in the past years. In 2018, REs already accounted for a third of the worldwide installed electricity capacity [1]. The planning of RE power plants has to be closely coordinated with power grid planning. By simultaneously considering grid and RE expansion, the costs of using local resources can be weighed against the costs of grid expansion to sites with higher RE potential [2]. However, many studies focus on just one of these aspects. For example, [3] and [4] concentrate on large-scale transmission grid planning with fixed generation capacities. At the same time, studies on transmission grid planning are often based on centralized RE generation [5-7].

However, [8] finds that a decentralised RE expansion could be economically favourable, largely due to higher required grid expansion costs in the centralized case. In fact, the expansion of RE resources is mainly decentralized due to their characteristics. Thus, the vast majority of the installed capacity of RE plants is connected to the distribution grid [9]. Related to this, in many countries, the owner structure of energy plants is changing: for example, the majority of German RE plants are actually owned and operated by private individuals, farmers and communities [10]. In this context, an increasing number of municipalities are striving for energy autonomy due to drivers like tax revenues and environmental awareness [11]. These municipalities mainly focus on annual municipal energy autonomy (AMEA), whereby the local RE generation exceeds the annual demand. In addition, some municipalities strive for complete municipal energy autonomy (CMEA), a state in which no energy is imported (i.e. "off-grid") [12].

For future power grid designs, the questions of whether, which, how many and at what cost municipalities could become completely energy autonomous is of interest. To this end, the whole energy system with all energy consumption sectors industrial, commercial, residential and transport - should be considered in municipal energy system analyses. Energy autonomy in municipalities has already been examined in [12-16]. Some of these studies are limited to the residential sector [12, 13]. Others also include further sectors. Thereby, the industrial energy demand is determined by surveys [16], interviews [15] or measurements of actual transformer substations [14]. Therefore, the application of these methods to other municipalities would require considerable effort. Since only individual municipalities or regions are considered in the studies, the results cannot be used to develop scenarios for future national energy systems. Furthermore, none of these studies investigates the impact of the flexibility through electric vehicles (EV) on costs

This paper aims to address the identified shortcomings of the studies on municipal energy autonomy. To this end, the energy systems of 15 municipalities are first analysed in detail with the

The authors gratefully acknowledge the financial support of the PhD College "Energy and Resource Efficiency" (ENRES), from the Federal State of Baden-Wuerttemberg, for funding the first author's PhD studentship. The contribution of the third author (MK) was supported by the Helmholtz Association under the Joint Initiative "Energy Systems Integration" (funding reference: ZT-0002).

J. M. Weinand, S. Ried, M. Kleinebrahm and W. Fichtner are with the Chair of Energy Economics, Institute for Industrial Production (IIP), Karlsruhe Institute of Technology (KIT), 76187 Karlsruhe, Germany (e-mails: jann.weinand@kit.edu, sabrina.ried@kit.edu, max.kleinebrahm@kit.edu and wolf.fichtner@kit.edu). R. McKenna is with the DTU Management, Technical University of Denmark, 2800 Kgs. Lyngby, Denmark (e-mail: rkenna@dtu.dk).

aid of an optimization model. This model is extended to include the personal transport, industrial and commercial sectors. Thereby, the assumption is made, that all vehicles of the personal transport sector are replaced by EVs. The optimization results are transferred to further municipalities by means of a regression model. Based on the results, energy scenarios could be derived for future planning of electricity grids. At the same time, the following research questions are addressed:

- How many and which municipalities can become energy autonomous?
- 2) Which cost increase would be associated with achieving energy autonomy in these municipalities compared to the optimized energy system without energy autonomy?
- 3) What impact does the consideration of the industrial, commercial and personal transport sector have on costs in off-grid municipalities?

Germany is selected as the case study for this paper since the developments in energy autonomy described in this introduction particularly apply to this country.

The organization of this paper is as follows: In section II, the methodology for optimizing energy systems in municipalities and transferring the results is presented. The results are then explained and discussed in section III and IV respectively, before the study concludes in section V.

II. METHODOLOGY

In this section, a method for determining the energy demand of the industrial, commercial and residential sector (cf. section II.A) as well as the RE potential (cf. section II.B) is presented. Section II.C explains how relevant municipalities for this study are identified using these demands and potentials. Subsequently, the RE³ASON model for energy system analysis is explained (cf. section II.D), which is extended by the industrial and commercial sectors (cf. section II.E) as well as personal transport sector (cf. section II.F). Finally, section II.G presents the methodology for transferring results of the RE³ASON model to other municipalities.

A. Demand of Energy Consumption Sectors

The assessment of the electricity demand for the residential, commercial and industrial sector is based on [17]. The assumption is made that the electricity demand of a municipality correlates with selected socio-economic indicators. Since the electricity demand and the corresponding indicators are known on a national level [18], the municipal electricity demand can be downscaled from the national level on the basis of the relative ratio of these indicators ("top-down scaling"). Based on the assumption that the significance of an indicator as a measure of size can be different for each sector, a weight matrix is used to indicate how strongly each indicator is weighted (cf. Table I). The weightings are determined by minimizing the mean square error in comparison to measured consumption values. For this purpose, 17 German municipalities are considered for which the annual electricity consumption is measured and published by sector [19-21].

The non-dimensional parameter sector size $(ss_{m,s})$ indicates the size of the respective sector s in the municipality m under consideration in relation to its size in Germany. The sector size is calculated on the basis of the weight matrix $w_{s,i}$, as well as the municipal $(IV_{m,i})$ and national $(IV_{N,i})$ values of all indicators I by means of the weight matrix:

$$ss_{m,s} = \sum_{i \in I} w_{s,i} \cdot \frac{IV_{m,i}}{IV_{N,i}} \tag{1}$$

The electricity demand ED in the municipality m and sector s is thus calculated by applying the sector size as a scaling factor for the corresponding national electricity demand ED_{N} :

ŀ

$$ED_{m,s} = ED_{N,s} \cdot ss_{m,s} \tag{2}$$

 TABLE I

 Weight Matrix $w_{5,i}$, for Assigning a Weight to each Indicator *i* to Calculate the Size of the Sector *s*. Sources of Indicators: [22–24].

Indicators	Residential	Commercial	Industrial
	sector	sector	sector
Area	0.075	0.000	0.787
Population	0.925	0.958	0.168
Number of industrial companies	0.000	0.000	0.015
Number of industrial employees	0.000	0.000	0.010
Gross salaries in industry	0.000	0.000	0.012
Number of employees with social security contributions	0.000	0.042	0.008

B. Renewable Energy Potential

The determination of the RE potentials in this section serves to select the municipality population to be investigated in this study (cf. section II.C). In [25], the potentials of residential rooftop photovoltaics [26] and wind energy [27] in Germany have been allocated at municipal level. As further potentials, the bioenergy and the deep geothermal energy potential in German municipalities are considered in this study using the methods from [17] and [12]. The bioenergy includes wood combustion plants and biogas plants. The data for forest area and agricultural land in the specific municipalities is taken from [24]. Furthermore, the fraction of the usable area is assumed to be 33% according to [28]. The hydrothermal temperatures for calculating the deep geothermal potential are taken from an open data set [29].

C. Selection of Municipality Population

By means of the methods in sections II.A and II.B, the annual electricity demand and potential RE electricity supply can be determined for each municipality in Germany. If the demand exceeds the supply, the respective municipality cannot achieve AMEA and thus especially not CMEA. Therefore, these municipalities are excluded from the municipality population beforehand. This calculation of AMEA neglects imports from neighbouring municipalities, which would be excluded anyway for CMEA.

The energy consumption patterns in the industrial sector show a high variety (cf. section II.D). Therefore, a standard load profile cannot adequately represent this sector. To minimize the impact on the results when using a standard load profile, only micro and small enterprises as defined by the European Commission [30] are taken into account. Therefore, municipalities with medium-sized and large industries (i.e. enterprises with more than 50 employees) are excluded from

the present analysis. When excluding the municipalities, the manufacturing industry serves as a representation for all economic sectors, since this branch accounts for the largest proportion of energy consumption [31] and the employment figures are only available for this sector at municipal level [32]. However, even small enterprises can be energy-intensive. Therefore, municipalities with companies from the *European Pollutant Release and Transfer Register* (PRTR) are additionally excluded. European companies must declare their emissions in this register, if the emission level exceeds certain thresholds [33] (100,000 tCO₂/a for greenhouse gases [34]).

D. RE³ASON Model

After determining the municipality population, the costs for achieving CMEA can be determined for these municipalities. The "Renewable Energies and Energy Efficiency Analysis and System OptimizatioN" (RE3ASON) model is used to calculate these costs, as it can be applied to any municipality in Germany without additional data collection. This is related to the fact that the model uses publicly available data to determine energy demand and potential energy supply. The optimization minimizes the total discounted system costs over the whole model horizon. Thereby, the types, dimensions and dispatch of the energy technologies and measures are optimized. The optimization takes a macroeconomic perspective and optimizes four years with 108 time slices each. The time horizon of the optimizations reaches until 2030 and the years 2015, 2020, 2025 and 2030 are optimized. Except for district heating, no explicit network infrastructure is considered in the model. Detailed information about the actual state of the model can be found in [17] and [13]. In the present study, the model is extended by the electricity demand of the commercial and industrial sector (cf. section II.E). Furthermore, EVs are implemented to represent the personal transport sector (cf. section II.F).

E. Implementation of Industrial and Commercial Sectors

Electricity load profiles enable to scale the calculated energy demand $ED_{m,s}$ to one year. For the commercial sector standard load profiles are used [35]. The electricity demand of the commercial sector c in a municipality m at hour t ($E_{m,e,t}$) is calculated as follows:

$$E_{m,c,t} = E_{c,t} * \frac{ED_{m,c}}{E_{c,sum}}$$
(3)

 $E_{e,t}$ is the electricity demand at time t and $E_{e,sum}$ the annual electricity demand of the standard load profile. The data set with industrial load profiles used in [36] contains three load profiles for small enterprises, which are used in the present study. The mean profile of the companies for *Shipping, Shaping of sheet* and *Iron casting* is used in this study as load profile for the industrial sector. Equation (3) can then be used to scale the demand profile analogously to the commercial sector.

F. Implementation of the Personal Transport Sector

This study assumes that all vehicles of the personal transport sector in a municipality are replaced by EVs. Thereby, the flexibility potential of the EV fleet is derived as follows. In a first step, flexibility potentials of single vehicles are generated with a model developed in [37]. The model uses representative mobility data of conventional vehicles in Germany [38] and simulates two extreme charging scenarios for each of them, given the assumption that an EV would replace them. The results include one-week time series for an as-soon-as-possible (ASAP) and an as-late-as-possible (ALAP) charging scenario, which can be considered as flexibility potentials for each vehicle. We assume that every vehicle has the possibility to charge both at home and at work, and that it is connected to the charging station throughout the parking duration.

The next step aims at aggregating the single vehicle flexibility potentials to a flexibility potential of one hypothetical battery which represents the municipality's EV fleet (cf. (4)). Therefore, we add the single vehicles' battery capacities C_{EV} to the fleet's battery capacity C_f . $N_{EV,m}$ is the number of EVs in a municipality, available from [39].

$$C_f = C_{EV} \cdot N_{EV,m} \tag{4}$$

In (5), the upper boundary for the fleet's battery state of charge $SoC_t^{max,f}$ in a time slice *t* is derived by totaling the single EVs' $SoC_{t,\nu}^{max,EV}$ which results when the vehicle is charged according to the ASAP-strategy. The single vehicle *v* is part of the total number of simulated EVs (*V*). In order to account for the representativeness of the vehicles in the dataset, the time series for $SoC_{t,\nu}^{max,EV}$ are weighted by the vehicle weightings w_{ν} . By dividing them by the sum of all weightings, the resulting weighted time series represents the average of the fleet. Finally, the flexibility potential is scaled on municipality level with the number of EVs in the municipality.

$$SoC_{t,m}^{max,f} = \sum_{\nu=1}^{V} \left(SoC_{t,\nu}^{max,EV} \cdot w_{\nu} \right) \cdot \left(\sum_{\nu=1}^{V} w_{\nu} \right)^{-1} \cdot N_{EV,m}$$
(5)

 $SOC_t^{min,f}$ is calculated accordingly based on the simulated ALAP-strategy. The power discharged from the EV battery by driving $P_t^{dr,f}$ and the available charging power $P_t^{max,f}$ are also determined analogously. The latter depends on the power of the charging station and whether the vehicle is parked at one of its charging locations or not.

As shown in [37], the driving and charging patterns vary for different degrees of urbanization. Since most municipalities from the preselected population are located in rural areas (cf. section III.A), the mobility data is preselected by geographic criteria. The data of vehicles in rural areas with higher and lower density, according to the municipality grouping by BBSR [40], are considered. The resulting flexibility potential pattern of the fleet is used for each municipality. Further assumptions are listed in Table II. The mean value of 3.7 – 22 kW charging power in low-voltage grids is used as available charging power.

Controlled bidirectional charging is selected as charging strategy for the EVs [42]. Therefore, the charging process is controlled by the municipal energy management system with regard to load, time and limitations by the mobility patterns. In addition, the battery can be discharged to feed electricity into the municipal grid, known as *vehicle-to-grid* (V2G). The main

modelling aspects of the EVs are listed below, for further information please refer to [42]. The SoC of the EV batteries (SoC_t^f) depends on the previous SoC (SoC_{t-1}^f) , the (dis)charging efficiency (η_{EV}) , the charge power $(P_t^{ch,f})$ as well as discharge power $(P_t^{v2G,f})$ and the power required for driving $(P_t^{dr,f})$ [42]:

$$SoC_t^f = SoC_{t-1}^f + \frac{(P_t^{ch,f} \cdot \eta_{EV} - P_t^{V2G,f} / \eta_{EV} - P_t^{dr,f}) \cdot dt}{C_f} \quad \forall t \in T(6)$$

At a SoC above 75%, the charging power reduction $(P_{t,max,red})$ increases linearly according to (7) [41]. P_{SE} is the available charging power of the supply equipment.

$$P_{t,max,red} \ge P_{SE} \left(4 \cdot SoC_t^f - 3 \right) \tag{7}$$

Equation (8) ensures that for each day d the EVs are charged with the energy required for driving $P_t^{dr,f}$. This implies that the load shift potential can only be exploited within one day and thus limits the usage of EV flexibility to a more conservative range.

$$\sum_{t=1}^{24} \left(P_t^{ch,f} - P_t^{\nu_{2G,f}} \right) \cdot dt \ge \sum_{t=1}^{24} P_t^{dr,f} \cdot dt \quad \forall d\epsilon D \qquad (8)$$

As in [42], the investment in EVs is assumed to be personal, preference-driven and for mobility reasons only. Therefore, this investment is not considered in the optimizations.

G. Transfer of Results

The RE³ASON model is applied to determine the costminimal energy system for preselected municipalities as case studies. On the one hand for the reference case without autonomy and on the other hand for the case with CMEA. In the reference case, the energy system is optimized without restricting imports and exports. Subsequently, the *Levelized Cost of Energy (LCOE)* are calculated for both cases and all preselected municipalities (cf. (9), [43]). Thereby the conversion factor for electricity into heat is assumed to be the heat pump's coefficient of performance (3.5) as in [43], since the heat load is taken into account for the residential sector.

$$LCOE = \frac{\sum_{y=1}^{Y} \frac{CAPEX_y + OPEX_y}{(1+r)^y}}{\sum_{y=1}^{Y} \frac{E_{m,total,y}}{(1+r)^y}}$$
(9)

The LCOEs are calculated depending on the investments (*CAPEX*), the operational and maintenance costs (*OPEX*), the total energy demand ($E_{m,total}$) and the year y. The interest rate r is assumed to be 5%.

A regression is used to transfer the results of the case studies to the entire municipality population. The dependent variable is the difference between LCOEs in the autonomous and in the reference case ($\Delta LCOE$). In the selection of the independent variables, those that correlate with other variables are eliminated. Therefore, for all correlations above |0.9| one variable is excluded.

To avoid an overfitting in the regression, a k-fold cross-

validation is applied [44]. Since our sample is small (n = 15), the *leave-one-out cross-validation* is used, with k = n = 15. 19 different methods are applied, ranging from linear regression models and support vector machines to Gaussian Process Regression models. From these methods, the model that results in the lowest root mean squared error is selected.

TABLE II	
C 3 C 1 11	DI CI

Assumptions	for Modell	ing EV Flexibility	у.
Parameter	Value	Unit	Source
Battery capacity (C_{EV})	50	kWh	[40]
EV energy consumption	10.2	kWh/100 km	[40]
EV battery efficiency (η_{EV})	90	%	[41]
Number of simulated EVs	229		[36, 37]
(V)			
Available charging power	13	kW	Assumption
$(P_{t,v}^{max,EV})$			
SoC-range during operation	5-100	%	Assumption

III. RESULTS

In section III.A, the municipality population examined in this study is presented. In addition, case studies are selected for investigation in the RE³ASON model. Subsequently, the optimization results of these case studies (cf. section III.B) as well as sensitivity analyses (cf. section III.C) are explained. Finally, section III.D presents the results of the regression.

A. Case Studies

The methodology described in section II.C results in an exclusion of 3,120 municipalities that are not suitable for AMEA, 2,656 municipalities with large industries and 616 municipalities from the PRTR Register (grey area in Fig. 4). The remaining 6,314 municipalities correspond to 56% of the municipalities, 14% of the population, 40% of the land area and 23% of the annual electricity consumption of Germany.

As case studies, municipalities which differ particularly with regard to the independent variables from the regression are selected from the municipality population. For each indicator, one municipality is selected that has the maximum or minimum value for this indicator. As for some indicators the municipalities are the same, a total of 15 different municipalities remain for examination, which are geographically distributed across Germany.

B. Energy System Optimization Results

For one of the 15 investigated municipalities, Prinzenmoor, the resulting LCOEs are shown in Fig. 1, for the reference case without CMEA (P1) and with CMEA (P2). Results of other scenarios (P3 to P6) are explained in section III.C.

Prinzenmoor is a small municipality with only 179 inhabitants. The electricity demand in the industrial, commercial and residential sectors is 2.9 GWh/a, 0.3 GWh/a and 0.2 GWh/a respectively.

The value range of the y-axis in Fig. 1 contains negative values (up to -0.1 C/kWh), since exports result in a small negative contribution to the LCOEs in P1. In the autonomy case P2, the energy system of Prinzenmoor changes greatly. Whereas in P1 the energy is provided by wind turbines and power grid, in P2 the entire energy is provided by deep geothermal energy. The geothermal plant is used for base load

operation, while the EV batteries in the municipality are discharged to cover peak loads.

For P2, the electricity supply and demand for a typical weekday in 2015 are given in Fig. 2. The high electricity demand of the residential sector at night is remarkable. This is due to the fact that buildings are considered as daily energy storages in RE³ASON. Therefore, during the night hours, when the electricity demand of the other sectors is low, a large part of the heat demand is covered by electric storage heaters and heat pumps. Due to the base load operation of the geothermal plant, electricity surpluses occur in time steps with low electricity demand. Parts of the surpluses are used to charge the batteries of the EVs. A comparison of the power from charging ($P_t^{ch,f}$) and V2G ($P_t^{V2G,f}$) with the maximum available charging or discharging power P_{SE} shows that no more than 60% of the battery flexibility is exploited in the various time steps of the optimization.

CMEA is associated with a high increase in LCOEs in Prinzenmoor (Δ LCOE = 0.45 \in kWh, cf. Fig. 1). In all 15 examined municipalities, the increase in LCOEs range between 82% and 487%, which corresponds to Δ LCOE between 0.19 \in kWh and 0.55 \in kWh. Biomass and additional battery storages are installed in almost every municipality. Geothermal plants are built in 10 of the 15 municipalities, partly supported by biomass, wind and solar energy.

In most municipalities and if CMEA has to be achieved, surplus electricity occurs in hours in which the generation exceeds the demand (cf. Fig. 2 for Prinzenmoor). The surpluses range from 0 to 4 GWh/a. In P2 the surpluses amount to about 50% of the total energy demand of the municipality, which is the highest share among all 15 municipalities. Between 2015 and 2030, the average surpluses are reducing from around 2 GWh to 1 GWh in all municipalities. This is due to the fact that several volatile generation technologies, which are installed in the municipalities in 2015, are replaced over time. The average CO₂ abatement costs for the 15 municipalities are around 3.7 k€/tCO₂. For a detailed discussion of RE³ASON model results, including demand and generation patterns, please refer to [13].

C. Sensitivity Analyses

In the sensitivity analyses (cf. Table III), the reference scenario for the autonomy case P2 is changed in order to examine the influence of model extensions and assumptions (P3, P5 and P6). Secondly, the influence of geothermal plants is investigated, as in many German municipalities no geothermal potential exists (P4).



Fig. 1: Technology-specific LCOE contributions for the optimal energy systems in the municipality Prinzenmoor for six different scenarios. The share "Other" includes costs for insulation, heating systems, appliances and lighting in the residential sector.

In order to quantify the uncertainty resulting from the chosen industrial load profiles (cf. section II.E), in scenario P3 the mean profile is replaced by the load profile of the *Iron Casting Company*, which shows higher peaks. The maximum peak is 105% higher than in the mean profile. Consequently, the LCOEs increase by 3% (cf. Fig. 1). In Prinzenmoor, the industrial sector accounts for about 80% of the energy demand. The change in the industrial load profiles therefore has a rather small influence on the results.

If no geothermal plant may be built in scenario P4, the energy is provided by wind and biomass (cf. Fig. 1). Due to the volatile wind energy production, additional storages are required. In addition, more efficiency measures are implemented, such as more efficient household appliances and the insulation of buildings. Especially the efficiency measures and further storages are responsible for the LCOE increase of 12% compared to P2. However, the low increase in costs shows that even municipalities without a large base load potential can become autonomous at comparable costs. Compared to P2, the electricity surpluses are 80% lower. This is because there is less base load in operation in P4 and therefore not so many time



Fig. 2: Electricity generation and demand patterns for a typical weekday in all four seasons in 2015.

steps with surplus electricity as in P2 (cf. Fig. 2).

Due to the electricity surpluses from the geothermal plant in P2, the storage capacities of the EVs are beneficial. However, if a scenario without EVs (P5) is considered, the construction of the geothermal plant is no longer economical. In this case, additional storages have to be installed in order to use much energy from the geothermal plant. Alternatively, the geothermal plant could be dimensioned smaller. However, this type of plant incurs very high fixed costs, and therefore this would not be economical either. This shows that, depending on the conditions of the energy system, the storage capacities of the EVs can moderately reduce costs (by 16% compared to P2). In an energy system without deep geothermal potential, though, the costs would increase only slightly (comparison between P4 and P5).

Due to the consideration of the industrial and commercial sectors in addition to the residential sector, energy autonomy is not feasible in every municipality (cf. section III.A). However, at the same time, the LCOEs for achieving autonomy are greatly reduced (comparison of P2 and P6), since fixed costs of the system are related to a significantly larger amount of energy. Prinzenmoor is an extreme case, due to the small size of the municipality. However, to a lesser extent, this statement can be applied to other municipalities as well.

 TABLE III

 Scenarios of the Sensitivity Analysis in Prinzenmoor.

 enario
 Differences to reference scenario P2

Scenario	Differences to reference scenario P2
P3	Load profile of industrial companies (Iron Casting)
P4	Without geothermal plants
P5	Without EVs
P6	Without industrial and commercial sector

D. Regression Results

After the correlation analysis, the following six indicators remained for the regression: industrial electricity demand, residential electricity demand, population density, technical geothermal potential, technical wind energy potential and technical bioenergy potential. From the 19 models of the 15 fold cross validation, *stepwise linear regression* proved to be the best method. The model whose results are shown in Fig. 3 yields the error measures in Table IV.

The technical bioenergy potential and the products of residential electricity demand and population density as well as residential electricity demand and technical geothermal potential are selected as features for the regression. The fact that the industrial electricity demand and the technical wind potential are not used in the regression could be related to the correlation above 0.8 with the technical bioenergy potential.

After applying the regression model to all 6,314 municipalities, 155 outliers downwards ($\Delta LCOE \leq 0.02 \in /kWh$) and 31 upwards ($\Delta LCOE \geq 1.50 \in /kWh$) are eliminated. This is done due to the high R² (0.86) as this could indicate a slight overfitting of the model. A lower bound of 0.02 \in /kWh was chosen as this corresponds to a cost increase of about 5% in relation to the LCOEs of the 15 investigated municipalities. Fig. 4 shows the 6,128 remaining municipalities and the distribution of $\Delta LCOE$. Among these municipalities, the mean value of

 $\Delta LCOE$ is 0.41 \in /kWh. The data of municipalities with $\Delta LCOE$, demand of the sectors and RE potential can be provided upon request.

When distributing the regression results according to the ten German municipality clusters from [26], the results seem plausible: The highest mean $\Delta LCOE$ is reached in cluster 2 (0.578 ϵ/kWh), which mainly contains cities with low RE potential. On the other hand, the lowest mean $\Delta LCOE$ s are achieved in clusters 3 (0.350 ϵ/kWh), 4 (0.349 ϵ/kWh) and 8 (0.379 ϵ/kWh), which contain mainly rural municipalities with particularly high potential for RE and especially deep geothermal energy.



Fig. 3: Results of the stepwise linear regression. The error margins between the results from the optimizations (true) and the predicted values from the regression are shown.

Б

TABLE IV			
tesulting Error Measures of the Stepwise Linear Regression.			
Root mean squared error	0.047		
Mean absolute percentage error	0.125		
R-squared	0.860		

IV. DISCUSSION AND CRITICAL APPRAISAL

In this study, a methodology for determining LCOEs for achieving energy autonomy in all municipalities of a country was presented. Germany was used as a case study, but the general methodology can also be applied to any other country.

The study has shown that achieving CMEA is associated with large additional costs of 0.41 ℓ/kWh on average. Thus the costs per kWh are more than doubled compared to an optimized energy system without autonomy. Therefore, future studies at a national level should investigate whether and where CMEA is worthwhile if grid expansion is taken into account. Thereby, the results of the present study can serve as a scenario in the design of transmission networks. For example, the assumption could be made that all municipalities with $\Delta LCOE$ less than the mean value (0.41 ℓ/kWh) will become autonomous. Then the demand and feed-in from these municipalities could be excluded from the analyses. Furthermore, simultaneous optimization of transmission grid expansion and selection of autonomous municipalities could be performed to determine the optimal future national energy system.

The LCOEs and ALCOEs have probably been overestimated

in this study. Firstly, due to the separate consideration of individual municipalities. The municipal boundaries represent administrative units that do not necessarily have to represent optimal boundaries for energy systems. In addition, the simultaneous optimization of neighbouring municipalities could lead to lower LCOEs than in individual cases. Instead, the electricity surpluses which could be used in neighbouring municipalities to cover parts of the demand are curtailed. Also, the reference case for the determination of the ALCOE is an optimized energy system without autonomy, which in reality does not exist in most municipalities. Furthermore, the expression of these costs in absolute terms, irrespective of the municipality size or energy system structure, could be misleading. An improvement could be to redistribute these costs per final consumer, in order to give a more meaningful and comparable indicator.



Fig. 4: Illustration of 6,128 [55%] German municipalities that can become completely autonomous and the associated Δ LCOE.

In addition, part of the costs could be underestimated as no grid infrastructures in or outside the municipality were considered. Also, a standard load profile was used to include the electricity demand of the industrial sector. The sensitivity analysis showed that a different structure of the load profile does not have a large influence on the costs. However, the load profiles in individual municipalities could differ greatly from those used in this study. Furthermore, the modelling of EV s could also be improved. Instead of an aggregated driving profile, individual or clustered driving profiles could be used The computing time of the energy system model would then be a particularly restrictive factor. All of these improvements should be explored in future studies.

Moreover, the regression was used to transfer the results to a large number of municipalities whose individual analysis would not be possible in a single study. However, optimization and subsequent regression do not replace detailed planning of

the energy system of a single municipality.

V. CONCLUSIONS

In the present study, a methodology was developed to determine the feasibility and costs for complete municipal energy autonomy. First, methods for estimating the energy demand and potential for renewable energies were proposed. On this basis, municipalities in which complete energy autonomy is not feasible could be excluded. Subsequently, an energy system optimization model was extended to include the personal transport, industrial and commercial sectors and applied to a number of municipalities in order to determine the costs for complete energy autonomy. In a final step, the results were transferred to further municipalities using selected indicators in a regression model.

In this paper, Germany has been selected as case study, where 6,314 (56%) municipalities were identified, in which complete energy autonomy could be technically feasible. Of these municipalities, 15 were selected as case studies, which differ greatly in terms of the indicators used in the regression analysis. The results of the optimizations showed the influence of individual technologies and measures on the levelized cost of energy (LCOE). Thereby, it became apparent that complete energy autonomy is always associated with a high cost increase. Furthermore, the integration of the industrial and commercial sectors has a reducing effect on the LCOEs, since fixed costs are distributed across a larger amount of energy. In addition, the flexibility of electric vehicles can moderately reduce LCOEs. Using a stepwise linear regression model (mean absolute percentage error = 12.5%), the results of the optimizations could finally be transferred to the 6,314 municipalities. On average, the additional LCOEs, which have to be paid in the autonomous compared to the reference (minimal cost) case, amount to 0.41 ${\ensuremath{\varepsilon}}/k{\ensuremath{W}} h$ Apart from energy demand, base load capable bioenergy and deep geothermal energy appear to have the greatest influence on the LCOEs.

The main areas for improving the methodology include the consideration of grid infrastructures and surplus electricity from neighbouring municipalities, as well as more detailed modelling of industrial demand. The method of calculating and comparing the costs of energy autonomy should be improved to express these costs per municipal end user. In future studies in which the national energy system or transmission grid expansion is planned, the results of this paper can be used as a possible scenario.

VI. REFERENCES

- IRENA, Renewable Energy New Accounts for a Third of Global Power Capacity. [Online] Available: https://www.irena.org/newsroom/ pressreleases/2019/Apr/Renewable-Energy-Now-Accounts-for-a-
- pressreleases/2019/A pr/Renewable-Energy-Now-Accounts-Renear-Third-of-Global-Power-Capacity. Accessed on Sep. 17 2019.
 [2] V. Kris Iman et al., "Co-optimization of electricity transmission and generation resources for planning and policy analysis: Review of concepts and modeling approaches," *Energy Syst*, vol. 7, no. 2, pp. 297-332, 2016.
- [3] S. Lumbreras, A. Ramos, and F. Banez-Chichano, "Optimal transmission network expansion planning in real-sized power systems with high networks be penetration," *Electric Power Systems Research*, vol. 149, pp. 76–88, 2017.
October 2019

- S. Lumbreras *et al.*, "Large-scale transmission expansion planning: From zonal results to a nodal expansion plan," *BT Generation, Transmission & Distribution*, vol. 11, no. 11, pp. 2778–2786, 2017. O. Ozdemir, F. D. Munoz, J. L. Ho, and B. F. Hobbs, "Economic [4]
- [5] O. Ozdemir, F. D. Munoz, J. L. Ho, and B. F. Hobos, "Economic Analysis of Transmission Expansion Planning With Price-Responsive Demand and Quadratic Losses by Successive LP," *IEEE Trans. Power Syst.*, vol. 31, no. 2, pp. 1096–1107, 2016. M. Moeini-Aghtaie, A. Abbaspour, and M. Fotuhi-Firuzabad, "Incorporating Large-Scale Distant Wind Farms in Probabilistic
- [6] Transmission Expansion Planning—Part II: Case Studies," *IEEE Trans. Power Syst.*, vol. 27, no. 3, pp. 1594–1601, 2012.
- A. Arabali, M. Ghofrani, M. Etezadi-Amoli, M. S. Fadali, and M. Moeini-Aghtaie, "A Multi-Objective Transmission Expansion Planning [7] Framework in Deregulated Power Systems With Wind Generation," IEEE Trans. Power Syst., vol. 29, no. 6, pp. 3003–3011, 2014.
- Reiner Lemoine Institut, Vergleich und Optimierung von zentral und dezentral orientierten Ausbaupfaden zu einer Stromversorgung aus Erneuerbaren Energien in Deutschland. [Online] Available: [8] https://www.bvmw.de/fileadmin/pdf-archiv/Studie_zur_dezentralen Energiewende.pdf.pdf. Accessed on: Sep. 07 2019.
- BMBF, New network structures: The ENSURE Kopernikus project. [Online] Available: https://www.kopernikus-projekte.de/en/projects/ [9]
- [Online] Available: https://www.kopernikus-projekte.de/en/projects/ new-network-structures. Accessed on: Sep. 17 2019. trend:research, *Bigentimerstruktur: Erneuerbare Energien. Entwicklung der Akteursvielfalt, Rolle der Energieversorger, Ausblick bis 2020.* [Online] Available: https://www.trendresearch.de/ studie.php?s=672. Accessed on: Sep. 06 2019. M. Engelken, B. Römer, M. Drescher, and I. Welpe, "Transforming the energy system: Why municipalities strive for energy self-sufficiency," *Energy Pachety vol.* 98, m 365–377, 2016. [10]
- [11]
- Energy Policy, vol. 98, pp. 365–377, 2016. J. M. Weinand, R. McKenna, M. Kleinebrahm, and K. Mainzer, [12] "Assessing the contribution of simultaneous heat and power generation from geothermal plants in off-grid municipalities," *Applied Energy*, vol. 255, 2019.
- Vol. 255, 2019.
 T. Jenssen, A. König, and L. Eltrop, "Bioenergy villages in Germany: Bringing a low carbon energy supply for rural areas into practice," *Renewable Energy*, vol. 61, pp. 74–80, 2014.
 E. Kötter, L. Schneider, F. Schnke, K. Ohnmeiss, and R. Schröer, "The Kerner distribution of the comparison of the second [13]
- [14] future electric power system: Impact of Power-to-Gas by interacting with other renewable energy components," Journal of Energy Storage, vol. 5, pp. 113-119, 2016.
- P. Peura, A. Haapanen, K. Reini, and H. Törmä, "Regional impacts of sustainable energy in western Finland," *Journal of Cleaner Production*, [15]
- vol. 187, pp. 85–97, 2018. J. Schmidt *et al.*, "Regional energy autarky: Potentials, costs and consequences for an Austrian region," Energy Policy, vol. 47, pp. 211-221, 2012
- [17] K. Mainzer, "Analyse und Optimierung urbaner Energiesysteme: K. Manizu, Analyse und Optimiering urbanet Energicesysteme: Entwicklung und Anwendung eines überragbaren Modellierungswerkzeugs zur nachhaltigen Systemgestaltung," Ph.D. dissertation, IIP, KIT, Karlsruhe, Germany, 2019. BMWi, Energiedaten: Gesamtausgabe (Energy data: Complete Denvice Content of Content o
- [18] edition). [Online] Available: https://www.bmwi.de/Redaktion/ DE/Downloads/Energiedaten/energiedaten-gesamt-pdf-grafiken. pdf?__blob=publicationFile&v=38. Accessed on: Mar. 06 2019.
- [19] Energieagentur Rheinland-Pfalz, Energieatlas Rheinland-Pfalz, [Online] Available: https://www.energieatlas.rlp.de/earp/daten/
- strom/stromverbrauch/. Accessed on: Mar. 06 2019. Stadt Karlsruhe, Klimaschutz in Karlsruhe: Zweiter Fortschrittsbericht [20] 2011. [Online] Available: https://www.karlsruhe.de/b3/natur_und umwelt/klimaschutz/klimaschutz/pt/pdf_dteien/HF_sections/content/1 487575573415/Zweiter_Fortschritsbericht_2011.pdf. Accessed on: Mar. 06 2019.
- [21] C. Neumann, A. Blumberg, H. Schwieder, and N. Sondermann, Energiebilanz für die Region Freiburg: Verbrauch und Potenziale. Endbericht. [Online] Available: http://www.wee100prozent.de/wp-content/uploads/2013/07/ Potenzialstudie-Energiebilanz-f%C3%BCr-die-Region-Freiburg-August-20121.pdf. Accessed on: Sep. 17 2019.
- Destatis, Zensusdatenbank des Zensus 2011. [Online] Available: https://ergebnisse.zensus2011.de/. Accessed on: Mar. 06 2019. [22]
- [23] Destatis, Gemeindeverzeichnis-Informationssystem GV-ISys. [Online] Available: https://www.destatis.de/DE/ZahlenFakten/Laender Regionen/Regionales/Gemeindeverzeichnis/GemeindeverzeichnisEinle itenderText.html. Accessed on: Mar. 06 2019

- Destatis, Regionaldatenbank Deutschland. [Online] Available: [24] https://www.regionalstatistik.de/genesis/online/data. Accessed on: Mar. 06 2019.
- J. M. Weinand, R. McKenna, and W. Fichtner, "Developing a [25] municipality typology for modelling decentralised energy systems," *Utilities Policy*, vol. 57, pp. 75–96, 2019.
- [26] K. Mainzer et al., "A high-resolution determination of the technical potential for residential-roof-mounted photovoltaic systems ir
- Germany," *Solar Energy*, vol. 105, pp. 715–731, 2014. R. McKenna, S. Hollnaicher, and W. Fichtner, "Cost-potential curves [27] for onshore wind energy: A high-resolution analysis for Germany, Applied Energy, vol. 115, pp. 103–115, 2014. Umweltbundesamt, Sustainable use of global land and biomass
- [28] resources. [Online] Available: https://www.umweltbundesamt.de/ sites/default/files/medien/419/publikationen/130617_englisch_lang_we b.pdf. Accessed on: Aug. 09 2019.
- J. M. Weinand, R. McKenna, and K. Mainzer, "Spatial high-resolution [29] socio-energetic data for municipal energy system analyses," Scientific Data, in press, 2019. European Commission, What is an SME? [Online] Available:
- [30] http://ec.europa.eu/growth/smes/business-friendly-environment/sme-definition_en. Accessed on: Mar. 06 2019.
- Umweltbundesamt, Branchenabhängiger Energieverbrauch des verarbeitenden Gewerbes. [Online] Available: https://www.umwelt [31] bundesamt.de/daten/umwelt-wirtschaft/industrie/branchenabhaengigerenergieverbrauch-des#textpart-1. Accessed on: Jul. 02 2019. Statistisches Bundesamt, Verarbeitendes Gewerbe, Bergbau und
- [32] Gewinnung von Steinen und Erden: Betriebe, Beschäftigte - Stichtag. 30.09. - regionale Tiefe: Gemeinden, Samt-/Verbandsgemeinden. [Online] Available: https://www.regionalstatistik.de/genesis/online/.
- [Online] Available: https://www.regionalstatistuk.de/genesisi/online/. Accessed on: Sep. 07 2017.
 EUR-lex, European Pollutant Release and Transfer Register (PRTR).
 [Online] Available: https://eur-lex.europa.eu/legal-content/EN/TXT/ ?uri=LEGISSUM%3 Al28149. Accessed on: Jul. 02 2019.
 R. Bodle and G. Homann, PRTR-Gesetz (SchadRegProtAG). [Online] Available: https://www.thru.de/fileadmin/SITE_MASTER/content/ [33]
- [34] Dokumente/Downloads/PRTR_Kommentar_final_2017.pdf. Accessed on: Jul. 02 2019.
- bdew, Standardlastprofile Strom. [Online] Available: https://www [35] bdew.de/energie/standardlastprofile-strom/. Accessed on: Jul. 01 2019. F. Braeuer, J. Rominger, R. McKenna, and W. Fichtner, "Battery
- [36] storage systems: An economic model-based analysis of parallel revenue streams and general implications for industry," Applied
- Energy, vol. 239, pp. 1424–1440, 2019. D. Heinz, "Erstellung und Auswertung repräsentativer Mobilitäts- und Ladeprofile für Elektrofahrzeuge in Deutschland," Working Paper [37] Series in Production and Energy 30, 2018. [Online] Available https://publikationen.bibliothek.kit.edu/1000086372.
- C. Weiß, B. Chlond, S. von Behren, T. Higert, and P. Vortisch, Deutsches Mobilität spanel (MOP) Wissenschaftliche Begleitung und Auswertungen, Bericht 2015/2016: Alltagsmobilität und Fahrleistung. [Online] Available: https://www.bmvi.de/SharedDocs/DE/Anlage/ [38] VerkehrUndMobilitaet/mop-jahresbericht-2015-2016.pdf?__blob=publicationFile. Accessed on: Jul. 18 2019. KBA, *Fahrzeuge (vehicles)*. [Online] Available: https://www.kba.de/
- [39] DE/Statistik/Fahrzeuge/fahrzeuge_node.html. Accessed on: Mar. 06 2019.
- BBSR, Laufende Raumbeobachtung Raumabgrenzungen. [Online] Available: https://www.bbsr.bund.de/BBSR/DE/Raumbeobachtung/ [40] Raumabgrenzungen/SiedlungsstrukturelleGebietstypen/Regionstypen/r egionstypen.html. Accessed on: Aug. 09 2019. A. Thielmann, A. Sauer, and M. Wietschel, *Produkt-Roadmap*
- [41] Energiespeicher für die Elektromobilität 2030. [Online] Available: https://www.isi.fraunhofer.de/content/dam/isi/dokumente/cct/lib/PRM-ESEM.pdf. Accessed on: Jul. 18 2019. T. Kaschub, P. Jochem, and W. Fichtner, "Solar energy storage in
- [42] German households: Profitability, load, changes and flexibility,"
 Energy Policy, vol. 98, pp. 520–532, 2016.
 D. Grosspietsch, P. Thömmes, B. Girod, and V. H. Hoffmann, "How,
- [43] When, and Where? Assessing Renewable Energy Self-Sufficiency at the Neighborhood Level," (eng), Environmental science & technology,
- G. Zhang and J. D. McCalley, "Estimation of Regulation Reserve Requirement Based on Control Performance Standard," *IEEE Trans.* [44] Power Syst., vol. 33, no. 2, pp. 1173-1183, 2018.